

Durham E-Theses

Holographic Correlators and Effective Actions for Quantum Gravity

ABL, THERESA

How to cite:

ABL, THERESA (2021) Holographic Correlators and Effective Actions for Quantum Gravity, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/14025/

Use policy

The full-text may be used and/or reproduced, and given to third parties in any format or medium, without prior permission or charge, for personal research or study, educational, or not-for-profit purposes provided that:

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Holographic Correlators and Effective Actions for Quantum Gravity

Theresa Abl

A Thesis presented for the degree of Doctor of Philosophy



Centre for Particle Theory Department of Mathematical Sciences Durham University United Kingdom

June 2021

Holographic Correlators and Effective Actions for Quantum Gravity

Theresa Abl

Submitted for the degree of Doctor of Philosophy June 2021

Abstract:

In this thesis we study holographic correlators in three different examples of the AdS/CFT correspondence. In particular, we consider the low-energy effective actions of quantum gravity, where the leading term describes supergravity and the corrections correspond to higher-derivative interactions. Firstly, we consider the duality between string theory in $AdS_5 \times S^5$ and 4d maximally supersymmetric Yang-Mills theory $(\mathcal{N} = 4 \text{ SYM})$. We propose a systematic procedure for obtaining all single-trace half-BPS correlators in $\mathcal{N} = 4$ SYM corresponding to the four-point tree-level amplitude for type IIB string theory in $AdS_5 \times S^5$. The underlying idea is to compute generalised ten-dimensional contact Witten diagrams, treating AdS and S on equal footing, which are obtained from a 10d scalar effective field theory in $AdS_5 \times S^5$. Next, we study holographic correlators in $AdS_7 \times S^4$. M-theory in this background is dual to the 6d(2,0) theory. In particular, we derive recursion relations for the anomalous dimensions of double-trace operators occurring in the conformal block expansion of four-point stress tensor correlators in the 6d (2,0) theory. These anomalous dimensions encode higher-derivative corrections to supergravity in $AdS_7 \times S^4$ arising from M-theory. Finally, we consider quantum gravity in $AdS_2 \times S^2$ whose dual CFT

has superconformal group SU(1,1|2). Firstly, we propose an $AdS_2 \times S^2$ effective action which describes both supergravity and higher-derivative corrections and compute the four-point half-BPS correlators using generalised 4d Witten diagrams, analogous to $AdS_5 \times S^5$ above. Moreover, it was recently shown that IIB supergravity in $AdS_5 \times S^5$ enjoys 10d conformal symmetry. We adapt this approach, which is complementary to the effective action approach, to quantum gravity in $AdS_2 \times S^2$. We show that the 1d supergravity and free theory correlators exhibit 4d conformal symmetry and discuss implications for higher-derivative corrections where the symmetry is generically broken, except for specific cases.

Declaration

The work in this thesis is based on research carried out in the Department of Mathematical Sciences at Durham University. The results in chapters 3, 4 and 5 are based on the following collaborative works

- Theresa Abl, Paul Heslop and Arthur E. Lipstein. 'Towards the Virasoro-Shapiro amplitude in AdS₅×S⁵'. In: *JHEP* 04 (2021), p. 237. doi:10.1007/JHEP04(2021)237. arXiv:2012.12091 [hep-th].
- Theresa Abl, Paul Heslop and Arthur E. Lipstein. 'Recursion relations for anomalous dimensions in the 6d (2,0) theory'. In: *JHEP* 04 (2019), p. 038. doi:10.1007/JHEP04(2019)038. arXiv:1902.00463 [hep-th].
- Theresa Abl, Paul Heslop and Arthur E. Lipstein. In preparation.

No part of this thesis has been submitted elsewhere for any degree or qualification.

Copyright © 2021 Theresa Abl.

"The copyright of this thesis rests with the author. No quotation from it should be published without the author's prior written consent and information derived from it should be acknowledged."

Acknowledgements

First and foremost, I would like to thank my supervisor Arthur Lipstein for sharing his knowledge, insight and ideas and for introducing me to these beautiful research topics. Furthermore, I am very grateful for his support and patience, for always taking time to answer my questions and to discuss research, as well as for his support in academic matters not directly project related and his advice and insight during the writing of this thesis. Secondly, I would like to thank Paul Heslop for working together on all of our projects. I am very grateful to him for sharing his knowledge, insight and ideas, for his patience and for always taking time for discussions and giving advice. I thoroughly enjoyed working on the projects presented in this thesis and I am extremely grateful for everything I have learnt from working with Arthur and Paul.

I would also like to thank my examiners Madalena Lemos and Arkady Tseytlin for very interesting discussions and comments as well as Simon Ross for chairing the viva.

Furthermore, I am very grateful for having been awarded the Durham Doctoral Studentship and would like to thank everyone involved in this process. I would also like to thank everyone in the Department of Mathematical Sciences for many interesting discussions as well as for sharing advice and providing support.

In particular, I would like to thank my office mates Andy, Danny, Kieran, Nam and Xiang, including Davide and Gabriele, for many interesting and inspiring discussions, as well as lunch breaks spent together, going to the pub or for dinner and more recently, for our zoom calls. I am grateful to my friends for always being there for me. I would like to especially thank Rachel for our regular walks and zoom calls during this last year which made this time so much more enjoyable. I want to thank Linda for being the best friend I could wish for, for always making me feel better whenever I hear from her and for always believing in me.

I would like to thank my family for their unconditional support, specifically my parents Marion and Michael, my sister Anna and brother Stefan, as well as my grandparents Walter and Annelies. Thank you for always believing in me.

Most important of all, I am immensely grateful to Severin, for all the support he has shown me during my whole studies and specifically during my time as a PhD student, for his patience and understanding and for making me laugh every single day.

Dedicated to

Severin

Contents

Al	Abstract			iii	
1	Introduction			1	
2	Rev	view			
	2.1	Conformal Co	orrelation Functions	13	
		2.1.1 Opera	tor-Product Expansion, Conformal Blocks and Bootstrap	p 15	
	2.2	The AdS/CF	T Correspondence	19	
	2.3	Holographic	Correlators and Witten Diagrams	23	
		2.3.1 Conta	ct Witten Diagrams	25	
		2.3.2 Highe	r-Derivative Corrections	28	
3	Tow	ards the Vir	casoro-Shapiro Amplitude in $\mathbf{AdS}_5 imes \mathbf{S}^5$	33	
	3.1	General Setu	p	36	
		3.1.1 Review	w: $\mathcal{N} = 4$ SYM/IIB Superstring Theory in AdS ₅ ×S ⁵	37	
		3.1.2 Half-H	3PS Correlators	43	
		3.1.3 Effect	ive Action	44	
		3.1.4 Gener	ralised Contact Witten Diagrams	47	
		3.1.5 AdS×	S Contact Diagrams in Mellin Space	51	

	3.2	α'^3 Corrections	52		
	3.3	Algorithm for Computing General α' Corrections	54		
		3.3.1 Generalised Witten Diagrams	54		
		3.3.2 Mellin Space	60		
	3.4	α'^5 Corrections	61		
	3.5	α'^6 Corrections	64		
	3.6	α'^7 Corrections	68		
	3.7	Conclusions and Future Directions	71		
4	AdS	$S_7 \times S^4$: Recursion Relations for Anomalous Dimensions in the 6d			
	(2,0) Theory				
	4.1	Review: 6d (2,0) Theory/M-Theory in $AdS_7 \times S^4$	79		
		4.1.1 11d Supergravity	80		
		4.1.2 Holographic Correlators	81		
	4.2	Toy Model	82		
		4.2.1 Recursion Relations	84		
		4.2.2 Solutions	88		
	4.3	(2,0) Theory	93		
		4.3.1 Recursion Relations	95		
		4.3.2 Solutions	98		
	4.4	Conclusions and Future Directions	.03		
5	Ads	$S_2 \times S^2$ Correlators: Effective Action and 4d Conformal Sym-			

-	, v	
metry		107
5.1 Review: A Hidden Ten-Dimensional Conformal Symmetry .		110
5.1.1 General Considerations and Higher-Derivative Correction	ons	113

5.2	Genera	al Setup	•	•		•	117
	5.2.1	Correlators in 1d and the Differential Operator $\Delta^{(2)}$.	•		•		118
	5.2.2	Conformal Blocks	•			•	122
	5.2.3	4d Scalar Effective Action	•				123
5.3	Free T	Theory	•				125
	5.3.1	4d Conformal Symmetry	•				126
5.4	Superg	gravity	•				130
	5.4.1	Lowest-Charge Supergravity Correlator	•				131
	5.4.2	4d Conformal Symmetry	•		•	•	134
	5.4.3	Effective Action	•			•	141
5.5	Confo	rmal Block Analysis and Double-Trace Spectrum	•				143
	5.5.1	Solving the Mixing Problem	•			•	150
5.6	Unmix	king Supergravity	•			•	154
	5.6.1	p = 0 Sector	•				154
	5.6.2	p = 1 Sector	•				158
	5.6.3	Anomalous Dimensions after Unmixing					159
5.7	Higher	r-Derivative Corrections					160
	5.7.1	Lowest-Charge Higher-Derivative Corrections	•			•	161
	5.7.2	Effective Action	•			•	163
	5.7.3	Breaking of 4d Conformal Symmetry	•			•	165
5.8	Unmix	xing Four-Derivative Corrections	•			•	170
	5.8.1	Unmixing for Odd t	•				171
	5.8.2	Unmixing for Even t	•				179
	5.8.3	Anomalous Dimensions after Unmixing	•			•	183
5.9	Conclu	usions and Future Directions				•	185

6	Conclusion	191
A	Mellin Space	195
в	The Polynomial $I(X_i, Y_i)$	199
С	Contact Diagrams in AdS×S and AdS, and Supergravity	201
D	α'^7 Ambiguities in AdS ₅ ×S ⁵	205
E	Conformal Blocks in 6d	211
F	Orthogonality of Hypergeometrics	213
G	Tree-Level Supergravity in the 6d $(2,0)$ Theory	215
н	Quadratic Super Casimir and Correlator of Descendants	219
Ι	Further Results for Unmixing of Four-Derivative Corrections	225
Bibliography		

Chapter 1

Introduction

Correlation functions are amongst the most fundamental observables in quantum field theories (QFTs). They characterise the theory and once all correlation functions of gauge invariant operators are known, the theory is considered solved. Conformal correlators, correlation functions in conformal field theories (CFTs), are especially interesting. In addition to Poincaré invariance, conformal correlators are also invariant under scaling and special conformal transformations. Therefore, these observables are strongly constrained by symmetries and show interesting structures.

In particular, conformal correlators can be studied through the AdS/CFT correspondence [1]. This is a remarkable duality between certain conformal field theories and quantum gravity in a specific curved background. The starting point is a stack of N D- or M-branes and in the low-energy limit the worldvolume theory describing this stack of branes is a conformal field theory. On the other hand, the stack of branes curves the spacetime and in the near-horizon limit the resulting geometry is Anti-de Sitter-space times a sphere (AdS×S) where the stack of branes is in the boundary of AdS. Therefore, string theory or M-theory in AdS×S is dual to the CFT in the boundary. This correspondence is very powerful as it relates quantum gravity in AdS to a non-gravitational theory in the boundary, and is therefore holographic. Gravity amplitudes in the so-called bulk are dual to stress tensor correlators in the boundary CFT. Studying these so-called holographic correlators can be of great

interest from the point of view of both sides of the duality. In theories with enough supersymmetry, which will be the focus of this thesis, stress tensor correlators can be related to certain scalar operators using supersymmetry. This is technically advantageous since correlators of scalars are much simpler than those of tensors. In this thesis, we will investigate different aspects of holographic correlators in three different examples of the AdS/CFT correspondence. In all of the considered examples the boundary theory is a supersymmetric CFT (SCFT) and in particular, we study four-point functions of half-BPS operators. These are operators in special representations of the superconformal group which are annihilated by half of the supersymmetry generators.

In the low-energy limit, quantum gravity can be described by an effective action, where the leading contribution is Einstein gravity coupled to matter. The matter and couplings can be chosen to give supergravity, which generally arises from the low-energy limit of string theory or M-theory. The subleading corrections take the form of interaction terms with derivatives and we will refer to them as higher-derivative corrections. In this thesis, we will study these effective actions and in particular the higher-derivative corrections in three different examples of the AdS/CFT correspondence. In the original AdS/CFT paper [1], three canonical examples of the correspondence are considered:

- CFT on D3-branes \leftrightarrow type IIB string theory in $\mathrm{AdS}_5{\times}\mathrm{S}^5$,
- CFT on M2-branes \leftrightarrow M-theory in AdS₄×S⁷,
- CFT on M5-branes \leftrightarrow M-theory in $AdS_7 \times S^4$.

In each of these dualities, the number of branes on the CFT side is N and moreover, there are N units of flux through the sphere on the gravity side. The first canonical example is the most studied and best understood of the three. It is a duality between type IIB superstring theory in $AdS_5 \times S^5$ and maximally supersymmetric Yang-Mills theory in four dimensions ($\mathcal{N} = 4$ SYM). We will consider holographic correlators in this duality in chapter 3. In particular we consider a 10d effective action describing higher-derivative corrections to the supergravity approximation arising from tree-level string theory.

The second canonical example is a duality between the worldvolume theory of N M2-branes and M-theory in $AdS_4 \times S^7$. M-theory is not well understood yet, but in the low-energy limit it can be described by 11d supergravity [2]. Its fundamental degrees of freedom are 2d and 5d objects, the M2- and M5-branes. The M2-brane worldvolume theory is understood by the ABJM theory, a 3d CFT that is dual to M-theory in $AdS_4 \times S^7$ [3], while the worldvolume theory of M5-branes is still mysterious. It is expected to be a 6d conformal field theory with (2,0) supersymmetry, the 6d (2,0) theory. We study aspects of holographic correlators in $AdS_7 \times S^4$ in chapter 4, which is the third of the canonical examples of the AdS/CFT correspondence. In particular we focus on higher-derivative corrections to the low-energy effective action of M-theory. This duality is less well understood than the previous examples since the boundary theory has no Lagrangian description and the bulk theory is not known beyond the supergravity approximation.

Finally, in chapter 5 we will apply the knowledge and techniques from previous considerations to the study of holographic correlators in $AdS_2 \times S^2$ which is expected to be dual to a one-dimensional SCFT in the boundary. We use superconformal symmetry, crossing and higher-dimensional symmetries to reconstruct tree-level supergravity and higher-derivative corrections which describe the low-energy limit of any theory of quantum gravity in this background. This duality is less well understood but of great interest because $AdS_2 \times S^2$ is the near-horizon geometry of extremal black holes in four dimensions [4, 5] and thus by studying holographic correlators in this background one can hope to gain insight that could be adapted to real world physics.

Note that in each of these examples we restrict to tree-level correlators, therefore we suppress all loop corrections by taking the Newton constant $G_N \to 0$. Trough the AdS/CFT correspondence $G_N \sim 1/c$ where c is the central charge of the boundary

CFT. The central charge is proportional to positive powers of N depending on the CFT, where N is the number of branes, therefore taking $N \rightarrow \infty$ suppresses all loop corrections. We will now briefly motivate the main ideas of each of the three main chapters, followed by a description of the structure of this thesis. More detailed introductions will then be given at the beginning of each chapter.

Holographic correlators in $AdS_5 \times S^5$

First, we study holographic correlators in $AdS_5 \times S^5$ which is the most studied canonical example of the AdS/CFT correspondence. Half-BPS correlators in \mathcal{N} = 4 SYM are dual to type IIB string theory scattering amplitudes in $AdS_5 \times S^5$. In flat space, the four-point tree-level amplitude of closed string theory takes a very compact form known as the Virasoro-Shapiro (VS) amplitude [6, 8]. This formula encodes many essential properties of string theory such as a Regge trajectory describing massive states with arbitrarily high spin, and exponential suppression at high-energy which was one of the earliest indications that string theory could be a promising candidate for quantum gravity. Given that the effects of quantum gravity are expected to become most important in curved backgrounds like the interior of black holes and the early Universe, it is therefore very important to understand how to generalise the VS amplitude beyond the flat space limit. At present it is technically challenging to calculate string amplitudes in curved backgrounds from first principles, but progress can be made in AdS backgrounds using holographic methods. In the limit $\alpha' \rightarrow 0$, where α' is related to the square of the string length, string theory in AdS₅×S⁵ can be approximated by supergravity. The subleading terms describe string corrections and take the form of higher-derivative interactions. We can also write the flat space VS amplitude as an infinite series in α' , where the leading term will describe supergravity while higher-order terms describe string corrections. These corrections

¹The Virasoro amplitude is the amplitude for four tachyonic scalars in bosonic string theory found by Virasoro [6] and generalised to n points by Shapiro [7]. The tree-level four-point amplitude in IIB string theory [8] which we will be considering here is given by the Virasoro amplitude multiplied by a kinematic factor and it has become the convention to refer to it as the Virasoro-Shapiro amplitude.

can be derived at all orders in α' from a simple effective field theory consisting of a scalar field with quartic interactions where the coefficients are fixed by comparing to the VS amplitude.

The goal is to generalise this to curved spacetime and remarkably, we will find that the interacting part of all single-trace half-BPS correlators in \mathcal{N} = 4 SYM can be obtained from a similar scalar effective action describing tree-level IIB string theory on $AdS_5 \times S^5$ (rather than a flat) background. The key idea that allows us to derive correlation functions from the 10d effective field theory is the use of a natural generalisation of contact Witten diagrams [9] (which are integrals over AdS space) to integrals over the full AdS×S space, treating AdS and S on equal footing. These manifestly 10d contact diagrams generate all four-point half-BPS correlators described by tree-level string theory, corresponding to string corrections at any order in α' . Note that we do not prove the existence of this 10d effective field theory in $AdS_5 \times S^5$. We propose its existence and derive the four-point correlators of singletrace half-BPS operators at different orders in α' . We show that it reproduces the known results for α'^3 and α'^5 corrections, which were previously obtained in [10– 15] using constraints imposed by superconformal and crossing symmetry as well as simplifications of the spectrum predicted by AdS/CFT. We also present a general algorithm for extending these predictions to arbitrarily high order in α' and use it to obtain new predictions at α'^6 and α'^7 . At the same time as we completed our work, the authors of [16] also obtained higher-order α' corrections in AdS₅×S⁵, which nicely complements our results.

Holographic correlators in $AdS_7 \times S^4$

Next, we consider the correspondence between the M5-brane worldvolume theory, the 6d (2,0) theory, and M-theory in $AdS_7 \times S^4$. Correlators in the 6d (2,0) theory are dual to M-theory amplitudes in $AdS_7 \times S^4$. The formulation of M-theory is one of the most important open questions in string theory and it is not well understood yet. It is an 11-dimensional theory of quantum gravity which, as mentioned before, is approximated by 11d supergravity in the low-energy limit. It arises in the strong coupling limit of type IIA string theory, where the size of the 11th dimension is proportional to the string coupling [17, 18]. Therefore, there is no tunable coupling constant and the only tunable parameter is the number of branes N. Further, there are no strings but its fundamental degrees of freedom are believed to be 2d and 5d objects, the M2- and M5-branes. Since there is no string length in this theory, the only length scale is the Planck length l_P which is related to the Newton constant as $G_N \sim l_P^9$. And because of the AdS/CFT identification $G_N \sim 1/c$, where $c \sim N^3$ is the central charge, the supergravity approximation $N \rightarrow \infty$ is also like taking $l_P \rightarrow 0$ which is like a low-energy limit. Moreover, this limit also implies $G_N \rightarrow 0$ which suppresses all loop corrections. Therefore, both, higherderivative corrections and loop corrections are suppressed by $N \to \infty$ and the leading contribution is tree-level supergravity. Understanding the worldvolume theory of a stack of M5-branes is an important open question in string theory and very little is known about it since it is intrinsically strongly coupled and it is believed to have no Lagrangian description in six dimensions. In [19] a 5d Lagrangian was proposed which is believed to describe the full 6d physics of the theory. Furthermore, the 6d (2,0)theory is also important because dimensional reduction along various manifolds gives various lower-dimensional theories, such as $\mathcal{N} = 4$ SYM. It also provides a geometric interpretation for their dualities, like S-duality in $\mathcal{N} = 4$ SYM [20]. A lot of progress in understanding the 6d (2,0) theory has been made by dimensionally reducing the theory or computing quantities protected by supersymmetry, but ultimately one wants to compute unprotected quantities in six dimensions.

A promising strategy to study the 6d (2,0) theory is the conformal bootstrap, where we try to use principle properties of the theory such as superconformal and crossing symmetry to constrain the correlators. This program was originally proposed in [21– 23] and brought back more than 30 years later in [24]. This approach was first applied to the 6d (2,0) theory in [25]. Our goal is to study the 6d (2,0) theory away from the strict large-N (or large central charge c) limit and thus study M-theory beyond the supergravity approximation. Important recent progress in this research area has been made in [26–28]. Moreover, the M-theory effective action can also be deduced from correlators of the ABJM theory [3], which is dual to M-theory in $AdS_4 \times S^7$ [27, 29].

We analyse four-point correlation functions of stress tensor multiplets in the 6d (2,0) theory in the large-*c* limit. The main aim is to derive the higher-derivative corrections to the effective action using conformal bootstrap methods without requiring knowledge of the explicit form of the correlators, which gives a very direct way of deriving higher-derivative corrections. We follow the strategy described in [30]. Starting from a 1/c expansion of the crossing equations, we derive recursion relations for the anomalous dimensions (1/c corrections to the scaling dimensions) of the operators in the conformal block expansion of the correlators. These anomalous dimensions encode the higher-derivative corrections to the supergravity effective action arising from M-theory.

Holographic correlators in $AdS_2 \times S^2$

Finally, we investigate holographic correlators in $AdS_2 \times S^2$ where the superconformal group of the boundary theory is SU(1,1|2). Hence, the 1d boundary SCFT is quarter-maximal. We will apply techniques from other examples of the AdS/CFT correspondence to this less well studied case. $AdS_2 \times S^2$ is of great interest because it is the near-horizon geometry of extremal black holes in four dimensions as mentioned before. Hence, understanding holographic correlators in this background can help to develop a description of quantum gravity in the real world. The AdS_2/CFT_1 correspondence has been studied e.g. in [31, 32]. More recently there has been a lot of interest in AdS_2/CFT_1 due to the relation between the 2d Jackiw-Teitelboim (JT) gravity [33, 34] and the 1d Sachdev-Ye-Kitaev (SYK) model [35, 36], e.g. in [37, 38], see [39] for a review. The relation to black holes in nature has been discussed in [40]. Another motivation to study holographic correlators in $AdS_2 \times S^2$ is that they are in many ways simpler than higher-dimensional analogues and hence, $AdS_2 \times S^2$ can act as a toy model for various aspects of holography.

It was recently observed in [41] that four-point tree-level correlators in $AdS_5 \times S^5$ exhibit a ten-dimensional conformal symmetry. The conjecture is that free theory and supergravity correlators can be obtained from a single object invariant under 10d conformal symmetry. This is a consequence of the fact that $AdS_5 \times S^5$ is conformally flat and further that the four-point tree-level supergravity amplitude is scale invariant and can therefore be transformed to flat space by conformal transformation. We aim to understand this higher-dimensional conformal symmetry more systematically and will start by showing that free theory and supergravity holographic correlators in $AdS_2 \times S^2$ exhibit 4d conformal symmetry. Furthermore we investigate higherderivative-corrections. For both, supergravity and higher-derivative corrections, we perform conformal block analyses of four-point half-BPS correlators. Generally, there are many different operators contributing to the same conformal block, i.e. there are many operators with the same scaling dimension and R-symmetry charge contributing to the spectrum. Thus the conformal block coefficients, and in particular the anomalous dimensions of exchanged operators, are degenerate and when unmixing these contributions one lifts the degeneracy which is called solving the mixing problem. After unmixing, the anomalous dimensions of double-trace operators in the spectrum exhibit a simple structure which can be interpreted in terms of the 4d conformal symmetry. The higher-dimensional conformal symmetry is generally broken for higher-derivative corrections, as was observed in [14] in the context of unmixing anomalous dimensions. We show that nevertheless an infinite set of specific correlators can be constructed from it.

On the other hand, we obtain all higher-derivative corrections from a 4d scalar effective action analogous to the one proposed above for $AdS_5 \times S^5$. We assume the existence of this effective action which, in the $AdS_2 \times S^2$ case, describes supergravity as well as higher-derivative corrections and justify it by comparing the results to ones obtained from other methods like the 4d conformal symmetry. These two approaches are complementary, as the 4d conformal symmetry describes all free

theory and supergravity tree-level half-BPS four-point correlators but not general higher-derivative corrections, while the 4d effective action generates all supergravity correlators and all higher-derivative corrections but does not describe free theory. Note that the effective action we found for $AdS_5 \times S^5$ describes higher-derivative corrections but not supergravity. In contrast, the effective action proposed for $AdS_2 \times S^2$ treats supergravity and higher-derivative corrections on equal footing.

Structure of this thesis

We begin with a review of important concepts in chapter 2, starting with a review of conformal correlators and the conformal bootstrap in section 2.1. Next, we will briefly review the AdS/CFT correspondence in section 2.2 followed by a discussion of holographic correlators in section 2.3. Specifically we review the evaluation of correlators from Witten diagrams in AdS and discuss higher-derivative corrections to the supergravity approximation. There is one appendix related to this chapter, appendix A, where we review some important concepts in Mellin space, in particular contact Witten diagrams.

The rest of this thesis consists of three chapters, each of which focusing on one of the examples of the AdS/CFT correspondence introduced above. In general, each of the chapters can be read independently, however some sections of chapter 5 make use of material presented in chapter 3. Where this is the case we will clearly refer to the relevant sections.

Chapter 3 is based on [42] and is organised as follows. In section 3.1 we provide an overview of the general strategy including a general discussion of the effective action and define generalised contact diagrams in AdS×S as well as their Mellin transforms. In section 3.2 we use these techniques to compute the leading correction to half-BPS correlators which occurs at α'^3 . In section 3.3 we develop an algorithm for extending these calculations to arbitrary order in α' . Using this algorithm, we reproduce previous results at α'^5 in section 3.4, and obtain new predictions at α'^6 and α'^7 in sections 3.5 and 3.6, respectively. We present conclusions and future directions in section 3.7. There are also several appendices related to this chapter. In appendix B, we present more details about the parametrisation of half-BPS correlators. In appendix C we discuss a relation between contact diagrams in AdS×S and AdS and present its implications for the tree-level supergravity prediction and in appendix D we list further results at order α'^7 .

Chapter 4 is based on [43] and its structure is described in the following. To start with, we review some important concepts, including M-theory and 11d supergravity, in section 4.1. In section 4.2 we derive recursion relations for anomalous dimensions in a toy 6d model and match the solutions against the conformal block expansion of Witten diagrams in AdS_7 . In section 4.3, we then adapt this analysis to the 6d (2,0) theory, and match the solutions of the supersymmetric recursion relations with the results obtained in [26]. In section 4.4 we present our conclusions and future directions. There are also several appendices. In appendix E, we provide formulas for the conformal blocks in terms of hypergeometric functions and in appendix F we derive inner products for these functions. Furthermore, in appendix G we perform the conformal block analysis of the supergravity solution.

Chapter 5 is based on [44], which at the time of submission of this thesis is in preparation for publication. We start by reviewing the ten-dimensional hidden conformal symmetry of $AdS_5 \times S^5$ in section 5.1 before going on to introduce the formalism for 1d superconformal correlators studied throughout this chapter in section 5.2. We will review the half-BPS correlators considered, the conformal blocks and conformal Casimirs relevant for the subsequent sections as well as propose a 4d scalar effective action from which we deduce the higher-derivative corrections to supergravity in $AdS_2 \times S^2$, similar to the one in chapter 3. Next, we discuss the free disconnected theory in the context of the higher-dimensional conformal symmetry in section 5.3. In section 5.4 we derive the supergravity correlator of lowest charge and then obtain all higher-charge correlators from the 4d conformal symmetry as well as from the 4d effective action point of view. In section 5.5 we discuss the double-trace spectrum of the conformal block decomposition of half-BPS correlators and then explain how to solve the mixing problem at different orders in the central charge and for higher-derivative corrections. In section 5.6 we solve the mixing problem for supergravity. Finally, we study higher-derivative corrections in section 5.7, where we derive general higher-derivative corrections from crossing symmetry before obtaining predictions for the correlators from the effective action approach. We also discuss the breaking of the four-dimensional conformal symmetry. In section 5.8 we solve the mixing problem for the four-derivative corrections, we can then analyse the anomalous dimensions in terms of a 4d effective spin. There are two appendices for this chapter. In appendix H we describe the derivation of the quadratic super Casimir of SU(1,1|2), which plays an important role in the 4d conformal symmetry, and will compute the correlator of descendants of the superconformal primaries. Finally, in appendix I we present further result from unmixing of four-derivative corrections.

Chapter 2

Review

In this chapter we review several concepts which play an important role throughout this thesis where the focus lies on four-point correlation functions. We start by reviewing conformal correlators, the operator-product expansion and the conformal bootstrap. Subsequently, we briefly review the AdS/CFT correspondence and then discuss holographic correlators including higher-derivative corrections, in particular their evaluation from Witten diagrams in AdS.

2.1 Conformal Correlation Functions

Firstly, let us review correlation functions in conformal field theories in more than two dimensions, for a more detailed review see e.g. [45]. For simplicity, we will consider scalar operators ϕ_i with scaling dimension Δ_i in what follows. The twopoint functions are completely fixed by the symmetries of the theory, i.e. Poincaré invariance, scaling invariance and invariance under special conformal transformations. They are given by

$$\langle \phi_i(x)\phi_j(y)\rangle = \frac{\delta_{ij}}{(x-y)^{2\Delta}},$$
 (2.1.1)

if ϕ_i and ϕ_j have the same scaling dimension Δ and zero otherwise. Similarly, three-point functions can be fixed by symmetries up to a constant as follows

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\rangle = \frac{\lambda_{123}}{|x_{12}|^{2\alpha_{123}}|x_{13}|^{2\alpha_{132}}|x_{23}|^{2\alpha_{231}}} ,$$

$$(2.1.2)$$

$$x_j \text{ and } \alpha_{ijk} = \frac{\Delta_i + \Delta_j - \Delta_k}{2}.$$

The four-point functions can no longer be fixed by symmetries but they can be significantly reduced to a function of only two conformally invariant variables u and v. Let us consider correlators of four identical operators for simplicity, the four-point functions are then given by

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\phi_4(x_4)\rangle = \frac{f(u,v)}{x_{12}^{2\Delta}x_{34}^{2\Delta}}, \qquad (2.1.3)$$

where u and v are conformal cross-ratios defined as

$$u = z\bar{z} = \frac{x_{12}^2 x_{34}^2}{x_{13}^2 x_{24}^2} , \qquad v = (1-z)(1-\bar{z}) = \frac{x_{14}^2 x_{23}^2}{x_{13}^2 x_{24}^2} . \tag{2.1.4}$$

It is useful to introduce the parameters (z, \bar{z}) and we will use (u, v) and (z, \bar{z}) interchangeably in later chapters. Note that in one-dimensional conformal field theories, which we discuss in chapter 5, the cross-ratios are not independent and reduce to

$$u_{\bar{z} \to z} = z^2 , \qquad v_{\bar{z} \to z} = (1 - z)^2 , \qquad (2.1.5)$$

see section 5.2.1 for more details. The rest of the discussion in this section is still valid for 1d CFTs if one keeps in mind that the cross-ratios are not independent and can be written in terms of a single variable z.

Importantly, the four-point function (2.1.3) is crossing symmetric, i.e. it is invariant under transformations of the four-point crossing symmetry group S_3 which is generated by

$$(u,v) \rightarrow (v,u)$$
, $(u,v) \rightarrow \left(\frac{u}{v}, \frac{1}{v}\right)$. (2.1.6)

Note that this follows from the fact that we consider four identical operators. As

where $x_{ij} = x_i - x_i$

a consequence also f(u, v) has to satisfy a specific crossing constraint. This can easily be seen by considering (2.1.3) which groups the external points (12) and (34) together. However, this choice is not unique and one can just as well group together (14) and (23). From (2.1.4) it is obvious that the exchange of $2 \leftrightarrow 4$ corresponds to $(u \leftrightarrow v)$ and thus

$$\frac{1}{x_{12}^{2\Delta}x_{34}^{2\Delta}}f(u,v) = \frac{1}{x_{14}^{2\Delta}x_{23}^{2\Delta}}f(v,u) .$$
 (2.1.7)

Multiplying this by $x_{14}^{2\Delta} x_{23}^{2\Delta}$ yields the following crossing equation for f(u, v):

$$\left(\frac{v}{u}\right)^{\Delta} f(u,v) = f(v,u) , \qquad (2.1.8)$$

which will play an important role in the following.

2.1.1 Operator-Product Expansion, Conformal Blocks and Bootstrap

An important concept in conformal field theories is the *operator-product expansion* (OPE). First, recall that primary operators are the lowest-weight operators in a representation of the conformal algebra, they are local operators annihilated by all generators of special conformal transformations and acting with translation generators on them gives the so-called descendants (derivatives of primaries)¹. A product of two primary operators can be expanded as a sum of all the primaries and descendants in the theory as follows:

$$\phi_1(x)\phi_2(0)|0\rangle = \sum_{\mathcal{O}} \lambda_{12\mathcal{O}} C_{\mathcal{O}}(x,\partial_y) \mathcal{O}(y)|_{y=0}|0\rangle , \qquad (2.1.9)$$

where we sum over all primary operators of the CFT. The coefficients $\lambda_{12\mathcal{O}}$ are called OPE coefficients and are the same as the constants λ_{123} in the three-point function (2.1.2). Furthermore, $C_{\mathcal{O}}(x, \partial_y)$ are polynomials of partial derivatives with

¹In fact we will study superconformal theories in this thesis, hence the operators we consider are superconformal primaries which in addition to the conformal symmetries are also annihilated by half of the supersymmetry generators.

respect to y acting on the primaries, which will generate all the descendants in the theory. Note that one can perform an OPE in any quantum field theory, but in general (2.1.9) only holds in the limit where the operators are close together, i.e. when $x \to 0$ whereas for a CFT the statement is much stronger. Since CFTs are scale-invariant we do not have a notion of distance and the OPE is valid for finite x.

Conformal field theories can be defined by its so-called *CFT data* which consists of a list of scaling dimensions Δ_i of all local primaries of the theory together with all OPE coefficients λ_{ijk} for any three primaries. Once one knows the whole CFT data one has solved the CFT in question. In principle we can construct all *n*-point correlation functions from the three-point functions, which are completely fixed up to coefficients, using the OPE expansion. In practice this is very difficult to realise and this is why we have to use other methods to constrain the CFT data. One such method is the *conformal bootstrap* which uses fundamental properties of the CFT, such as crossing symmetry and conformal invariance, to constrain the CFT data and fix the correlation functions.

The starting point is the four-point function, where we again consider four identical fields, written in terms of OPEs of two pairs of primaries as

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\phi_4(x_4)\rangle = \sum_{\mathcal{O}} \lambda_{12\mathcal{O}} \lambda_{34\mathcal{O}} [C_{\mathcal{O}}(x_{12},\partial_y) C_{\mathcal{O}}(x_{34},\partial_z) \langle \mathcal{O}(y) \mathcal{O}(z)\rangle],$$
(2.1.10)

where $y = \frac{x_1+x_2}{2}$, $z = \frac{x_3+x_4}{2}$ and the functions in the square brackets are completely fixed by conformal symmetry. Since the form of the four-point function is known to be $\frac{f(u,v)}{x_{12}^{2\Delta}x_{34}^{2\Delta}}$, it follows that the right hand side of (2.1.10) has to have the same transformation properties which yields

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\phi_4(x_4)\rangle = \sum_{\mathcal{O}} \lambda_{12\mathcal{O}} \lambda_{34\mathcal{O}} \frac{G_{\mathcal{O}}(u,v)}{x_{12}^{2\Delta} x_{34}^{2\Delta}} .$$
(2.1.11)

The conformally invariant functions $G_{\mathcal{O}}(u, v)$ are called *conformal blocks*. Decomposing a four-point function into conformal blocks, the blocks represent the contributions of the separate conformal primary operators in the theory together with their descendants, where the operators and their corresponding blocks are labelled by their scaling dimension and spin. To make this more precise, the conformal block expansion of a correlator of four operators with scaling dimensions Δ_i , where $i = 1, \ldots, 4$, can be written as

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\phi_4(x_4)\rangle = \mathcal{P}^{\Delta_i} \sum_{\Delta,l} A^{\Delta_i}_{\Delta,l} G^{\Delta_i}_{\Delta,l}(u,v) , \qquad (2.1.12)$$

where $G_{\Delta,l}^{\Delta_i}(u, v)$ are the conformal blocks describing the contributions of operators in the theory belonging to a multiplet whose conformal primary has scaling dimension Δ and spin l. The coefficients $A_{\Delta,l}^{\Delta_i}$ are squares of OPE coefficients and \mathcal{P}^{Δ_i} is a prefactor depending on the spacetime coordinates. The labels Δ_i indicate a dependence on the scaling dimensions of the four external operators. As an example of operators contributing to the conformal block expansion, consider double-trace operators which will be studied in detail in later chapters. These operators are constructed from two half-BPS operators with dimensions p, q and have the schematic form $\mathcal{O}_p \partial^l \square^n \mathcal{O}_q$. Their classical scaling dimension is p+q+2n+l, where the spin l counts the number of partial derivatives and n counts the number of boxes acting on one of the half-BPS operators. It is also common to refer to the twist of the exchanged operator which is defined as its scaling dimension minus its spin. Double-trace operators of the above form have twist p+q+2n.

Conformal blocks have been computed explicitly for many theories. For four-point correlators of scalar operators of arbitrary scaling dimensions in any even dimension they were derived in [46]. The authors obtained the conformal blocks by solving for the eigenfunctions of the Casimir operator of the conformal group SO(d, 2) in d dimensions. The blocks for d = 4 and d = 6 can be written in terms of hypergeometric functions in a simple way (see e.g. section 5.4.2 for 4d blocks and appendix E for blocks in 6d), whereas the blocks for general even dimensions are given as an infinite sum of so-called Jack polynomials. Superconformal blocks in six dimensions were given in [25, 47]. Conformal blocks for odd dimensions were obtained in [48]. Finally, the superconformal blocks relevant for 1d supersymmetric CFTs which we study in

chapter 5 were derived in [49] (see section 5.2.2).

Once the blocks are known, this allows us to perform a so-called conformal block analysis of the theory. Decomposing the four-point correlators into conformal blocks gives information about the spectrum of operators in the theory, their scaling dimensions and spins. We will perform conformal block analyses of correlators in 1d CFTs in chapter 5 where we compute three-point coefficients and anomalous dimensions of operators in the double-trace spectrum of the theory in the context of a higher-dimensional conformal symmetry as mentioned in the introduction. We also perform a conformal block analysis of correlators in a 6d SCFT in chapter 4 where we compute anomalous dimensions which encode information about higherderivative corrections to the bulk low-energy effective action. However, the main focus of chapter 4 is to compute anomalous dimensions of double-trace operators in the spectrum of half-BPS correlators without requiring knowledge of the explicit form of the four-point functions but rather deriving a recursion relation for anomalous dimensions using the conformal bootstrap, which we describe in the following.

Let us start by writing the above equation (2.1.11) diagrammatically:

$$\langle \phi_1(x_1)\phi_2(x_2)\phi_3(x_3)\phi_4(x_4)\rangle = \sum_{\mathcal{O}} \lambda_{12\mathcal{O}}\lambda_{34\mathcal{O}}$$

Notice that in the double OPE of the four-point function we chose the OPE channel (12)(34), we could as well have chosen the channel (14)(23) and the result must be the same. This leads to the conformal bootstrap or crossing symmetry condition

$$\sum_{\mathcal{O}} \lambda_{12\mathcal{O}} \,\lambda_{34\mathcal{O}} \frac{G_{\mathcal{O}}(u,v)}{x_{12}^{2\Delta} x_{34}^{2\Delta}} = \sum_{\mathcal{O}'} \lambda_{14\mathcal{O}'} \,\lambda_{23\mathcal{O}'} \frac{G_{\mathcal{O}'}(v,u)}{x_{14}^{2\Delta} x_{23}^{2\Delta}} , \qquad (2.1.13)$$

or diagrammatically:



After imposing these conditions on all four-point functions of a theory no more conditions will arise from higher-point functions. The bootstrap conditions (also called OPE associativity or crossing equations) are used to classify CFTs using the following statement.

A CFT is a set of CFT data which satisfies the crossing equations for all four-point functions [21, 23].

To summarise, crossing and (super)conformal symmetry impose powerful constraints on four-point correlators which can be used to determine CFT data analytically or numerically. When a holographic dual exists, the CFT data can be constrained even more by combining the conformal bootstrap with knowledge about the holographic dual.

2.2 The AdS/CFT Correspondence

In this section, we briefly review the AdS/CFT correspondence [1, 9, 50]. For a detailed review see e.g. [51, 52].

Through the AdS/CFT correspondence quantum gravity in AdS is described by a CFT in the boundary. Considering a stack of D- or M-branes, in the low-energy limit, the worldvolume quantum field theory describing the stack of branes is a CFT. Besides, the stack of branes warps the surrounding spacetime and in the near-horizon limit the geometry of this curved space is $AdS_p \times S^q$ and the stack of branes is located in the boundary of AdS. Hence, the boundary CFT is dual to string or M-theory

in $\operatorname{AdS}_p \times \operatorname{Sq}$, which is a very remarkable duality between a theory of gravity and a quantum field theory without gravity in different dimensions. It is conjectured that the two theories in the duality are equivalent to each other, i.e. its operator observables, states, correlation functions as well as the full dynamics of the theories are equivalent. This means that each of these objects can be computed from two completely different calculations and both will lead to the same result. Because the correspondence is between a theory in AdS×S and a CFT on the boundary of AdS, it is also called a holographic duality. The stack of N branes creates a flux through the sphere, thus integrating the flux and using Gauss' law gives the number of branes. When the stack is coincident and in flat space, the gauge group of the boundary theory is expected to be SU(N) in the case of $\mathcal{N} = 4$ SYM.

In Maldacena's original paper [1] there are three canonical examples, as we have seen in the introduction. The most well studied duality is between type IIB string theory in $AdS_5 \times S^5$ and $\mathcal{N} = 4$ SYM in the boundary which we study in chapter 3. Another canonical example is the focus of chapter 4, the correspondence between the worldvolume theory of N M5-branes and M-theory in $AdS_7 \times S^4$. The third canonical example is a duality between M-theory in $AdS_4 \times S^7$ and the worldvolume theory of N M2-branes which can be described by a 3d CFT, the ABJM theory [3]. In chapter 5 we focus on quantum gravity in $AdS_2 \times S^2$ which is expected to be dual to a 1d SCFT with superconformal symmetry group SU(1, 1|2).

Let us have a closer look at the two canonical examples we study in later chapters and their parameters, starting with $\mathcal{N} = 4$ SYM and string theory in AdS₅×S⁵. The parameters of the two theories are identified with one another as

$$g_{\rm YM}^2 = 4\pi g_s , \qquad R^4 = 4\pi g_s N(\alpha')^2 , \qquad (2.2.1)$$

where $g_{\rm YM}$ is the Yang-Mills coupling, g_s is the closed string coupling, R is the curvature radius of the AdS₅ space and the S⁵ sphere and α' is related to the string

length l_s as $\alpha' = l_s^2$. With the 't Hooft coupling $\lambda = g_{YM}^2 N$ this yields

$$R = \lambda^{1/4} l_s , \qquad (2.2.2)$$

which relates R/l_s , the AdS and S radius in units of string length, on the bulk side of the duality to the 't Hooft coupling of the boundary CFT on the other side. This makes the duality very powerful since it relates a weakly coupled field theory which can be studied perturbatively to a string theory in a strongly curved background which makes computations very difficult. On the other hand, when the string theory is in a weakly curved background and thus calculations are simpler because it can be approximated by supergravity, the boundary CFT is strongly coupled. Thus, problems that are intractable on the one side of the correspondence can be much simpler to compute on the other side. Note that even when both sides of the duality are non-perturbative, as is the case in the M5-brane case (see chapter 4), this duality is very useful as for example in the CFT we can use conformal bootstrap methods to study correlation functions which in turn gives insight into the dual quantum gravity theory where computations are very difficult.

The second relation in (2.2.1) is derived by constructing extremal black D3-brane solutions, where the near-horizon geometry of N coincident D3-branes is $AdS_5 \times S^5$. A similar relation for M5-branes is obtained by constructing extremal black M5-brane solutions:

$$r_5^3 = \pi N_5 l_P^3 \ . \tag{2.2.3}$$

The near-horizon geometry of N_5 coincident M5-branes is $AdS_7 \times S^4$, where the radius of the four-sphere is r_5 and the radius of AdS_7 is $2r_5$. Note that there are no strings in M-theory and thus no concept of string length, therefore M-theory is manifestly non-perturbative and has only one length scale, the Planck length l_P .

The AdS/CFT correspondence is complicated when considering the full quantum gravity theory in the bulk, however there are significant simplifications when considering certain limits. Taking the limit $N \to \infty$ is like taking the Newton constant $G_N \to 0$ and therefore this limit suppresses all loop corrections and restricts to tree-
level gravity. This can be seen from the AdS/CFT identification $G_N \sim 1/c$, where c is the central charge of the CFT which is proportional to positive powers of N depending on the theory. In all of the following chapters we will restrict to this limit and only consider tree-level correlators.

Note that in the case of $\mathcal{N} = 4$ SYM one can take $N \to \infty$ while keeping the 't Hooft coupling λ constant, this corresponds to a topological expansion in 1/N where the leading contribution is the planar limit. Furthermore, it restricts to tree-level string theory where quantum corrections described by string loops are suppressed since $g_s \to 0$. Additionally, one can consider the low-energy limit $\alpha' \to 0$ which corresponds to the strong-coupling limit of $\mathcal{N} = 4$ SYM, where $\lambda \to \infty$, and describes supergravity with all stringy corrections suppressed. When studying $\mathcal{N} = 4$ SYM away from the strict $\alpha' \to 0$ limit one adds stringy corrections to the supergravity approximation. Hence, a particularly simple limit of the duality is accessible in $\mathcal{N} = 4$ SYM, a duality between a strongly coupled planar CFT and weakly coupled classical supergravity.

In the case of M-theory, these limits are not available separately since there is only one length-scale, the Planck length l_P . As described before, taking $N \to \infty$ suppresses all loop corrections. Since the Newton constant is proportional to l_P , $G_N^{11d} \sim l_P^9$ in eleven dimensions, the $N \to \infty$ limit also corresponds to $l_P \to 0$ and is thus a low-energy limit. Further, M-theoretic corrections are also suppressed in the limit $l_P \to 0$ (or $N \to \infty$), since this is the only length scale available. Therefore, both, loop corrections and higher-derivative corrections are described by a large-Nexpansion and $N \to \infty$ corresponds to the tree-level supergravity approximation. In this thesis we will not discuss loop corrections but focus on tree-level supergravity and higher-derivative corrections arising from quantum gravity.

2.3 Holographic Correlators and Witten Diagrams

In each of the following chapters we consider four-point correlation functions of half-BPS operators, which are annihilated by half of the supercharges and thus form a short multiplet. On the quantum gravity side, in the large-N limit, these short multiplets correspond to supergravity multiplets plus an infinite tower of Kaluza-Klein (KK) excitations on the sphere. These excitations arise from dimensionally reducing the corresponding theory of quantum gravity on the sphere. Thus, in the 4d, 6d and 1d examples considered in the chapters below the half-BPS operators are dual to harmonics on the S⁵, S⁴ and S² sphere respectively. These spherical harmonics transform as totally symmetric and traceless rank-n tensors under SO(6), SO(5) and SO(3) transformations in the case of $AdS_5 \times S^5$, $AdS_7 \times S^4$ and $AdS_2 \times S^2$ respectively.

A supermultiplet can be characterised by its primary operator, which is annihilated by all generators of special conformal transformations and conformal supercharges. The descendants in the supermultiplet are obtained by acting on the primary with Poincaré supercharges. In the case of short multiplets, the primary operators, in addition to being annihilated by the conformal symmetries, are also annihilated by half of the supercharges. These primaries are called *chiral primary operators* (half-BPS operators). It will be useful to describe schematically how chiral primary operators are constructed in general superconformal theories. To construct a chiral primary operator, consider the trace of a product of n scalar operators

$$\mathcal{O}^{I_1 I_2 \dots I_n} = \operatorname{Tr}(\phi^{I_1} \phi^{I_2} \cdots \phi^{I_n}) , \qquad (2.3.1)$$

with R-symmetry indices I_i , where in supersymmetric theories R-symmetry transforms different supercharges into each other. Next, symmetrise all n indices in (2.3.1) and remove all traces to make a traceless symmetric tensor. Thus, consider operators of the form $\phi^{(I_1}\phi^{I_2}...\phi^{I_n})$, which are totally symmetric and it is understood that all traces are removed. More precisely, in the case of $\mathcal{N} = 4$ SYM the scalar operators ϕ^{I_i} are the six real scalars of $\mathcal{N} = 4$ SYM (see subsection 3.1.1) where the interacting fields can be written as perturbations around the free fields. Moreover, the trace in $\mathcal{N} = 4$ SYM is defined in terms of the gauge group SU(N). In the 6d (2,0) theory on the other hand, there is no gauge group and there is no expression for the interacting fields in terms of the free fields due to the lack of a small coupling constant, nevertheless these operators are expected to exist. Importantly, nowhere in chapter 4 is the definition of these operators required and it will be enough to know the superconformal blocks for this theory. Let us now go back to the general case.

The operator $\phi^{(I}\phi^{J)}$ is the superconformal primary of the stress tensor multiplet and, through the AdS/CFT correspondence, encodes the graviton in AdS. Traceless symmetric operators of the form $\phi^{(I_1}\phi^{I_2}...\phi^{I_n)}$ with more than two indices are highercharge operators which encode higher KK modes on the sphere. These operators are so-called single-trace operators and (2.3.1) forms a complete list of single-trace chiral primary operators for $n \leq N$. For n > N they can be related to so-called multi-trace operators which are products of single-trace operators. We will study correlation functions of single-trace chiral primary operators in the large-N limit in the following chapters.

Through the AdS/CFT correspondence, the dimensions Δ of any scalar operator on the conformal field theory side correspond to the mass of the dual bulk state on the gravity side. The relation between these quantities is given by

$$\Delta(\Delta - d) = m^2 R^2 , \qquad (2.3.2)$$

where m is the mass of the scalar field in the bulk, R is the AdS radius and d is the spacetime dimension of the CFT. Considering the masses of Kaluza-Klein excitations on the sphere and comparing to the dimensions of the chiral primary operators in the dual conformal field theory will satisfy this relation.

Finally, correlation functions of half-BPS operators correspond to scattering amp-

litudes of scalar fields in the bulk with mass m (2.3.2). Therefore, by studying AdS scattering amplitudes we can learn more about CFT correlation functions and vice versa [53–56]. As mentioned before, we are interested in studying the tree-level supergravity approximation of quantum gravity. This is described by an effective action whose leading contribution describes supergravity and the subleading corrections are contained in an infinite tower of higher-derivative terms. These corrections correspond to local interaction vertices in the tree-level low-energy effective action in AdS, where we focus on quartic interactions in this thesis. These quartic interactions can be computed from contact Witten diagrams in AdS and we will review them in the following subsections starting with interaction vertices without derivatives followed by a discussion of Witten diagrams for higher-derivative corrections.

2.3.1 Contact Witten Diagrams

Supergravity scattering in AdS can be computed from Witten diagrams. They are most conveniently expressed using embedding coordinates for AdS:

$$\hat{X}^{2} = -\left(\hat{X}^{-1}\right)^{2} - \left(\hat{X}^{0}\right)^{2} + \sum_{i=1}^{d} \left(\hat{X}^{i}\right)^{2} = -1 , \qquad (2.3.3)$$

where d is the spacetime dimension of the boundary CFT. In terms of these coordinates, covariant derivatives can be defined using projection tensors

$$\mathcal{P}_A^B = \delta_A^B + \hat{X}_A \hat{X}^B , \qquad (2.3.4)$$

which satisfy the useful identities

$$\mathcal{P}_A^B \hat{X}^A = 0 , \quad \mathcal{P}_A^B \mathcal{P}_B^C = \mathcal{P}_A^C . \tag{2.3.5}$$

In particular, the covariant derivative of a tensor is given by [57, 58]

$$\nabla_A \mathbf{T}_{\mathbf{A}_1\dots\mathbf{A}_N} = \mathcal{P}_A^{\mathbf{C}} \mathcal{P}_{\mathbf{A}_1}^{\mathbf{C}_1} \dots \mathcal{P}_{\mathbf{A}_N}^{\mathbf{C}_N} \frac{\partial}{\partial \hat{\mathbf{X}}^{\mathbf{C}}} \left(\mathcal{P}_{\mathbf{C}_1}^{\mathbf{E}_1} \dots \mathcal{P}_{\mathbf{C}_N}^{\mathbf{E}_N} \mathbf{T}_{\mathbf{E}_1\dots\mathbf{E}_N} \right) .$$
(2.3.6)

As an application, let us consider two transverse tensors T and U of rank N + 1 and N, respectively. Using the chain rule, we see that

$$T^{BA_1...A_N} \nabla_B U_{A_1...A_N} = -\nabla_B T^{BA_1...A_N} U_{A_1...A_N} + \dots$$
(2.3.7)

where the ellipsis denotes

$$\partial_C \left(T^{BA_1...A_N} \mathcal{P}_B^C U_{A_1...A_N} \right) - T^{BA_1...A_N} \partial_C \left(\mathcal{P}_B^C \mathcal{P}_{A_1}^{C_1} ... \mathcal{P}_{A_N}^{C_N} \right) U_{C_1...C_N} .$$
(2.3.8)

When we act on projection tensors with derivatives, this gives terms which vanish when contracted with the transverse tensors, so the second term vanishes. Since the first term is a total derivative, (2.3.7) implies that Lagrangians written in embedding coordinates enjoy the same equivalence relations as flat space Lagrangians under integration by parts.

The AdS contact Witten diagrams in embedding space are defined as integrals over AdS_{d+1} of products of bulk-to-boundary propagators. These propagators are given by

$$G(\hat{X}, X_i) = \frac{\mathcal{C}_{\Delta_i}}{(-2\hat{X}.X_i)^{\Delta_i}} , \qquad (2.3.9)$$

properly normalised to yield a delta-function at the boundary, the bulk-to-boundary propagator will include a normalisation² [57, 59]

$$\mathcal{C}_{\Delta} = \frac{\Gamma(\Delta)}{2\pi^{\frac{d}{2}}\Gamma(\Delta - \frac{d}{2} + 1)} .$$
(2.3.10)

Note that $x_{ij}^2 = -2X_i \cdot X_j$ and the X_i are boundary points which satisfy

$$X_i^2 = 0 \ . \tag{2.3.11}$$

Using the definition (2.3.6) and acting on the bulk-to-boundary propagator we get the equations of motion

$$\nabla^2 G = \Delta \left(\Delta - d \right) G , \qquad (2.3.12)$$

²These are normally omitted from the definition of the contact diagrams or *D*-functions and we do so here. We will also later absorb factors of C_{Δ_i} into the definition of the Mellin amplitude in chapter 3.



Figure 2.1: Contact Witten diagram for a conformal four-point function dual to a quartic contact interaction of the bulk scalar fields. Bulk-to-boundary propagators $G(\hat{X}, X_i)$ connect the boundary points X_i to the bulk interaction point \hat{X} . A general quartic interaction vertex includes derivatives acting on the bulk-to-boundary propagators (see section 2.3.2).

which agrees with the mass from the AdS/CFT prediction (2.3.2) (where we set R = 1). For $d = \Delta$ the propagator obeys massless equations of motion $\nabla^2 G = 0$.

To compute a four-point function associated with a ϕ^4 quartic interaction in AdS, we take the AdS integral over four bulk-to-boundary propagators as follows:

$$D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}^{(d)}(X_i) = \frac{1}{(-2)^{2\Sigma_{\Delta}}} \int_{\text{AdS}} \frac{d^{d+1}\hat{X}}{(\hat{X}.X_1)^{\Delta_1}(\hat{X}.X_2)^{\Delta_2}(\hat{X}.X_3)^{\Delta_3}(\hat{X}.X_4)^{\Delta_4}} , \quad (2.3.13)$$

where $\Sigma_{\Delta} = (\Delta_1 + \Delta_2 + \Delta_3 + \Delta_4)/2$. This is the definition of the *D*-functions. They describe quartic contact diagrams and will play an important role in each of the following chapters. The powers of minus 2 can be absorbed into the propagators as $(-2\hat{X}.X_i)$, but for notational simplicity we pull them out. In figure 2.1 a general four-point Witten diagram is illustrated.

There is a particularly useful representation for contact diagrams, the Mellin representation. The above *D*-functions have the following form in Mellin space [57] (see appendix A for more details on Mellin space and contact diagrams)

$$D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}^{(d)}(X_i) = \mathcal{N}_{\Delta_i}^{\mathrm{AdS}_{d+1}} \times \int \frac{d\delta_{ij}}{(2\pi i)^2} \prod_{i < j} \frac{\Gamma(\delta_{ij})}{(X_i \cdot X_j)^{\delta_{ij}}} , \quad \text{with } \sum_{i < j} \delta_{ij} = \Delta_j , \quad (2.3.14)$$

where the normalisation is given by

$$\mathcal{N}_{\Delta_i}^{\mathrm{AdS}_{d+1}} = \frac{\frac{1}{2}\pi^{d/2}\Gamma(\Sigma_{\Delta} - d/2)}{(-2)^{\Sigma_{\Delta}}\prod_i \Gamma(\Delta_i)} \ . \tag{2.3.15}$$

For later use we define normalised *D*-functions without the factor $\mathcal{N}_{\Delta_i}^{\mathrm{AdS}_{d+1}}$ as

$$D_{\Delta_i}(X_i) = \frac{1}{\mathcal{N}_{\Delta_i}^{\text{AdS}_{d+1}}} D_{\Delta_i}^{(d)}(X_i) . \qquad (2.3.16)$$

Note that the normalised *D*-functions are independent of the spacetime dimension d as can be seen from (2.3.14) and they are distinguished by the presence or not of the superscript (d).

To perform explicit calculations in position space it is useful to rewrite the $D^{(d)}$ functions in terms of \overline{D} -functions, which only depend on the conformal cross-ratios u, v [60]:

$$\frac{\prod_{i=1}^{4} \Gamma(\Delta_{i})}{\Gamma(\Sigma_{\Delta} - \frac{d}{2})} \frac{2}{\pi^{\frac{d}{2}}} D^{(d)}_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}(x_{i}) = \frac{(x_{14}^{2})^{\Sigma_{\Delta} - \Delta_{1} - \Delta_{4}} (x_{34}^{2})^{\Sigma_{\Delta} - \Delta_{3} - \Delta_{4}}}{(x_{13}^{2})^{\Sigma_{\Delta} - \Delta_{4}} (x_{24}^{2})^{\Delta_{2}}} \bar{D}_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}(u, v) .$$
(2.3.17)

The simplest of these functions D_{1111} is given by the standard four-dimensional box integral which can be written explicitly in terms of dilogarithms depending on u, v. Higher \overline{D} -functions can be computed by acting with differential operators on the box integral. For more details on these functions we refer the reader to e.g. [60].

2.3.2 Higher-Derivative Corrections

In this subsection we discuss higher-derivative corrections to the low-energy effective action of the quantum gravity theory under consideration which correspond to subleading terms in the low-energy expansion of correlators in the boundary CFT. Let us illustrate this with a schematic low-energy effective action. Supergravity is described by a supersymmetric version of the Einstein-Hilbert action which contains the Riemann curvature tensor \mathcal{R} . Including quantum gravity corrections corresponds to including higher-derivative interaction terms where we restrict to quartic interactions of the schematic form $\mathcal{D}^{2k}\mathcal{R}^4$:

$$\mathcal{L} \sim \frac{1}{G_N} \left(\mathcal{R} + \sum_k c_k \, \mathcal{D}^{2k} \mathcal{R}^4 + \dots \right) \,, \qquad (2.3.18)$$

with the Newton constant G_N . Recall that the Riemann tensor contains two derivatives, therefore the first correction \mathcal{R}^4 has six more derivatives than supergravity. Note that in later chapters \mathcal{R}^4 will be referred to as the zero-derivative correction, even though it has more derivatives than supergravity, as we label the corrections in terms of the number of derivatives acting on \mathcal{R}^4 .

In [30] the authors considered higher-derivative corrections for generic 2d and 4d CFTs in the large-N expansion at first subleading order which corresponds to treelevel supergravity plus higher-derivative corrections. Their strategy was to solve the crossing equations for four-point correlators by performing a conformal block analysis (as in (2.1.12)) and truncating the expansion to spin L. They found that there are (L+2)(L+4)/8 solutions to the crossing equations for each spin-L truncation. Furthermore, they provided holographic arguments and came to the same conclusion from a bulk point of view. In particular, the authors considered a massive scalar field in AdS with local quartic interactions (which can be thought of as a toy model for the low-energy effective action of quantum gravity in AdS, schematically described in (2.3.18)) and showed that, up to integration by parts and equations of motion, the quartic bulk interactions are in one-to-one correspondence with the solutions to the crossing equations described above. There are L/2 + 1 independent interactions which can create or annihilate a state of at most spin L, with the total number of derivatives ranging from 2L to 3L in intervals of two. These can be written as

$$\left(\nabla_{\underline{\mu}}^{L/2}\phi\right)\left(\nabla_{\underline{\nu}}^{L/2}\phi\right)\left(\nabla_{\underline{\rho}}^{k}\phi\right)\left(\nabla_{\underline{\mu}\nu\rho}^{L+k}\phi\right) , \quad k = 0, 1, \dots, L/2 , \qquad (2.3.19)$$

where the underscores denote sets of Lorentz indices. Note that the first two scalars in isolation have L free Lorentz indices as do the last two and so they can create a spin-L state. Hence, there is for example one spin-0 interaction vertex ϕ^4 , and two spin-2 interaction vertices equivalently written $\phi^2 (\nabla_{\mu} \nabla_{\nu} \phi)^2$ and $\phi^2 (\nabla_{\mu} \nabla_{\nu} \nabla_{\rho} \phi)^2$ which contain four and six derivatives, respectively. The total number of interactions up to spin L is then given by $\sum_{l=0}^{L/2} (l+1) = (L+2)(L+4)/8$. The diagram in (2.3.13) describes a zero-derivative interaction and thus corresponds to a spin-0 interaction vertex ϕ^4 (or to \mathcal{R}^4 in (2.3.18)).

The counting of independent interaction vertices can be nicely seen in Mellin space. In [61] a basis of solutions for higher-derivative corrections is given in Mellin space by

$$P_{p,q} = \sigma_2^p \sigma_3^q , \quad \text{with} \quad \sigma_2 = s^2 + t^2 + u^2 , \quad \sigma_3 = s^3 + t^3 + u^3 , \quad (2.3.20)$$

with non-negative integers p, q and s, t, u are Mellin variables, analogous to Mandelstam variables for four-particle scattering. The solutions $P_{p,q}$ correspond to solutions of the crossing equations with conformal block expansions truncated to spin L = 2(p+q), hence for each L there are L/2 + 1 solutions.

Remarkably, in [30] the authors argued that the number of derivatives in the bulk interaction vertex can be deduced from the large-twist behaviour of the anomalous dimensions of double-trace operators in the conformal block expansions of half-BPS correlators. From this analysis it is possible to obtain the large-N scaling of the solutions to the crossing equations by dimensional analysis from comparing the largetwist limit of the anomalous dimensions corresponding to spin-L corrections to those of supergravity. This was implemented for $\mathcal{N} = 4$ SYM in [61] and for the 6d (2,0) theory in [26, 43] (see chapter 4 for more details).

In the previous subsection we have seen how to obtain contact Witten diagrams corresponding to a quartic bulk interaction with no derivatives (2.3.14). Let us illustrate the evaluation of contact diagrams with covariant derivatives acting on the bulk-to-boundary propagators in a couple of simple examples. The first non-trivial quartic interaction with derivatives has four derivatives

$$\left(\nabla\phi\right)^2 \left(\nabla\phi\right)^2 \,, \qquad (2.3.21)$$

which corresponds to a spin-2 solution, see (2.3.19). This is a correction with four more derivatives than the ϕ^4 interaction and thus corresponds to the $D^4 \mathcal{R}^4$ term in the low-energy effective action. The interaction with two derivatives can be reduced to ϕ^4 by using integration by parts and equations of motion (2.3.12). The contact diagram for a quartic interaction with four derivatives acting on the bulk-to-boundary propagator is then

$$\sum_{\text{perms}} \int_{\text{AdS}} d\hat{X} \nabla_A \left(-2\hat{X}.X_1 \right)^{-\Delta} \nabla^A \left(-2\hat{X}.X_2 \right)^{-\Delta} \nabla_B \left(-2\hat{X}.X_3 \right)^{-\Delta} \nabla^B \left(-2\hat{X}.X_4 \right)^{-\Delta} \\ \propto \int_{\text{AdS}} d\hat{X} \prod_{i=1}^4 (-2\hat{X}.X_i)^{-\Delta} \sum_{\text{perms}} \left(\frac{X_1.X_2 X_3.X_4}{\hat{X}.X_1 \hat{X}.X_2 \hat{X}.X_3 \hat{X}.X_4} + \dots \right) , \qquad (2.3.22)$$

where we use (2.3.6) and the ellipsis denotes terms that can be reduced to ϕ^4 by use of integration by parts and equations of motion.³ The new contribution can be written in terms of \bar{D} -functions as

$$(1+u+v)D_{\Delta+1\,\Delta+1\,\Delta+1}$$
. (2.3.23)

Performing a conformal block analysis of this correlator one finds that indeed the sum truncates to spin-2. Furthermore, analysing the large-twist behaviour of the anomalous dimensions gives insight into the large-N scaling of this solution to the crossing equations, see chapter 4.

The next higher-derivative correction is given by a six-derivative interaction vertex $(\nabla \phi)^2 (\nabla_\mu \nabla_\nu \phi)^2$ which corresponds to another spin-2 solution as explained above. The contact diagram is

$$\sum_{\text{perms}} \int_{\text{AdS}} d\hat{X} \nabla_A \left(-2\hat{X}.X_1 \right)^{-\Delta} \nabla^A \left(-2\hat{X}.X_2 \right)^{-\Delta} \nabla_B \nabla_C \left(-2\hat{X}.X_3 \right)^{-\Delta} \nabla^B \nabla^C \left(-2\hat{X}.X_4 \right)^{-\Delta} \\ \propto \int_{\text{AdS}} d\hat{X} \prod_{i=1}^4 \left(-2\hat{X}.X_i \right)^{-\Delta} \sum_{\text{perms}} \left(\frac{X_1.X_2 \left(X_3.X_4 \right)^2}{\hat{X}.X_1 \hat{X}.X_2 \left(\hat{X}.X_3 \right)^2 \left(\hat{X}.X_4 \right)^2} + \dots \right), \quad (2.3.24)$$

where the ellipsis denotes terms that can be reduced to the zero- and four-derivative contributions. The new six-derivative contribution is given in terms of \overline{D} -functions as follows

$$\bar{D}_{\Delta+2\Delta+1\Delta+2\Delta+1} + \bar{D}_{\Delta+1\Delta+2\Delta+1\Delta+2} + u^2 \bar{D}_{\Delta+2\Delta+2\Delta+1\Delta+1} + u \bar{D}_{\Delta+1\Delta+1\Delta+2\Delta+2} + v^2 \bar{D}_{\Delta+1\Delta+2\Delta+2\Delta+1} + v \bar{D}_{\Delta+2\Delta+1\Delta+1\Delta+2} .$$
(2.3.25)

³These additional terms will become important in chapter 3 where we introduce higherdimensional generalised bulk-to-boundary propagators which treat AdS and S on equal footing. These propagators lead to new AdS×S contact Witten diagrams and the corresponding covariant derivatives no longer commute, therefore ambiguities appear in the place of the ellipsis above and they can no longer be disregarded.

Again, performing the conformal block expansion one finds that indeed the sum truncates to spin-2. This analysis can be generalised to higher derivatives by acting with the appropriate number of covariant derivatives on the bulk-to-boundary propagators as in the above example. The results can be represented in position space in terms of *D*-functions or in Mellin space according to appendix A.

We will encounter such local quartic bulk interactions dual to conformal correlators with spin-truncated conformal block expansions again in all three main chapters. Such correlators can be obtained from contact Witten diagrams only and no exchange diagrams need to be considered to compute higher-derivative corrections [30]. This makes the calculation much simpler and all Witten diagrams (with any number of derivatives) are given in terms of *D*-functions (or their Mellin transforms). Note that for supergravity the conformal block expansion does not generally truncate in spin and therefore it is not included in the above discussion. However, in the 1d case considered in chapter 5 supergravity is actually described by a ϕ^4 bulk interaction and can thus be obtained from contact diagrams.

We have now reviewed important concepts which will play a role in the following chapters. In the next chapter we discuss higher-derivative corrections to the low-energy effective action of type IIB string theory in $AdS_5 \times S^5$ dual to half-BPS correlators in the small α' expansion in $\mathcal{N} = 4$ SYM.

Chapter 3

Towards the Virasoro-Shapiro Amplitude in $AdS_5 \times S^5$

This chapter is based on [42] and we follow the paper closely. As explained in the introduction 1, in flat space, four-point amplitudes of closed strings are given by the Virasoro-Shapiro amplitude. It is of great interest to generalise this to curved spacetime and our aim is to obtain an analogue of the flat space VS amplitude in $AdS_5 \times S^5$.

The AdS/CFT correspondence relates IIB gravity amplitudes to $\mathcal{N} = 4$ SYM singletrace¹ half-BPS correlators. From the early days of the AdS/CFT correspondence, many direct calculations of four-point AdS amplitudes at tree-level and in the supergravity limit have been performed, resulting in predictions for the corresponding correlators on the CFT side [60, 63, 65–74]. Although the action for superstrings in AdS₅×S⁵ is known using the Green-Schwarz [75, 76] and pure spinor [77] formalisms, explicit construction of vertex operators is not fully understood. Hence, computing amplitudes beyond the supergravity approximation in this background directly from string theory remains challenging (see [78–80] for recent progress). On the other

¹In fact the operators dual to supergravity are only single-trace in the large-N limit but have multi-trace corrections at subleading order [62, 63]. These have recently been given explicitly to all orders in N [64]. Here however, we work at leading order and so these multi-trace corrections will not play a role.

hand, a great deal of progress has recently been achieved on the CFT side despite the CFT being strongly coupled, using the constraints imposed by superconformal and crossing symmetry as well as the simplification of the spectrum predicted by AdS/CFT (hereby summarised as 'bootstrap methods'). All tree-level single-trace half-BPS correlators in the supergravity limit have been obtained in this way [81– 83] and more recently string corrections have also been bootstrapped [10–16] with groundwork laid in [30, 61]. Loop corrections to four-point AdS amplitudes have also been obtained via bootstrap methods both in the supergravity limit [84–89] as well as string corrections [11, 90–92]. The more recent of these works have also made use of a hidden 10d conformal symmetry [41] which we will discuss further in the context of $AdS_2 \times S^2$ in chapter 5.

The program in this chapter can be viewed as partly going back to the direct calculation approach but in a hugely simplified form. We notice that the tree-level string corrections obtained via bootstrap methods can be obtained via AdS×S contact diagrams arising from a simple 10d scalar effective action. The starting point is the observation that if we write the flat space VS amplitude as an infinite series in α' , the leading term will describe supergravity while higher-order terms describe string corrections. These corrections can be derived from a simple effective field theory consisting of a scalar field with quartic interactions. For example, the first string correction is simply a constant proportional to α'^3 which arises from a ϕ^4 interaction, and the next correction is $\mathcal{O}(\alpha^{\prime 5})$ and quadratic in the Mandelstam variables so can be derived from a four-derivative interaction $(\partial \phi . \partial \phi)^2$. In this way, we can construct the four-field piece of the linearised (about flat space) effective action at all orders in α' , fixing coefficients by comparing to the VS amplitude. This can be made more precise. All the fields of type IIB supergravity can be described with a chiral scalar superfield, ϕ , in 10d $\mathcal{N} = 2$ superspace [93], and it is this scalar superfield that appears in the superaction. The Virasoro-Shapiro amplitude for IIB string theory is a superamplitude containing a factor $\delta^{16}(Q)$ [94]. Similarly the corresponding linearised effective action is a superaction and one integrates a scalar

superfield (prepotential) over 16 Grassmann odd variables $\int d^{16}\theta$ [95]. The action of four Grassmann derivatives on the scalar produces the Riemann curvature and so ϕ^4 in the effective superpotential produces the familiar \mathcal{R}^4 correction to supergravity. We propose a generalisation of the flat space VS amplitude by uplifting its small α' expansion to AdS₅×S⁵ by replacing the derivatives in flat space by covariant derivatives in AdS_5 and S^5 . We will find that this 10d scalar effective action describing tree-level string theory in $AdS_5 \times S^5$ generates the interacting part of all single-trace half-BPS correlators. The resulting correlators are naturally packaged together into a 10d structure. This 10d structure is very reminiscent of and indeed was partly inspired by the 10d conformal structure of these correlators observed in [41]. However, here the 10d *conformal* structure is not apparent and does not play a role. We can read off some coefficients of the AdS×S effective action directly from the flat space one, but not all terms can be read off in this way. Firstly, since covariant derivatives will no longer commute in general, there is the possibility of commutator terms which vanish in flat space. Furthermore, it is also possible to add terms proportional to the curvature which vanish in the flat space limit. The effective action will therefore have additional terms with unfixed coefficients.

We do not here prove the existence of the effective field theory in $AdS_5 \times S^5$, but justify it a posteriori by showing that it reproduces all known results for four-point correlators of single-trace half-BPS operators at orders α'^3 and α'^5 , which were previously obtained via bootstrap methods in [10–15]. Furthermore, we present a general algorithm to obtain four-point correlators to any order in α' and use it to derive new predictions at α'^6 and α'^7 . At the same time as we completed our work, the authors of [16] also obtained higher-order α' corrections in $AdS_5 \times S^5$, using bootstrap methods in Mellin space to arrive at the higher-derivative corrections. Their results nicely complement ours.

As mentioned in the introduction, the key technical tool to derive correlation functions from the 10d effective field theory in $AdS_5 \times S^5$ are generalised 10d Witten diagrams which treat AdS and S on equal footing. Usual contact Witten diagrams

describe supergravity scattering and are defined as integrals over AdS space, we will generalise them to integrals over the full AdS×S space to manifestly include the spherical harmonics on the five-sphere. We are not aware of such generalised Witten diagrams directly appearing in the literature before, although similar structures on the sphere are given in [96] where analogues of geodesic Witten diagrams (which give conformal blocks) on the sphere were considered. The generalised Witten diagrams involve introducing propagators connecting the (5+5)-dimensional bulk of AdS₅×S⁵ to a generalised notion of a boundary. Although the five-sphere is compact, we can formally define its boundary using embedding coordinates analogous to those of AdS_5 . This definition is physically sensible when describing half-BPS operators since it essentially encodes the condition that they are traceless and symmetric in R-symmetry indices. Expanding the 10d Witten diagrams in modes on the S^5 then gives a prediction for all four-point correlators of single-trace half-BPS operators corresponding to a fixed order in the α' expansion of tree-level string theory in AdS₅×S⁵. Comparing these results to those obtained using localisation techniques² [12, 97, 98] allows us to fix some ambiguities in the effective action.

Before we go on and compute these higher-derivative corrections from the effective action we describe the general setup for our analysis in the following section.

3.1 General Setup

In this section we describe the basic ingredients that we will use in this chapter. We start with a review of $\mathcal{N} = 4$ super Yang-Mills theory, type IIB string theory and the corresponding holographic correlators. This is followed by a review of half-BPS correlators in $\mathcal{N} = 4$ SYM, which will be the analogue of the Virasoro-Shapiro amplitude in AdS₅×S⁵. Next, we describe our strategy for deducing an effective action from the VS amplitude in flat space and translating it to AdS₅×S⁵. Subsequently, we

²Supersymmetric localisation is a method to exactly compute correlation functions of supersymmetric operators in certain supersymmetric quantum field theories by reducing the path integral to finite-dimensional integrals.

introduce embedding space formalism for the sphere together with sphere-analogues of AdS contact diagrams. In the next subsection we then show how to compute contact diagrams directly in AdS×S using novel bulk-to-boundary propagators which are manifestly ten-dimensional. For a given order in the α' expansion of the VS amplitude, this will allow us to compute the infinite tower of half-BPS correlators by computing Witten diagrams from a 10d effective action and expanding them in modes on the sphere. The correlators are most elegantly expressed in Mellin space, which we review in the last subsection. In particular, we find that expanding our 10d Witten diagrams in terms of spherical coordinates gives rise to a spherical analogue of the Mellin transform and implies a generalised Mellin amplitude where AdS₅ and S⁵ are on equal footing. The question of stringy corrections will be addressed in subsequent sections.

3.1.1 Review: $\mathcal{N} = 4$ SYM/IIB Superstring Theory in AdS₅×S⁵

Let us start by reviewing the ingredients of the correspondence between $\mathcal{N} = 4$ SYM and string theory in AdS₅×S⁵. We start by introducing the field content and action of $\mathcal{N} = 4$ SYM before considering type IIB supergravity. For more detailed reviews of these concepts see e.g [52, 99, 100]. Finally, we discuss holographic correlators in this example of the AdS/CFT correspondence.

\mathcal{N} = 4 super Yang-Mills theory

The conformal field theory we study in this chapter, $\mathcal{N} = 4$ supersymmetric Yang-Mills theory, is a theory of great interest to the scientific community. On the one hand, it is the interacting four-dimensional gauge theory with the highest amount of supersymmetry and it is conformal even at the quantum level [101, 102]. It is also dual to IIB string theory in $AdS_5 \times S^5$ and is the classical example of the AdS/CFTcorrespondence. Moreover, it is believed to be integrable, i.e. exactly solvable at all values of the coupling constant, at least in the planar limit [103]. On the other hand, $\mathcal{N} = 4$ SYM is a four-dimensional interacting theory and its tree-level scattering amplitudes are equivalent to those of massless QCD [104], therefore it is a useful toy model for developing techniques that can be applied to less symmetric QFTs, such as QCD. We will briefly introduce the field theory in the following.

Denoted by $\mathcal{N} = 4$ SYM is the maximally supersymmetric four-dimensional Yang-Mills theory with gauge group SU(N). The field content of $\mathcal{N} = 4$ SYM consists of one gluon field A^a_{μ} , with $a = 1, \ldots N^2 - 1$ with four fermions as superpartners, $\psi^a_{\alpha A}$, $\bar{\psi}^{\dot{\alpha} a}_A$ with $\alpha, \dot{\alpha} = 1, 2$ and A = 1, 2, 3, 4. Additionally, closure of the supersymmetry algebra requires that there are six real scalars ϕ^{aI}_{YM} , with $I = 1, \ldots, 6$, transforming in the fundamental representation of $SO(6) \sim SU(4)$. Hence, there are 4×2 fermionic and 6+2 bosonic degrees of freedom. Due to supersymmetry, all fields have to transform in the adjoint representation of SU(N). The action of $\mathcal{N} = 4$ SYM was obtained in [105]

$$S = \int d^4x \operatorname{Tr} \left(-\frac{1}{4g_{YM}^2} F_{\mu\nu} F^{\mu\nu} + \frac{\theta}{16\pi^2} F_{\mu\nu} \tilde{F}^{\mu\nu} - (D_\mu \varphi_{AB}) (D^\mu \varphi^{AB}) + i \bar{\psi}^A_{\dot{\alpha}} \sigma^{\dot{\alpha}\alpha}_{\mu} D^\mu \psi_{\alpha A} - \frac{i g_{YM}}{2} \psi^{\alpha}_A [\varphi^{AB}, \psi_{\alpha B}] - \frac{i g_{YM}}{2} \bar{\psi}^A_{\dot{\alpha}} [\varphi_{AB}, \bar{\psi}^{\dot{\alpha}B}] - \frac{g_{YM}^2}{2} [\varphi_{AB}, \varphi_{CD}] [\varphi^{AB}, \varphi^{CD}] \right),$$

$$(3.1.1)$$

where we grouped the six real scalar fields ϕ_{YM}^{aI} into six complex scalar fields $\varphi^{aAB} = -\varphi^{aBA}$ which transform in the fully antisymmetric two-index representation of SU(4). The field strength is defined as $F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu} + i g_{\text{YM}} [A_{\mu}, A_{\nu}]$ and the covariant derivative is $D_{\mu} = \partial_{\mu} - i g_{\text{YM}} [A_{\mu}, A_{\nu}]$. In the second term, θ is a real coupling, the so-called Yang-Mills theta-angle, and $\tilde{F}^{\mu\nu}$ is the Hodge dual of the field strength $\tilde{F}^{\mu\nu} = \frac{1}{2} \epsilon^{\mu\nu\rho\sigma} F_{\mu\nu}$. It is standard to combine the real couplings g_{YM} and θ into the single complex Yang-Mills coupling

$$\tau = \frac{\theta}{2\pi} + \frac{4\pi i}{g_{\rm YM}^2} . \tag{3.1.2}$$

The matrices $(\bar{\sigma}^{\mu})^{\dot{\alpha}\alpha} = (\mathrm{Id}, -\sigma)^{\dot{\alpha}\alpha}$ and $(\sigma^{\mu})_{\alpha\dot{\alpha}} = \epsilon_{\alpha\beta}\epsilon_{\dot{\alpha}\dot{\beta}}(\bar{\sigma})^{\dot{\beta}\beta} = (\mathrm{Id}, \sigma)_{\alpha\dot{\alpha}}$, where σ^{μ} are the usual Pauli matrices and ϵ_{ab} is the Levi-Civita tensor. The only tunable

parameters in the action are the gauge coupling $g_{\rm YM}$ and the rank of the gauge group SU(N). Taking the limit $N \to \infty$, $g_{\rm YM} \to 0$, while holding the 't Hooft coupling $\lambda = g_{\rm YM}^2 N$ fixed, gives a topological expansion in 1/N where the leading term corresponds to the planar limit.

A remarkable property of this theory is that it is ultraviolet finite, i.e. that the coupling $g_{\rm YM}$ is not renormalised at the quantum level. Hence, $\mathcal{N} = 4$ SYM is a conformal field theory even at the quantum level [101, 102]. Thus, the theory is invariant under the conformal group $SO(2,4) \sim SU(2,2)$ and supersymmetry enhances this to the superconformal group PSU(2,2|4). The bosonic part of PSU(2,2|4) then consists of the conformal group SU(2,2) and the R-symmetry group SU(4). The fermionic part of the supergroup are the four supersymmetry generators and their superconformal partners.

Type IIB supergravity

The holographic dual of $\mathcal{N} = 4$ SYM is type IIB superstring theory in AdS₅×S⁵. Type IIB string theory was found in [106] and in the low-energy limit it can be approximated by type IIB supergravity [107]. Let us review the field content and action of 10d IIB supergravity here.

First, since it is a theory of gravity, there is a graviton, hence the metric $g_{\mu\nu}$ which is a supersymmetric traceless tensor of the symmetry group SO(8) and thus has 35 bosonic degrees of freedom (counting the number of independent components of a symmetric 8×8 matrix and subtracting one due to tracelessness). The rest of the bosonic part of the spectrum consists of the axion-dilaton $C + i\Phi$ with two bosonic degrees of freedom, a rank-2 antisymmetric tensor $B_{\mu\nu} + iA_{2\mu\nu}$ with 56, and a rank-4 antisymmetric tensor $A^+_{4\mu\nu\rho\sigma}$ with 35 bosonic degrees of freedom. Furthermore, there are two Majorana-Weyl gravitinos $\psi^I_{\mu\alpha}$, where I = 1, 2, with 112 fermionic degrees of freedom and two Majorana-Weyl dilatinos λ^I_{α} , I = 1, 2, with 16 fermionic degrees of freedom. Thus, there are 128 bosonic and 128 fermionic degrees of freedom, as required by supersymmetry. The + superscript of the rank-4 antisymmetric tensor $A_{4\mu\nu\rho\sigma}^{+}$ indicates that it has a self-dual field strength, the five-form field strength \tilde{F}_{5} defined below. The two gravitinos have the same chirality and so do the two dilatinos, however theirs is opposite to that of the gravitinos. Therefore, the theory is chiral.

The type IIB supergravity action was found in [93, 107]. Because of the self-dual field strength \tilde{F}_5 it is not straightforward to write down a classical action. This is because the term $|\tilde{F}_5|^2$ in the action counts twice the desired amount of physical degrees of freedom since it does not contain the self-duality constraint. This can be resolved by imposing the self-duality constraint as an additional field equation. The action is then given by

$$4\kappa_B^2 S = \int d^{10}x \sqrt{-g} e^{-\Phi} \left(2\mathcal{R} + 8\partial_\mu \Phi \partial^\mu \Phi - |H_3|^2 \right) - \int \left[d^{10}x \sqrt{-g} \left(|F_1|^2 + |\tilde{F}_3|^2 + \frac{1}{2} |\tilde{F}_5|^2 \right) + A_4^+ \wedge H_3 \wedge F_3 \right] + \text{fermions} , \quad (3.1.3)$$

with the scalar curvature \mathcal{R} , $g = \det g_{\mu\nu}$ and the coupling constant κ_B is related to the string length l_s and the 10d Newton constant as follows:

$$2\kappa_B^2 = \frac{1}{2\pi} \left(2\pi l_s\right)^8 , \qquad 16\pi G_N^{10d} = \frac{1}{2\pi} \left(2\pi l_s\right)^8 g_s^2 = 2\kappa_B^2 g_s^2 . \qquad (3.1.4)$$

Moreover, the field strengths are defined as:

$$F_{1} = dC , \quad H_{3} = dB , \quad F_{3} = dA_{2} , \quad F_{5} = dA_{4}^{+} ,$$

$$\tilde{F}_{3} = F_{3} - C H_{3} , \quad \tilde{F}_{5} = F_{5} - \frac{1}{2}A_{2} \wedge H_{3} + \frac{1}{2}B \wedge F_{3} , \qquad (3.1.5)$$

and the supplementary self-duality constraint is

$$\star \tilde{F}_5 = \tilde{F}_5 . \tag{3.1.6}$$

Finally, the quantities $|F_p|^2$ are defined as

$$|F_p|^2 = \frac{1}{p!} g^{\mu_1 \nu_1} \dots g^{\mu_p \nu_p} \bar{F}_{\mu_1 \dots \mu_p} F_{\nu_1 \dots \nu_p} , \qquad (3.1.7)$$

where \overline{F} is the complex conjugate of F. It is worth noting that by dimensional

analysis this theory is non-renormalisable which generally speaking is true for all effective supergravity actions. This suggests that they do not approximate welldefined quantum theories. However, this is no problem when considering these theories as low-energy effective actions of more fundamental quantum theories (like string theory or M-theory). The non-renormalisability can be seen from conventional power counting. In the example of the 10d theory above this goes as follows. The curvature \mathcal{R} has dimension (length)⁻². By dimensional analysis the square of the gravitational coupling constant κ_B^2 then has to have dimensions (length)⁸, or for general dimensions d, (length)^{d-2}. It is now easy to see that the coupling constant has negative mass dimension for d > 2 and is therefore non-renormalisable. As mentioned before, this is no problem because we only consider it as an effective theory describing the low-energy physics of string theory.

Type IIB supergravity has $SL(2,\mathbb{R})$ symmetry. This is most manifest in the Einstein frame, where we redefine fields from the string metric $g_{\mu\nu}$ to the Einstein metric $g_{\mu\nu}^E$

$$g_{\mu\nu} = e^{\Phi/2} g^E_{\mu\nu} , \qquad (3.1.8)$$

which leads to a transformation of the scalar curvature term such that it contains the usual Einstein-Hilbert term. Furthermore, combining the axion C and dilaton Φ into a new complex scalar field gives the axion-dilaton field

$$\tau \equiv C + i e^{-\Phi} . \tag{3.1.9}$$

The metric and A_4^+ fields are invariant under $SL(2,\mathbb{R})$ transformations and the axion-dilaton field transforms under a Möbius transformation

$$\tau \to \frac{a\tau + b}{c\tau + d}$$
, with $ad - bc = 1$, $a, b, c, d \in \mathbb{R}$. (3.1.10)

Moreover, the fields $B_{\mu\nu}$ and $A_{2\mu\nu}$ rotate into each other under the above linear transformation. When considering the full quantum theory, there is a quantisation condition $\tau \sim \tau + 1$. Thus, the symmetry group of type IIB superstring theory is the $SL(2,\mathbb{Z})$ subgroup of $SL(2,\mathbb{R})$, where $a, b, c, d \in \mathbb{Z}$ in the Möbius transformation.

Holographic correlators

We have discussed general holographic correlators in section 2.3, were we have seen that correlators of chiral primary (half-BPS) operators are dual to scattering amplitudes of bulk scalars in AdS. In particular, we have described how these correlators can be computed from AdS Witten diagrams. In this chapter we will study four-point correlators of chiral primary operators in $\mathcal{N} = 4$ SYM. We have described the form of general chiral primary operators in section 2.3. Now consider the superconformal primary of the stress tensor multiplet in $\mathcal{N} = 4$ SYM. It is a scalar operator of protected dimension two and transforms in the two-index traceless symmetric representation **20** of the R-symmetry group SO(6). It is constructed from scalar fields as follows

$$\mathcal{O}_2 = \operatorname{Tr}(\phi_{\mathrm{YM}}^I \phi_{\mathrm{YM}}^J) - \frac{\delta_{IJ}}{6} \operatorname{Tr}(\phi_{\mathrm{YM}}^K \phi_{\mathrm{YM}K}) . \qquad (3.1.11)$$

Its supergravity dual is a scalar in the graviton multiplet in AdS₅ with mass $m_2^2 = -4$ (see (2.3.2), where we set the AdS radius R = 1). Traceless symmetric operators of the form $\phi_{\rm YM}^{(I_1} \phi_{\rm YM}^{I_2} \dots \phi_{\rm YM}^{I_p)}$ with more than two indices, where all p R-symmetry indices are symmetrised and all traces removed, are dual to higher Kaluza-Klein modes on the five-sphere. These modes arise from dimensionally reducing type IIB gravity on S⁵ which leads to an infinite tower of protected scalar operators with scaling dimension $\Delta = p$, where p is the SO(6) R-symmetry charge describing the KK modes on the sphere. A chiral primary operator with SO(6) charge p is thus dual to a KK mode with mass $m_p^2 = p(p-4)$.

In the following, we consider the low-energy approximation of string theory, which corresponds to a small α' expansion, where the leading contribution is supergravity and the subleading terms describe stringy corrections. These higher-derivative corrections to the supergravity approximation are the focus of this chapter. Furthermore, taking $N \to \infty$ (which is like taking $G_N \to 0$) suppresses loop corrections and restricts to classical gravity. Through the AdS/CFT correspondence $G_N \sim 1/c$, where c is the central charge, and in the present theory $c = (N^2 - 1)/4$. Hence, we consider a large-c expansion. See the review in 2.2 for further discussions of these limits and the relations between the parameters of the bulk and boundary theory.

We now understand the correlators from a holographic point of view, in the next subsection we consider the half-BPS correlators further in the specific context of the considerations in this chapter.

3.1.2 Half-BPS Correlators

As we have seen, there are six real scalars in $\mathcal{N} = 4$ SYM transforming in the adjoint representation of SU(N) and the fundamental of SO(6), $\phi_{YM}^{aI}(X)^3$. Here we view the 4d Minkowski space via null 6d embedding coordinates X^A with X.X = 0manifesting the conformal SO(2,4) symmetry (see section 2.3.1 for a review on AdS embedding space). We also project with a null 6d coordinate Y_I , Y.Y = 0 to obtain $\phi_{YM}(X,Y) = \phi_{YM}^I(X)Y_I$ manifesting the internal SO(6) symmetry. Then the single-trace half-BPS operators are defined as

$$\mathcal{O}_p(X,Y) = \frac{1}{pN^{p/2}} \text{Tr}(\phi_{\text{YM}}^p) ,$$
 (3.1.12)

which looks like the chiral primary operators discussed in the previous subsection but with the R-symmetry indices contracted and with an additional normalisation. Note that we normalise the operators with an additional factor of $1/\sqrt{p}$ compared to the normalisation giving a normalised two-point function, first derived in [108]. This normalisation is inspired by the ten-dimensional conformal symmetry of [41] and will be discussed further in chapter 5.

It is then useful to collect together the four-point functions of all single-trace half-BPS operators $\mathcal{O}_p(X,Y)$ into a single object $\langle \mathcal{OOOO} \rangle$ as follows

$$\langle \mathcal{OOOO} \rangle = \sum_{p,q,r,s=2}^{\infty} \langle \mathcal{O}_p \mathcal{O}_q \mathcal{O}_r \mathcal{O}_s \rangle_{\text{int}} ,$$
 (3.1.13)

³We drop the index *a* from now on and it will be understood that the $\phi_{\text{YM}}^{I}(X)$ transform in the adjoint of SU(N).

where $\langle \mathcal{O}_p \mathcal{O}_q \mathcal{O}_r \mathcal{O}_s \rangle_{\text{int}}$ represents the interacting part of the correlator, which always contains a particular factor $I(X_i, Y_i)$ due to superconformal symmetry [109] which we thus divide out

$$\langle \mathcal{O}_p \mathcal{O}_q \mathcal{O}_r \mathcal{O}_s \rangle_{\text{int}} = \frac{\langle \mathcal{O}_p \mathcal{O}_q \mathcal{O}_r \mathcal{O}_s \rangle - \langle \mathcal{O}_p \mathcal{O}_q \mathcal{O}_r \mathcal{O}_s \rangle_{\text{free}}}{I(X_i, Y_i)} .$$
(3.1.14)

From now on we will usually drop the explicit 'int' subscript at the end of the correlators. Here I is a polynomial in X_i and Y_i , the so-called Intriligator polynomial, which is a common factor of all interacting half-BPS four-point functions [109]. It is the counterpart of the $\delta^{16}(Q)$ factor of flat space superamplitudes [110] and we give its explicit form in appendix B.

Now that we have specified what correlators we are studying in this chapter we can go on and discuss the Virasoro-Shapiro amplitude that describes string scattering in flat space and how to lift it to curved spacetime.

3.1.3 Effective Action

As explained in the introduction, the four-point amplitude of closed string theory takes a very compact form in flat space, the Virasoro-Shapiro amplitude:

$$A_{\rm VS}(S,T) = \frac{1}{STU} \frac{\Gamma(1 - \frac{\alpha'S}{4})\Gamma(1 - \frac{\alpha'T}{4})\Gamma(1 - \frac{\alpha'U}{4})}{\Gamma(1 + \frac{\alpha'S}{4})\Gamma(1 + \frac{\alpha'T}{4})\Gamma(1 + \frac{\alpha'U}{4})} , \qquad S + T + U = 0 ,$$
(3.1.15)

where S, T, U are the standard four-point kinematic invariants. Note that we have factored out a supermomentum delta-function which encodes all the external supergravity states. In $AdS_5 \times S^5$ the analogue is to factor out an Intriligator polynomial from the interacting part of half-BPS correlators in the boundary, as we explained in the previous subsection. Our goal will then be to construct a bosonic 10d effective action which describes the remaining quantity. A priori it is not obvious that such an effective action should exist in curved background, but we justify it by showing that it reproduces previous results. It is important to note that the fact that we can factor out the supermomentum delta-function, or the Intriligator polynomial which contains all the supersymmetry properties of the correlators, is the crucial aspect of the theory that allows us to obtain correlators using a higher-dimensional effective action.

The flat space VS amplitude in (3.1.15) has expansion

$$A_{VS}(S,T) = \frac{1}{STU} \exp\left(\sum_{n=1}^{\infty} 2\left(\frac{\alpha'}{4}\right)^{2n+1} \frac{\zeta_{2n+1}}{2n+1} (S^{2n+1} + T^{2n+1} + U^{2n+1})\right)$$

$$= \frac{1}{STU} + 2\zeta_3 (\frac{\alpha'}{4})^3 + (S^2 + T^2 + U^2)\zeta_5 (\frac{\alpha'}{4})^5 + 2STU(\zeta_3)^2 (\frac{\alpha'}{4})^6$$

$$+ \frac{1}{2} (S^2 + T^2 + U^2)^2 \zeta_7 (\frac{\alpha'}{4})^7 + \dots, \qquad (3.1.16)$$

where ζ_n are Riemann-zeta functions. Excluding the first term, which corresponds to supergravity, we can view the remaining terms as arising from a scalar effective action. From this point of view, the α'^3 correction which gives a constant contribution to the four-point amplitude, comes from a ϕ^4 interaction. Higher-order terms can then be obtained by applying derivatives to the ϕ^4 interaction corresponding to the invariants S, T, U. So $S = -2k_1 \cdot k_2 \rightarrow 2\partial_\mu \phi \partial^\mu \phi \phi^2$, $T = -2k_1 \cdot k_3 \rightarrow 2\partial_\mu \phi \phi \partial^\mu \phi \phi$ etc.

Specifically then the VS amplitude is equivalent to the following four-field terms in an effective superpotential for supergravity linearised about flat space:

$$V_{\rm VS}^{\rm flat}(\phi) = \frac{1}{2^3 \times 4!} \Big(2\zeta_3 (\frac{\alpha'}{2})^3 \phi^4 + 3\zeta_5 (\frac{\alpha'}{2})^5 (\partial\phi.\partial\phi)^2 + 2(\zeta_3)^2 (\frac{\alpha'}{2})^6 (\partial\phi.\partial\phi) (\partial_\mu\partial_\nu\phi\partial^\mu\partial^\nu\phi) + 3\zeta_7 (\frac{\alpha'}{2})^7 (\partial_\mu\partial_\nu\phi\partial^\mu\partial^\nu\phi)^2 + \dots \Big) .$$
(3.1.17)

We now uplift the effective superpotential to an $AdS_5 \times S^5$ background by replacing the flat derivatives with covariant AdS×S derivatives. This uplift is not unique however. Firstly the covariant derivatives no longer commute with each other leading to ambiguities. Secondly there could be terms involving lower number of derivatives, compensated by the AdS radius R which would vanish in the flat space limit. So to $\mathcal{O}(\alpha'^7)$ the superpotential translates to

$$V_{\rm VS}^{\rm AdS \times S}(\phi) = \frac{1}{8 \times 4!} \Big[\Big(\frac{\alpha'}{2} \Big)^3 A \phi^4 + \Big(\frac{\alpha'}{2} \Big)^5 \Big(3B (\nabla \phi . \nabla \phi)^2 + 6C \nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2 \Big) \\ + \Big(\frac{\alpha'}{2} \Big)^6 \Big(D (\nabla \phi . \nabla \phi) (\nabla_\mu \nabla_\nu \phi \nabla^\mu \nabla^\nu \phi) + 6E \nabla_\mu \nabla^2 \nabla_\nu \phi \nabla^\mu \nabla^\nu \phi \phi^2 \Big) \\ + \Big(\frac{\alpha'}{2} \Big)^7 \Big(6F (\nabla_\mu \nabla_\nu \phi \nabla^\mu \nabla^\nu \phi)^2 \\ + 6G_1 (\nabla^\mu \nabla^\nu \nabla_\mu \nabla^\rho \nabla^\sigma \nabla_\rho \phi) (\nabla_\nu \nabla_\sigma \phi) \phi^2 + \dots \Big) + \dots \Big] .$$
(3.1.18)

There are four more eight-derivative terms with coefficients G_2, G_3, G_4, G_5 whose explicit expressions are given in (3.6.2) and appendix D. The ambiguities with coefficients C, E, G_i are multiplied by symmetry factors for later convenience. Here, unlike in flat space, the coefficients A, B, C, \ldots themselves can have an expansion in the dimensionless parameter α'/R^2 where R is the radius of AdS (or S). So whereas in flat space 2k-derivative terms only occur at order α'^{k+3} , in AdS×S, 2k-derivative terms occur at α'^{k+3} and all higher orders in principle.

One could also imagine replacing the coefficient of 1/STU in (3.1.16) with an expansion in α'/R^2 , which is not included in (3.1.18), however this is forbidden by superconformal symmetry of $\mathcal{N} = 4$ SYM correlators. In more detail, the non-renormalisation results of [109] imply that supergravity correlators must contain a contribution from free theory and there is a non-trivial cancellation between the two terms which links them together. Since free theory does not receive α' corrections, there cannot be α'/R^2 corrections to the coefficient of 1/STU.

The zeroth-order terms in the expansion of A, B, D, F are then determined by the Virasoro-Shapiro amplitude. Specifically,

$$A(\alpha') = 2\zeta_3 + A_1 \frac{\alpha'}{2R^2} + A_2 \left(\frac{\alpha'}{2R^2}\right)^2 + \dots \qquad E(\alpha') = E_0 + E_1 \frac{\alpha'}{2R^2} + \dots$$
$$B(\alpha') = \zeta_5 + B_1 \frac{\alpha'}{2R^2} + \dots \qquad F(\alpha') = \frac{1}{2}\zeta_7 + F_1 \frac{\alpha'}{2R^2} + \dots$$
$$C(\alpha') = C_0 + C_1 \frac{\alpha'}{2R^2} + \dots \qquad G_i(\alpha') = G_{i;0} + G_{i;1} \frac{\alpha'}{2R^2} + \dots \text{ for } i = 1, \dots, 5.$$
$$D(\alpha') = 2(\zeta_3)^2 + D_1 \frac{\alpha'}{2R^2} + \dots \qquad (3.1.19)$$

For simplicity, we will set R = 1 from now on throughout this chapter, but it will be

understood that these higher-order terms vanish in the flat space limit.

Computing 10d Witten diagrams using novel generalised bulk-to-boundary propagators and expanding them in terms of S^5 coordinates will give all single-trace half-BPS four-point correlators in $\mathcal{N} = 4$ SYM described by tree-level string theory in AdS₅×S⁵. We introduce the generalised bulk-to-boundary propagators and Witten diagrams in the following subsection.

3.1.4 Generalised Contact Witten Diagrams

Standard AdS Witten diagrams were reviewed in section 2.3 and in this subsection we define analogous objects on the sphere (following similar ideas in [96]) and finally we introduce a generalisation of Witten diagrams using bulk-to-boundary propagators which are intrinsically ten-dimensional and treat AdS and S on equal footing. This will have a big pay-off since we will obtain the whole tower of half-BPS correlators by expanding the Witten diagrams in spherical coordinates.

As we have seen in subsection 2.3.1, Witten diagrams are most conveniently expressed in embedding coordinates. In addition to embedding coordinates in AdS_{d+1} we also introduce embedding coordinates for S^{d+1} :

$$\hat{Y}^2 = \sum_{i=-1}^d \left(\hat{Y}^i \right)^2 = 1 .$$
(3.1.20)

In the present context, d = 4. As for AdS (2.3.4), covariant derivatives in terms of embedding coordinates are defined using projection tensors

$$\mathcal{P}_I^J = \delta_I^J - \hat{Y}_I \hat{Y}^J , \qquad (3.1.21)$$

which satisfy the useful identities

$$\mathcal{P}_I^J \hat{Y}^J = 0 , \qquad \mathcal{P}_I^J \mathcal{P}_J^K = \mathcal{P}_I^K . \tag{3.1.22}$$

To distinguish AdS and S projection tensors we use labels A, B, C... for AdS in section 2.3.1 and I, J, K... for the sphere analogues here. Recall that the covariant

derivative of a tensor is then given by [57, 58]

$$\nabla_A \mathbf{T}_{\mathbf{A}_1\dots\mathbf{A}_N} = \mathcal{P}_{\mathbf{A}}^{\mathbf{C}} \mathcal{P}_{\mathbf{A}_1}^{\mathbf{C}_1} \dots \mathcal{P}_{\mathbf{A}_N}^{\mathbf{C}_N} \partial_{\mathbf{C}} \left(\mathcal{P}_{\mathbf{C}_1}^{\mathbf{E}_1} \dots \mathcal{P}_{\mathbf{C}_N}^{\mathbf{E}_N} \mathbf{T}_{\mathbf{E}_1\dots\mathbf{E}_N} \right) . \tag{3.1.23}$$

This can equally be applied to the sphere case by simply sending A, B, C, \ldots indices to I, J, K, \ldots etc.

Now first recall that the standard AdS contact Witten diagrams in embedding space are defined as integrals over AdS_{d+1} of products of bulk-to-boundary propagators $G(\hat{X}, X_i) = \frac{C_{\Delta_i}}{(-2\hat{X}.X_i)^{\Delta_i}}$, see (2.3.13). At four points this yields:

$$D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}^{(d)}(X_i) = \frac{1}{(-2)^{2\Sigma_{\Delta}}} \int_{\text{AdS}} \frac{d^{d+1}\hat{X}}{(\hat{X}.X_1)^{\Delta_1}(\hat{X}.X_2)^{\Delta_2}(\hat{X}.X_3)^{\Delta_3}(\hat{X}.X_4)^{\Delta_4}} \quad (3.1.24)$$

Recall that $\Sigma_{\Delta} = (\Delta_1 + \Delta_2 + \Delta_3 + \Delta_4)/2$. These *D*-functions have the following form in Mellin space [57] (see also appendix A)

$$D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}^{(d)}(X_i) = \mathcal{N}_{\Delta_i}^{\mathrm{AdS}_{d+1}} \times \int \frac{d\delta_{ij}}{(2\pi i)^2} \prod_{i < j} \frac{\Gamma(\delta_{ij})}{(X_i \cdot X_j)^{\delta_{ij}}} , \quad \text{with } \sum_{i < j} \delta_{ij} = \Delta_j ,$$

$$(3.1.25)$$

where the normalisation is given in (2.3.15) and for later use we define normalised D-functions without the factor $\mathcal{N}_{\Delta_i}^{\mathrm{AdS}_{d+1}}$ which are independent of the spacetime dimension d:

$$D_{\Delta_i}(X_i) = \frac{1}{\mathcal{N}_{\Delta_i}^{\text{AdS}_{d+1}}} D_{\Delta_i}^{(d)}(X_i) . \qquad (3.1.26)$$

We can also consider direct analogues of these contact diagrams on the sphere. Bulk-to-boundary propagators on the sphere were introduced in [96]

$$G(\hat{Y}, Y_i) \propto (-2\hat{Y}.Y_i)^{p_i} \tag{3.1.27}$$

and in this context it is then very natural to introduce functions $B_{p_1p_2p_3p_4}^{(d)}(Y_i)$, spherical analogues of the contact Witten diagrams $D_{\Delta_1\Delta_2\Delta_3\Delta_4}^{(d)}$, as:

$$B_{p_1p_2p_3p_4}^{(d)}(Y_i) = (-2)^{2\Sigma_p} \int_{\mathcal{S}} d^{d+1} \hat{Y}(\hat{Y}.Y_1)^{p_1} (\hat{Y}.Y_2)^{p_2} (\hat{Y}.Y_3)^{p_3} (\hat{Y}.Y_4)^{p_4} , \qquad (3.1.28)$$

where $\Sigma_p = (p_1 + p_2 + p_3 + p_4)/2$. Even though the sphere is compact, we can formally

define a boundary when describing half-BPS operators in $\mathcal{N} = 4$ SYM since the condition Y.Y = 0 simply encodes tracelessness of the R-symmetry indices. The *B*-functions are polynomials in the Y_i and can be explicitly evaluated purely combinatorially, following similar techniques to those found in the appendix of [96] (where the two- and three-point analogues were obtained):

$$B_{p_1p_2p_3p_4}^{(d)}(Y_i) = \mathcal{N}_{p_i}^{S^{d+1}} \sum_{\{d_{ij}\}} \prod_{i < j} \frac{(Y_i \cdot Y_j)^{d_{ij}}}{\Gamma(d_{ij} + 1)} , \quad \text{with } \sum_{i < j} d_{ij} = p_j , \quad (3.1.29)$$

where

$$\mathcal{N}_{p_i}^{\mathrm{S}^{d+1}} = 2 \times 2^{\Sigma_p} \frac{\pi^{d/2+1} \prod_i \Gamma(p_i+1)}{\Gamma(\Sigma_p + d/2+1)} \ . \tag{3.1.30}$$

For later use let us also define a normalised B-function which does not depend on the dimension d:

$$B_{p_i}(Y_i) = \frac{1}{\mathcal{N}_{p_i}^{\mathrm{S}^{d+1}}} B_{p_i}^{(d)}(Y_i) \ . \tag{3.1.31}$$

In (3.1.29) the sum is over all sets of numbers $d_{ij} = d_{ji}$ such that

$$\left\{ (d_{12}, d_{13}, d_{14}, d_{23}, d_{24}, d_{34}) : 0 \le d_{ij} = d_{ji}, \quad d_{ii} = 0, \quad \sum_{i=1}^{4} d_{ij} = p_j \right\}.$$
(3.1.32)

These constraints on d_{ij} leave just two free parameters. Note the close similarity this explicit expansion of the *B*-functions (3.1.29) has with the Mellin transform of the AdS contact terms (3.1.25). The *B*-function can be seen as a discrete *D*-function, which is expected since it lives in a compact space. We can thus view the expansion parameters d_{ij} as analogues of the Mellin variables δ_{ij} .

It is now natural to combine the above AdS and S bulk-to-boundary propagators into one 10d object, which we refer to as a generalised bulk-to-boundary propagator in AdS×S:

$$G(\hat{X}, \hat{Y}; X, Y) = \left(-2\hat{X}.X - 2\hat{Y}.Y\right)^{-\Delta} , \qquad (3.1.33)$$

where X and Y satisfy

$$X^2 = Y^2 = 0 (3.1.34)$$

Using the definition in (3.1.23), we see that

$$\nabla^2 G = \left(\nabla_{\hat{X}}^2 + \nabla_{\hat{Y}}^2\right) G = \Delta(\Delta - d) \left((-2\hat{X}.X)^2 - (-2\hat{Y}.Y)^2\right) \left(-2\hat{X}.X - 2\hat{Y}.Y\right)^{-\Delta - 2} .$$
(3.1.35)

Hence, the propagator obeys massless equations of motion when $d = \Delta$:

$$\nabla^2 G = 0 \quad , \tag{3.1.36}$$

which will become important in later sections when computing ambiguities. Whereas X describes the boundary of AdS, Y is not a boundary point since the sphere is compact.

We will derive predictions for four-point correlators of half-BPS operators from an effective action by computing analogues of Witten diagrams directly in the product geometry AdS×S. For now we will just develop some general properties of $AdS_{d+1} \times S^{d+1}$ contact Witten diagrams which are defined simply as ⁴

$$D_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}^{\mathrm{AdS}_{d+1}\times\mathrm{S}^{d+1}}(X_{i},Y_{i}) = \frac{1}{(-2)^{2\Sigma_{\Delta}}} \int_{\mathrm{AdS\times S}} \frac{d^{d+1}\hat{X}d^{d+1}\hat{Y}}{(P_{1}+Q_{1})^{\Delta_{1}}(P_{2}+Q_{2})^{\Delta_{2}}(P_{3}+Q_{3})^{\Delta_{3}}(P_{4}+Q_{4})^{\Delta_{4}}}$$
(3.1.37)

where we introduce the shorthand

$$P_i = \hat{X}.X_i , \qquad Q_i = \hat{Y}.Y_i .$$
 (3.1.38)

The contact diagrams in AdS×S are related to the standard AdS contact diagrams in an intriguing way. This relation is explained in appendix C together with a discussion of tree-level supergravity which is not expected to arise from a superpotential⁵ but can be discussed using the aforementioned relation.

It is now straightforward to expand this AdS×S contact diagram into an infinite number of standard AdS contact diagrams multiplied by sphere analogues. In

⁴We keep d and Δ_i general here but we will be focussing on the case $\Delta_i = d = 4$ later in this chapter. The case $\Delta_i = d = 1$ will be studied in chapter 5.

⁵Remarkably, in the case of 1d CFTs studied in chapter 5 we find that supergravity correlators correspond to a ϕ^4 interaction and can therefore be computed from a 4d effective superpotential.

particular, using

$$\frac{1}{(P+Q)^{\Delta}} = \sum_{p=0}^{\infty} (-1)^p \frac{(p+1)_{\Delta-1}}{\Gamma(\Delta)} \frac{Q^p}{P^{p+\Delta}}$$
(3.1.39)

four times and then inserting (3.1.24) and (3.1.28) gives the expansion:

$$D_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}^{\text{AdS}\times\text{S}}(X_{i},Y_{i}) = \sum_{p_{i}=0}^{\infty} \prod_{i=1}^{4} (-1)^{p_{i}} \frac{(p_{i}+1)_{\Delta_{i}-1}}{\Gamma(\Delta_{i})} \times D_{p_{1}+\Delta_{1}p_{2}+\Delta_{2}p_{3}+\Delta_{3}p_{4}+\Delta_{4}}^{(d)}(X_{i}) B_{p_{1}p_{2}p_{3}p_{4}}^{(d)}(Y_{i}) .$$
(3.1.40)

Expanding the 10d Witten diagrams in spherical modes gives all half-BPS correlators and in (3.1.40) it is given in terms of AdS and S contact diagrams for a quartic interaction with no derivatives. We will discuss derivative interactions in section 3.3 but first let us discuss how to express the higher-dimensional Witten diagrams in Mellin space.

3.1.5 AdS×S Contact Diagrams in Mellin Space

Inserting the expression for the AdS contact term, $D^{(d)}$, as a Mellin integral (3.1.25) and the sphere analogue $B^{(d)}$ as an expansion (3.1.29) into the expression for the AdS×S contact term (3.1.40), we get, after some simplifications, a Mellin representation for the AdS×S contact term:

$$D_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}^{\mathrm{AdS}_{d+1}\times\mathrm{S}^{d+1}}(X_{i},Y_{i}) = \frac{\pi^{d+1}}{(-2)^{\Sigma_{\Delta}}\prod_{i}\Gamma(\Delta_{i})}$$

$$\times \sum_{p_{i}=0}^{\infty} (-1)^{\Sigma_{p}} \int \frac{d\delta_{ij}}{(2\pi i)^{2}} \sum_{\{d_{ij}\}} \left(\prod_{i< j} \frac{(Y_{i}.Y_{j})^{d_{ij}}}{(X_{i}.X_{j})^{\delta_{ij}}} \frac{\Gamma(\delta_{ij})}{\Gamma(d_{ij}+1)}\right) \times (\Sigma_{p}+d/2+1)_{\Sigma_{\Delta}-d-1} ,$$
where $\sum_{i< j} \delta_{ij} = p_{j} + \Delta_{j} , \qquad \sum_{i< j} d_{ij} = p_{j} ,$

$$(3.1.41)$$

and $x_n = \Gamma(x+n)/\Gamma(x)$ is the Pochhammer-symbol.

We thus define the $\operatorname{AdS}_{d+1} \times \operatorname{S}^{d+1}$ Mellin amplitude, $\mathcal{M}_{\Delta_i}[f](\delta_{ij}, d_{ij})$, for any such four-point expression, $f(X_i, Y_i)$, via a similar expression

$$f(X_i, Y_i) = \frac{1}{4!} \frac{\pi^{d+1}}{(-2)^{\Sigma_{\Delta}}} \left(\prod_i \frac{\mathcal{C}_{\Delta_i}}{\Gamma(\Delta_i)} \right)$$
$$\times \sum_{p_i=0}^{\infty} (-1)^{\Sigma_p} \int \frac{d\delta_{ij}}{(2\pi i)^2} \sum_{\{d_{ij}\}} \left(\prod_{iwhere $\sum_{i (3.1.42)$$$

Hence, the Mellin amplitude of an AdS×S contact diagram with no derivatives is not in general a constant as for the AdS case, but rather a Pochhammer:

$$\frac{1}{4!} \left(\prod_{i} \mathcal{C}_{\Delta_{i}} \right) \times D^{\mathrm{AdS}_{d+1} \times \mathrm{S}^{d+1}}_{\Delta_{1} \Delta_{2} \Delta_{3} \Delta_{4}} (X_{i}, Y_{i}) \qquad \leftrightarrow \qquad \mathcal{M}_{\Delta_{i}} (\delta_{ij}, d_{ij}) = (\Sigma_{p} + d/2 + 1)_{\Sigma_{\Delta} - d - 1} .$$

$$(3.1.43)$$

We now have all the tools at our disposal to discuss the first correction in the α' expansion of the correlators.

3.2 α'^3 Corrections

Having outlined the general procedure for computing stringy corrections to tree-level four-point half-BPS correlators in $\mathcal{N} = 4$ SYM using an effective action in AdS₅×S⁵, we will now illustrate how this works for the first correction to the supergravity prediction which occurs at order α'^3 .

In particular, the first term of the effective action (3.1.18) is just a ϕ^4 interaction:

$$S_{\alpha'^{3}} = \frac{1}{8 \times 4!} \left(\frac{\alpha'}{2}\right)^{3} \times 2\zeta_{3} \times \int_{\text{AdS} \times \text{S}} d^{5} \hat{X} d^{5} \hat{Y} \phi(\hat{X}, \hat{Y})^{4} .$$
(3.2.1)

To obtain the corresponding CFT correlators we mimic the standard AdS/CFT procedure for obtaining correlators from AdS, but in a fully 10d covariant way, including the sphere manifestly. Using the generalised bulk-to-boundary propagators in (3.1.33) we obtain the AdS×S Witten diagram for this contact interaction, yielding the following proposal for the α'^3 corrections to the correlators:

$$\begin{aligned} \langle \mathcal{OOOO} \rangle |_{\alpha'^{3}} \\ &= \frac{1}{8 \times 4!} \left(\frac{\alpha'}{2}\right)^{3} \times 2\zeta_{3} \times \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\mathrm{AdS} \times \mathrm{S}} \frac{d^{5} \hat{X} d^{5} \hat{Y}}{(P_{1} + Q_{1})^{4} (P_{2} + Q_{2})^{4} (P_{3} + Q_{3})^{4} (P_{4} + Q_{4})^{4}} \\ &= \frac{1}{8 \times 4!} \left(\frac{\alpha'}{2}\right)^{3} (\mathcal{C}_{4})^{4} \times 2\zeta_{3} \times D_{4444}^{\mathrm{AdS}_{5} \times \mathrm{S}^{5}} . \end{aligned}$$
(3.2.2)

We can now extract any specific half-BPS correlator from (3.2.2) by expanding to the appropriate power in Y_i (see (3.1.13)). First note that the 10d bulk-to-boundary propagator Taylor expands as

$$(P_i + Q_i)^{-4} = \sum_{p=2}^{\infty} (-1)^p \frac{(p-1)_3}{3!} (P_i)^{-p-2} (-Q_i)^{p-2} .$$
(3.2.3)

So the individual correlators are given by⁶:

$$\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle |_{\alpha'^3}$$

$$= \frac{1}{8 \times 4!} \left(\frac{\alpha'}{2}\right)^3 \times 2\zeta_3 \times \frac{(\mathcal{C}_4)^4}{(-2)^{16}} \prod_i \frac{(p_i - 1)_3}{3!} \int_{\text{AdS}} d^5 \hat{X} \prod_i \frac{1}{(P_i)^{p_i + 2}} \times \int_{\text{S}} d^5 \hat{Y} \prod_i (Q_i)^{p_i - 2}$$

$$= \frac{1}{8 \times 4!} \left(\frac{\alpha'}{2}\right)^3 (\mathcal{C}_4)^4 \times 2\zeta_3 \times \left(\prod_i \frac{(p_i - 1)_3}{3!}\right)$$

$$\times D_{p_1 + 2p_2 + 2p_3 + 2p_4 + 2}^{(4)} (X_i) \times B_{p_1 - 2p_2 - 2p_3 - 2p_4 - 2}^{(4)} (Y_i) .$$

$$(3.2.4)$$

To see what it looks like in Mellin space we plug the Mellin transform of D (3.1.25) and the expansion of B (3.1.29) into this expression (or just use (3.1.41)) giving the Mellin amplitude (defined in (3.1.42))

$$\mathcal{M}_{\alpha'^3} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^3 \times 2\zeta_3 \times (\Sigma_p - 1)_3 . \tag{3.2.5}$$

This correctly reproduces the results of [10, 11, 13] for the Mellin amplitude of half-BPS correlators at this order.

⁶This is (3.1.40) with $\Delta_i = d = 4$ and with $p_i \rightarrow p_i - 2$ to account for the fact that the lowest correlator is labelled with $p_i = 2$ rather than $p_i = 0$. We do not need to worry about the minus signs in the factors $(-1)^p$ in (3.2.3) since $B_{p_1p_2p_3p_4} = 0$ if $p_1 + p_2 + p_3 + p_4$ is odd.

3.3 Algorithm for Computing General α' Corrections

At higher orders in α' the effective action (3.1.18) has terms with covariant derivatives acting on the scalar field. Thus before proceeding, we describe an efficient way to evaluate generalised contact diagrams with derivatives in AdS×S in position space. Then we present a general formula for converting them to Mellin space.

3.3.1 Generalised Witten Diagrams

Computing the action of the covariant derivatives quickly becomes complicated and so it is useful to develop an algorithm to do this automatically. We will motivate the algorithm by building up from simple cases. First, we consider the application of multiple covariant derivatives at a single point in AdS. From (3.1.23) this is given recursively as

$$\nabla_{\!A}\nabla_{\!B}...\nabla_{\!C}\phi = \mathcal{P}_{\!A}^{A'}\mathcal{P}_{\!B}^{B'}...\mathcal{P}_{\!C}^{C'}\partial_{A'}\left(\nabla_{\!B'}...\nabla_{\!C'}\phi\right) , \qquad \nabla_{\!A}\phi = \mathcal{P}_{\!A}^{A'}\partial_{A'}\phi . \qquad (3.3.1)$$

So the application of two covariant derivatives gives

$$\nabla_{\!B}\nabla_{\!A}\phi = \mathcal{P}_B^{B'}\mathcal{P}_A^{A'}\partial_{B'}(\mathcal{P}_{A'}^{A''}\partial_{A''}\phi) = \mathcal{P}_B^{B'}\mathcal{P}_A^{A'}\partial_{B'}\partial_{A'}\phi + \mathcal{P}_{BA}\hat{X}.\partial\phi .$$
(3.3.2)

The first term arises from the partial derivative $\partial_{B'}$ being commuted through $\mathcal{P}_{A'}^{A''}$ whereas the second term arises from the partial derivative hitting $\mathcal{P}_{A'}^{A''}$. To arrive at this form, one then uses the definition of \mathcal{P} given in (2.3.4) as well as the useful formulae (2.3.5). We denote this result graphically as

$$\nabla_B \nabla_A = \begin{array}{c} B \bullet \\ A \bullet \end{array} + \begin{array}{c} B \\ A \end{array} \right), \qquad (3.3.3)$$

where each vertex corresponds to an index ordered vertically such that the bottom one is the index of the first derivative to act. An isolated vertex at position Adenotes $(\mathcal{P}.\partial)_A$ (with the understanding that the derivative has been commuted all the way to the right) whereas an edge between vertices A and B denotes $\mathcal{P}_{AB}\dot{X}.\partial$.

Next, consider three covariant derivatives. Here we obtain

$$\nabla_{C} \nabla_{B} \nabla_{A} \phi$$

$$= \mathcal{P}_{C}^{C'} \mathcal{P}_{B}^{B'} \mathcal{P}_{A}^{A'} \partial_{C'} \left(\mathcal{P}_{B'}^{B''} \mathcal{P}_{A'}^{A''} \partial_{B''} \partial_{A''} + \mathcal{P}_{B'A'} \hat{X} . \partial \right) \phi$$

$$= \left(\mathcal{P}_{C}^{C'} \mathcal{P}_{B}^{B'} \mathcal{P}_{A}^{A'} \partial_{C'} \partial_{B'} \partial_{A'} + \mathcal{P}_{AC} \mathcal{P}_{B}^{B'} \hat{X} . \partial \partial_{B'} + \mathcal{P}_{CB} \mathcal{P}_{A}^{A'} \hat{X} . \partial \partial_{A'} + \mathcal{P}_{AB} \mathcal{P}_{C}^{C'} \hat{X} . \partial \partial_{C'} \right)$$

$$+ \mathcal{P}_{AB} \mathcal{P}_{C}^{C'} \partial_{C'} \phi$$

$$= \begin{pmatrix} B \bullet & + & B \bullet \\ A \bullet & A \end{pmatrix} + \begin{pmatrix} C \bullet & C \bullet \\ B \bullet & + & B \bullet \\ A \bullet & A \end{pmatrix} + \begin{pmatrix} C \bullet & C \bullet \\ B \bullet & + & B \bullet \\ A \bullet & A \end{pmatrix} + \begin{pmatrix} C \bullet & C \bullet \\ B \bullet & + & B \bullet \\ A \bullet & A \end{pmatrix} , \quad (3.3.4)$$

and we give the corresponding diagrammatic form in the same order as the terms above. All terms apart from the last arise either from the derivative, ∂_C , hitting a \mathcal{P} (which we denote with a solid line) or commuting through (leaving an isolated vertex at C). The last term arises from the derivative, ∂_C , hitting the $\hat{X}.\partial$ term associated with the solid line between A and B. We denote this by a dotted line from C to B. Thus, a solid line with a dotted line attached to the top of it loses its decoration, $\hat{X}.\partial$. For the general case of several derivatives acting at a point we can work recursively: each additional derivative either commutes through everything, corresponding to an isolated vertex, or it hits a \mathcal{P} corresponding to a solid line, or it hits a $\hat{X}.\partial$, denoted by a dotted line. We add all such lines in all possible ways. Hence, the *n*-derivative term is given diagrammatically by summing all graphs containing *n* vertices in a vertical line, with any number of solid edges between any two points, such that no vertex is attached to more than one solid edge, and with any number of dotted edges from the vertex at the top of a solid edge to a higher vertex either isolated or at the bottom of a solid edge.

The above examples are already enough to illustrate the key ingredients of the general algorithm for obtaining an explicit expression for several covariant derivatives at a point, $\nabla_{A_1} \nabla_{A_2} \dots \nabla_{A_n}$ by summing over all possible graphs.

Algorithm for $\nabla_{A_1} \nabla_{A_2} \dots \nabla_{A_n} \phi$

- 1. Draw *n* vertices vertically. Each corresponds to an embedding space index ordered so the bottom one corresponds to A_n and the top one to A_1 .
- 2. Draw any number of solid edges between any two vertices such that each vertex is connected to at most one solid edge.
- 3. Draw any number of dotted edges from the upper vertex of a solid edge up to either a higher disconnected vertex or a higher vertex that is the lower vertex of a solid edge. No vertex can be attached to more than one dotted edge.
- 4. Sum over all the resulting graphs with the following interpretation:

So solid edges come with a decoration $\hat{X}.\partial$ unless they have a dotted line attached to the top in which case the decoration is removed (otherwise the dotted lines can be ignored).

A general derivative interaction term consists of covariant derivatives acting on different scalars with indices contracted together pairwise. We denote this graphically by putting together two or more of the above vertical graphs and adding grey edges corresponding to the contractions. So for example we obtain $\nabla^B \nabla^A \phi_1 \nabla_B \nabla_A \phi_2$ by taking two copies of all the two-derivative diagrams (3.3.3) and gluing the corresponding vertices together

$$\nabla^{B}\nabla^{A}\phi_{1}\nabla_{B}\nabla_{A}\phi_{2} = \bigoplus_{\phi_{1}} + \bigoplus_{\phi_{2}} + \mathcal{P}^{AB}(\partial_{A}\partial_{B}\phi_{1})\hat{X}.\partial\phi_{2} + \mathcal{P}^{AB}(\partial_{A}\partial_{B}\phi_{2})\hat{X}.\partial\phi_{1} + \mathcal{P}^{A}_{A}(\hat{X}.\partial\phi_{1})(\hat{X}.\partial\phi_{2}). \quad (3.3.6)$$

Similarly we obtain $\nabla^C \nabla^B \nabla^A \phi_1 \nabla_A \phi_2 \nabla_B \phi_3 \nabla_C \phi_4$ by taking the three-derivative diagram (3.3.4) together with three more vertices to the right and gluing the vertices correspondingly



The general algorithm for interaction terms is then a straightforward extension of the one above for covariant derivatives acting on a single scalar.

Algorithm for contact interactions in AdS

- 1. For each scalar ϕ_i with n_i covariant derivatives acting on it, draw all the corresponding contributing vertical graphs using the above algorithm. Place the graphs for each scalar next to each other horizontally (taking the outer product over the list of graphs at each point).
- 2. Draw grey lines between corresponding contracted vertices in the interaction term.
- 3. Finally sum over all the resulting graphs with the following interpretation:
- 4. Each connected path of solid and grey lines with end points in the vertical line above ϕ_i and ϕ_j corresponds to $\mathcal{P}^{AB}\partial_A\phi_i\partial_B\phi_j$.
- 5. Each solid line above ϕ_i corresponds to $\hat{X} \cdot \partial \phi_i$, as long as it does not have a dotted line attached to its upper vertex (if it does have such a dotted line it has no additional contribution).

See the above two examples (3.3.6) and (3.3.7).

So far we have only discussed AdS covariant derivatives. The above rules can be used with the obvious modifications if instead we are viewing the action on a sphere (i.e. A, B indices become I, J indices, $\hat{X} \rightarrow \hat{Y}$ and $\mathcal{P}^{AB} \rightarrow \mathcal{P}^{IJ}$ in (2.3.4)). However, our main purpose here is to consider AdS×S covariant derivatives. Thus, each vertex now represents a 10d index $\mu = (A, I)$ but there needs to be some non-trivial re-interpretation in the case of the product geometry.

Algorithm for contact interactions in AdS×S

The first three steps of the algorithm are as for the AdS case above. Then

- 4. Each connected path of solid and grey lines with end points in the vertical line above ϕ_i and ϕ_j respectively corresponds to $\mathcal{P}^{\mu\nu}\partial_{\mu}\phi_i\partial_{\nu}\phi_j$, but:
- 5. Each solid line above ϕ_i , as long as it does not have a dotted line attached to its upper vertex, breaks this manifest 10d structure by contributing a multiplicative factor $\hat{X}^A \partial_A \phi_i$, if the index running through it is in AdS or $-\hat{Y}^I \partial_I \phi_i$ if the index running through is in the sphere. (The minus sign appears in the latter case since this term arises from a derivative hitting \mathcal{P} in (2.3.4) or (3.1.21) which has a minus sign for the internal case.)
- 6. Finally, there is an additional subtlety related to the dotted lines. The dotted line ties together the index type corresponding to the otherwise potentially disconnected parts of the graph, and then contributes a factor of +1 if the

index running through is in AdS or -1 if the index running through is in the sphere. (Recall that the dotted lines arise from derivatives $\partial_{\hat{X}}$ or $\partial_{\hat{Y}}$ hitting the decoration $\hat{X}^A \partial_A \phi_i$ or $-\hat{Y}^I \partial_I \phi_i$. Thus firstly, this vanishes unless the derivative type (AdS or S) is the same as that of the solid line (hence tying together the index type), and secondly it gives ± 1 depending on whether it is AdS or S.)

Thus, for example the AdS×S covariant version of (3.3.7) is, with each of the five lines corresponding to the five graphs in (3.3.7),

$$\nabla^{\rho}\nabla^{\nu}\nabla^{\mu}\phi_{1}\nabla_{\mu}\phi_{2}\nabla_{\nu}\phi_{3}\nabla_{\rho}\phi_{4}$$

$$= \mathcal{P}^{\mu\mu'}\mathcal{P}^{\nu\nu'}\mathcal{P}^{\rho\rho'}(\partial_{\mu}\partial_{\nu}\partial_{\rho}\phi_{1})(\partial_{\mu'}\phi_{2})(\partial_{\nu'}\phi_{3})(\partial_{\rho'}\phi_{4})$$

$$+ \mathcal{P}^{\nu\nu'}\mathcal{P}^{AC}(\hat{X}.\partial_{\hat{X}}\partial_{\nu}\phi_{1})(\partial_{A}\phi_{2})(\partial_{\nu'}\phi_{3})(\partial_{C}\phi_{4}) - \mathcal{P}^{\nu\nu'}\mathcal{P}^{IK}(\hat{Y}.\partial_{\hat{Y}}\partial_{\nu}\phi_{1})(\partial_{I}\phi_{2})(\partial_{\nu'}\phi_{3})(\partial_{K}\phi_{4})$$

$$+ \mathcal{P}^{\mu\mu'}\mathcal{P}^{BC}(\hat{X}.\partial_{\hat{X}}\partial_{\mu}\phi_{1})(\partial_{\mu'}\phi_{2})(\partial_{B}\phi_{3})(\partial_{C}\phi_{4}) - \mathcal{P}^{\mu\mu'}\mathcal{P}^{JK}(\hat{Y}.\partial_{\hat{Y}}\partial_{\mu}\phi_{1})(\partial_{\mu'}\phi_{2})(\partial_{J}\phi_{3})(\partial_{K}\phi_{4})$$

$$+ \mathcal{P}^{\rho\rho'}\mathcal{P}^{AB}(\hat{X}.\partial_{\hat{X}}\partial_{\rho}\phi_{1})(\partial_{A}\phi_{2})(\partial_{B}\phi_{3})(\partial_{\rho'}\phi_{4}) - \mathcal{P}^{\rho\rho'}\mathcal{P}^{IJ}(\hat{Y}.\partial_{\hat{Y}}\partial_{\rho}\phi_{1})(\partial_{I}\phi_{2})(\partial_{J}\phi_{3})(\partial_{\rho'}\phi_{4})$$

$$+ \mathcal{P}^{CC'}\mathcal{P}^{AB}(\partial_{C}\phi_{1})(\partial_{A}\phi_{2})(\partial_{B}\phi_{3})(\partial_{C'}\phi_{4}) - \mathcal{P}^{KK'}\mathcal{P}^{IJ}(\partial_{K}\phi_{1})(\partial_{I}\phi_{2})(\partial_{J}\phi_{3})(\partial_{K'}\phi_{4}) .$$
(3.3.8)

In particular, note that only the first line is manifestly 10d covariant (has only 10d μ, ν indices). Also compare carefully the penultimate with the final line. These arise from similar graphs (the last two in (3.3.7)) but one with a dotted line and one without. In the final line, as well as the decoration $\hat{X}.\partial_{\hat{X}}$ or $\hat{Y}.\partial_{\hat{Y}}$ being absent, the dotted line has tied together the two otherwise disconnected parts of the graph, meaning for example that all indices are either AdS or S, with no mixed ones, unlike the penultimate line.

Finally, note that in practice for our purposes here the derivatives will always be acting on bulk-to-boundary propagators (3.1.33) and thus partial derivatives acting on a single scalar, $\partial_{\mu_1}\partial_{\mu_2}\ldots\partial_{\mu_{n_i}}\phi_i$, give $(-1)^{n_i}(\Delta_i)_{n_i}X^{\mu_1}\ldots X^{\mu_{n_i}}$ etc.

3.3.2 Mellin Space

The previous subsection gave an algorithm for obtaining explicit expressions for the integrands of generalised Witten diagrams in AdS×S coming from contact interactions with derivatives. This will result in integrands corresponding to decorations of the (no-derivative) contact diagram D (3.1.37). The decorations are in the form of polynomials in $X_i.X_j$, $Y_i.Y_j$, Q_i and P_i which are homogeneous at each point (i.e. scale the same under the local scaling $X_i.X_j \rightarrow e_ie_jX_i.X_j$, $Y_i.Y_j \rightarrow e_ie_jY_i.Y_j$, $Q_i \rightarrow e_iQ_i$ and $P_i \rightarrow e_iP_i$). Each term of such a decoration thus has the form

$$\frac{1}{4!} \frac{\prod_i \mathcal{C}_{\Delta_i}}{(-2)^{2\Sigma_{\Delta}}} \int_{\mathrm{AdS} \times \mathrm{S}} d^{d+1} \hat{X} d^{d+1} \hat{Y} \left(\prod_i \frac{Q_i^{n_i^Q} P_i^{n_i^P} \times (\Delta_i)_{n_i}}{(P_i + Q_i)^{\Delta_i + n_i}} \right) \times \left(\prod_{i < j} (X_i \cdot X_j)^{n_{ij}^X} (Y_i \cdot Y_j)^{n_{ij}^Y} \right) ,$$

$$(3.3.9)$$

with $n_i = n_i^P + n_i^Q + \sum_j n_{ij}^X + \sum_j n_{ij}^Y$. We define Σ_X, Σ_Y to represent the sum of all the n_{ij}^X, n_{ij}^Y respectively, Σ_Q, Σ_P represents *half* the sum of all the n_i^Q, n_i^P and Σ_n half the sum of the n_i , so $\Sigma_n = \Sigma_P + \Sigma_Q + \Sigma_X + \Sigma_Y$. Such a decorated integral will modify (3.1.40) to

$$(-2)^{2\Sigma_{X}+2\Sigma_{Y}} \sum_{p_{i}=0}^{\infty} \left(\prod_{i=1}^{4} (-1)^{p_{i}} \frac{(p_{i}+1)_{\Delta_{i}+n_{i}-1}}{\Gamma(\Delta_{i})} D_{p_{i}+\Delta_{i}+n_{i}-n_{i}^{P}}^{(d)}(X_{i}) B_{p_{i}+n_{i}^{Q}}^{(d)}(Y_{i}) \right) \times \left(\prod_{i< j} (X_{i}.X_{j})^{n_{ij}^{X}} (Y_{i}.Y_{j})^{n_{ij}^{Y}} \right).$$

$$(3.3.10)$$

Inserting the Mellin transform of D (3.1.25) and expansion of B (3.1.29) and performing some re-definitions and simplifications then gives the Mellin amplitude (defined in (3.1.42)):

$$\mathcal{M}_{\Delta_{i}}[(3.3.9)] = (-2)^{\Sigma_{X}} 2^{\Sigma_{Y}} (-1)^{2\Sigma_{Q}} \left(\prod_{i < j} (\delta_{ij})_{n_{ij}^{X}} (d_{ij} - n_{ij}^{Y} + 1)_{n_{ij}^{Y}} \right) \\ \times \left(\prod_{i} (p_{i} + n_{i}^{X} + \Delta_{i})_{n_{i}^{P}} (p_{i} - n_{i}^{Q} - n_{i}^{Y} + 1)_{n_{i}^{Q}} \right) (\Sigma_{p} - \Sigma_{Y} + \frac{d}{2} + 1)_{\Sigma_{\Delta} - d - 1 + \Sigma_{X} + \Sigma_{Y}} ,$$

where $\sum_{i < j} \delta_{ij} = p_{j} + \Delta_{j} , \qquad \sum_{i < j} d_{ij} = p_{j} .$ (3.3.11)

We will use this general formula, in conjunction with the algorithm of the previous subsection, to compute higher-order terms in the α' expansion of half-BPS correlators

in the next sections.

3.4 α'^5 Corrections

After α'^3 , the next terms in the effective action for string corrections occur at α'^5 . In the flat space limit, such terms contain four derivatives, so first we consider all the possible terms in the effective action in AdS×S involving four derivatives. At first there are many terms one can write down, but using integration by parts as well as the equations of motion (3.1.36) reduces the number down quickly. We find that in fact there are only two linearly independent terms one can write down involving four derivatives:

$$(\nabla \phi . \nabla \phi)^2$$
 and $\nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2$. (3.4.1)

These are the two terms appearing in the effective action (3.1.18). Any other fourderivative term can be written in terms of these, using integration by parts and the equations of motion. For example

$$\nabla_{\mu}\nabla_{\nu}\phi\nabla^{\mu}\phi\nabla^{\nu}\phi\phi \sim -\frac{1}{2}\left(\nabla\phi.\nabla\phi\right)^{2} ,$$

$$\left(\nabla_{\mu}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi\right)\phi^{2}\sim\left(\nabla\phi.\nabla\phi\right)^{2}-\nabla^{2}\nabla_{\mu}\phi\nabla^{\mu}\phi\phi^{2} .$$
(3.4.2)

Although at this level the independent integrands can be obtained by hand, they can also be nicely checked on a computer by using the algorithm of the previous section and converting to Mellin space where the integration by parts identities are made manifest. Simply list all possible four-derivative integrands on the computer, use the algorithm to obtain the corresponding integrand, convert them to Mellin amplitudes, and then solve for the independent ones.

We see here for the first time that the effective action has an ambiguity - a term not determined by the Virasoro-Shapiro amplitude: in the flat space limit the second integrand in (3.4.1) will vanish (as we can commute the Laplacian through so it acts directly on ϕ giving zero by the equations of motion) and so remains undetermined. The complete effective action at this order is thus (see (3.1.18))

$$S_{\alpha'^{5}} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^{5} \left(\zeta_{5} S_{\alpha'^{5}}^{\text{main}} + C_{0} S_{\alpha'^{5}}^{\text{amb}} + A_{2} S_{\alpha'^{3}}^{\text{main}}\right) , \qquad (3.4.3)$$

where

$$S_{\alpha'^{5}}^{\text{main}} = \frac{3}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} (\nabla\phi.\nabla\phi) (\nabla\phi.\nabla\phi) ,$$

$$S_{\alpha'^{5}}^{\text{amb}} = \frac{6}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} \nabla^{2} \nabla_{\mu}\phi \nabla^{\mu}\phi\phi^{2} ,$$

$$S_{\alpha'^{3}}^{\text{main}} = \frac{1}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y}\phi^{4} .$$
(3.4.4)

The ambiguity in the third line of (3.4.4) vanishes in the flat space limit since it comes from a $1/R^2$ expansion of the coefficient of the α'^3 correction in (3.1.18).

Replacing the scalar fields by bulk-to-boundary propagators and applying the covariant derivatives directly on them then gives a prediction for the half-BPS correlators at this order in α' . First consider the main contribution (3.4.4)

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{5};\text{main}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} d^{5} \hat{X} d^{5} \hat{Y} \frac{N_{12}N_{34} + N_{13}N_{24} + N_{14}N_{23}}{(P_{1} + Q_{1})^{5}(P_{2} + Q_{2})^{5}(P_{3} + Q_{3})^{5}(P_{4} + Q_{4})^{5}} \times 4^{4} , \quad (3.4.5)$$

where

$$N_{ij} = X_i \cdot X_j + Y_i \cdot Y_j + P_i P_j - Q_i Q_j . (3.4.6)$$

This can then be straightforwardly expanded to give any correlator directly and explicitly in position space in terms of AdS and S contact diagram functions, as is done for a general integral in (3.3.10). The corresponding Mellin amplitude can also be read off directly from (3.3.11)

$$\mathcal{M}_{\alpha'^{5}}^{\text{main}} = 4 \left[\left(\Sigma_{p} - 1 \right)_{5} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) + \left(\Sigma_{p} - 1 \right)_{4} \left(-10 \left(\tilde{s} \, \mathbf{s} + \tilde{t} \, \mathbf{t} + \tilde{u} \, \mathbf{u} \right) - 5 \left(c_{s} \, \mathbf{s} + c_{t} \, \mathbf{t} + c_{u} \, \mathbf{u} \right) \right) + \left(\Sigma_{p} - 1 \right)_{3} \left(20 \left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) + 4 \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) + 20 \left(\tilde{s} \, c_{s} + \tilde{t} \, c_{t} + \tilde{u} \, c_{u} \right) \right) + \left(\Sigma_{p} - 1 \right)_{3} \left(-12 \, \Sigma_{p}^{2} \right) \right].$$
(3.4.7)

Here we have used (3.3.11) to obtain the Mellin amplitude (with $\Delta_i = 4, d = 4$ and

 $p_i \rightarrow p_i - 2$) and then solved the constraints

$$\sum_{i < j} \delta_{ij} = p_j + 2 , \qquad \sum_{i < j} d_{ij} = p_j - 2 , \qquad (3.4.8)$$

in terms of new variables (s, t, u) and $(\tilde{s}, \tilde{t}, \tilde{u})$, which are defined as follows [15]:

$$\begin{split} \delta_{12} &= -s + c_s \ , & \delta_{14} &= -t + c_t \ , & \delta_{13} &= -u \ , \\ \delta_{23} &= -t \ , & \delta_{24} &= -u + c_u \ , & \delta_{34} &= -s \ , \\ d_{12} &= \tilde{s} + c_s \ , & d_{14} &= \tilde{t} + c_t \ , & d_{13} &= \tilde{u} \ , \\ d_{23} &= \tilde{t} \ , & d_{24} &= \tilde{u} + c_u \ , & d_{34} &= \tilde{s} \ , \\ \mathbf{s} &= s + \tilde{s} \ , & \mathbf{t} &= t + \tilde{t} \ , & \mathbf{u} &= u + \tilde{u} \ , & (3.4.9) \end{split}$$

where $s + t + u = -p_3 - 2$, $\tilde{s} + \tilde{t} + \tilde{u} = p_3 - 2$ and $\mathbf{s} + \mathbf{t} + \mathbf{u} = -4$. We also define

$$c_s = \frac{p_1 + p_2 - p_3 - p_4}{2} , \quad c_t = \frac{p_1 + p_4 - p_2 - p_3}{2} , \quad c_u = \frac{p_2 + p_4 - p_3 - p_1}{2} . \quad (3.4.10)$$

Note that for any CFT with a string theory or M-theory dual, the leading terms of the Mellin amplitude in the limit $s, t \rightarrow \infty$ can be compared to the appropriate higher-dimensional string-/M-theory scattering amplitude. In our case, the first line in (3.4.7) is leading in the flat space limit and is fixed from the flat space VS amplitude.

Now let us take a closer look at the ambiguity in the second line of (3.4.4). Using the equations of motion (3.1.36), the integrand can be written as

$$\nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2 = \left[\nabla_{\hat{X}}^2, \nabla_A \right] \phi \nabla^A \phi \phi^2 + \left[\nabla_{\hat{Y}}^2, \nabla_I \right] \phi \nabla^I \phi \phi^2 .$$
(3.4.11)

Moreover, after some algebra we find that

$$\left[\nabla_{\hat{X}}^{2}, \nabla_{A}\right]\phi = -d\nabla_{A}\phi , \quad \left[\nabla_{\hat{Y}}^{2}, \nabla_{I}\right]\phi = d\nabla_{I}\phi , \qquad (3.4.12)$$

so the ambiguity can be written as

$$\nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2 = -d \left(\left(\nabla_{\hat{X}} \phi \right)^2 - \left(\nabla_{\hat{Y}} \phi \right)^2 \right) \phi^2 . \tag{3.4.13}$$

The corresponding Witten diagram expression is given by

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{5};\text{amb}} = -\frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} \frac{d^{5}\hat{X}d^{5}\hat{Y}}{\prod_{i} (P_{i}+Q_{i})^{4}} \sum_{i< j} \frac{L_{ij}}{(P_{i}+Q_{i})(P_{j}+Q_{j})} \times 4^{3} ,$$
(3.4.14)

where

$$L_{ij} = X_i X_j + P_i P_j - Y_i Y_j + Q_i Q_j . (3.4.15)$$

This takes a very simple form in Mellin space

$$\mathcal{M}_{\alpha'^5}^{\text{amb}} = 4\left(\Sigma_p - 1\right)_3 \left(c_s^2 + c_t^2 + c_u^2 + \Sigma_p^2 - 16\right) . \tag{3.4.16}$$

Moreover, after multiplying the α'^3 term in (3.2.5) by $(\alpha'/(2R^2))^2$ (where we set R = 1), it can be thought of as an additional ambiguity at α'^5 , which is the origin of the third line in (3.4.4) as we have seen before. Restoring the prefactors in (3.4.3), the α'^5 correction to the Mellin amplitude for half-BPS correlators can be written as a sum over three terms:

$$\mathcal{M}_{\alpha'^5} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^5 \left(\zeta_5 \mathcal{M}_{\alpha'^5}^{\mathrm{main}} + C_0 \mathcal{M}_{\alpha'^5}^{\mathrm{amb}} + A_2 \mathcal{M}_{\alpha'^3}^{\mathrm{main}}\right) , \qquad (3.4.17)$$

where $\mathcal{M}_{\alpha^{\prime 3}}^{\text{main}} = (\Sigma_p - 1)_3$ (it is given in (3.2.5) but here we take it without the explicit normalisation there). The coefficients of the subleading terms can be fixed by comparing to the localisation result in [12] and are given by

$$C_0 = -\frac{3}{2}\zeta_5 , \quad A_2 = -30\zeta_5 .$$
 (3.4.18)

We find perfect agreement with the results from bootstrap methods of [14] (rewritten in this notation in [15])⁷.

3.5 α'^6 Corrections

At order α'^6 we have to consider all possible terms in the effective action involving six derivatives. Using a computer, it is straightforward to enumerate all possibilities

 $^{^7\}mathrm{We}$ thank Francesco Aprile for explicitly checking this agreement.

and compute their Mellin amplitudes using the algorithm explained in section 3.3 to find all linearly independent terms. After doing so, we find that there are only two linearly independent terms involving six derivatives:

$$(\nabla\phi.\nabla\phi)(\nabla_{\mu}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi)$$
 and $\nabla_{\mu}\nabla^{2}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi\phi^{2}$, (3.5.1)

which appear in the effective action in (3.1.18). The first term is the main correction at α'^6 while the second term is an ambiguity which vanishes in the flat space limit and is thus not determined by the flat space Virasoro-Shapiro amplitude.

The complete action at order α'^6 is given by (see (3.1.18))

$$S_{\alpha'^6} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^6 \left(2(\zeta_3)^2 S_{\alpha'^6}^{\text{main}} + E_0 S_{\alpha'^6}^{\text{amb}} + B_1 S_{\alpha'^5}^{\text{main}} + C_1 S_{\alpha'^5}^{\text{amb}} + A_3 S_{\alpha'^3}^{\text{main}}\right) , \quad (3.5.2)$$

where

$$S_{\alpha'^{6}}^{\text{main}} = \frac{1}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} (\nabla\phi.\nabla\phi) (\nabla_{\mu}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi) ,$$

$$S_{\alpha'^{6}}^{\text{amb}} = \frac{6}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} \nabla_{\mu}\nabla^{2}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi\phi^{2} , \qquad (3.5.3)$$

and the rest was defined in (3.4.4) (in particular, they arise from taking all the terms contributing at α'^5 and multiplying them with $\alpha'/(2R^2)$ with unfixed numerical coefficients and setting R = 1). We then find that the main contribution to half-BPS correlators at this order is

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^6;\text{main}} = \frac{1}{4!} \frac{1}{6} \frac{(\mathcal{C}_4)^4}{(-2)^{16}} \int_{\text{AdS} \times \text{S}} \frac{d^5 \hat{X} d^5 \hat{Y}}{\prod_i (P_i + Q_i)^5} \left[\frac{N_{12} M_{34}}{(P_3 + Q_3) (P_4 + Q_4)} + \text{perms} \right] \times 4^4 \times 5^2 , \quad (3.5.4)$$

where the correlator is understood to come from the first line of (3.5.3), N_{ij} was defined in (3.4.6) and

$$M_{ij} = (X_i \cdot X_j + P_i P_j + Y_i \cdot Y_j - Q_i Q_j)^2 - \frac{1}{5} (P_i P_j - Q_i Q_j)^2 \quad . \tag{3.5.5}$$

Before discussing the Mellin amplitude of the main contribution, let us briefly de-

scribe the ambiguity whose integrand can be written as

$$\nabla_{\mu}\nabla^{2}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi\phi^{2} = -d\left(\left(\nabla_{A}\nabla_{B}\phi\right)^{2} - \left(\nabla_{I}\nabla_{J}\phi\right)^{2}\right)\phi^{2}, \qquad (3.5.6)$$

where A and I indices label \hat{X} and \hat{Y} coordinates, respectively. We obtained the right hand side by commuting the ∇^2 with ∇_{ν} and using the equations of motion as we did in the previous subsection. The Witten diagram expression associated with (3.5.6) is

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{6}; \text{amb}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS} \times \text{S}} \frac{d^{5} \hat{X} d^{5} \hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \sum_{i < j} \frac{K_{ij}}{(P_{i} + Q_{i})^{2} (P_{j} + Q_{j})^{2}} \times 4^{3} \times 5^{2} , \qquad (3.5.7)$$

where

$$K_{ij} = (X_i \cdot X_j + P_i P_j)^2 - (Y_i \cdot Y_j - Q_i Q_j)^2 - \frac{1}{5} \left((P_i P_j)^2 - (Q_i Q_j)^2 \right) .$$
(3.5.8)

Converting this to Mellin space gives the ambiguity

$$\mathcal{M}_{\alpha'^{6}}^{\text{amb}} = -32 \left[\left(\Sigma_{p} - 1 \right)_{5} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) + \left(\Sigma_{p} - 1 \right)_{4} \frac{1}{2} \left(\mathbf{s} c_{s}^{2} + \mathbf{t} c_{t}^{2} + \mathbf{u} c_{u}^{2} \right) - \left(\Sigma_{p} - 1 \right)_{4} \left(\Sigma_{p} + 3 \right) \left[2 \left(\mathbf{s} \, \tilde{s} + \mathbf{t} \, \tilde{t} + \mathbf{u} \, \tilde{u} \right) + \left(\mathbf{s} \, c_{s} + \mathbf{t} \, c_{t} + \mathbf{u} \, c_{u} \right) \right] - \left(\Sigma_{p} - 1 \right)_{3} \left(\left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3} \right) + 2 \left(c_{s}^{2} \, \tilde{s} + c_{t}^{2} \, \tilde{t} + c_{u}^{2} \, \tilde{u} \right) - 10 \, \Sigma_{p} \left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) \right) + \left(\Sigma_{p} - 1 \right)_{3} \left(10 \, \Sigma_{p} \left(\tilde{s} \, c_{s} + \tilde{t} \, c_{t} + \tilde{u} \, c_{u} \right) + 3 \, \Sigma_{p} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) \right) + \left(\Sigma_{p} - 1 \right)_{3} \left(-2 \, \Sigma_{p}^{3} - 16 \, \Sigma_{p} \right) \right].$$

$$(3.5.9)$$

Converting (3.5.4) to Mellin space, the Mellin amplitude of the main contribution is

$$\mathcal{M}_{\alpha'^6}^{\text{main}} = \hat{\mathcal{M}}_{\alpha'^6}^{\text{main}} - \frac{1}{12} \mathcal{M}_{\alpha'^6}^{\text{amb}}, \qquad (3.5.10)$$

where

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha^{\prime 6}}^{\text{main}} &= \frac{8}{3} \Big[\left(\Sigma_p - 1 \right)_6 \left(\mathbf{s}^3 + \mathbf{t}^3 + \mathbf{u}^3 \right) \\ &+ \left(\Sigma_p - 1 \right)_5 \left(6 \Sigma_p \left(\mathbf{s}^2 + \mathbf{t}^2 + \mathbf{u}^2 \right) - 18 \left(\mathbf{s}^2 \,\tilde{s} + \mathbf{t}^2 \,\tilde{t} + \mathbf{u}^2 \,\tilde{u} \right) \right) \end{aligned}$$

$$+ (\Sigma_{p}-1)_{5} \left(-9 \left(\mathbf{s}^{2} c_{s} + \mathbf{t}^{2} c_{t} + \mathbf{u}^{2} c_{u}\right)\right) + (\Sigma_{p}-1)_{4} \left(90 \left(\mathbf{s} \tilde{s}^{2} + \mathbf{t} \tilde{t}^{2} + \mathbf{u} \tilde{u}^{2}\right) + \frac{39}{2} \left(\mathbf{s} c_{s}^{2} + \mathbf{t} c_{t}^{2} + \mathbf{u} c_{u}^{2}\right)\right) + (\Sigma_{p}-1)_{4} \left(90 \left(\mathbf{s} \tilde{s} c_{s} + \mathbf{t} \tilde{t} c_{t} + \mathbf{u} \tilde{u} c_{u}\right) \right) + (\Sigma_{p}-1)_{4} \left(-60 \Sigma_{p} \left(\mathbf{s} \tilde{s} + \mathbf{t} \tilde{t} + \mathbf{u} \tilde{u}\right) - 30 \Sigma_{p} \left(\mathbf{s} c_{s} + \mathbf{t} c_{t} + \mathbf{u} c_{u}\right)\right) + (\Sigma_{p}-1)_{3} \left(-120 \left(\tilde{s}^{3} + \tilde{t}^{3} + \tilde{u}^{3}\right) - 9 \left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3}\right)\right) + (\Sigma_{p}-1)_{3} \left(-180 \left(\tilde{s}^{2} c_{s} + \tilde{t}^{2} c_{t} + \tilde{u}^{2} c_{u}\right) - 78 \left(c_{s}^{2} \tilde{s} + c_{t}^{2} \tilde{t} + c_{u}^{2} \tilde{u}\right)\right) + (\Sigma_{p}-1)_{3} \left(120 \Sigma_{p} \left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2}\right) + 27 \Sigma_{p} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2}\right)\right) + (\Sigma_{p}-1)_{3} \left(120 \Sigma_{p} \left(\tilde{s} c_{s} + \tilde{t} c_{t} + \tilde{u} c_{u}\right) - 50 \Sigma_{p}^{3} - 16 \Sigma_{p}\right)\right].$$
(3.5.11)

This Mellin amplitude shows a similar structure as (3.4.7). Every line is multiplied by a Pochhammer depending on the power of $\{\mathbf{s}, \mathbf{t}, \mathbf{u}\}$ and the rest is at most cubic in the variables $\{\mathbf{s}, \mathbf{t}, \mathbf{u}, \tilde{s}, \tilde{t}, \tilde{u}, c_s, c_t, c_u, \Sigma_p\}$. Additionally, after multiplying the three terms which span the α'^5 correction in (3.4.17) by α' , they become additional ambiguities at α'^6 , see the expansion (3.1.19). The complete Mellin amplitude for half-BPS correlators at order α'^6 can then be written as a sum over five terms:

$$\mathcal{M}_{\alpha'^{6}} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^{6} \left(2 \left(\zeta_{3}\right)^{2} \mathcal{M}_{\alpha'^{6}}^{\text{main}} + E_{0} \mathcal{M}_{\alpha'^{6}}^{\text{amb}} + B_{1} \mathcal{M}_{\alpha'^{5}}^{\text{main}} + C_{1} \mathcal{M}_{\alpha'^{5}}^{\text{amb}} + A_{3} \mathcal{M}_{\alpha'^{3}}^{\text{main}}\right), \quad (3.5.12)$$

where we restore the coefficients from (3.5.2).⁸

We can fix two of the coefficients by comparing the Mellin amplitude to the result from localisation in [97, 98]. To compare (3.5.12) to [97] we take $s \rightarrow \frac{s}{2} - 2$, $t \rightarrow \frac{t}{2} - 2$ and specialise to $p_i = 2$ (where $\tilde{s} = \tilde{t} = \tilde{u} = 0$):

$$\mathcal{M}_{\alpha'^{6}}^{2222} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^{6} \times 60 \left(672(\zeta_{3})^{2} s t u + 14 \left(3 B_{1} + 4 \left((\zeta_{3})^{2} - 6 E_{0}\right)\right) \left(s^{2} + t^{2} + u^{2}\right) + A_{3} - 96 B_{1} + 768 E_{0} - 3200(\zeta_{3})^{2}\right), \qquad (3.5.13)$$

where u = 4 - s - t. We can now compare this expression to the result in [97] and

⁸Note that the number of ambiguities is consistent with the number obtained via the bootstrap method. We thank Francesco Aprile, James Drummond, Hynek Paul and Michele Santagata for discussions on this.

partially fix the coefficients to

$$E_0 = \frac{B_1}{8} + \frac{2(\zeta_3)^2}{3} , \quad A_3 = 0 , \qquad (3.5.14)$$

which leads to the α'^6 correction to the correlator for $p_i = 2$:

$$\mathcal{M}_{\alpha'^6}^{2222} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^6 \times 2(\zeta_3)^2 \times (3)_6 \left[st\,u - \frac{1}{4}\left(s^2 + t^2 + u^2\right) - 4\right] \,. \tag{3.5.15}$$

It is noteworthy that localisation predicts the coefficient $A_3 = 0$. Localisation also predicts the absence of any α'^4 corrections i.e. $A_1 = 0$ [12]. Indeed, as we discuss in the conclusions, it is natural to expect all odd terms in the expansion of the coefficients in α'/R^2 (see (3.1.19)) to vanish in which case we would have $B_1 = C_1 = 0$ and then the α'^6 correction to the Mellin amplitude in (3.5.12) is completely fixed.

3.6 α'^7 Corrections

Using the algorithm explained in section 3.3 to find all linearly independent terms in the effective action involving eight derivatives, we find that there are six independent terms, notably the main contribution

$$(\nabla_{\mu}\nabla_{\nu}\phi\nabla^{\mu}\nabla^{\nu}\phi)^2 , \qquad (3.6.1)$$

and five ambiguities:

$$(\nabla^{\mu}\nabla^{\nu}\nabla_{\mu}\nabla^{\rho}\nabla^{\sigma}\nabla_{\rho}\phi) (\nabla_{\nu}\nabla_{\sigma}\phi) \phi^{2} , \quad (\nabla^{2}\nabla^{\mu}\nabla^{\nu}\nabla^{\rho}\nabla_{\nu}\phi) (\nabla_{\mu}\nabla_{\rho}\phi) \phi^{2} , (\nabla^{2}\nabla^{\mu}\nabla^{\nu}\nabla^{\rho}\nabla_{\nu}\phi) (\nabla_{\rho}\phi) (\nabla_{\mu}\phi) \phi , \quad (\nabla^{\mu}\nabla^{\nu}\nabla^{\rho}\nabla_{\nu}\nabla_{\rho}\phi) (\nabla^{\sigma}\nabla_{\mu}\phi) (\nabla_{\sigma}\phi) \phi , (\nabla^{\mu}\nabla^{\nu}\nabla^{\rho}\nabla^{\sigma}\nabla_{\rho}\phi) (\nabla_{\mu}\nabla_{\sigma}\phi) (\nabla_{\nu}\phi) \phi .$$

$$(3.6.2)$$

See appendix D for details on the ambiguities. The complete effective action at this order is then given by (see (3.1.18))

$$S_{\alpha'^7} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^7 \left(\frac{1}{2} \zeta_7 S_{\alpha'^7}^{\text{main}} + G_{1;0} S_{\alpha'^7}^{\text{amb}_1} + G_{2;0} S_{\alpha'^7}^{\text{amb}_2} + G_{3;0} S_{\alpha'^7}^{\text{amb}_3} + G_{4;0} S_{\alpha'^7}^{\text{amb}_4} + G_{5;0} S_{\alpha'^7}^{\text{amb}_5} + D_1 S_{\alpha'^6}^{\text{main}} + E_1 S_{\alpha'^6}^{\text{amb}} + B_2 S_{\alpha'^5}^{\text{main}} + C_2 S_{\alpha'^5}^{\text{amb}} + A_4 S_{\alpha'^3}^{\text{main}}\right),$$
(3.6.3)

where the main contribution is

$$S_{\alpha^{\prime 7}}^{\text{main}} = \frac{6}{4!} \int_{\text{AdS} \times \text{S}} d^5 \hat{X} d^5 \hat{Y} (\nabla_\mu \nabla_\nu \phi \nabla^\mu \nabla^\nu \phi)^2 , \qquad (3.6.4)$$

and the contributions from the five α'^7 ambiguities in (3.6.2) to the effective action are given in appendix D together with their Witten diagram expressions and Mellin amplitudes. The contributions from lower α' orders were defined in (3.4.4) and (3.5.3). The prediction for the main contribution to the half-BPS correlator in position space at this order is:

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7};\text{main}} = \frac{2}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} \frac{d^{5}\hat{X}d^{5}\hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{6}} [M_{12}M_{34} + \text{perms}] \times 4^{4} \times 5^{4},$$
(3.6.5)

where M_{ij} was defined in (3.5.5). The Mellin amplitude of the main contribution is

$$\mathcal{M}_{\alpha'^{7}}^{\text{main}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{main}} + \frac{63}{8} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{1}} - \frac{31}{4} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{2}} - \frac{25}{32} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{3}} - \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{4}} - 32 \mathcal{M}_{\alpha'^{5}}^{\text{main}} - \frac{85}{2} \mathcal{M}_{\alpha'^{5}}^{\text{amb}} - 1024 \mathcal{M}_{\alpha'^{3}}^{\text{main}} , \qquad (3.6.6)$$

where $\hat{\mathcal{M}}_{\alpha'^7}^{\mathrm{main}}$ is:

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha'7}^{\text{main}} &= 32 \left[\left(\Sigma_p - 1 \right)_7 \left(\mathbf{s}^4 + \mathbf{t}^4 + \mathbf{u}^4 \right) \right. \\ &+ \left(\Sigma_p - 1 \right)_6 \left(8 \Sigma_p \left(\mathbf{s}^3 + \mathbf{t}^3 + \mathbf{u}^3 \right) - 28 \left(\mathbf{s}^3 \,\tilde{s} + \mathbf{t}^3 \,\tilde{t} + \mathbf{u}^3 \,\tilde{u} \right) - 14 \left(\mathbf{s}^3 \,c_s + \mathbf{t}^3 \,c_t + \mathbf{u}^3 \,c_u \right) \right) \\ &+ \left(\Sigma_p - 1 \right)_5 \left(\Sigma_p \left(26 \,\Sigma_p + 9 \right) \left(\mathbf{s}^2 + \mathbf{t}^2 + \mathbf{u}^2 \right) + 252 \left(\mathbf{s}^2 \,\tilde{s}^2 + \mathbf{t}^2 \,\tilde{t}^2 + \mathbf{u}^2 \,\tilde{u}^2 \right) \right. \\ &+ 252 \left(\mathbf{s}^2 \,\tilde{s} \,c_s + \mathbf{t}^2 \,\tilde{t} \,c_t + \mathbf{u}^2 \,\tilde{u} \,c_u \right) + 57 \left(\mathbf{s}^2 \,c_s^2 + \mathbf{t}^2 \,c_t^2 + \mathbf{u}^2 \,c_u^2 \right) \\ &- 144 \,\Sigma_p \left(\mathbf{s}^2 \,\tilde{s} + \mathbf{t}^2 \,\tilde{t} + \mathbf{u}^2 \,\tilde{u} \right) - 72 \,\Sigma_p \left(\mathbf{s}^2 \,c_s + \mathbf{t}^2 \,c_t + \mathbf{u}^2 \,c_u \right) \right) \\ &+ \left(\Sigma_p - 1 \right)_4 \left(- 840 \left(\mathbf{s} \,\tilde{s}^3 + \mathbf{t} \,\tilde{t}^3 + \mathbf{u} \,\tilde{u}^3 \right) - 75 \left(\mathbf{s} \,c_s^3 + \mathbf{t} \,c_t^3 + \mathbf{u} \,c_u^3 \right) \\ &- 1260 \left(\mathbf{s} \,\tilde{s}^2 \,c_s + \mathbf{t} \,\tilde{t}^2 \,c_t + \mathbf{u} \,\tilde{u}^2 \,c_u \right) - 570 \left(\mathbf{s} \,c_s^2 \,\tilde{s} + \mathbf{t} \,c_t^2 \,\tilde{t} + \mathbf{u} \,c_u^2 \,\tilde{u} \right) \\ &+ 720 \,\Sigma_p \left[\left(\mathbf{s} \,\tilde{s}^2 + \mathbf{t} \,\tilde{t}^2 + \mathbf{u} \,\tilde{u}^2 \right) + \left(\mathbf{s} \,\tilde{s} \,c_s + \mathbf{t} \,\tilde{t} \,c_t + \mathbf{u} \,\tilde{u} \,c_u \right) \right] \\ &+ \frac{1}{2} \left(336 \,\Sigma_p - 1 \right) \left(\mathbf{s} \,c_s^2 + \mathbf{t} \,\tilde{t}^2 + \mathbf{u} \,c_u^2 \right) \\ &- \Sigma_p \left(139 \,\Sigma_p + 27 \right) \left[2 \left(\mathbf{s} \,\tilde{s} + \mathbf{t} \,\tilde{t} + \mathbf{u} \,\tilde{u} \right) + \left(\mathbf{s} \,c_s + \mathbf{t} \,c_t + \mathbf{u} \,c_u \right) \right] \right) \\ &+ \left(\Sigma_p - 1 \right)_3 \left(840 \left(\tilde{s}^4 + \tilde{t}^4 + \tilde{u}^4 \right) + \frac{191}{8} \left(c_s^4 + c_t^4 + c_u^4 \right) + 1680 \left(\tilde{s}^3 \,c_s + \tilde{t}^3 \,c_t + \tilde{u}^3 \,c_u \right) \right) \end{aligned}$$

$$+ 1140 \left(\tilde{s}^{2} c_{s}^{2} + \tilde{t}^{2} c_{t}^{2} + \tilde{u}^{2} c_{u}^{2}\right) + 300 \left(c_{s}^{3} \tilde{s} + c_{t}^{3} \tilde{t} + c_{u}^{3} \tilde{u}\right) - 960 \Sigma_{p} \left(\tilde{s}^{3} + \tilde{t}^{3} + \tilde{u}^{3}\right) - \frac{191}{2} \Sigma_{p} \left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3}\right) - 1440 \Sigma_{p} \left(\tilde{s}^{2} c_{s} + \tilde{t}^{2} c_{t} + \tilde{u}^{2} c_{u}\right) - 671 \Sigma_{p} \left(c_{s}^{2} \tilde{s} + c_{t}^{2} \tilde{t} + c_{u}^{2} \tilde{u}\right) + 610 \Sigma_{p}^{2} \left[\left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2}\right) + \left(\tilde{s} c_{s} + \tilde{t} c_{t} + \tilde{u} c_{u}\right) \right] + \frac{573}{4} \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2}\right) - \frac{1471}{8} \Sigma_{p}^{4} - 116 \Sigma_{p}^{2} \right) \right], \qquad (3.6.7)$$

and the Mellin amplitudes of the ambiguities are given in appendix D. Note that this exhibits a similar structure to (3.4.7) and (3.5.11), since every line is multiplied by a Pochhammer depending on the power of $\{\mathbf{s}, \mathbf{t}, \mathbf{u}\}$ and the rest is at most quartic in the variables $\{\mathbf{s}, \mathbf{t}, \mathbf{u}, \tilde{s}, \tilde{t}, \tilde{u}, c_s, c_t, c_u, \Sigma_p\}$. Collecting all possible contributions at this order, the complete Mellin amplitude for the half-BPS correlator at α'^7 is given by eleven terms:

$$\mathcal{M}_{\alpha'^{7}} = \frac{1}{8} \left(\frac{\alpha'}{2}\right)^{7} \left(\frac{1}{2} \zeta_{7} \mathcal{M}_{\alpha'^{7}}^{\text{main}} + G_{1;0} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{1}} + G_{2;0} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{2}} + G_{3;0} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{3}} + G_{4;0} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{4}} + G_{5;0} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{5}} + D_{1} \mathcal{M}_{\alpha'^{6}}^{\text{main}} + E_{1} \mathcal{M}_{\alpha'^{6}}^{\text{amb}} + B_{2} \mathcal{M}_{\alpha'^{5}}^{\text{main}} + C_{2} \mathcal{M}_{\alpha'^{5}}^{\text{amb}} + A_{4} \mathcal{M}_{\alpha'^{3}}^{\text{main}} \right) . \quad (3.6.8)$$

The coefficients of the subleading terms remain unfixed at this order, to fix them we would need additional information. As an example, let us look at the lowest-charge correlator with $p_i = 2$ (as in the previous section we shift $s \rightarrow \frac{s}{2} - 2$, $t \rightarrow \frac{t}{2} - 2$):

$$\mathcal{M}_{\alpha'^{7}}^{2222} = \frac{1}{8} \left(\frac{\alpha'}{2} \right)^{7} \times 60 \left(a_{1} \left(s^{2} + t^{2} + u^{2} \right)^{2} + a_{2} s t u + a_{3} \left(s^{2} + t^{2} + u^{2} \right) + a_{4} \right), \quad (3.6.9)$$

with u = 4 - s - t and

$$a_{1} = 1512 \zeta_{7}, \quad a_{2} = 336 \left(D_{1} + 48 \left(G_{5;0} - 2 \zeta_{7} \right) \right) ,$$

$$a_{3} = 42 B_{2} + 28 \left(D_{1} - 6 \left(2 E_{1} - 18 G_{1;0} - 20 G_{2;0} + 40 G_{3;0} - 12 G_{5;0} + 23 \zeta_{7} \right) \right) ,$$

$$a_{4} = A_{4} - 32 \left(3 B_{2} + 50 D_{1} - 12 \left(2 E_{1} - 18 G_{1;0} - 20 G_{2;0} + 40 G_{3;0} - 204 G_{5;0} + 335 \zeta_{7} \right) \right) .$$

$$(3.6.10)$$

3.7 Conclusions and Future Directions

To conclude, in this chapter, we have postulated a simple effective field theory in ten dimensions describing all four-point tree-level string interactions in $AdS_5 \times S^5$. To obtain the interaction terms in the $AdS_5 \times S^5$ effective action we start from the flat space VS amplitude written as an infinite series in small α' , where the leading term is supergravity and all subleading terms describe string corrections and can be obtained from a scalar effective action in flat space. Lifting the interaction terms in this flat space action to $AdS_5 \times S^5$ by replacing the flat space derivatives with 10d covariant derivatives containing both AdS and S derivatives we obtain the curved space analogue of the scalar effective field theory. The main new tool to obtain correlators from this effective action was to use a new formulation of Witten diagrams and the Mellin transform which is manifestly 10d and treats AdS and S on equal footing. We have shown that this simple description reproduces previous results for all four-point correlators of half-BPS operators in \mathcal{N} = 4 SYM up to order α'^5 , and have proposed a general algorithm for extending this to arbitrarily high order. From this algorithm we obtained new predictions at α'^6 and α'^7 . The coefficients of the effective action can be determined by comparing to the flat space VS amplitude, although there are curvature-dependent ambiguities which cannot be fixed in this way and need additional input from other methods such as localisation. After fixing all the coefficients in the effective action, the 10d Mellin amplitudes derived from it can be thought of as the analogue of the VS amplitude in $AdS_5 \times S^5$.

Note that in the considerations above we have focused on the limit of tree-level string theory for which all orders in the α' effective action are known in flat space. However, the coefficients of the first three terms in the flat space effective action (3.1.17) (i.e. up to $\partial^6 \phi^4$) are actually known at the full non-perturbative level as functions of the string coupling [111–115]⁹. These results imply that the coefficients in (3.1.19) can be promoted to full functions of the complex Yang-Mills coupling $\tau = \theta/(2\pi) + 4\pi i N \alpha'^2$

 $^{^{9}\}mathrm{We}$ thank Congkao Wen for drawing our attention to this.

(see (3.1.2), where θ is the Yang-Mills theta-angle and we recall that $N\alpha'^2 = g_{YM}^{-2}$ if we set the AdS radius R = 1). Specifically they are promoted as

$$\begin{pmatrix} \frac{\alpha'}{2} \end{pmatrix}^{3} A_{0} = \left(\frac{\alpha'}{2}\right)^{3} 2\zeta_{3} \rightarrow \frac{1}{(2^{4}\pi N)^{3/2}} \times E\left(\frac{3}{2}, \tau, \bar{\tau}\right) , \\ \left(\frac{\alpha'}{2}\right)^{5} B_{0} = \left(\frac{\alpha'}{2}\right)^{5} \zeta_{5} \rightarrow \frac{1}{(2^{4}\pi N)^{5/2}} \times \frac{1}{2} E\left(\frac{5}{2}, \tau, \bar{\tau}\right) , \\ \left(\frac{\alpha'}{2}\right)^{6} D_{0} = \left(\frac{\alpha'}{2}\right)^{6} 2(\zeta_{3})^{2} \rightarrow \frac{1}{(2^{4}\pi N)^{3}} \times 3 \mathcal{E}\left(3, \frac{3}{2}, \frac{3}{2}, \tau, \bar{\tau}\right) ,$$
 (3.7.1)

where A_0, B_0, D_0 are the leading coefficients in the first, second, and fourth lines of the left column in (3.1.19),

$$E(s,\tau,\bar{\tau}) = 2\zeta_{2s}(\Im(\tau))^{s}(1+...)$$
(3.7.2)

are non-holomorphic Eisenstein series and

$$\mathcal{E}(3, \frac{3}{2}, \frac{3}{2}, \tau, \bar{\tau}) = \frac{2}{3} (\zeta_3)^2 (\Im(\tau))^3 (1 + \dots)$$
(3.7.3)

is a generalised Eisenstein series. In the above two equations the ellipses denote perturbative and non-perturbative terms which vanish when $\Im(\tau) \to \infty$. The precise definitions of the functions can be found for example in [98]. The modular functions in (3.7.1) are a consequence of the $SL(2,\mathbb{Z})$ symmetry of IIB string theory (see subsection 3.1.1), which can be understood from compactifying M-theory on a torus and identifying the IIB coupling τ with the complex structure of the torus [116, 117].

Furthermore, recently in [98, 110, 118] the corresponding dual (but lowest-charge only) correlators were considered and completely fixed via localisation to all orders. This then fixes the remaining ambiguities at this order (assuming $B_1 = C_1 = 0$ as we discuss around (3.5.13) and the second bullet point below) in terms of the above functions as

$$C_0 = -\frac{3}{2}B_0$$
, $A_2 = -30B_0$, $E_0 = \frac{D_0}{3}$. (3.7.4)

These relations follow from the earlier results in (3.4.18) and (3.5.14), respectively. In summary, the 10d effective action in (3.1.18) appears to be a very useful way to describe IIB string theory in $AdS_5 \times S^5$ and a powerful tool for computing four-point correlators in $\mathcal{N} = 4$ SYM. We will propose a similar effective action for $AdS_2 \times S^2$ in chapter 5.

Future directions

Let us now discuss some interesting questions for future research.

- Firstly, we have not proven the existence of the 10d scalar effective field theory but rather justified it by showing that it reproduces known results for string corrections to IIB supergravity in AdS₅×S⁵ which were previously obtained using bootstrap methods in [11–15]. Hence, it would be very interesting to prove the existence of this local effective field theory and to do so one would consider the superspace formalism. As on flat background [93], IIB supergravity linearised on the AdS×S superspace background is again described by a chiral scalar superfield with a certain fourth-order constraint [119, 120]. It presumably then makes sense to integrate a superpotential consisting of a holomorphic function of this scalar in chiral AdS×S superspace. This then leads to the question of the existence of an effective chiral superpotential describing the full nonlinear theory. Such an object has been discussed before [121–123] and it would be interesting to explore this point further.
- As we have seen above, the effective action has ambiguities corresponding to curvature corrections which vanish in the flat space limit. For low orders in α' , we find that these ambiguities can be fixed by comparing to results from localisation. It would be interesting to understand whether one could find a systematic way to fix all the ambiguities. If this were possible, the next question would be whether we can resum the α' expansion to obtain a compact form analogous to the flat space VS amplitude. If so, how does the analytic structure become modified in curved background? Note here that, as observed below (3.5.13), the explicit results for these ambiguities obtained via

localisation are completely consistent with all odd powers in the expansion of α'/R^2 vanishing. Since the curvature has opposite sign for AdS (~ $-1/R^2$) and S (~ $1/R^2$), it is perhaps quite natural to expect that only even powers of the curvature should contribute.

- Another important direction would be to extend this approach to other backgrounds. As explained above, in $\mathcal{N} = 4$ SYM the supersymmetry factors out of certain correlators in a very simple way making it possible to derive them from a 10d scalar theory in the bulk. We expect this factorisation to hold when the bulk geometry is $AdS_p \times S^q$ with p = q, but not when $p \neq q$. And we will propose the existence of such an action for $AdS_2 \times S^2$ in chapter 5 and deduce the consequences. For $AdS_q \times S^q$ with q = 3, 5, it was recently shown that supergravity correlators enjoy conformal symmetry which can be used to lift the lowest-charge half-BPS four-point correlator to all higher-charge correlators [41, 124, 125]. It would be interesting to investigate the relation of this higher-dimensional conformal symmetry with the explicit higher-dimensional integrals (AdS×S Witten diagrams) we write down here. We will investigate this further in chapter 5 where we study holographic correlators in $AdS_2 \times S^2$ in the context of a 4d hidden conformal symmetry as well as derive higherderivative corrections from a 4d scalar effective action analogous to the one in this chapter.
- It would also be interesting to extend this approach to higher-point correlators. The four-point AdS×S contact diagrams have a direct generalisation to npoints. Note that an important feature of half-BPS four-point correlators in $\mathcal{N} = 4$ SYM that allowed us to write down a simple effective action was the ability to factor out a polynomial which encodes all the supersymmetry. This is analogous to factoring out a supersymmetric delta-function $\delta^{16}(Q)$ from a maximally supersymmetric four-point superamplitude in flat space. Therefore, it is not obvious how to generalise this approach to generic *n*-point functions.

However, there are specific cases where similar properties hold as for example for the *n*-point maximally nilpotent correlators – those with fermionic degree n - 4 [126–128] which have recently been studied at strong coupling in [110]. Thus, one might expect them to be computable from a 10d scalar effective action just as for the four-point ones.

- It would also be conceptually very satisfying to derive the effective action directly from CFT without assuming local spacetime description in the bulk. A systematic approach to such a derivation was achieved in the context of a toy model consisting of a scalar field in AdS in [30] using crossing and conformal symmetry of boundary CFT correlators. This calculation was adapted to stress tensor correlators in N = 4 SYM in [61]. The fact that IIB string theory in AdS₅×S⁵ can be reduced to a simple 10d effective field theory therefore suggests that this program might be realised for a full-blown theory of quantum gravity.
- In recent years, loop corrections to the amplitudes have been obtained via bootstrap methods on the CFT side of the duality [11, 85, 87–92]. It would be very interesting to perform loop computations directly on the gravitational side. This is usually technically very difficult and one could hope that the simplicity at tree-level uncovered from the approach in this chapter could help to give new insight also at loop-level.

Chapter 4

AdS₇×S⁴: Recursion Relations for Anomalous Dimensions in the 6d (2,0) Theory

This chapter is based on [43] and we mainly follow the structure of this paper. We study anomalous dimensions of double-trace operators in the conformal block expansion of stress tensor correlators in the 6d (2,0) theory. As mentioned in the introduction, the 6d (2,0) theory is dual to M-theory in $AdS_7 \times S^4$ and studying the holographic correlators is a promising way of understanding M-theory better. In this chapter we are interested in the stable 5d objects whose worldvolume theory is the 6d (2,0) theory, the M5-branes. In the low-energy limit, $l_P \rightarrow 0$, M-theory can be approximated by 11d supergravity which on the CFT side of the duality corresponds to studying correlation functions in the limit $N \rightarrow \infty$, since in $AdS_7 \times S^4$ we have the relation $l_P \sim N^{-1/3}$ and N is the number of M5-branes. Recent progress has been made in computing all supergravity tree-level correlators in $AdS_7 \times S^4$ in Mellin space using constraints from the symmetry of the problem and the analytic properties of the amplitudes [129, 130]. When going away from the strict large-N limit and studying subleading terms we can learn something about higher-derivative corrections to the 11d supergravity effective action. Since the CFT in question is non-Lagrangian and can therefore not be studied perturbatively, using conformal bootstrap methods is a very promising approach (see chapter 2 for a review on conformal bootstrap). We will use these methods to compute anomalous dimensions of the operators in the conformal block expansion which will in turn tell us something about the higher-derivative corrections.

Our main strategy, adapted from the seminal work of [30], is to expand the crossing equations in the inverse central charge c, where it has been shown using holographic methods, that $c \sim N^3$ [131]. Next, one takes a certain limit of the conformal cross-ratios, the so-called light-cone limit, to isolate the terms in the conformal block expansion corresponding to anomalous dimensions of double-trace operators. We then truncate the conformal block expansion in spin and use an orthogonality relation of the hypergeometric functions in the superconformal blocks to derive a recursion relation for the anomalous dimensions. For truncated spin L, we find that the solution to the recursion relation depends on (L + 2)(L + 4)/8 free parameters, in agreement with holographic arguments of [30] (see section 2.3.2) and with the explicit four-point functions found in [26]. In particular, they can be thought of as the coefficients of higher-derivative corrections to supergravity in AdS₇×S⁴ arising from M-theory [26] (see [10, 61] for similar results in $\mathcal{N} = 4$ SYM).

A strategy for fixing these coefficients using a chiral algebra conjecture [132] was proposed in [28]. Moreover, the M-theory effective action can also be deduced from correlators of the ABJM theory [3], which is dual to M-theory in $AdS_4 \times S^7$ [27, 29]. As a warm-up for our analysis in the (2,0) theory, we first derive recursion relations for anomalous dimensions in an abstract non-supersymmetric 6d CFT, which we match against the conformal block expansion of Witten diagrams for a massive scalar field in AdS_7 . The recursion relations we obtain for both the toy model and the (2,0) theory can be efficiently solved using a computer.

We start with a brief review of the 6d (2,0) Theory/M-Theory in AdS₇×S⁴ correspondence, followed by the derivation of recursion relations for anomalous dimensions in a 6d toy model. Finally, we consider the 6d (2,0) theory, where we compute anom-

alous dimensions which encode the higher-derivative corrections to the supergravity effective action.

4.1 Review: 6d (2,0) Theory/M-Theory in $AdS_7 \times S^4$

In this section we briefly review some aspects of the correspondence between the 6d (2,0) theory and M-theory in $AdS_7 \times S^4$. Other important concepts used in this chapter are conformal bootstrap methods and higher-derivative corrections which were both reviewed in chapter 2. For a more detailed review on M-theory see e.g. [51].

The 6d (2,0) theory is the six-dimensional worldvolume theory of M5-branes with $OSp(8^*|4)$ symmetry, which is the maximal supersymmetry in 6d. The worldvolume theory of a single M5-brane can be formulated in terms of an abelian (2,0) tensor multiplet [133–135]. The field content in this case consists of five scalars ϕ^I , eight fermions and a self-dual two-form gauge field. Generalising this to a higher number of branes and thus considering the interacting 6d(2,0) theory is very difficult. It is believed to be non-Lagrangian, since a 6d local Lagrangian can be ruled out by powercounting as it would contain non-renormalisable and unbounded interactions. In [19] a 5d Lagrangian, which is believed to capture the full 6d physics, was proposed. The 6d (2,0) theory is manifestly non-perturbative and the only tunable parameter is N, which is the number of M5-branes. Therefore, a very promising approach to study correlators in this theory is to constrain the CFT data using conformal bootstrap methods. This in turn will tell us something about the bulk dual, Mtheory in $AdS_7 \times S^4$, or more specifically about the higher-derivative corrections to the low-energy approximation, which is 11d supergravity. Note that higher-derivative corrections can also be deduced from uplifting string calculations [111, 136, 137]. Let us briefly review the field content and action of 11d supergravity before we discuss holographic correlators in this example of the AdS/CFT duality.

4.1.1 11d Supergravity

We start by considering the field content of 11d supergravity, where we focus on the bosonic fields for our considerations. Firstly, the graviton which is a supersymmetric traceless tensor of the symmetry group SO(9) has 44 bosonic degrees of freedom, which counts the number of independent components of a symmetric 9×9 matrix subtracting one because of tracelessness. For the theory to be supersymmetric, the number of physical bosonic and fermionic degrees of freedom needs to be the same. The only fermion field in the theory is the gravitino Ψ_M which has, in addition to the vector index, an implicit spinor index. It is a 32-component Majorana spinor for each value of the index M and has 128 fermionic degrees of freedom (see e.g. [51]) for more details on the fermionic degrees of freedom). To get the right number of bosonic degrees of freedom (compared to 128 fermionic ones) one needs to include a rank-3 antisymmetric tensor, represented by a three-form A_3 . The theory is then invariant under gauge transformations $A_3 \rightarrow A_3 + d\Lambda_2$, where Λ_2 is a two-form. As a consequence of the gauge invariance the indices for the independent physical polarisations are transverse (as for any antisymmetric tensor gauge field). Hence, the three-form in 11 dimensions has $9 \times 8 \times 7/3! = 84$ degrees of freedom and combining this with the graviton gives 44 + 84 = 128 bosonic degrees of freedom which agrees with the fermionic ones.

The bosonic part of the action of 11d supergravity is then given by

$$2\kappa_{11}^2 S = \int d^{11}x \sqrt{-G} \left(\mathcal{R} - \frac{1}{2} |F_4|^2 \right) - \frac{1}{6} \int A_3 \wedge F_4 \wedge F_4 , \qquad (4.1.1)$$

with the scalar curvature \mathcal{R} , the field strength $F_4 = dA_3$ and the 11d gravitational coupling constant κ_{11} . The coupling constant is related to the 11d Newton constant G_N^{11d} and the 11d Planck length l_P as follows:

$$16\pi G_N^{11d} = 2\kappa_{11}^2 = \frac{1}{2\pi} (2\pi l_P)^9 . \qquad (4.1.2)$$

G is defined as det G_{MN} where G_{MN} is the metric combination

$$G_{MN} = \eta_{AB} E_M^A E_N^B , \qquad (4.1.3)$$

with η_{AB} being the flat metric and E_M^A a vielbein field where the indices M, N, \ldots describe curved base-space vectors in 11d and transform non-trivially under general coordinate transformations. The indices A, B, \ldots on the other hand describe flat tangent-space vectors which transform non-trivially under local Lorentz transformations. The quantity $|F_4|^2$ is defined as

$$|F_4|^2 = \frac{1}{4!} G^{M_1 N_1} \dots G^{M_4 N_4} F_{M_1 \dots M_4} F_{N_1 \dots N_4} .$$
(4.1.4)

Note that the goal of this chapter is to go beyond the supergravity approximation and obtain the form of higher-derivative corrections to the low-energy effective action (4.1.1). We approach this by computing anomalous dimensions of operators in the conformal block decomposition of the four-point stress tensor correlator in the dual conformal field theory.

4.1.2 Holographic Correlators

We are interested in the study of four-point stress tensor correlators. The stress tensor belongs to a half-BPS multiplet whose superconformal primary, T_{IJ} , is a dimension-4 scalar in the two-index symmetric traceless representation **14** of the R-symmetry group SO(5). As described in the review 2.3, general chiral primary operators are constructed from scalar fields as $\phi^{(I_1}\phi^{I_2}\dots\phi^{I_k)}$, which is totally symmetric and all traces are understood to be removed. The superconformal primary of the stress tensor multiplet in the 6d (2,0) theory is then constructed from the scalar fields ϕ^I in the abelian (2,0) tensor multiplet as follows

$$T_{IJ} = \operatorname{Tr}(\phi^{I}\phi^{J}) - \frac{\delta_{IJ}}{5}\operatorname{Tr}(\phi^{K}\phi_{K}) , \qquad (4.1.5)$$

where k = 2 and its scaling dimension is 2k = 4. Let us now consider the holographic duals of these operators. Dimensionally reducing the bulk dual on the four-sphere yields a Kaluza-Klein tower of scalars in AdS₇ with masses $m_k^2 = 4k(k-3)$ in units of the inverse AdS radius [138], which agrees with (2.3.2). The KK modes are the holographic duals of chiral primary operators with k indices which have scaling dimension 2k. The superconformal primaries of the stress tensor multiplet T_{IJ} (4.1.5), which are the focus of this chapter, correspond to the bottom of the tower, k = 2, and are dual to bulk scalars with mass $m^2 = -8$.

Four-point correlators of stress tensor multiplets were computed in the supergravity approximation in [139], and a conformal block decomposition of these results was subsequently carried out in [47]. More recently, corrections to the supergravity approximation were deduced in [26] by constructing solutions to the crossing equations whose conformal block expansion is truncated in spin. In section 4.3, we will derive recursion relations for the anomalous dimensions appearing in the conformal block expansions of these solutions. These recursion relations allow one to directly compute the OPE data of these solutions without having to know them explicitly, and can be straightforwardly implemented on a computer.

Before we go on and study the four-point stress tensor correlator in the (2,0) theory we consider a general bosonic six-dimensional CFT to illustrate our strategy of deriving recursion relations for anomalous dimensions and interpreting them to give insight into the higher-derivative corrections to the low-energy effective action.

4.2 Toy Model

In [30] the authors considered four-point correlators of scalar operators in an abstract non-supersymmetric CFT in two and four dimensions, and showed that the solutions to the crossing equations whose conformal block expansion is truncated in spin are in one-to-one correspondence with local quartic interactions of a massive scalar field in AdS (modulo integration by parts and equations of motion). In this section, we will carry out a similar analysis for a toy model in six dimensions as a warm up for our analysis of the 6d (2,0) theory in the next section. In particular, we will analyse four-point correlators of a scalar operator \mathcal{O} with classical dimension Δ_0 . Recall the form of four-point correlators [46]

$$\langle \mathcal{O}_1(x_1)\mathcal{O}_2(x_2)\mathcal{O}_3(x_3)\mathcal{O}_4(x_4)\rangle = \frac{F(u,v)}{(x_{12}^2)^{\Delta_0}(x_{34}^2)^{\Delta_0}},$$
 (4.2.1)

where x_i is the position of the *i*th operator, $x_{ij}^2 = (x_i - x_j)^2$, and *F* is a function of the conformal cross-ratios

$$u = \frac{x_{12}^2 x_{34}^2}{x_{13}^2 x_{24}^2} = z\bar{z} , \qquad v = \frac{x_{14}^2 x_{23}^2}{x_{13}^2 x_{24}^2} = (1-z)(1-\bar{z}) , \qquad (4.2.2)$$

where we will use the variables (u, v) interchangeably with (z, \bar{z}) , see also section 2.1. Note that exchanging x_2 with x_4 corresponds to exchanging u and v, or (z, \bar{z}) with $(1 - z, 1 - \bar{z})$. Invariance of the correlator under this exchange (known as crossing symmetry, see (2.1.6)) then implies the following constraint on F:

$$v^{\Delta_0} F(u, v) = u^{\Delta_0} F(v, u) .$$
(4.2.3)

In this model, the primary double-trace operators are schematically

$$\mathcal{O}_{n,l} = \mathcal{O}\partial_{\mu_1}...\partial_{\mu_l}\partial^{2n}\mathcal{O} , \qquad (4.2.4)$$

which have scaling dimension $\Delta = 2n + l + 2\Delta_0 + \mathcal{O}(1/c)$, spin l and naive twist $2n + 2\Delta_0$.

The conformal block expansion of F(u, v) is then given by the following sum over primary operators:

$$F(u,v) = \sum_{n,l \ge 0} A_{n,l} G^{\rm B}_{\Delta,l}(z,\bar{z}) , \qquad (4.2.5)$$

where $A_{n,l}$ are OPE coefficients and $G^{\text{B}}_{\Delta,l}$ are six-dimensional bosonic conformal blocks, which implicitly depend on n through the scaling dimensions of the conformal primary operator Δ . They are given in terms of hypergeometric functions and we spell them out in appendix E. Note that $A_{n,l} = 0$ when l is odd since operators with an odd number of derivatives in the OPE of two identical operators correspond to descendants. The function F can be obtained from tree-level Witten diagrams for supergravity (see [30], and section 2.3). However, the main point of this chapter is to derive the anomalous dimensions that appear in the conformal block expansion of the correlator directly from a recursion relation derived from (4.2.5) without requiring knowledge of the explicit form of the correlators.

The first step is to expand the OPE data in 1/c:

$$A_{n,l} = A_{n,l}^{(0)} + \frac{1}{c}A_{n,l}^{(1)} + \dots, \qquad \Delta = 2n + l + 2\Delta_0 + \frac{1}{c}\gamma_{n,l} + \dots$$
(4.2.6)

Firstly, let us compute the free OPE coefficients $A_{n,l}^{(0)}$ from the conformal block expansion of the free disconnected part of the four-point correlator which is given by

$$F^{\text{free}}(u,v) = 1 + u^{\Delta_0} + \frac{u^{\Delta_0}}{v^{\Delta_0}} .$$
(4.2.7)

The leading contribution to the OPE coefficients is then given by

$$A_{n,l}^{(0)} = \frac{2 (l+2) (2\Delta_0 + l+2n-2) (2\Delta_0 + l+2n-3) ((\Delta_0 + n-3)!)^2}{((\Delta_0 - 3)!)^2 ((\Delta_0 - 1)!)^2 n! (l+n+2)! (2\Delta_0 + 2n-6)! (2\Delta_0 + 2l+2n-2)!} \times ((\Delta_0 + l+n-1)!)^2 (2\Delta_0 + n-6)! (2\Delta_0 + l+n-4)!, \qquad (4.2.8)$$

for even l and zero otherwise. In the next subsection, we will derive recursion relations for the anomalous dimensions $\gamma_{n,l}$ in (4.2.6). After solving the recursion relations, we can then deduce the 1/c correction to the OPE coefficients $A_{n,l}^{(1)}$ using the following formula:

$$A_{n,l}^{(1)} = \frac{1}{2} \partial_n \left(A_{n,l}^{(0)} \gamma_{n,l} \right) .$$
 (4.2.9)

This formula was first found in two and four dimensions [30, 140] and was subsequently observed to hold in six dimensions [26].

4.2.1 Recursion Relations

In this subsection, we will derive a formula for the anomalous dimensions of doubletrace operators in the toy 6d CFT described above following the method developed for 2d and 4d CFTs in [30]. This formula will be written as a sum over the spin of the operators and will depend on two non-negative integers p and q. Truncating the sum over spin to maximum spin L and choosing p and q appropriately will then give rise to recursion relations for the anomalous dimensions, which can be solved for arbitrary twist and spin $l \leq L$ in terms of (L+2)(L+4)/8 free parameters, in agreement with counting of solutions in lower dimensions and holographic arguments, as we will describe in subsection 4.2.2.

After expanding the OPE data in 1/c in (4.2.6), expanding the conformal block decomposition (the right hand side of (4.2.5)) in 1/c and inserting this into the crossing equation (4.2.3) then gives

$$v^{\Delta_0} \sum_{n,l \ge 0} \left[A_{n,l}^{(1)} G^{\rm B}_{\Delta,l}(z,\bar{z}) + \frac{1}{2} A_{n,l}^{(0)} \gamma_{n,l} \partial_n G^{\rm B}_{\Delta,l}(z,\bar{z}) \right] - (u \leftrightarrow v) = 0 .$$
 (4.2.10)

Note that in general there will be degeneracy in the free theory, i.e. more than one operator with each given naive dimension and spin. Consequently, in (4.2.10) the free conformal block coefficient gives a sum over these operators of three-point coefficients squared, $A_{n,l}^{(0)} = \sum_i \langle \mathcal{O}_{\Delta_0} \mathcal{O}_{\Delta_0} \mathcal{O}_i \rangle^2$. Then $\gamma_{n,l}$ is in reality the so-called 'averaged anomalous dimension' $\gamma_{n,l} = (\sum_i \langle \mathcal{O}_{\Delta_0} \mathcal{O}_{\Delta_0} \mathcal{O}_i \rangle^2 \gamma_{n,l,i})/(\sum_i \langle \mathcal{O}_{\Delta_0} \mathcal{O}_{\Delta_0} \mathcal{O}_i \rangle^2)$ where $\gamma_{n,l,i}$ are the anomalous dimensions of the individual operators. To obtain the individual anomalous dimensions requires more data, for example four-point functions of operators with different dimensions. We will solve mixing problems like these for 1d correlators in chapter 5 but in the present chapter our focus lies on the averaged anomalous dimensions since they contain information about the higher-derivative terms in the low-energy effective action.

The exact form of the conformal blocks is given in appendix E, but they are given as a sum of products of hypergeometrics with the following schematic form

$$G^{\rm B}_{\Delta,l}(z,\bar{z}) \sim \sum \frac{u^n}{(z-\bar{z})^3} k_{\alpha}(z) k_{\beta}(\bar{z}) , \qquad (4.2.11)$$

where

$$k_{\beta}(z) = {}_{2}F_{1}\left(\frac{\beta}{2}, \frac{\beta}{2}, \beta, z\right)$$
 (4.2.12)

From this we see that $\partial_n G^{\rm B}_{\Delta,l}(z,\bar{z})$ gives a contribution of the form $\log(u) = \log(z\bar{z})$,

and the analogous term in the cross channel will contribute $\log((1-z)(1-\bar{z}))$. As a result, we can isolate the terms containing the anomalous dimensions in both channels simultaneously by taking the $\log(z)\log(1-\bar{z})$ coefficient of the crossing equation as $z \to 0$ and $\bar{z} \to 1$. This limit is referred to as the light-cone limit. In order for the crossing equation to be consistent, the $\log(z)$ coming from $\partial_n G^{\rm B}_{\Delta,l}(z,\bar{z})$ must thus be accompanied by a $\log(1-\bar{z})$. Such terms indeed arise from the hypergeometrics depending on \bar{z} after making use of the relation

$$k_{\beta}(\bar{z}) = \log(1-\bar{z})\,\tilde{k}_{\beta}(1-\bar{z}) + \text{holomorphic at } \bar{z} = 1 , \qquad (4.2.13)$$

where

$$\tilde{k}_{\beta}(z) = -\frac{\Gamma(\beta)}{\Gamma\left(\frac{\beta}{2}\right)^2} {}_2F_1\left(\frac{\beta}{2}, \frac{\beta}{2}, 1, z\right) .$$
(4.2.14)

Similarly, the hypergeometrics depending on 1 - z in the cross channel will give rise to $\log(z)$.

In summary, we take the $\log(z) \log(1 - \overline{z})$ coefficient of (4.2.10) as $z \to 0$ and $\overline{z} \to 1$ yielding the refined crossing equation:

$$v^{\Delta_{0}} \sum_{n,l \ge 0} A_{n,l}^{(0)} \gamma_{n,l} \left(\partial_{n} G_{2n+l+2\Delta_{0},l}^{\mathrm{B}}(z,\bar{z}) \right) \Big|_{\log z \log(1-\bar{z})} = u^{\Delta_{0}} \sum_{n,l \ge 0} A_{n,l}^{(0)} \gamma_{n,l} \left(\partial_{n} G_{2n+l+2\Delta_{0},l}^{\mathrm{B}}(1-z,1-\bar{z}) \right) \Big|_{\log z \log(1-\bar{z})} , \qquad (4.2.15)$$

into which we insert (the precise forms of) (4.2.11) and (4.2.13) to obtain sums of terms of the form $k_{\alpha}(z)\tilde{k}_{\beta}(1-\bar{z})$ and $k_{\alpha}(1-\bar{z})\tilde{k}_{\beta}(z)$. To extract a purely numerical recursion relation we use an orthogonality relation between hypergeometric functions obtained in [30]

$$\delta_{m,m'} = \oint \frac{dz}{2\pi i} \, z^{m-m'-1} \, k_{2m+4}(z) \, k_{-2m'-2}(z) \,, \qquad (4.2.16)$$

for more details on this relation please see appendix F. To use it, one has to multiply (4.2.15) in terms of sums of the schematic form $k \tilde{k}$ by

$$\frac{k_{-2q}(z)}{z^{5-\Delta_0+q}} \times \frac{k_{-2p}(1-\bar{z})}{(1-\bar{z})^{5-\Delta_0+p}} , \qquad (4.2.17)$$

which leads to a sum of terms of the schematic form $k k k \tilde{k}$, where p and q are arbitrary non-negative integers. Then, performing the contour integrals $\oint \frac{dz}{2\pi i} \oint \frac{d\bar{z}}{2\pi i}$, where the contours encircle $(z, \bar{z}) = (0, 1)$, one can use the orthogonality relation (4.2.16), which takes two of the k's per term and turns them into numerical expressions. Additionally, to make sense of the terms of the form $k \tilde{k}$ we define the following integral

$$\mathcal{I}_{m,m'} = \oint \frac{dz}{2\pi i} \frac{(1-z)^{m-\Delta_0+3}}{z^{m'-\Delta_0+5}} \tilde{k}_{2m}(z) \, k_{-2m'}(z) \; . \tag{4.2.18}$$

Finally, we arrive at the following equation:

$$0 = \sum_{l=0}^{L} \sum_{n=0}^{\infty} A_{n,l}^{(0)} \gamma_{n,l} \Big[(l+1) \left(\delta_{q,l+n+3} \mathcal{I}_{\Delta_0+n-3,p+\Delta_0-4} - \delta_{q,n} \mathcal{I}_{\Delta_0+l+n,p+\Delta_0-4} \right) \\ + (l+3) \left(\delta_{q,n+1} \mathcal{I}_{\Delta_0+l+n-1,p+\Delta_0-4} - \delta_{q,l+n+2} \mathcal{I}_{\Delta_0+n-2,p+\Delta_0-4} \right) \\ + P_{n,l} \left(\delta_{q,l+n+3} \mathcal{I}_{\Delta_0+n-1,p+\Delta_0-4} - \delta_{q,n+2} \mathcal{I}_{\Delta_0+l+n,p+\Delta_0-4} \right) \\ + Q_{n,l} \left(\delta_{q,n+1} \mathcal{I}_{\Delta_0+l+n+1,p+\Delta_0-4} - \delta_{q,l+n+4} \mathcal{I}_{\Delta_0+n-2,p+\Delta_0-4} \right) - (q \leftrightarrow p) \Big], \quad (4.2.19)$$

where

$$P_{n,l} = \frac{(l+3)(\Delta_0 + n - 2)^2(2\Delta_0 + l + 2n - 4)}{4(2\Delta_0 + 2n - 5)(2\Delta_0 + 2n - 3)(2\Delta_0 + l + 2n - 2)},$$

$$Q_{n,l} = \frac{(l+1)(\Delta_0 + l + n)^2(2\Delta_0 + l + 2n - 4)}{4(2\Delta_0 + l + 2n - 2)(2\Delta_0 + 2l + 2n - 1)(2\Delta_0 + 2l + 2n + 1)}.$$
(4.2.20)

Note that we have truncated the sum over spins in (4.2.19) to a maximum spin L. Recursion relations for the anomalous dimensions are then obtained by making particular choices of p and q, and the solutions are labelled by L. In the next subsection we will explain how to solve the recursion relations for L = 0, 2 before describing a general algorithm to solve the recursion relations for any L.

4.2.2 Solutions

Let us first consider the L = 0 spin truncation in (4.2.19). In this case, setting q = 0 leads to the following recursion relation in terms of p:

$$\mathcal{I}_{\Delta_0, p+\Delta_0-4} A_{0,0}^{(0)} \gamma_{0,0} = \sum_{a=0}^4 C_a A_{p-a,0}^{(0)} \gamma_{p-a,0} , \qquad (4.2.21)$$

where

$$C_{0} = \mathcal{I}_{p+\Delta_{0},\Delta_{0}-4} ,$$

$$C_{1} = -3\mathcal{I}_{p+\Delta_{0}-2,\Delta_{0}-4} - \frac{(p+\Delta_{0}-1)^{2}(p+\Delta_{0}-3)\mathcal{I}_{p+\Delta_{0},\Delta_{0}-4}}{4(p+\Delta_{0}-2)(2p+2\Delta_{0}-1)(2p+2\Delta_{0}-3)} ,$$

$$C_{2} = 3\mathcal{I}_{p+\Delta_{0}-4,\Delta_{0}-4} + \frac{3(p+\Delta_{0}-4)^{3}\mathcal{I}_{p+\Delta_{0}-2,\Delta_{0}-4}}{4(p+\Delta_{0}-3)(2p+2\Delta_{0}-7)(2p+2\Delta_{0}-9)} ,$$

$$C_{3} = -\mathcal{I}_{p+\Delta_{0}-6,\Delta_{0}-4} - \frac{3(p+\Delta_{0}-5)^{3}\mathcal{I}_{p+\Delta_{0}-4,\Delta_{0}-4}}{4(p+\Delta_{0}-4)(2p+2\Delta_{0}-9)(2p+2\Delta_{0}-11)} ,$$

$$C_{4} = \frac{(p+\Delta_{0}-4)^{2}(p+\Delta_{0}-6)\mathcal{I}_{p+\Delta_{0}-6,\Delta_{0}-4}}{4(p+\Delta_{0}-5)(2p+2\Delta_{0}-9)} .$$
(4.2.22)

This recursion relation can be solved for all $\gamma_{n,0}$ with n > 0 in terms of $\gamma_{0,0}$ as follows:

$$\gamma_{n,0}^{\text{spin-0}}(\Delta_0) = \gamma_{0,0} \frac{(2\Delta_0 - 3) (2\Delta_0 - 1) (n+1) (n+2) (\Delta_0 + n-2) (\Delta_0 + n-1)}{8 (\Delta_0 - 2)^2 (\Delta_0 - 1) (2\Delta_0 + 2n-5) (2\Delta_0 + 2n-3)} \times \frac{(2\Delta_0 + n-5) (2\Delta_0 + n-4)}{(2\Delta_0 + 2n-1)} , \qquad (4.2.23)$$

where we divided by $A_{n,0}^{(0)}$, see (4.2.8).

For L = 2, first choose (p, q) = (1, 0) to obtain $\gamma_{1,0}$ in terms of three unfixed parameters $\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}$. For p > 1, one can then solve the equations with $q \in \{0, 1\}$ for $\gamma_{p,l}$ with $l \in \{0, 2\}$ in terms of $\gamma_{p',l'}$ with p' < p and $l' \in \{0, 2\}$. In the end, we obtain a solution for all $\gamma_{n,l}$ with $l \in \{0, 2\}$ in terms of $\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}$. We find the following solutions for general scaling dimension Δ_0 :

$$\gamma_{n,0}^{\text{spin-2}}(\Delta_0) = \frac{\gamma_{n,0}^{\text{spin-0}}(\Delta_0)}{\gamma_{0,0}} \left(\gamma_{0,0} + \gamma_{0,2} f_1(n, \Delta_0) + \gamma_{1,2} f_2(n, \Delta_0)\right) , \qquad (4.2.24)$$

$$\gamma_{n,2}^{\text{spin-2}}(\Delta_{0}) = -\frac{\gamma_{n,0}^{\text{spin-0}}(\Delta_{0})}{\gamma_{0,0}} \frac{(2\Delta_{0}+1)^{2} (2\Delta_{0}+3) (n-1) (n+3) (n+4)}{4 (\Delta_{0}-1) \Delta_{0}^{4} (\Delta_{0}+1)^{2}} \\ \times \frac{(\Delta_{0}+n) (\Delta_{0}+n+1) (2\Delta_{0}+n) (2\Delta_{0}+n-3) (2\Delta_{0}+n-2)}{(2\Delta_{0}-3) (2\Delta_{0}+2n+1) (2\Delta_{0}+2n+3)} \\ \times \left(\gamma_{0,2}-\gamma_{1,2} \frac{4\Delta_{0} (2\Delta_{0}+3) (2\Delta_{0}+5) n (2\Delta_{0}+n-1)}{(\Delta_{0}+1) (\Delta_{0}+2)^{2} (2\Delta_{0}-1) (2\Delta_{0}+1) (n-1) (2\Delta_{0}+n)}\right),$$

$$(4.2.25)$$

where

$$f_{1}(n, \Delta_{0}) = \frac{(2\Delta_{0}+1)^{2} (2\Delta_{0}+3) n (2\Delta_{0}+n-3)}{(\Delta_{0}-1) \Delta_{0}^{4} (\Delta_{0}+1)^{2} (2\Delta_{0}-3) (2\Delta_{0}+2n-7) (2\Delta_{0}+2n+1)} \\ \times \left(5n^{6}+15 (2\Delta_{0}-3) n^{5}+(89\Delta_{0}^{2}-161\Delta_{0}+127) n^{4}+(2\Delta_{0}-3)\right) \\ \times (78\Delta_{0}^{2}-22\Delta_{0}+29) n^{3}+2 (82\Delta_{0}^{4}-143\Delta_{0}^{3}-107\Delta_{0}^{2}+117\Delta_{0}-39) n^{2} \\ + (2\Delta_{0}-3) (48\Delta_{0}^{4}-14\Delta_{0}^{3}-215\Delta_{0}^{2}-33\Delta_{0}-6) n \\ + \frac{6 (\Delta_{0}-1) \Delta_{0}^{2} (2\Delta_{0}-7) (4\Delta_{0}^{3}+12\Delta_{0}^{2}+5\Delta_{0}-1)}{2\Delta_{0}+1}\right),$$

$$f_{2}(n, \Delta_{0}) = \frac{(2\Delta_{0}+1)(2\Delta_{0}+3)^{2}(2\Delta_{0}+5)n(2\Delta_{0}+n-3)}{(3-2\Delta_{0})(\Delta_{0}-1)\Delta_{0}^{3}(\Delta_{0}+1)^{3}(\Delta_{0}+2)^{2}(2\Delta_{0}-1)(2\Delta_{0}+2n-7)} \times \frac{1}{(2\Delta_{0}+2n+1)} \Big(20n^{6}+60(2\Delta_{0}-3)n^{5}+(4\Delta_{0}(89\Delta_{0}-199)+508)n^{4} + 4(2\Delta_{0}-3)(78\Delta_{0}^{2}-98\Delta_{0}+29)n^{3}+8(\Delta_{0}(2\Delta_{0}(\Delta_{0}(41\Delta_{0}-131)+104) -27)-39)n^{2}+4(2\Delta_{0}-3)(\Delta_{0}(\Delta_{0}(4\Delta_{0}(12\Delta_{0}-25)-41)+21)-6)n + 24(\Delta_{0}-1)\Delta_{0}^{2}(\Delta_{0}+1)(2\Delta_{0}-7)(2\Delta_{0}-1)\Big).$$

$$(4.2.26)$$

We will discuss these solutions and solutions with general L below, but first let us describe an algorithm to solve recursion relations for general L.

Algorithm to solve recursion relations for any spin-L truncation

Recursion relations for the anomalous dimensions of double-trace operators for a general spin-L truncation are encoded in (4.2.19) for the bosonic toy model and below in (4.3.19) for the (2,0) theory, respectively. They are obtained by specifying a spin truncation L and making appropriate choices of non-negative integers p and

- q. The general algorithm for solving the recursion relation for any L is as follows:
 - For each $1 \le p \le L/2$, write down the equations for $0 \le q \le p-1$.
 - Solve these equations for $\gamma_{p,l}$ with $0 \le l \le 2p 2$ in terms of $\gamma_{p',l'}$ with $(p' \le p 1, l' \le L)$ and $(p' = p, 2p \le l' \le L)$.
 - For each $p \ge L/2 + 1$, write down the equations for $0 \le q \le L/2$.
 - Solve these equations for $\gamma_{p,l}$ with $0 \leq l \leq L$ in terms of $\gamma_{p',l'}$ with $(p' \leq p 1, l' \leq L).$

In the end, this algorithm will give all $\gamma_{n,l}$ with $l \leq \min(2n-2,L)$ in terms of all $\gamma_{n',l'}$ with $2n' \leq l' \leq L$, which correspond to (L+2)(L+4)/8 free parameters as depicted in figure 3 of [30] (for a review of the holographic arguments of [30] see subsection 2.3.2). This algorithm can easily be implemented on a computer¹ by generating all the free parameters for a given L, writing down the equations for every $p \geq 1$ and $0 \leq q \leq \min(p-1, L/2)$, replacing (L+2)(L+4)/8 of the anomalous dimensions by the free parameters, and solving these equations for the remaining anomalous dimensions.

Discussion of solutions

For a spin-L truncation, we find that the solution depends on (L+2)(L+4)/8 unfixed parameters, as described above for general L and illustrated in the L = 0, 2 cases. This agrees with the holographic arguments of [30] (see subsection 2.3.2). Recall that the four-point correlators are dual to local quartic interactions in the bulk. Specifically, there are L/2+1 independent interactions which can create or annihilate a state of at most spin L, with the total number of derivatives ranging from 2L to 3L in intervals of two. For the cases we considered in the discussion of the recursion relation above, there is one spin-0 interaction vertex with no derivatives ϕ^4 , and

¹This can be seen in the auxiliary Mathematica file 6drecursion.nb of [43].

two spin-2 interaction vertices $\phi^2 (\nabla_\mu \nabla_\nu \phi)^2$ and $\phi^2 (\nabla_\mu \nabla_\nu \nabla_\rho \phi)^2$ with four and six derivatives, respectively, see (2.3.19). The total number of interactions up to spin L is then given by $\sum_{l=0}^{L/2} (l+1) = (L+2)(L+4)/8$.

Note that this is the same number as there are unfixed parameters in the solutions to the recursion relations above. Thus, these unfixed parameters can be identified with coefficients of the bulk interaction vertices. Indeed, we have verified that the solution in (4.2.23) reproduces the anomalous dimensions in the conformal block expansion of a Witten diagram for a ϕ^4 interaction (see [30] and subsection 2.3.1)

$$F^{\text{spin-0}}(u,v) = C^{(0)} \bar{D}_{\Delta_0 \,\Delta_0 \,\Delta_0 \,\Delta_0} , \qquad (4.2.27)$$

for the following choice of the free parameter:

$$\gamma_{0,0} = -\frac{C^{(0)} \left((\Delta_0 - 1)! \right)^4}{(2\Delta_0 - 1)!} , \qquad (4.2.28)$$

where the coefficient $C^{(0)}$ is unfixed and the definition of \overline{D} -functions was reviewed in chapter 2 around (2.3.17). Note that the anomalous dimensions of $F^{\text{spin-0}}$ are obtained by expanding this function according to (4.2.10) but without considering the crossing equation.

Moreover, the L = 2 solution encodes the anomalous dimensions in the conformal block expansion of Witten diagrams with four- and six-derivative interactions (again see [30] and section 2.3.2)

$$F_{4}^{\text{spin-2}}(u,v) = C_{4}^{(2)}(1+u+v)\bar{D}_{\Delta_{0}+1\,\Delta_{0}+1\,\Delta_{0}+1\,\Delta_{0}+1}, \qquad (4.2.29)$$

$$F_{6}^{\text{spin-2}}(u,v) = C_{6}^{(2)}\left(\bar{D}_{\Delta_{0}+2\,\Delta_{0}+1\,\Delta_{0}+2\,\Delta_{0}+1} + \bar{D}_{\Delta_{0}+1\,\Delta_{0}+2\,\Delta_{0}+1\,\Delta_{0}+2} + u^{2}\bar{D}_{\Delta_{0}+2\,\Delta_{0}+2\,\Delta_{0}+1\,\Delta_{0}+1} + u\bar{D}_{\Delta_{0}+1\,\Delta_{0}+2\,\Delta_{0}+2} + v^{2}\bar{D}_{\Delta_{0}+1\,\Delta_{0}+2\,\Delta_{0}+2} + v\bar{D}_{\Delta_{0}+2\,\Delta_{0}+1} + v\bar{D}_{\Delta_{0}+2\,\Delta_{0}+1\,\Delta_{0}+2}\right), \qquad (4.2.30)$$

for the following choice of free parameters:

$$\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}_{4} = C_{4}^{(2)} \left\{ -\frac{4\left(\Delta_{0}!\right)^{3}\left(\Delta_{0}+1\right)!}{\left(2\Delta_{0}+2\right)!}, -\frac{2\Delta_{0}!\left(\left(\Delta_{0}+1\right)!\right)^{2}\left(\Delta_{0}+2\right)!}{3\left(2\Delta_{0}+1\right)\left(2\Delta_{0}+4\right)!}, -\frac{\left(\Delta_{0}+1\right)\left(2\Delta_{0}-1\right)\left(\Delta_{0}-1\right)!\left(\left(\Delta_{0}+2\right)!\right)^{2}\left(\Delta_{0}+3\right)!}{3\left(2\Delta_{0}+3\right)\left(2\Delta_{0}+6\right)!} \right\}, \quad (4.2.31)$$

$$\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}_{6} = C_{6}^{(2)} \left\{ -\frac{4\left(\Delta_{0}!\right)^{2}\left((\Delta_{0}+1)!\right)^{2}}{\left(2\Delta_{0}+2\right)!}, -\frac{2\left(3\Delta_{0}+2\right)\Delta_{0}!\left((\Delta_{0}+1)!\right)^{2}\left(\Delta_{0}+2\right)!}{3\left(2\Delta_{0}+1\right)\left(2\Delta_{0}+4\right)!}, -\frac{\left(\Delta_{0}+1\right)\left(6\Delta_{0}^{2}+7\Delta_{0}-2\right)\left(\Delta_{0}-1\right)!\left((\Delta_{0}+2)!\right)^{2}\left(\Delta_{0}+3\right)!}{3\left(2\Delta_{0}+3\right)\left(2\Delta_{0}+6\right)!} \right\},$$

$$(4.2.32)$$

where the coefficients $C_{4,6}^{(2)}$ are unfixed. Even though the contact interactions can be obtained from Witten diagrams in AdS case by case for spin-*L* truncations and this could be used to obtain the anomalous dimensions of double-trace operators in the OPE, computing the anomalous dimensions from a recursion relation is much more efficient. The recursion relation does not require the knowledge of the exact form of the correlators, one only needs the conformal blocks (and the free theory coefficients) and can derive the averaged anomalous dimensions for any spin-*L* truncation easily. Furthermore, the algorithm for general *L* solutions described in subsection 4.2.2 can be implemented on a computer which makes it very efficient to obtain anomalous dimensions for any *L*.

Importantly, the number of derivatives in the bulk interactions can be read off from the large-twist behaviour of the corresponding anomalous dimensions. Indeed, the anomalous dimensions of $F^{\text{spin-0}}$ scale like n^3 in the large-n limit, while those of $F_4^{\text{spin-2}}$ and $F_6^{\text{spin-2}}$ scale like n^7 and n^9 , respectively. In other words, the anomalous dimensions associated with four- and six-derivative interactions scale like n^4 and n^6 compared to those of the ϕ^4 interaction. Studying the anomalous dimensions obtained from the recursion relations, there is a subtlety for L > 0. The solutions (4.2.24) and (4.2.25) both scale like n^9 , so they both correspond to six-derivative interactions. We know however, that for L = 2 there should also be a four-derivative interaction corresponding to anomalous dimensions scaling like n^7 and it turns out that this is included in the solutions as follows. For a specific choice of the ratio between the coefficients $\gamma_{0,2}$ and $\gamma_{1,2}$, the scaling of (4.2.25) reduces to n^7 , which can be easily deduced by imposing the vanishing of the large-*n* limit of the last line in (4.2.25). The choice of free parameters is

$$\gamma_{1,2} = \frac{(\Delta_0 + 1) (\Delta_0 + 2)^2 (2\Delta_0 - 1) (2\Delta_0 + 1)}{4\Delta_0 (2\Delta_0 + 3) (2\Delta_0 + 5)} \gamma_{0,2} .$$
(4.2.33)

Note that the solution in (4.2.31) is consistent with this constraint. More generally, for a spin-L solution one can deduce L/2 constraints on the coefficients (corresponding to bulk interactions with the number of derivatives ranging from 2L to 3L - 2 in intervals of 2) by analysing the large-twist limit. Unconstrained coefficients then encode the freedom to add solutions with lower spin or subleading large-twist behaviour.

We will discuss this further in the 6d (2,0) case in subsection 4.3.2 where the anomalous dimensions encode information about the higher-derivative corrections to the low-energy 11d supergravity action. Let us now go on and derive recursion relations for the supersymmetric theory.

4.3 (2,0) **Theory**

In this section we will adapt the analysis of the previous section to four-point stress tensor correlators of the 6d (2,0) theory. Recall from the review in section 4.1.2 that the superconformal primary of the half-BPS multiplet is the dimension-four scalar T_{IJ} in the two-index symmetric traceless representation of the R-symmetry group SO(5). Its holographic dual corresponds to the bottom of the KK tower, k = 2, and has mass $m^2 = -8$.

As shown in [47, 141], superconformal symmetry constrains the four-point function of stress tensor multiplets in the 6d (2,0) theory in terms of a prepotential $F(z, \bar{z})$
as follows:

$$(z - \bar{z})^4 (g_{13}g_{24})^{-2} \langle T_1 T_2 T_3 T_4 \rangle = \mathcal{D} \left(\mathcal{S}F(z, \bar{z}) \right) + \mathcal{S}_1^2 F(z, z) + \mathcal{S}_2^2 F(\bar{z}, \bar{z}) , \quad (4.3.1)$$

where $\mathcal{D} = -(\partial_z - \partial_{\bar{z}} + (z - \bar{z})\partial_z\partial_{\bar{z}})(z - \bar{z})$ and the variables z, \bar{z} are defined in terms of the spacetime cross-ratios (4.2.2). We have introduced auxiliary variables Y^I to contract the SO(5) indices of T_{IJ} via $T_i = T_{IJ}Y_i^IY_i^J$. Using these internal coordinates, we then define superpropagators $g_{ij} = Y_i \cdot Y_j/x_{ij}^4$ and internal conformal cross-ratios

$$y\bar{y} = \frac{Y_1 \cdot Y_2 Y_3 \cdot Y_4}{Y_1 \cdot Y_3 Y_2 \cdot Y_4} , \qquad (1-y)\left(1-\bar{y}\right) = \frac{Y_1 \cdot Y_4 Y_2 \cdot Y_3}{Y_1 \cdot Y_3 Y_2 \cdot Y_4} , \qquad (4.3.2)$$

in terms of which we define $S_1 = (z - y)(z - \bar{y})$, $S_2 = (\bar{z} - y)(\bar{z} - \bar{y})$, and $S = S_1S_2$. Crossing symmetry implies that

$$F(u,v) = F(v,u)$$
. (4.3.3)

This is important because as there is a crossing symmetric prepotential in terms of which the correlator is constrained, one does not need to consider the full correlator but studying the much simpler prepotential is enough. Moreover, we can write F(u, v) as

$$F(z,\bar{z}) = \frac{A}{u^2} + \frac{g(z) - g(\bar{z})}{u(z - \bar{z})} + (z - \bar{z})G(z,\bar{z}) , \qquad (4.3.4)$$

where each function in this decomposition encodes certain contributions to the OPE. Roughly speaking, A encodes the unit operator, g encodes protected operators, and G encodes non-protected double-trace operators, which will be our main interest. In more detail, these operators have the schematic form $T\partial^l \square^n T$ with $n \ge 0$ and scaling dimension $\Delta = 2n + l + 8 + \mathcal{O}(1/c)$. They contribute to the conformal block expansion of G as follows ²

$$(z - \bar{z})^2 G(z, \bar{z}) = \sum_{n,l \ge 0} A_{n,l} G^{\rm S}_{\Delta,l}(z, \bar{z}) , \qquad (4.3.5)$$

where the supersymmetric conformal blocks $G^{\rm S}_{\Delta,l}(z,\bar{z})$ are given in appendix E, and

²Note that the conformal block expansion of G also contains protected double-trace operators, which correspond to $n \in \{-1, -2\}$ in our conventions, but we will not need to consider these operators. For more details, see [47].

implicitly depend on n through Δ . Note that equations (4.3.3) and (4.3.4) imply that G obeys the following crossing relation:

$$G(z,\bar{z}) = -G(1-z,1-\bar{z}) . \tag{4.3.6}$$

Again we start by computing the leading free theory coefficients which are obtained from the free disconnected part of the four-point correlator. This can be computed in the abelian theory and corresponds to the following prepotential:

$$F^{\text{free-disc}}(u,v) = 1 + \frac{1}{u^2} + \frac{1}{v^2}$$
 (4.3.7)

Decomposing this function according to (4.3.4) and computing the conformal block expansion of the piece encoding the non-protected operators according to (4.3.5)then gives the following formula for the leading contribution to the OPE coefficients:

$$A_{n,l}^{(0)} = \frac{(l+2)(n+3)!(n+4)!(l+2n+9)(l+2n+10)(l+n+5)!(l+n+6)!}{72(2n+5)!(2l+2n+9)!}.$$
(4.3.8)

As we have established, our focus in this chapter are the higher-derivative corrections to the supergravity approximation which we will discuss in the following subsections. Nevertheless, it is interesting to analyse the tree-level supergravity result as well, and this is done in appendix G.

4.3.1 Recursion Relations

To derive recursion relations for the anomalous dimensions of the double-trace operators described above, we follow the same procedure as section 4.2. First, expand the OPE data in 1/c:

$$A_{n,l} = A_{n,l}^{(0)} + \frac{1}{c}A_{n,l}^{(1)} + \dots , \qquad \Delta = 2n + l + 8 + \frac{1}{c}\gamma_{n,l} + \dots$$
(4.3.9)

Focusing on the part of the prepotential which describes non-protected operators and expanding the crossing equation (4.3.6) to first order in 1/c then gives³

$$\sum_{n,l\geq 0} \left[A_{n,l}^{(1)} G_{\Delta,l}^{\rm S}(z,\bar{z}) + \frac{1}{2} A_{n,l}^{(0)} \gamma_{n,l} \partial_n G_{\Delta,l}^{\rm S}(z,\bar{z}) \right] + (u \leftrightarrow v) = 0 .$$
 (4.3.10)

In the supersymmetric case, the conformal blocks have the schematic form

$$G^{\rm S}_{\Delta,l}(z,\bar{z}) \sim \sum u^n h_\alpha(z) h_\beta(\bar{z}) , \qquad (4.3.11)$$

where

$$h_{\beta}(z) = {}_{2}F_{1}\left(\frac{\beta}{2}, \frac{\beta}{2} - 1, \beta, z\right) , \qquad (4.3.12)$$

see appendix \mathbf{E} for the explicit form of the superconformal blocks. Following the same reasoning described in the previous section, the term $\partial_n G^{\mathrm{S}}_{\Delta,l}(z,\bar{z})$ in (4.3.10) gives a contribution proportional to $\log(z)$ and the analogous term in the cross channel will give $\log(1-\bar{z})$, so we can isolate the terms containing anomalous dimensions by taking the light-cone limit $z \to 0$ and $\bar{z} \to 1$. In this case, the hypergeometrics depending on \bar{z} and 1-z will give rise to $\log(1-\bar{z})$ and $\log(z)$ using the relation

$$h_{\beta}(\bar{z}) = \log(1-\bar{z})(1-\bar{z})\tilde{h}_{\beta}(1-\bar{z}) + \text{holomorphic at } \bar{z} = 1 , \qquad (4.3.13)$$

where

$$\tilde{h}_{\beta}(z) = \frac{\Gamma(\beta)}{\Gamma\left(\frac{\beta}{2}\right)\Gamma\left(\frac{\beta}{2}-1\right)} {}_{2}F_{1}\left(\frac{\beta}{2}+1,\frac{\beta}{2},2,z\right) .$$

$$(4.3.14)$$

We thus consider the $\log(z) \log(1 - \overline{z})$ coefficient of (4.3.10) in the limit $z \to 0$ and $\overline{z} \to 1$:

$$\sum_{n,l\geq 0} A_{n,l}^{(0)} \gamma_{n,l} \left(\partial_n G_{\Delta,l}^{\rm S}(z,\bar{z}) \right) \Big|_{\log z \log(1-\bar{z})} = -\sum_{n,l\geq 0} A_{n,l}^{(0)} \gamma_{n,l} \left(\partial_n G_{\Delta,l}^{\rm S}(1-z,1-\bar{z}) \right) \Big|_{\log z \log(1-\bar{z})} , \qquad (4.3.15)$$

into which we insert the (precise forms of) (4.3.11) and (4.3.13) to obtain sums of terms involving $h_{\alpha}(z)\tilde{h}_{\beta}(1-\bar{z})$ and $h_{\alpha}(1-\bar{z})\tilde{h}_{\beta}(z)$. To obtain a purely numeric

³Note that as in the toy model case, here $\gamma_{n,l}$ are averaged anomalous dimensions.

crossing equation, first multiply this equation by

$$\frac{h_{-2q}(z)}{z^q (1-z)} \times \frac{h_{-2p}(1-\bar{z})}{(1-\bar{z})^p \bar{z}} , \qquad (4.3.16)$$

where p and q are arbitrary non-negative integers, and perform the contour integrals $\oint \frac{dz}{2\pi i} \oint \frac{d\bar{z}}{2\pi i}$, which encircle $(z, \bar{z}) = (0, 1)$. Again we use an orthogonality relation of the hypergeometric functions and we prove this new relation in appendix F. It is given by

$$\delta_{m,m'} = \oint \frac{dz}{2\pi i} \frac{z^{m-m'-1}}{1-z} h_{2m+4}(z) h_{-2m'-2}(z) , \qquad (4.3.17)$$

and defining the integral

$$\mathcal{I}_{m,m'} = \oint \frac{dz}{2\pi i} \frac{(1-z)^{m-3}}{z^{m'-1}} \tilde{h}_{2m}(z) h_{-2m'}(z) , \qquad (4.3.18)$$

finally leads to the following equation:

$$0 = \sum_{l=0}^{L} \sum_{n=0}^{\infty} A_{n,l}^{(0)} \gamma_{n,l} \Big[P_{n,l} \left(\delta_{q,n} \mathcal{I}_{n+l+6,p+2} - \delta_{q,n+l+3} \mathcal{I}_{n+3,p+2} \right) + Q_{n,l} \left(\delta_{q,n+2} \mathcal{I}_{n+l+6,p+2} - \delta_{q,n+l+3} \mathcal{I}_{n+5,p+2} \right) + R_{n,l} \left(\delta_{q,n+l+2} \mathcal{I}_{n+4,p+2} - \delta_{q,n+1} \mathcal{I}_{n+l+5,p+2} \right) + S_{n,l} \left(\delta_{q,n+l+4} \mathcal{I}_{n+4,p+2} - \delta_{q,n+1} \mathcal{I}_{n+l+7,p+2} \right) - (q \leftrightarrow p) \Big], \qquad (4.3.19)$$

where we have truncated the sum over spins and defined

$$P_{n,l} = \frac{l+1}{(n+3)(n+l+5)}, \quad Q_{n,l} = \frac{(l+3)(n+5)(2n+l+8)}{4(2n+7)(2n+9)(n+l+5)(2n+l+10)},$$

$$R_{n,l} = \frac{l+3}{(n+3)(n+l+5)}, \quad S_{n,l} = \frac{(l+1)(n+l+7)(2n+l+8)}{4(n+3)(2n+l+10)(2n+2l+11)(2n+2l+13)}.$$
(4.3.20)

As we explain in the next subsection and in 4.2.2, recursion relations for the anomalous dimensions are obtained from (4.3.19) by making appropriate choices for p and q, and solutions are labelled by the spin truncation L.

4.3.2 Solutions

In this subsection, we will describe solutions to the recursion relations for low spin truncations and match them with results previously obtained in [26], an algorithm to obtain solutions for any L was described in subsection 4.2.2. For spin truncation L = 0, setting q = 0 in (4.3.19) gives the following recursion relation in terms of p:

$$\frac{1}{15}\mathcal{I}_{6,p+2}A^{(0)}_{0,0}\gamma_{0,0} = \sum_{a=0}^{4} C_a A^{(0)}_{p-a,0}\gamma_{p-a,0} , \qquad (4.3.21)$$

where

$$C_{0} = \frac{\mathcal{I}_{p+6,2}}{(p+3)(p+5)},$$

$$C_{1} = -\frac{3\mathcal{I}_{p+4,2}}{(p+2)(p+4)} - \frac{(p+3)(p+6)\mathcal{I}_{p+6,2}}{4(p+2)(p+4)(2p+9)(2p+11)},$$

$$C_{2} = \frac{3\mathcal{I}_{p+2,2}}{(p+1)(p+3)} + \frac{3(p+2)\mathcal{I}_{p+4,2}}{4(p+3)(2p+3)(2p+5)},$$

$$C_{3} = -\frac{\mathcal{I}_{p,2}}{p(p+2)} - \frac{3(p+1)\mathcal{I}_{p+2,2}}{4(p+2)(2p+1)(2p+3)},$$

$$C_{4} = \frac{p(p+3)\mathcal{I}_{p,2}}{4(p-1)(p+1)(2p+3)(2p+5)}.$$
(4.3.22)

This can be solved for all $\gamma_{n,0}$ in terms of $\gamma_{0,0}$ to give

$$\gamma_{n,0}^{\text{spin}-0} = \gamma_{0,0} \frac{11(n+1)_8 (n+2)_6}{2304000(2n+7)(2n+9)(2n+11)} , \qquad (4.3.23)$$

where we divided by $A_{n,0}^{(0)}$, see (4.3.8), and recall that $x_n = \Gamma(x+n)/\Gamma(x)$ is the Pochhammer symbol. It is interesting to compare this with the bosonic solution (4.2.23) for $\Delta_0 = 4$:

$$(\gamma_{\rm bos})_{n,0}^{\rm spin-0} = (\gamma_{\rm bos})_{0,0} \frac{35(n+1)_4(n+2)_2}{96(2n+3)(2n+5)(2n+7)} . \tag{4.3.24}$$

Similarly, following the procedure described in subsection 4.2.2, we obtain solutions for L = 2 in terms of three unfixed parameters { $\gamma_{0,0}$, $\gamma_{0,2}$, $\gamma_{1,2}$ }, which are :

$$\gamma_{n,0}^{\text{spin-2}} = \frac{\gamma_{n,0}^{\text{spin-0}}}{\gamma_{0,0}} \left(\gamma_{0,0} + \gamma_{0,2} f_1(n) + \gamma_{1,2} f_2(n)\right) , \qquad (4.3.25)$$

$$\gamma_{n,2}^{\text{spin-2}} = -\frac{\gamma_{n,0}^{\text{spin-0}}}{\gamma_{0,0}} \frac{845(n-1)(n+5)(n+6)(n+8)(n+9)^2(n+10)(n+12)}{4064256(2n+13)(2n+15)} \times \left(\gamma_{0,2} - \gamma_{1,2} \frac{51n(n+11)}{364(n-1)(n+12)}\right), \qquad (4.3.26)$$

where

$$f_{1}(n) = \frac{325n(n+9)}{1016064(2n+5)(2n+13)} \times (13n^{6} + 351n^{5} + 6201n^{4} + 64233n^{3} + 385476n^{2} + 1251666n + 1512620) ,$$

$$f_{2}(n) = -\frac{1105n(n+9)}{9483264(2n+5)(2n+13)} \times (5n^{6} + 135n^{5} + 2157n^{4} + 20601n^{3} + 117468n^{2} + 370494n + 441700) .$$

$$(4.3.27)$$

We discuss these solutions and what they tell us about the bulk interactions in the following.

Discussion of solutions

Again, for spin truncation L, the solution will depend on (L + 2)(L + 4)/8 free parameters, in agreement with the counting of solutions in section 4.2.2. Moreover, our results for the anomalous dimensions agree with those obtained in [26], which deduced solutions to the crossing equations whose conformal block expansions are truncated in spin. In particular, the anomalous dimensions in (4.3.23) can be obtained from the conformal block expansion of [26]

$$F^{\text{spin-0}}(u,v) = C^{(0)}(z-\bar{z})^2 u v \bar{D}_{5755} , \qquad (4.3.28)$$

where the coefficient $C^{(0)}$ is unfixed. Decomposing $F^{\text{spin-0}}$ according to (4.3.4) and performing the conformal block expansion according to (4.3.10) gives the anomalous dimensions in (4.3.23) if we choose the free parameter to be

$$\gamma_{0,0} = -\frac{7200 \, C^{(0)}}{77} \ . \tag{4.3.29}$$

Chapter 4. $AdS_7 \times S^4$: Recursion Relations for Anomalous Dimensions 100 in the 6d (2,0) Theory

Following the holographic arguments of [26, 30, 61], which were reviewed in section 2.3.2, $F^{\text{spin-0}}$ should arise from an \mathcal{R}^4 correction to supergravity in $\text{AdS}_7 \times \text{S}^4$, where \mathcal{R} is the Riemann tensor. This can be seen by comparing the large-n limit of the anomalous dimensions of the spin-L solutions to the anomalous dimensions of the tree-level supergravity solutions at large n, see appendix G where one can see that for supergravity the anomalous dimensions scale like n^5 . The difference in the powers of n then predicts how many derivatives the spin-L interaction has compared to supergravity. Note that in the large-n limit the anomalous dimensions scale like n^{11} which is n^6 times the anomalous dimensions obtained in the supergravity approximation, indicating that the corresponding interaction vertex has six more derivatives than supergravity. Since \mathcal{R} has two derivatives and we are only considering quartic interactions, the spin-0 correction is connected to a \mathcal{R}^4 term.

Note that F is a prepotential from which many four-point component correlators (corresponding to different choices of Y_i) are obtained by applying a differential operator according to (4.3.1). This differential operator can be rewritten in terms of u, v derivatives and if the prepotential is expressed in terms of \overline{D} -functions then so will all the component correlators. Whilst this does not prove that the prepotential can always be expressed in terms of \overline{D} -functions, this property holds in all the examples we have considered, and it is natural to conjecture that it should hold in general. A similar conjecture was made in [129] for four-point correlators of more general half-BPS operators in the supergravity approximation.

For the L = 2 spin truncation, [26] found the following solutions to the crossing equation:

$$F_4^{\text{spin-2}}(u,v) = 2C_4^{(2)}\left(z-\bar{z}\right)^2 uv\left(\bar{D}_{6776} + \bar{D}_{7676} + \bar{D}_{7766}\right) , \qquad (4.3.30)$$

$$F_6^{\text{spin-2}}(u,v) = 6 C_6^{(2)} \left(z - \bar{z}\right)^2 u v \bar{D}_{7777} , \qquad (4.3.31)$$

where the coefficients $C_{4,6}^{(2)}$ are unfixed and the subscripts indicate the number of additional derivatives compared to the bulk interaction vertex associated with the L = 0 solution. The first solution corresponds to a $D^4 \mathcal{R}^4$ correction and the second one corresponds to a $D^6 \mathcal{R}^4$ correction to supergravity in AdS₇×S⁴, which can be read off from the large-twist behaviour of the corresponding anomalous dimensions. In the large-*n* limit, the first solution scales like n^{15} while the second one scales like n^{17} , which corresponds to four and six more derivatives than the spin-0 solution which goes like n^{11} . The anomalous dimensions of these two solutions are reproduced from the general solution in (4.3.25) and (4.3.26) for the choice of parameters

$$\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}_4 = C_4^{(2)} \left\{ -\frac{5 \times 72000}{1001} , \frac{80640}{1859} , \frac{5 \times 150528}{2431} \right\} , \qquad (4.3.32)$$

$$\{\gamma_{0,0}, \gamma_{0,2}, \gamma_{1,2}\}_6 = C_6^{(2)} \left\{ \frac{54 \times 72000}{1001} , -\frac{3 \times 80640}{1859} , -\frac{33 \times 150528}{2431} \right\} .$$
(4.3.33)

Note that in both cases, $\gamma_{0,0}$ has the opposite sign of $\gamma_{0,2}$ and $\gamma_{1,2}$, in contrast to what we found for the toy model in (4.2.31) and (4.2.32), where all three parameters had the same sign.

Similar to the toy model, considering the anomalous dimensions from the recursion relations for L = 2, both solutions scale like n^{17} in the large-n limit. It follows that the corresponding bulk interactions have six more derivatives than the spin-0 interaction. However, we also expect to find solutions corresponding to interactions with four more derivatives, which scale like n^{15} for large n. We obtain these solutions for the choice of free parameters

$$\gamma_{1,2} = \frac{364}{51} \gamma_{0,2} , \qquad (4.3.34)$$

which comes from imposing that the last line of (4.3.26) vanishes in the large-*n* limit. The solution in (4.3.32) is consistent with this constraint. Having discussed the solutions for L = 0, 2 using the algorithm in section 4.2.2 one can solve the recursion relations up to any desired spin truncation for any twist.

Although solutions to the recursion relations have unfixed coefficients, it is possible to deduce their leading 1/c-dependence using holographic reasoning, as described in [26]. First, note that since we solve the recursion relations by truncating in spin, this restricts to contact interactions in the bulk (interactions involving bulk-to-bulk propagators will not truncate in spin). The effective action then has the schematic form

$$\mathcal{L} \sim \frac{1}{G_N^{11d}} \left[\left(\partial \phi \right)^2 + \sum_D l_P^{D-2} \partial^D \phi^4 \right] \,, \tag{4.3.35}$$

where ϕ represents a graviton field, G_N^{11d} is the 11d Newton constant, and the Planck length l_P is inserted by dimensional analysis. After rescaling the graviton by $\sqrt{G_N^{11d}}$ in order to have canonical kinetic terms, the four-point interactions will acquire a factor of $G_N^{11d} \sim 1/c$ (this is the origin of the 1/c in (4.3.9)). Recalling that $G_N^{11d} \sim l_P^9$ in eleven dimensions (see (4.1.2)), we see that a four-point contact interaction with D derivatives must therefore have coefficient $G_N^{11d} l_P^{D-2} \sim c^{-(D+7)/9}$. Moreover, the number of derivatives in a contact interaction can be read off from the large-twist behaviour of the corresponding solution to the crossing equations [30]. In particular, if the anomalous dimensions of the solution scale like n^{α} , then the corresponding bulk interaction must have $D = (\alpha - 5) + 2 = \alpha - 3$ derivatives (recalling that anomalous dimensions scale like n^5 in the supergravity approximation).

In summary, a solution whose anomalous dimensions scale like n^{α} must have a coefficient $c^{-(\alpha+4)/9}$. For example, the spin-0 solution in (4.3.23) will have a coefficient of $c^{-5/3}$ and the spin-2 solutions (4.3.25) and (4.3.26) which scale like n^{15} and n^{17} (with a specific choice of parameters) will have coefficients of $c^{-19/9}$ and $c^{-7/3}$, respectively. Note that the spin-0 correction is the leading correction to the low-energy tree-level effective action, whereas the spin-2 corrections are subleading to the one-loop supergravity correction which goes like $(G_N^{11d})^2 \sim c^{-2}$.

Similar reasoning applies to conformal field theories with string theory duals, like $\mathcal{N} = 4$ SYM with any fixed finite value of the string coupling. In that case, a contact interaction with D derivatives will have a coefficient of $G_N \alpha'^{(D-2)/2}$, where α' is related to the square of the string length. Writing this prefactor in terms of the central charge and string coupling, and fixing the latter at some finite value will then give an expansion in 1/c analogous to M-theory.

4.4 Conclusions and Future Directions

In this chapter, we derive recursion relations for anomalous dimensions of doubletrace operators in the 6d (2,0) theory. Given that no Lagrangian description is presently known for this model, our strategy is to use superconformal and crossing symmetry of four-point correlators of stress tensor multiplets. In particular, we expand the crossing equation to first order in the inverse central charge and then take the light-cone limit of the conformal cross-ratios to isolate the terms containing anomalous dimensions. Recursion relations then follow from truncating the conformal block expansion in spin and taking inner products of the resulting equation with certain hypergeometric functions, where we make use of an orthogonality relation of the hypergeometrics. These recursion relations can then be solved to obtain anomalous dimensions for arbitrary twist and spin, reproducing the results for low spin truncations previously obtained in [26]. As a warm-up, we derive analogous recursion relations in a toy model corresponding to an abstract bosonic 6d CFT, and match the results with the conformal block expansion of Witten diagrams in AdS_7 , confirming the holographic arguments of [30]. We describe an algorithm to solve these recursion relations and compute anomalous dimensions in both the bosonic and supersymmetric theories to any desired twist and spin truncation. We note that this method for extracting anomalous dimensions is much more efficient than extracting them using a conformal block expansion of a known four-point function.

The anomalous dimensions are physically significant because they encode higherderivative corrections to supergravity in $AdS_7 \times S^4$. The number of derivatives in each term of the effective action can be read off from the large-twist behaviour of the corresponding anomalous dimensions. Furthermore, the coefficients of these higherderivative terms correspond to free parameters of the solutions to the recursion relations. Therefore, these coefficients cannot be determined by our approach and we will discuss this further below.

Future directions

There are several interesting questions for future research related to the results discussed above.

- The first open question is the problem of fixing the coefficients of the higherderivative corrections in the low-energy effective action. These coefficients correspond to the free parameters in the solutions of the recursion relations for anomalous dimensions derived in this chapter and are thus not fixed by our approach. In the flat space limit, the coefficients of the R⁴ and D⁶R⁴ terms in the M-theory effective action have been deduced by uplifting string theory amplitudes (note that the D⁴R⁴ term vanishes in 11 dimensions) [114, 136], but the coefficient of the D⁸R⁴ term (which arises from a truncated spin-4 solution in our approach) is unknown. It would therefore be desirable to develop methods for fixing these coefficients using CFT techniques.
- A strategy for doing so was proposed in [28], and used to fix the coefficient of the R⁴ term and argue that the D⁴R⁴ term vanishes. This was achieved by applying the chiral algebra conjecture in [132] to four-point correlators of the form (kkkk) with k = 3, where k refers to a half-BPS scalar operator in the k-index symmetric traceless representation of the SO(5) R-symmetry group with scaling dimension 2k (note that k = 2 is the case considered in this chapter). It would therefore be interesting to find truncated spin solutions to the crossing equations for higher-charge correlators, compute the corresponding anomalous dimensions in their conformal block expansions, and ultimately fix the coefficients of higher-derivative terms in the M-theory effective action. Correlators of the form (kkkk) and (n + k, n k, k + 2, k + 2) were computed in the supergravity approximation in [129, 142] before all supergravity correlators were obtained in [130]. Moreover new solutions to the conformal Ward identities in Mellin space have been found for (kkkk) with k = 2, 3 in [28], so it would be

interesting to see how those methods are related to the ones developed in this chapter.

- Since the conformal blocks for higher-charge correlators appear to be much simpler in 4d [46, 49], it may be instructive to first carry out the analysis described above for 4d N = 4 SYM (for which a chiral algebra description was also proposed in [143]), and use it to deduce terms in the effective action for IIB string theory in AdS₅×S⁵ (which was the focus of the previous chapter).
- Finally, it would be very interesting to explore the loop expansion in M-theory in AdS₇×S⁴ using conformal bootstrap techniques, following on from the recent success in AdS₅×S⁵ [11, 41, 84–88, 90–92, 144, 145]. Since the loop expansion in M-theory is an expansion in 1/c, as is the expansion in higher-derivative corrections, the loop corrections have to be disentangled from the M-theoretic corrections. The first loop corrections to holographic correlators in the 6d (2,0) theory have been obtained in [146].

Chapter 5

AdS₂×S² Correlators: Effective Action and 4d Conformal Symmetry

This chapter is based on [44], which at the time of submission of this thesis is in preparation for publication. We will study different aspects of superconformal correlators in one dimension which are dual to tree-level scattering in quantum gravity in AdS₂×S². The Kaluza-Klein spectrum of supergravity in AdS₂×S² is described in [147, 148]. In particular, in [147] the authors start from 11d supergravity (see subsection 4.1.1) and dimensionally reduce it to obtain 4d supergravity. After compactifying on AdS₂×S² one then obtains 4d $\mathcal{N} = 2$ supergravity. Finally, compactifying 4d supergravity on S² yields an infinite tower of KK modes in AdS₂ which form representations of the SU(1,1|2) superalgebra. The holographic dual of $\mathcal{N} = 2$ supergravity in AdS₂×S² is expected to be a one-dimensional SCFT with superconformal group SU(1,1|2). We do not specify the underlying theory of quantum gravity beyond its symmetries in this chapter, besides the M-theory origin described above supergravity in AdS₂×S² could also originate from superstring theory in AdS₂×S²×T⁶. This is e.g. discussed in [149], where the authors consider embedding the AdS₂×S² background into 10d type IIA or IIB superstring theory. The AdS₂×S²×T⁶ background can be obtained either from a quarter-supersymmetric type IIB background describing four intersecting D3-branes [150] or from a type IIA background e.g. describing a superposition of three D4-branes and one D0-brane. As mentioned in the introduction, the $AdS_2 \times S^2$ background is interesting to study because it is the near-horizon geometry of extremal black holes in four dimensions. Furthermore, superconformal correlators in 1d are in many ways simpler than higher-dimensional analogues, such as the ones considered in the previous chapters. Therefore, they are an excellent playing ground for the study of various aspects of holographic correlators.

Firstly, we investigate a hidden conformal symmetry which was first discovered in $AdS_5 \times S^5$ in [41] and later investigated in $AdS_3 \times S^3$ in [124, 125]. In the supergravity approximation, four-point tree-level correlators of half-BPS operators dual to scalars in AdS_5 were found to correspond to 10d supergravity scattering amplitudes in the flat space limit. Consequently, it is conjectured that all supergravity tree-level fourpoint correlators exhibit a surprising 10d conformal symmetry and that correlators of any spherical harmonics can be generated from a single SO(10,2)-invariant function. This conjecture is also true for free theory and predictions can be made for loop corrections, however we will leave the discussion of loop corrections in 1d to [44]. The higher-dimensional conformal symmetry arises because four-point supergravity correlators in $AdS_5 \times S^5$ correspond to superamplitudes in 10d flat space which have a dimensionless coupling and are thus conformally invariant. Hence, this symmetry can only arise for specific four-point functions and furthermore only on conformally flat backgrounds, $AdS_q \times S^q$. In this chapter we aim to understand this hidden conformal symmetry more systematically and we will discuss it in the context of free theory, supergravity and higher-derivative corrections in 1d. The symmetry is generally broken for higher-derivative corrections and this was recently confirmed in [14], where the authors studied the double-trace spectrum of half-BPS correlators described by tree-level string theory in $AdS_5 \times S^5$. We will discuss this breaking of the symmetry and investigate higher-derivative corrections in $AdS_2 \times S^2$. We show that an infinite set of correlators for specific charges can be reconstructed from the

higher-dimensional conformal symmetry, while the symmetry is broken for general KK modes. We will briefly discuss the implications for α'^3 corrections in $\mathcal{N} = 4$ SYM. Even though the higher-dimensional conformal symmetry is only a symmetry of specific four-point functions in $\mathrm{AdS}_q \times \mathrm{S}^q$, it is a remarkable symmetry which implies powerful constraints on four-point functions and nicely complements other methods such as the effective action approach we proposed in chapter 3. We will consider a 4d analogue of this effective action in this chapter.

We propose a scalar effective action in four dimensions similar to the one in $AdS_5 \times S^5$. We do not prove the existence of the effective field theory but we deduce half-BPS correlators in the supergravity approximation as well as higher-derivative corrections from it and analyse the results. Where possible we compare the results to those from different methods, such as the 4d conformal symmetry, where we find agreement in the supergravity sector. The coefficients in the effective action encode the underlying quantum theory and they are unfixed in our considerations. To deduce all half-BPS correlators of any R-symmetry charge we again evaluate generalised Witten diagrams in AdS×S. These manifestly four-dimensional Witten diagrams contain all spherical harmonics and one can obtain all higher-derivative corrections and in the 1d case, also the supergravity correlators. Note that from the effective action approach in $AdS_2 \times S^2$ we can deduce supergravity and all higher-derivative corrections but neither free theory nor loop corrections. This is nicely complemented by the 4d hidden conformal symmetry which describes free theory and supergravity but is generically broken by higher-derivative corrections (it also makes predictions for loop corrections which we will not discuss in this thesis).

We start by reviewing the ten-dimensional conformal symmetry of supergravity correlators in $AdS_5 \times S^5$ before going on to consider holographic correlators in $AdS_2 \times S^2$.

5.1 Review: A Hidden Ten-Dimensional Conformal Symmetry

In [41] a surprising ten-dimensional conformal symmetry was found for tree-level supergravity correlators in $AdS_5 \times S^5$. Of course, the correlators have four-dimensional conformal symmetry and are dual to bulk scattering in AdS_5 with an infinite tower of spherical harmonics from compactifying on the sphere. However, it is not expected that they also have a ten-dimensional conformal symmetry manifestly including the modes on the sphere. This is very powerful since all higher-charge supergravity correlators can be packaged into a single generating function which is related to the flat space superamplitude in ten dimensions.

The first hint towards a higher-dimensional conformal symmetry was observed in [144], where the authors perform a conformal block analysis of four-point correlators in the large-c expansion in $\mathcal{N} = 4$ SYM described by tree-level supergravity in $AdS_5 \times S^5$. They consider half-BPS operators which are dual to the infinite tower of spherical harmonics on the five-sphere and make use of a formula for correlators of general charge in Mellin space, conjectured in [81]. As mentioned in subsection 4.2.1, when performing a conformal block decomposition of correlators there are usually many operators in the spectrum which are described by the same quantum numbers (scaling dimension and spin), thus there is a degeneracy. To resolve this degeneracy is called to solve the mixing problem and we will describe this in subsection 5.5.1before we discuss it in detail for supergravity and higher-derivative corrections in sections 5.6 and 5.8 respectively. It involves constructing matrices of conformal block coefficients for correlators with different R-symmetry representation and the same quantum numbers and solving an eigenvalue problem. In [144] the mixing problem for the anomalous dimensions of double-trace operators in the conformal block expansion of half-BPS correlators described by tree-level supergravity was solved and

a remarkably simple structure was observed:

$$\gamma_{4d}^{\text{sugra}} = \frac{-2M_t M_{t+l+1}}{(l_{10d}+1)_6} = \frac{\delta^{(8)}}{(l_{10d}+1)_6} , \qquad (5.1.1)$$

where

$$M_t = (t-1)(t+a)(t+a+b+1)(t+2a+b+2) , \qquad (5.1.2)$$

where a, b describe the R-symmetry representation of the double-trace operators which is given by the Dynkin labels [a, b, a]. The spin l and twist t are the quantum numbers of the exchanged operators, where the twist is defined as the scaling dimension minus the spin (see the discussion around (2.1.12)). Note that the anomalous dimensions are rational numbers and their simple structure can be interpreted in terms of an effective ten-dimensional spin

$$l_{10d} = l + a + 2(i+r) - 1 - \frac{1 + (-1)^{a+l}}{2} , \qquad (5.1.3)$$

where i, r label the degenerate operators (see [14] for details on the definition of these labels). The numerator in (5.1.1) can be understood as the eigenvalue $\delta^{(8)}$ of a special eighth-order differential operator $\Delta^{(8)}$ with the conformal blocks as eigenfunctions. The significance of this differential operator which is derived from conformal Casimirs will become clear later in this subsection.

It is very surprising that the anomalous dimensions after unmixing are rational numbers. They are eigenvalues of non-trivial matrices of OPE coefficients and there is no reason to expect them to be rational and to show such a simple structure. This was the first hint that tree-level supergravity in $AdS_5 \times S^5$ exhibits more symmetries than at first expected. In [41] the authors observed a striking similarity of (5.1.1) to coefficients of the partial wave expansion of the ten-dimensional supergravity scattering amplitude in flat space. This suggests that the structure in (5.1.1) arises from the conformal flatness of $AdS_5 \times S^5$ and a ten-dimensional conformal symmetry of the tree-level supergravity scattering amplitude. Let us look at this heuristic argument in more detail. The ten-dimensional IIB supergravity amplitude in flat space is given by

$$A_{10} = \frac{G_N \delta^{16}(Q)}{stu} \to G_N \frac{s^3}{tu}$$

$$(5.1.4)$$

where we have taken the dilaton component. Note that $G_N \delta^{16}(Q)$ is dimensionless, which is the origin of the 10d conformal symmetry of this amplitude. The partial wave expansion in general dimensions is given by

$$A_d(s, \cos \theta) = \frac{1}{s^{(d-4)/2}} \sum_l (l+1)_{d-4} C_l(\cos \theta) \mathcal{A}_l^d(s) , \qquad (5.1.5)$$

where the scattering angle $\cos \theta = 1 + \frac{2t}{s}$ and the partial waves C_l can be expressed in terms of Gegenbauer polynomials (see e.g. [151]). Expressing the 10d amplitude in terms of s and θ , we find

$$A_{10}(s,\cos\theta) = \frac{4G_N s}{\sin^2\theta} . \tag{5.1.6}$$

The single power of s in the numerator indicates two-derivative interactions, as expected for supergravity. Comparing this to (5.1.5) implies that the 10d partial wave coefficients are

$$\mathcal{A}_{l}^{10}(s) \sim 1 + \frac{R^{8}}{c} \frac{s^{4}}{(l+1)_{6}} ,$$
 (5.1.7)

where the one is put in by hand and we used the fact that the Newton constant $G_N \sim R^8/c$ in 10d, where R is the AdS radius and c is the central charge. Now compare this to the $\mathcal{N} = 4$ SYM anomalous dimensions after unmixing (5.1.1), where we note that γ^{sugra} contributes at order 1/c:

$$e^{\frac{1}{c}\gamma_{4d}^{\mathrm{sugra}}} \sim 1 + \frac{1}{c} \frac{\delta^{(8)}}{(l_{10d}+1)_6}$$
 (5.1.8)

The eighth-order Casimir encodes s^4 in the flat space limit and we see that the Pochhammers precisely match, which justifies the definition of the effective 10d spin in (5.1.3). The anomalous dimensions after unmixing can be interpreted in terms of this higher-dimensional spin and we will discuss this in sections 5.6 and 5.8 for supergravity and higher-derivative corrections in $AdS_2 \times S^2$ respectively. The above similarity is suggestive of a direct relation between the 4d CFT and flat space 10d supergravity. Importantly, this higher-dimensional conformal symmetry allows for all the correlators of external operators with any R-symmetry charge to be combined into one big 10d object which acts as a generating function. We will briefly review the 10d conformal symmetry for correlators in the supergravity approximation and in free theory in $\mathcal{N} = 4$ SYM below, followed by some general considerations including higher-derivative corrections.

One goal of this chapter is to understand the hidden higher-dimensional conformal symmetries more systematically and to approach this, we study a 4d conformal symmetry of holographic correlators in $AdS_2 \times S^2$. In this background the analogue of the dimensionless coupling $G_N \delta^{16}(Q)$, which is the origin of the 10d conformal symmetry above, is the dimensionless coupling $G_N \delta^4(Q)$ of 4d $\mathcal{N} = 2$ supergravity amplitudes in flat space. We will consider supergravity and free theory correlators and show that they exhibit a 4d conformal symmetry. Moreover, we extend our discussions to higher-derivative corrections for which the symmetry is generally broken, and discuss how some of the higher-dimensional conformal structure could still be intact even for higher-derivative corrections.

5.1.1 General Considerations and Higher-Derivative Corrections

First, consider four-point tree-level correlators in the supergravity limit which, as explained above, correspond to a flat space 10d supergravity amplitude with a dimensionless coupling $G_N \delta^{16}(Q)$ and are thus conformally invariant. It is then conjectured that all half-BPS correlators can be combined into a 10d generating function containing all KK modes. This conjecture can be extended to free theory. The higher-dimensional conformal symmetry arises when the correlator corresponds to a conformally invariant flat space scattering amplitude with dimensionless coupling. For tree-level supergravity this is the case for four-point functions in $AdS_5 \times S^5$ which is conformally flat. If the correlator does not correspond to such an amplitude, which is the case for free theory, it can be rescaled by acting with differential operators of appropriate powers on the correlator. This can be understood from dimensional analysis.

As we have seen, for supergravity the object which exhibits 10d conformal symmetry is the correlator itself. In the case of $AdS_5 \times S^5$, supergravity comes with a factor of $G_N \sim l_P^8$. Since free theory is proportional to l_P^0 , to construct an object of the right dimensions, one needs to act with a specific eighth-order differential operator on the correlator to get something of order l_{P}^{8} . The right differential operator to consider is $\Delta^{(8)}$ which is the same operator whose eigenvalue appears in the anomalous dimensions in (5.1.1). This shows the significance of $\Delta^{(8)}$ which is derived from quadratic Casimir operators. Acting with $\Delta^{(8)}$ on free theory correlators gives something that looks exactly like free theory of superconformal descendants (see [41] and section 5.3 below). Indeed, the correlator of descendants is the object that plays the leading role in the hidden conformal symmetry. In the case of supergravity, which is of order 1/c, to uncover the 10d conformal symmetry one has to divide the tree-level superamplitude by the dimensionless coupling $G_N \delta^{16}(Q)$ which according to the discussion around (5.1.7) can be identified with $\Delta^{(8)}/c$. Hence, in the case of supergravity the correlator itself is the relevant object which exhibits 10d conformal symmetry while for free theory it is the correlator of descendants. Moreover, this discussion can be extended to loop corrections. These corrections contribute at higher negative orders in c and to obtain the corresponding 10d invariant object one would act with the appropriate negative powers of $\Delta^{(8)}$ on the relevant correlators.

In this chapter we study the hidden conformal symmetry in the context of onedimensional SCFTs with hidden four-dimensional conformal symmetry. This case is much simpler than the 4d/10d case and therefore a good starting point to understand this surprising symmetry in more detail. Analogue to the above discussions for 10d, 4d $\mathcal{N} = 2$ supergravity amplitudes in flat space have a dimensionless coupling $G_N \delta^4(Q)$. Thus, the corresponding 1d tree-level supergravity correlators of four chiral primaries dual to KK modes in AdS₂ are expected to exhibit a 4d conformal symmetry. We will show that this is indeed the case in subsection 5.4.2. Supergravity in AdS₂×S² comes with a factor of $G_N \sim l_P^2$, this is how one sees whether the object has the right dimensions to be invariant under 4d conformal symmetry or not. Since 4d $\mathcal{N} = 2$ supergravity is connected to a scaleless coupling in the flat space limit, whenever a correlator comes with $\sim l_P^2$ in 1d it is connected to a flat space amplitude with dimensionless coupling and thus has 4d conformal symmetry. If it does not, a 4d conformal object can be constructed by acting with differential operators derived from conformal Casimirs on the correlator. Therefore, free theory, which goes like l_P^0 , needs to be acted on by a second-order differential operator $\Delta^{(2)}$ (analogue to $\Delta^{(8)}$ above) to get something invariant under 4d conformal symmetry. As in the 10d case above, $\Delta^{(2)}$ acting on the free theory correlator gives exactly free theory of superconformal descendants. We will show this in detail in section 5.3. This can be extended to loop corrections, but we leave this discussion to [44].

Finally, we attempt to extend these considerations to higher-derivative corrections, which were not a part of the considerations in [41]. Generally, the higher-dimensional conformal symmetry is broken for higher-derivative corrections, which can be seen specifically when unmixing anomalous dimensions in the double-trace spectrum of the conformal block expansion of half-BPS correlators corresponding to higherderivative corrections. These anomalous dimensions after unmixing still show a simple structure and many of them are rational numbers as was the case in the supergravity approximation. However, there are anomalous dimensions which contain square roots and this indicates a breaking of the higher-dimensional conformal symmetry. These anomalous dimensions were obtained for $\mathcal{N} = 4$ SYM in [13, 14] up to four derivatives and in [16] for higher derivatives. We will study them in detail in section 5.8 for the four-derivative corrections in 1d. It is reasonable to expect that some of the hidden conformal symmetry is still intact for higher-derivative corrections since, as mentioned before, many of the anomalous dimensions are still rational numbers. Remarkably, for an infinite set of correlators with specific charges, one can indeed construct an object that does have higher-dimensional conformal symmetry. The breaking of the symmetry for generic higher-derivative corrections can be anticipated because when an interaction vertex contains derivatives and they are reduced on the sphere, this gives several terms with different numbers of derivatives in AdS which cannot all be rescaled simultaneously. This will be discussed further in subsection 5.7.3.

Let us now consider the dimensional analysis for higher-derivative corrections in $AdS_2 \times S^2$. Recall that higher-derivative corrections contribute to the low-energy effective action as interaction vertices with k+2 derivatives¹, where supergravity has two derivatives and we restrict to quartic interactions in this thesis. The vertices with ∂^{k+2} go like $G_N l_P^k$, so to get an interaction term with the right scaling, i.e. $G_N \sim l_P^2$, one has to multiply it by l_P^{-k} . To achieve this, one acts with $(\Delta^{(2)})^{-\frac{k}{2}}$ on the correlators. In practice this means that we consider objects which can be lifted to four dimensions to generate all higher-charge correlators (which turn out to be 4d conformal blocks) and then act with differential operators of positive order k on them to reconstruct the higher-charge versions of the higher-derivative corrections. We analyse this for the case of four-derivative corrections in subsection 5.7.3 where it turns out that this can be constructed for all correlators of four external operators with the charges $\{p_i\} = pp11$, where 1 is the lowest mode on the two-sphere and p corresponds to a general KK mode, and crossing versions but is broken for general correlators. Note that in $AdS_2 \times S^2$ the zero-derivative term describes supergravity but in $AdS_5 \times S^5$ the analogue term describes the first higher-derivative correction to supergravity arising from string theory, which has six more derivatives than supergravity itself.

To summarise, the higher-dimensional conformal symmetry can be realised for specific

¹Note that we often count the number of derivatives not necessarily compared to supergravity but compared to the correlator dual to the zero-derivative bulk scalar interaction ϕ^4 . In AdS₂×S² this is indeed supergravity but in higher dimensions, ϕ^4 corresponds to the term \mathcal{R}^4 in the lowenergy effective action (2.3.18) which describes the first higher-derivative correction and has six more derivatives than supergravity. This correction is referred to as the zero-derivative correction and subleading corrections are interactions with k - 6 derivatives acting on \mathcal{R}^4 , where k + 2 is the total number of derivatives in the interaction.

four-point correlators in conformally flat backgrounds. It is not just a symmetry in the supergravity case but can be understood from constructing interactions with the required scaling by acting on the correlator with appropriate differential operators. However, for higher-derivative corrections the symmetry breaks in general due to covariant derivatives acting on the interaction vertices. Nevertheless, there is still an infinite number of correlators with specific R-symmetry charges which can be constructed from the hidden conformal symmetry. Note that on the other hand, all higher-derivative corrections can be deduced from the 4d effective action, see subsection 5.7.2.

5.2 General Setup

In this chapter we study four-point correlators of half-BPS operators in a large central charge c expansion in 1d which correspond to tree-level quantum gravity scattering in the bulk. In the large-c limit, the half-BPS operators are dual to bulk scalars in AdS₂ described by an infinite tower of KK modes on the two-sphere. Furthermore, we consider the theory in the low-energy approximation, which corresponds to 4d $\mathcal{N} = 2$ supergravity, where we also consider subleading contributions in the form of higher-derivative corrections. Since we do not specify the bulk theory nor the boundary theory beyond its symmetries we formulate the low-energy approximation in terms of an expansion in a small parameter a, where $a \rightarrow 0$ corresponds to the strict low-energy limit. Considering this double-expansion, free theory correlators are proportional to a^0c^0 , supergravity correlators contribute at order a^0c^{-1} and the higher-derivative corrections to the supergravity effective action contribute at orders $c^{-1}a^{k-1}$, where 2k is the number of derivatives in the bulk scalar interaction and $k = 2, 3, \ldots$

5.2.1 Correlators in 1d and the Differential Operator $\Delta^{(2)}$

Let us start by looking at the general structure of four-point functions in 1d CFTs. The local operators are defined on a line and are invariant under the SO(2,1)conformal group. This symmetry allows us to fix three points on the line, therefore there is only one free real parameter left, describing the position of the fourth point. As a consequence, the usual conformal cross-ratios u, v are not independent and there is only one cross-ratio, x. Starting from the usual conformal cross-ratios:

$$u = z\bar{z} = \frac{x_{12}^2 x_{34}^2}{x_{13}^2 x_{24}^2} , \quad v = (1-z)(1-\bar{z}) = \frac{x_{23}^2 x_{14}^2}{x_{13}^2 x_{24}^2} , \quad (5.2.1)$$

where $x_{ij} = x_i - x_j$ and defining $z = \overline{z} = x$, the 1d cross-ratio x is

$$x = \frac{x_{12}x_{34}}{x_{13}x_{24}} , \qquad (5.2.2)$$

and the usual u, v are given in terms of x by

$$u = x^2$$
, $v = (1 - x)^2$. (5.2.3)

Hence, 1d correlators correspond to a holomorphic limit.

To discuss the half-BPS correlators of interest, we follow the formalism of [49] with m = n = 1. The relevant superspace is the super Grassmannian Gr(1|1, 2|2) of (1|1) planes in (2|2) dimensions. Coordinates on this Grassmannian can be given as

$$X_{i} = \begin{pmatrix} x_{i} & \theta_{i} \\ \bar{\theta}_{i} & y_{i} \end{pmatrix}, \qquad (5.2.4)$$

where x_i is the 1d spacetime coordinate, y_i is a (complex) internal coordinate used to deal with the SU(2) structure and $\theta_i, \bar{\theta}_i$ are Grassmann odd coordinates. Since we will be dealing with correlators of four operators on this space, we added a subscript i = 1, 2, 3, 4 to denote the particle number.

The field content of 4d \mathcal{N} = 2 supergravity is one graviton, six gravitinos, 15 vector and 10 (complex) hypermultiplets [147]. Each of these multiplets contains four bosonic and four fermionic degrees of freedom and compactifying on the two-sphere gives an infinite tower of KK modes for each multiplet. The 4d $\mathcal{N} = 2$ hypermultiplet is the simplest to understand from a higher-dimensional perspective. This multiplet on an AdS₂×S² background is AdS/CFT dual to an infinite tower of fermionic 1d half-BPS multiples [147]. These are fermionic superfields of scaling dimension Δ and SU(2) representation Δ . They can be written on the above super Grassmannian and decompose into the following component fields

$$\Psi_{\Delta}(x_i,\theta_i,\bar{\theta}_i,y_i) = \psi_{\Delta} + \theta_i\phi_{\Delta+\frac{1}{2}} + \bar{\theta}_i\bar{\phi}_{\Delta+\frac{1}{2}} + \theta_i\bar{\theta}_i\lambda_{\Delta+1} , \qquad \Delta = \frac{1}{2}, \frac{3}{2}, \frac{5}{2}, \dots , \quad (5.2.5)$$

where the subscript denotes the dimension of the field. The field ψ_{Δ} has SU(2)representation Δ , ϕ_{Δ} has SU(2) representation $\Delta - 1$, λ_{Δ} has SU(2) representation $\Delta - 2$. For the special multiplet $\Psi_{1/2}$ the descendant $\lambda_{3/2}$ is absent. Expanding in the y_i coordinates manifests the SU(2) indices for these representations

$$\psi_{\Delta}(x_{i}, y_{i}) = \psi_{I_{1}...I_{2\Delta}}(x_{i})y_{i}^{I_{1}}...y_{i}^{I_{2\Delta}}, \qquad \Delta = \frac{1}{2}, \frac{3}{2}, ..., \phi_{\Delta}(x_{i}, y_{i}) = \phi_{I_{1}...I_{2\Delta-2}}(x_{i})y_{i}^{I_{1}}...y_{i}^{I_{2\Delta-2}}, \qquad \Delta = 1, 2, ..., \lambda_{\Delta}(x_{i}, y_{i}) = \lambda_{I_{1}...I_{2\Delta-4}}(x_{i})y_{i}^{I_{1}}...y_{i}^{I_{2\Delta-4}}, \qquad \Delta = \left(\frac{3}{2}\right), \frac{5}{2}, ...,$$
(5.2.6)

where

$$y_i^I = (1, y_i) . (5.2.7)$$

We are interested in the four-point functions of half-BPS superconformal primaries:

$$\langle \psi_{\Delta_1}(x_1, y_1)\psi_{\Delta_2}(x_2, y_2)\psi_{\Delta_3}(x_3, y_3)\psi_{\Delta_4}(x_4, y_4)\rangle .$$
 (5.2.8)

However, it will also be useful to consider the four-point function of superdescendants

$$\langle \phi_{\Delta_1}(x_1, y_1) \phi_{\Delta_2}(x_2, y_2) \bar{\phi}_{\Delta_3}(x_3, y_3) \bar{\phi}_{\Delta_4}(x_4, y_4) \rangle$$
 (5.2.9)

Indeed it is the latter which plays the leading role in the hidden higher-dimensional conformal symmetry as described in section 5.1. Both correlators transform covariantly under the bosonic subgroup of the superconformal group $SU(1,1) \times SU(2) \subset$

SU(1,1|2) which means they have the form

$$\langle \psi_{\Delta_1}(x_1, y_1)\psi_{\Delta_2}(x_2, y_2)\psi_{\Delta_3}(x_3, y_3)\psi_{\Delta_4}(x_4, y_4)\rangle = P_{\Delta_i}G_{\psi_{\Delta_i}}(x, y) , \langle \phi_{\Delta_1}(x_1, y_1)\phi_{\Delta_2}(x_2, y_2)\bar{\phi}_{\Delta_3}(x_3, y_3)\bar{\phi}_{\Delta_4}(x_4, y_4)\rangle = \frac{P_{\Delta_i - \frac{1}{2}}}{x_{12}x_{34}y_{12}y_{34}}G_{\phi_{\Delta_i}}(x, y) ,$$
 (5.2.10)

where the prefactor P_{Δ_i} is

$$P_{\Delta_{i}} = g_{12}^{\Delta_{1}+\Delta_{2}} g_{34}^{\Delta_{3}+\Delta_{4}} \left(\frac{g_{24}}{g_{14}}\right)^{\Delta_{21}} \left(\frac{g_{14}}{g_{13}}\right)^{\Delta_{43}} ,$$

with $g_{ij} = \frac{y_{ij}}{x_{ij}} , \quad y_{ij} = y_{i} - y_{j} ,$ (5.2.11)

and $\Delta_{ij} = \Delta_i - \Delta_j$. This prefactor by itself transforms correctly as a $\langle \psi_{\Delta_1} \psi_{\Delta_2} \psi_{\Delta_3} \psi_{\Delta_4} \rangle$ correlator under the bosonic subgroup, leaving a remaining function $G_{\psi}(x, y)$ which is conformally invariant. Thus, it is a function of the conformal cross-ratios x, y, the spacetime and internal cross-ratios in one dimension:

$$x = \frac{x_{12}x_{34}}{x_{13}x_{24}}$$
, $y = \frac{y_{12}y_{34}}{y_{13}y_{24}}$. (5.2.12)

The spacetime cross-ratio was discussed above, around (5.2.2), and equivalently for the internal cross-ratios, there is only a single independent one, y.

The two component correlators (5.2.10) are related to each other by superconformal symmetry. Indeed both can be obtained from the same supercorrelator $\langle \Psi_{\Delta_1} \Psi_{\Delta_2} \Psi_{\Delta_3} \Psi_{\Delta_4} \rangle$ as

$$\langle \psi_{\Delta_1}\psi_{\Delta_2}\psi_{\Delta_3}\psi_{\Delta_4}\rangle = \langle \Psi_{\Delta_1}\Psi_{\Delta_2}\Psi_{\Delta_3}\Psi_{\Delta_4}\rangle|_{\theta_i=\bar{\theta}_i=0} ,$$

$$\langle \phi_{\Delta_1+\frac{1}{2}}\phi_{\Delta_2+\frac{1}{2}}\bar{\phi}_{\Delta_3+\frac{1}{2}}\bar{\phi}_{\Delta_4+\frac{1}{2}}\rangle = \partial_{\theta_1}\partial_{\theta_2}\partial_{\bar{\theta}_3}\partial_{\bar{\theta}_4}\langle \Psi_{\Delta_1}\Psi_{\Delta_2}\Psi_{\Delta_3}\Psi_{\Delta_4}\rangle|_{\theta_i=\bar{\theta}_i=0} .$$

$$(5.2.13)$$

Superconformal invariance implies that the full supercorrelator only depends nontrivially on two bosonic variables. This then implies that the above relations reduce to the following direct differential relation between the two component correlators

$$\left\langle \phi_{\Delta_{1}+\frac{1}{2}}\phi_{\Delta_{2}+\frac{1}{2}}\bar{\phi}_{\Delta_{3}+\frac{1}{2}}\bar{\phi}_{\Delta_{4}+\frac{1}{2}} \right\rangle = \mathcal{I}^{-1}\mathcal{C}_{1,2}^{SU(1,1|2)} \left\langle \psi_{\Delta_{1}}\psi_{\Delta_{2}}\psi_{\Delta_{3}}\psi_{\Delta_{4}} \right\rangle , \qquad (5.2.14)$$

where we leave the derivation of this result to appendix H. Here $\mathcal{C}_{1,2}^{SU(1,1|2)}$ denotes

the superconformal quadratic Casimir given in terms of the second-order differential operator $\Delta^{(2)}$ by

$$\mathcal{C}_{1,2}^{SU(1,1|2)} = P_{p_i} \times \frac{x - y}{xy} \Delta^{(2)} \frac{xy}{x - y} P_{p_i}^{-1} ,$$

$$\Delta^{(2)} = \mathcal{D}_x^{(p_{12}, p_{43})} - \mathcal{D}_y^{(-p_{12}, -p_{43})} ,$$

$$\mathcal{D}_x^{(p_{12}, p_{43})} = x^2 \partial_x (1 - x) \partial_x + (p_{12} + p_{43}) x^2 \partial_x - p_{12} p_{43} x , \qquad (5.2.15)$$

where $p_{ij} = p_i - p_j$. For a derivation of the quadratic Casimir see appendix H. Furthermore, \mathcal{I} is the special polynomial

$$\mathcal{I} = x_{12}x_{34}y_{13}y_{24} - y_{12}y_{34}x_{13}x_{24} , \qquad (5.2.16)$$

which is completely antisymmetric under crossing symmetry.

The second-order differential operator $\Delta^{(2)}$ is the analogue of $\Delta^{(8)}$ playing an important role in the 10d conformal symmetry of $\mathcal{N} = 4$ SYM correlators. We have discussed the significance of $\Delta^{(2)}$ in subsection 5.1.1. Recall that the object that has 4d conformal symmetry in $\mathrm{AdS}_2 \times \mathrm{S}^2$ in the supergravity limit is the half-BPS correlator itself, whereas for free theory acting with $\Delta^{(2)}$ on the correlator yields the 10d conformally invariant object which is the correlator of descendant bosonic scalars ϕ . In 1d this is related to the correlator of chiral primaries ψ by acting on the latter with $\Delta^{(2)}$, see (5.2.14).

Throughout this chapter we will consider four-point functions of half-BPS operators and in the free theory case also the correlators of descendants. The half-BPS operators are fermionic primaries ψ_{Δ} with dimensions and SU(2) charges $\Delta = \frac{1}{2}, \frac{3}{2}, \ldots$ However, it will turn out to be most useful to label the correlators in terms of the bosonic descendant operators $\phi_{\Delta+1/2}$ with dimensions $1, 2, \ldots$ Therefore, we define the half-BPS operators as

$$\mathcal{O}_p = (-1)^{\frac{p}{2}} \frac{\psi_{(2p-1)/2}}{\sqrt{2p-1}} , \qquad (5.2.17)$$

where p = 1, 2, ... In this convention the half-BPS operators have dimension and SU(2) representation (2p-1)/2. Furthermore, it is useful to introduce a normalisa-

tion $(-1)^{p/2} (2p-1)^{-1/2}$ inspired by the higher-dimensional conformal symmetry [41]. It will become clear why this is useful in the discussions below (5.3.12) and (5.3.20). We then denote the descendants of \mathcal{O}_p by L_p, \bar{L}_p :

$$\mathbf{O}_p(x,y) = \mathcal{O}_p(x,y) + \theta L_p(x,y) + \theta \bar{L}_p(x,y) , \qquad (5.2.18)$$

where L_p has dimension p and SU(2) charge p-1.

5.2.2 Conformal Blocks

In this subsection we discuss the conformal blocks in 1d. We will consider four-point functions of half-BPS operators \mathcal{O}_{p_i} (5.2.17) with $i = 1, \ldots, 4$. In the OPE of two such half-BPS operators one finds also long multiplets. These can all be represented as $\mathcal{O}_{\Delta,p}$, where Δ is the dilatation weight and p is the SU(2) representation of the exchanged operators.

The four-point function can be expanded in superconformal blocks $B_{\Delta,p}(x,y)$ as

$$\langle \mathcal{O}_{p_1}(x_1, y_1) \mathcal{O}_{p_2}(x_2, y_2) \mathcal{O}_{p_3}(x_3, y_3) \mathcal{O}_{p_4}(x_4, y_4) \rangle$$

= $\sum_{\Delta=1}^{\infty} \sum_{p=0}^{p_1+p_2-2} A_{\Delta,p}^{p_1p_2p_3p_4} g_{12}^{p_1+p_2-1} g_{34}^{p_3+p_4-1} \left(\frac{g_{24}}{g_{14}}\right)^{p_{21}} \left(\frac{g_{14}}{g_{13}}\right)^{p_{43}} B_{\Delta,p,p_{12},p_{34}}(x, y) , \quad (5.2.19)$

where the coefficient $A_{\Delta,p}$ is given in terms of a sum of squares of OPE coefficients as

$$A^{p_1 p_2 p_3 p_4}_{\Delta, p} = \sum_{\mathcal{O}^{\Delta, p}, \tilde{\mathcal{O}}^{\Delta, p}} C^{\mathcal{O}}_{p_1 p_2} C^{\tilde{\mathcal{O}}}_{p_3 p_4} C_{\mathcal{O}\tilde{\mathcal{O}}} .$$
(5.2.20)

Further

$$\tilde{g}_{ij} = \frac{y_{ij}}{|x_{ij}|} \tag{5.2.21}$$

is the two-point function of charge 1 half-BPS operators. Note that the two-point function \tilde{g}_{ij} is antisymmetric under exchange *i* and *j*, as expected for fermions. It

will also be convenient to define the bosonic two-point function

$$g_{ij} = \frac{y_{ij}}{x_{ij}} \ . \tag{5.2.22}$$

The superconformal blocks are derived from a general formula for superconformal blocks with symmetry group SU(m, m|2n) in [49]. In our case m = n = 1 and the blocks for long multiplets are given by

$$B_{\Delta,p,p_{12},p_{34}}^{\text{long}}(x,y) = -(x-y)(-x)^{\Delta}{}_{2}F_{1}(\Delta+1-p_{12},\Delta+1+p_{34};2\Delta+2;x)$$
$$\times y^{-p-1}{}_{2}F_{1}(-p+p_{12},-p-p_{34};-2p;y) .$$
(5.2.23)

For half-BPS multiplets $\Delta = p$ and the blocks are

$$B_{p,p_{12},p_{34}}^{\text{half-BPS}}(x,y) = \left(-\frac{x}{y}\right)^p \left(1 + (x-y)\sum_{i=1}^k [x^{-i} {}_2F_1(p+1-i-p_{12},p+1-i+p_{34};2p+2-2i;x)] \times y^{i-1} {}_2F_1(i-p+p_{12},i-p-p_{34};2i-2p;y)\right), \qquad (5.2.24)$$

where $k = \min(p - p_{12}, p + p_{34})$ and the square bracket means we take the regular piece as $x \to 0$.

It is interesting to note that the block of an operator with dimension Δ and SU(2) representation p contributes as follows to the four-point function

$$\langle \mathcal{O}_{p_1}(x_1, y_1) \mathcal{O}_{p_2}(x_2, y_2) \mathcal{O}_{p_3}(x_3, y_3) \mathcal{O}_{p_4}(x_4, y_4) \rangle \sim \sum_{\Delta, p} A_{\Delta, p}^{p_1 p_2 p_3 p_4} g_{12}^{p_1 + p_2 - 1} g_{34}^{p_3 + p_4 - 1} \left(\frac{g_{24}}{g_{14}}\right)^{p_{21}} \left(\frac{g_{14}}{g_{13}}\right)^{p_{43}} (-x)^{\Delta} y^{-p} \left(1 + \mathcal{O}(x, y)\right) , \quad (5.2.25)$$

where the higher orders in x, y correspond to spacetime or SU(2) descendants.

5.2.3 4d Scalar Effective Action

In addition to studying the 4d hidden conformal symmetry in this chapter, we also obtain the correlators described by tree-level supergravity and higher-derivative corrections in $AdS_2 \times S^2$ from a higher-dimensional effective action analogous to the 10d effective action proposed in chapter 3. There a 10d scalar effective action was introduced which generates all half-BPS four-point correlators in $\mathcal{N} = 4$ SYM described by tree-level string theory in $AdS_5 \times S^5$ [42].

For the present case, we consider the following four-field terms in an effective superpotential for supergravity linearised about flat 4d space, with the expansion parameter a (where $a \rightarrow 0$ corresponds to the low-energy limit):

$$V^{\text{flat}}(\phi) = \frac{1}{4!} \left(A \phi^4 + B a \left(\partial \phi \partial \phi \right)^2 + D a^2 \left(\partial \phi \partial \phi \right) \left(\partial_\mu \partial_\nu \phi \partial^\mu \partial^\nu \phi \right) + \dots \right), \quad (5.2.26)$$

where in this chapter we mainly focus on the first two terms. In subsection 5.4.2 we will see that in $AdS_2 \times S^2$ the zero-derivative term describes supergravity and thus the ϕ^4 term in the effective action is of order a^0 . The fact that the supergravity approximation is included in the effective action is very powerful since we can compute all half-BPS supergravity correlators from the 4d effective action alongside all higher-derivative corrections, whereas in $AdS_5 \times S^5$ only higher-derivative corrections can be obtained from the effective action. We organise the expansion in a such that 2k-derivative terms contribute at order a^{k-1} starting from k = 2, where k = 0 terms are excluded since they describe supergravity and not higher-derivative corrections.

To uplift the effective superpotential to an $AdS_2 \times S^2$ background the flat derivatives are replaced with covariant AdS×S derivatives, see (2.3.6). It is important to note that this uplift is not unique as we have seen in the $AdS_5 \times S^5$ case. There are ambiguities because the covariant derivatives no longer commute with each other and there could be contributions from interactions with a lower number of derivatives, compensated by the AdS radius R, which would vanish in the flat space limit. So to $\mathcal{O}(a^2)$ the superpotential translates to

$$V^{\text{AdS}\times\text{S}}(\phi) = \frac{1}{4!} \Big[A\phi^4 + a \Big(3B(\nabla\phi,\nabla\phi)^2 + 6C\nabla^2\nabla_\mu\phi\nabla^\mu\phi\phi^2 \Big) \\ + a^2 \Big(6D(\nabla\phi,\nabla\phi)(\nabla_\mu\nabla_\nu\phi\nabla^\mu\nabla^\nu\phi) + 6E\nabla_\mu\nabla^2\nabla_\nu\phi\nabla^\mu\nabla^\nu\phi\phi^2 \Big) + \dots \Big] .$$
(5.2.27)

As in the $AdS_5 \times S^5$ case (3.1.18), the coefficients B, C, \ldots themselves can have an expansion in the dimensionless parameter a/R^2 where R is the radius of AdS (or S). So whereas in flat space 2k-derivative terms only contribute at order a^{k-1} , in AdS×S,

2k-derivative terms can contribute at a^{k-1} and all higher orders in principle. These expansions are

$$B(a) = B_0 + B_1 \frac{a}{R^2} + \dots$$

$$C(a) = C_0 + C_1 \frac{a}{R^2} + \dots$$
...
(5.2.28)

For simplicity, we will set R = 1 from now on throughout this chapter, but it will be understood that these higher-order terms vanish in the flat space limit. The supergravity coefficient A does not allow for a a/R^2 expansion because of superconformal symmetry (see also subsection 3.1.3).

5.3 Free Theory

In this section we will derive the 1d free theory correlators. As explained in subsection 5.1.1, the object that has 4d conformal symmetry in free theory is the correlator of descendants which is obtained by acting on the half-BPS correlator with the second-order differential operator $\Delta^{(2)}$. One can then construct a four-dimensional generating function which contains all 1d half-BPS correlators in free theory.

First, write the correlators according to (5.2.10) in terms of a prefactor and a function only depending on the conformal cross-ratios (5.2.12) as

$$\langle \mathcal{O}_{p_1}(x_1, y_1) \mathcal{O}_{p_2}(x_2, y_2) \mathcal{O}_{p_3}(x_3, y_3) \mathcal{O}_{p_4}(x_4, y_4) \rangle = P_{p_i} \times G_{p_1 p_2 p_3 p_4}(x, y) ,$$
 (5.3.1)

where the prefactor is given by

$$P_{p_i} = g_{12}^{p_1 + p_2 - 1} g_{34}^{p_3 + p_4 - 1} \left(\frac{g_{24}}{g_{14}}\right)^{p_{21}} \left(\frac{g_{14}}{g_{13}}\right)^{p_{43}} .$$
 (5.3.2)

It is useful to decompose the correlator in such a way that the solution to the superconformal Ward identities becomes straightforward. The superconformal Ward identities in 1d are $\partial_x G(x,y)|_{x=y} = 0$, i.e. simply that G(x,x) is independent of x.

This has the straightforward solution

$$G_{p_i}(x,y) = k_{p_i}(x,y) + \frac{x-y}{xy} H_{p_i}(x,y) ,$$

where $k_{p_i}(x,y) = \kappa \left(\frac{x}{y}\right)^{p_{43}} ,$ (5.3.3)

with H(x,x) finite and $k_{p_i}(x,y)$ is defined such that $\mathcal{C}_{1,2}^{SU(1,1|2)}(P_{p_i}k_{p_i}) = 0$ with a constant κ and the superconformal Casimir is given in (5.2.15).

In this decomposition the Casimir only sees the interacting piece of the correlator, the so-called reduced correlator H_{p_i} :

$$\mathcal{C}_{1,2}^{SU(1,1|2)} \langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle = P_{p_i} \frac{x-y}{xy} \Delta^{(2)} H_{p_i}(x,y) = P_{p_i} \frac{x-y}{xy} \tilde{H}_{p_i}(x,y) , \quad (5.3.4)$$

where we define

$$\tilde{H}_{p_i}(x,y) = \Delta^{(2)} H_{p_i}(x,y) .$$
(5.3.5)

To construct an object which has the right dimensions to transform like a correlator which has 4d conformal symmetry, we define the correlator of descendants in terms of \tilde{H} like

$$\langle L_{p_1} L_{p_2} \bar{L}_{p_3} \bar{L}_{p_4} \rangle = \frac{P_{p_i}}{x_{12} x_{34} y_{12} y_{34}} \tilde{H}_{p_i}(x, y) , \qquad (5.3.6)$$

where L_p is the descendant of \mathcal{O}_p with dimension p and SU(2) charge p-1, as described in (5.2.18). Comparing equations (5.3.4) and (5.3.6) we get

$$\langle L_{p_1} L_{p_2} \bar{L}_{p_3} \bar{L}_{p_4} \rangle = \mathcal{I}^{-1} \mathcal{C}_{1,2}^{SU(1,1|2)} \langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle , \qquad (5.3.7)$$

which agrees with the relation in (5.2.14).

5.3.1 4d Conformal Symmetry

Let us now construct the four-dimensional generating function for free theory correlators, starting with the correlators of half-BPS operators of equal charge. The free theory correlator for equal charge operators is given in terms of the fermionic two-point functions \tilde{g}_{ij} (5.2.21) as

$$\begin{aligned} \langle \mathcal{O}_{p}(x_{1}, y_{1}) \mathcal{O}_{p}(x_{2}, y_{2}) \mathcal{O}_{p}(x_{3}, y_{3}) \mathcal{O}_{p}(x_{4}, y_{4}) \rangle |_{c^{0}} \\ &= \frac{1}{\mathcal{N}_{pppp}^{(0)}} \left[\left(\tilde{g}_{12} \tilde{g}_{34} \right)^{2p-1} - \left(\tilde{g}_{13} \tilde{g}_{24} \right)^{2p-1} + \left(\tilde{g}_{14} \tilde{g}_{23} \right)^{2p-1} \right] \\ &= \frac{1}{\mathcal{N}_{pppp}^{(0)}} \left(\tilde{g}_{12} \tilde{g}_{34} \right)^{2p-1} \left[1 + \left(\frac{x}{y} \right)^{2p-1} \left((-\operatorname{sgn} x) + \left(\frac{1-y}{1-x} \right)^{2p-1} \operatorname{sgn} \left[x \left(1-x \right) \right] \right) \right] , \quad (5.3.8) \end{aligned}$$

where the normalisation

$$\mathcal{N}_{p_i}^{(0)} = (-1)^{\Sigma_p} \sqrt{(2p_1 - 1)(2p_2 - 1)(2p_3 - 1)(2p_4 - 1)}$$
(5.3.9)

comes from the normalisation of the half-BPS operators in (5.2.17) and recall that $\Sigma_p = \frac{1}{2} (p_1 + p_2 + p_3 + p_4)$. Choosing $0 \le x \le 1$, we then find that

$$G_{pppp}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pppp}^{(0)}} \left[1 + \left(\frac{x}{y}\right)^{2p-1} \left(-1 + \left(\frac{1-y}{1-x}\right)^{2p-1} \right) \right] .$$
(5.3.10)

This decomposes into

$$k_{pppp}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pppp}^{(0)}}, \quad H_{pppp}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pppp}^{(0)}} \frac{x\,y}{x-y} \left(\frac{x}{y}\right)^{2p-1} \left[-1 + \left(\frac{1-y}{1-x}\right)^{2p-1} \right]. \quad (5.3.11)$$

The action of $\Delta^{(2)}$ then yields

$$\tilde{H}_{pppp}^{(0)} = \frac{x^{2p}}{y^{2p-2}} \left(1 + \frac{(1-y)^{2p-2}}{(1-x)^{2p}} \right) , \qquad (5.3.12)$$

which indeed looks like the free theory of dimension p charge p-1 operators you would get as the descendant. Acting with $\Delta^{(2)}$ on H gives a factor of $(2p-1)^2$ which gets cancelled by the normalisation $1/\mathcal{N}_{pppp}^{(0)}$. This is how the normalisation in (5.2.17) was chosen, up to the sign which will become clear shortly.

Note that for the analysis of free theory in terms of the 4d conformal symmetry this decomposition is not necessary since acting with $\Delta^{(2)} \frac{xy}{x-y}$ on G(x, y) only sees the interacting piece H anyway. Nevertheless, the discussion below can be written in terms of H in a simple way. Furthermore, the decomposition is useful as it will allow us to consider the 1/c expansion of H in (5.5.2) later on.

Finally, consider the correlator of descendants from which we construct the generating

function with 4d conformal symmetry. From (5.3.6) and (5.3.12) we get

$$\langle L_p L_p \bar{L}_p \bar{L}_p \rangle_{c^0} = g_{14}^{2p-2} g_{23}^{2p-2} \frac{1}{x_{14}^2 x_{23}^2} + g_{13}^{2p-2} g_{24}^{2p-2} \frac{1}{x_{13}^2 x_{24}^2} .$$
 (5.3.13)

Note that the correlator of bosonic descendants is given in terms of bosonic two-point functions g_{ij} (5.2.22). Take the lowest case p = 1 where we have

$$\langle L_1 L_1 \bar{L}_1 \bar{L}_1 \rangle_{c^0} = \frac{1}{x_{12}^2 x_{34}^2} \tilde{H}_{1111}^{(0)} = \frac{1}{x_{14}^2 x_{23}^2} + \frac{1}{x_{13}^2 x_{24}^2}$$
 (5.3.14)

To construct the generating function, now lift this to four dimensions by replacing $x_{ij}^2 \rightarrow x_{ij}^2 + y_{ij}^2 = x_{ij}^2 (1 + g_{ij}^2)$ which gives

$$\left\langle LL\bar{L}\bar{L}\bar{L}\right\rangle_{c^{0}}^{4d} = \frac{1}{x_{14}^{2}x_{23}^{2}} \frac{1}{(1+g_{14}^{2})(1+g_{23}^{2})} + \frac{1}{x_{13}^{2}x_{24}^{2}} \frac{1}{(1+g_{13}^{2})(1+g_{24}^{2})} \quad (5.3.15)$$

This is a 4d object which contains all 1d free theory correlators. To obtain the specific correlators Taylor expand in g_{ij} and take the appropriate coefficients. To check this for the case of equal charges expand out $(1+g^2)^{-1} = 1 - g^2 + g^4 - g^6 + ...$ and keep the two terms proportional to $g_{14}^{2p-2}g_{23}^{2p-2}$ and $g_{13}^{2p-2}g_{24}^{2p-2}$, this indeed reproduces the prediction in (5.3.13).

Next, let us investigate correlators of unequal charge and check that they can be predicted from the generating function (5.3.15), which is the main result of this section. First, consider $\{p_i\} = pqpq$, where p > q. These correlators will be important in later sections, together with the equal charge ones, when performing conformal block analyses of correlators in the supergravity approximation and at the order of higher-derivative corrections. The free theory correlator is given by

$$\langle \mathcal{O}_p(x_1, y_1) \mathcal{O}_q(x_2, y_2) \mathcal{O}_p(x_3, y_3) \mathcal{O}_q(x_4, y_4) \rangle_{c^0} = -\frac{1}{\mathcal{N}_{pqpq}^{(0)}} \tilde{g}_{13}^{2p-1} \tilde{g}_{24}^{2q-1} ,$$
 (5.3.16)

and
$$G_{pqpq}^{(0)}(x,y) = -\frac{1}{\mathcal{N}_{pqpq}^{(0)}} \left(\frac{x}{y}\right)^{p+q-1}$$
, (5.3.17)

where the operators have dimensions and charges (2p-1)/2, (2q-1)/2 and we

decompose $G_{pqpq}^{(0)}$ into

$$k_{pqpq}^{(0)}(x,y) = -\frac{1}{\mathcal{N}_{pqpq}^{(0)}} \left(\frac{x}{y}\right)^{q-p} , \quad H_{pqpq}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pqpq}^{(0)}} \frac{x\,y}{x-y} \left(\frac{x}{y}\right)^{q-p} \left[1 - \left(\frac{x}{y}\right)^{2p-1}\right].$$
(5.3.18)

Acting on this with $\Delta^{(2)}$ gives

$$\tilde{H}_{pqpq}^{(0)}(x,y) = (-1)^{p+q} \frac{x^{p+q}}{y^{p+q-2}} , \qquad (5.3.19)$$

which does look like free theory with dimensions p, q and charges p-1, q-1 as we would expect for descendants. Using (5.3.6) yields

$$\langle L_p L_q \bar{L}_p \bar{L}_q \rangle_{c^0} = (-1)^{p+q} g_{13}^{2p-2} g_{24}^{2q-2} \frac{1}{x_{13}^2 x_{24}^2} , \qquad (5.3.20)$$

which agrees with the term proportional to $g_{13}^{2p-2}g_{24}^{2q-2}$ in the expansion of the generating function (5.3.15). Thus the choice of signs in our normalisation of the half-BPS operators (5.2.17) was inspired by the realisation of the 4d conformal symmetry in free theory, similar to the rest of the normalisation.

Consider one more correlator of mixed charges $\{p_i\} = pqqp$, where p > q and the free theory correlator is given by

$$\langle \mathcal{O}_p(x_1, y_1) \mathcal{O}_q(x_2, y_2) \mathcal{O}_q(x_3, y_3) \mathcal{O}_p(x_4, y_4) \rangle_{c^0} = \frac{1}{\mathcal{N}_{pqqp}^{(0)}} \tilde{g}_{14}^{2p-1} \tilde{g}_{23}^{2q-1}$$

and $G_{pqqp}^{(0)}(x, y) = \frac{1}{\mathcal{N}_{pqqp}^{(0)}} \left(\frac{x}{y}\right)^{p+q-1} \left(\frac{1-y}{1-x}\right)^{2q-1} .$ (5.3.21)

The operators again have dimensions and charges (2p-1)/2, (2q-1)/2 as expected. The correlator decomposes into

$$k_{pqqp}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pqqp}^{(0)}} \left(\frac{x}{y}\right)^{p-q} ,$$

$$H_{pqqp}^{(0)}(x,y) = \frac{1}{\mathcal{N}_{pqqp}^{(0)}} \frac{x\,y}{x-y} \left[\left(\frac{1-y}{1-x}\right)^{2q-1} \left(\frac{x}{y}\right)^{p+q-1} - \left(\frac{x}{y}\right)^{p-q} \right] .$$
(5.3.22)

The action of $\Delta^{(2)}$ then yields

$$\tilde{H}_{pqqp}^{(0)}(x,y) = (-1)^{p+q} \frac{x^{p+q}}{y^{p+q-2}} \frac{(1-y)^{2q-2}}{(1-x)^{2q}} , \qquad (5.3.23)$$
which as expected looks like free theory of descendants with dimensions p, q and charges p-1, q-1. Using (5.3.6) we finally get the correlator of descendants

$$\langle L_p L_q \bar{L}_q \bar{L}_p \rangle_{c^0} = (-1)^{p+q} g_{14}^{2p-2} g_{23}^{2q-2} \frac{1}{x_{14}^2 x_{23}^2} ,$$
 (5.3.24)

which agrees with the term proportional to $g_{14}^{2p-2}g_{23}^{2q-2}$ in the expansion of (5.3.15).

In this section we have shown that free theory correlators in 1d dual to quantum gravity in $AdS_2 \times S^2$ have four-dimensional conformal symmetry. The generating function can be constructed by uplifting the correlator of descendants with lowest charge to four dimensions. In the following section we will study correlators in the supergravity limit in the context of the four-dimensional conformal symmetry as well as the 4d scalar effective action described in subsection 5.2.3.

5.4 Supergravity

As explained in subsection 5.1.1, for supergravity the object which plays the leading role in the four-dimensional conformal symmetry is the half-BPS correlator itself. We will study correlators in the supergravity approximation in this section. To construct a 4d generating function of all the 1d half-BPS correlators dual to treelevel supergravity from the 4d conformal symmetry, we first need to determine the lowest-charge supergravity correlator. This will be derived in subsection 5.4.1 using only crossing symmetry and $x \to 0$ behaviour. In subsection 5.4.2 we will then discuss the 4d conformal symmetry of supergravity correlators. Furthermore, in subsection 5.4.3 we will obtain all half-BPS supergravity correlators from the 4d effective action (5.2.27) by evaluating generalised $AdS_2 \times S^2$ Witten diagrams analogous to chapter 3.

5.4.1 Lowest-Charge Supergravity Correlator

In this subsection we determine the supergravity correlator for $p_i = 1$ using crossing symmetry and the $x \to 0$ behaviour. The free disconnected correlator of operators with equal charges $p_i = 1$ is given in terms of \tilde{g}_{ij} as

$$\langle \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_4, y_4) \rangle |_{c^0} = \tilde{g}_{12} \tilde{g}_{34} - \tilde{g}_{13} \tilde{g}_{24} + \tilde{g}_{14} \tilde{g}_{23} .$$
(5.4.1)

At next order in the large-c expansion, which corresponds to supergravity, we would expect an expression of the form

$$\langle \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_4, y_4) \rangle |_{1/c} = \tilde{g}_{12} \tilde{g}_{34} \frac{x}{y} (x - y) a(x) ,$$

however this does not turn out to be the most natural definition, which can be seen by considering the crossed version. Indeed, if we exchange positions 1 and 3, which takes $x \to 1 - x$ and $y \to 1 - y$, we get

$$\begin{aligned} \langle \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_4, y_4) \rangle |_{1/c} &= -\tilde{g}_{32} \tilde{g}_{14} \frac{1-x}{1-y} (x-y) a (1-x) \\ &= \tilde{g}_{12} \tilde{g}_{34} \frac{x}{y} (x-y) a (1-x) . \end{aligned}$$

Setting this equal to minus the original expression (by Fermi statistics) then implies the crossing equation

$$a(1-x) = -a(x) \operatorname{sgn} [x(1-x)] .$$
 (5.4.2)

For 0 < x < 1, this reduces to a(1-x) = -a(x). We expect the supergravity correlator should be given in terms of *D*-functions, or \overline{D} -functions to get concrete expressions in position space. Note that since there is only one conformal spacetime cross-ratio xin 1d, one has to consider holomorphic \overline{D} -functions, which we denote by $\overline{D}^{\text{hol}}$. These functions can be reached by taking the limit $\overline{z} \to z$ of the usual \overline{D} -functions in terms of z, \overline{z} (2.3.17) and setting z = x, see (5.2.3). However, the crossing condition (5.4.2) is not consistent with the behaviour of $\overline{D}^{\text{hol}}$ -functions under crossing.

This can be resolved by constructing the four-point function using bosonic two-point

functions g_{ij} (5.2.22)², which yields

$$\langle \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_4, y_4) \rangle |_{1/c} = g_{12} g_{34} \frac{x}{y} (x - y) a(x) .$$
 (5.4.3)

In this case, from $1 \leftrightarrow 3$, we find the crossing constraint

$$a(1-x) = a(x) . (5.4.4)$$

We shall take (5.4.3) as our ansatz. Before considering crossing versions of the fourpoint function, it is important to carefully consider some properties of four-point functions in one dimension. As mentioned in subsection 5.2.1, conformal symmetry allows us to fix three points on the line, i.e. we can set $x_1 = 0$, $x_3 = 1$ and $x_4 = \infty$ [152]. The position of the third point x_2 is then equal to x and ranges over all real numbers. Let us refer to the interacting part of the four-point function, $x^2a(x)$, by $\mathcal{A}(x)$ for the purpose of this discussion. The function $\mathcal{A}(x)$ is singular for values of x where two points coincide, i.e. $x = 0, 1, \infty$. Thus, $\mathcal{A}(x)$ splits into three different functions for different regions of the real x line:

$$\mathcal{A}(x) = \begin{cases} \mathcal{A}^{-}(x) \text{ for } x \in (-\infty, 0) ,\\ \mathcal{A}^{0}(x) \text{ for } x \in (0, 1) ,\\ \mathcal{A}^{+}(x) \text{ for } x \in (1, \infty) , \end{cases}$$
(5.4.5)

see also [153] for similar discussions. The functions are related to each other by the Fermi symmetry of the four-point function, where we have $\mathcal{A}^{0,\pm}(x) = x^2 a^{0,\pm}(x)$. As mentioned above, we choose the ordering 0 < x < 1 where \mathcal{A}^0 is the relevant function and thus in all the above equations we replace a(x) by $a^0(x)$. When considering crossing symmetry however, the other regions become relevant.

Let us now consider exchanging 2 with 3, which takes $x \to 1/x$ and $y \to 1/y$:

$$\langle \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_4, y_4) \rangle |_{1/c} = g_{12} g_{34} \frac{x}{y} (x - y) \left(-\frac{1}{x^2} a^+ (1/x) \right) .$$

$$(5.4.6)$$

 $^{^{2}}$ Although we are using bosonic two-point functions, the four-point function is still a valid solution to the Ward identities with the expected symmetries.

Equating this with minus the original expression (due to Fermi statistics) then implies

$$a^{+}\left(\frac{1}{x}\right) = x^{2}a^{0}(x)$$
 (5.4.7)

Finally, for the third crossing condition, consider exchanging 1 with 3 in (5.4.6):

$$\langle \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_4, y_4) \rangle |_{1/c} = g_{12}g_{34}\frac{x}{y}(x-y)\frac{1}{(1-x)^2}a^+\left(\frac{1}{1-x}\right)$$

Equating this with the expression in (5.4.3) then gives the condition

$$a^{+}\left(\frac{1}{1-x}\right) = (1-x)^{2}a^{0}(x) , \qquad (5.4.8)$$

which follows from (5.4.4) and (5.4.7).

An additional constraint comes from the fact that the subleading part of the correlator in the large-c expansion should not encode exchange of the identity operator. To use this conditions, consider the conformal block expansion of the correlator

$$\langle \mathcal{O}_{1}(x_{1}, y_{1}) \mathcal{O}_{1}(x_{2}, y_{2}) \mathcal{O}_{1}(x_{3}, y_{3}) \mathcal{O}_{1}(x_{4}, y_{4}) \rangle$$

= $\sum_{\Delta=1}^{\infty} A_{\Delta,0}^{1111} g_{12} g_{34} B_{\Delta,0}^{\log}(x, y)$
~ $\sum_{\Delta=1}^{\infty} A_{\Delta,0}^{1111} g_{12} g_{34} (-x)^{\Delta} (1 + \mathcal{O}(x))$ (5.4.9)

and compare the approximation in the last line to the ansatz (5.4.3) considering the $x \to 0$ limit. This implies that $a^0(x)$ must satisfy the additional constraint

$$a^0(x) = \mathcal{O}(1)$$
 . (5.4.10)

With these constraints at our hand, assume that $a^0(x)$, $a^+(x)$ take the form

$$a^{0}(x) = p^{0}(x) \log x^{2} + p^{0}(1-x)\log(1-x)^{2} + r^{0}(x) \quad \text{for } x \in (0,1) ,$$

$$a^{+}(x) = p^{+}(x)\log x^{2} + p^{+}(1-x)\log(1-x)^{2} + r^{+}(x) \quad \text{for } x \in (1,\infty) , \quad (5.4.11)$$

for rational functions p^0, p^+, r^0, r^+ . Note that for $x \in (0, 1)$ and $x \in (1, \infty)$ for $a^0(x)$ and $a^+(x)$ respectively the logarithms in the above ansatz will not pose a problem when considering crossing symmetry. Furthermore, this ansatz makes sense since we expect the correlator to be written in terms of $\overline{D}^{\text{hol}}$ -functions, and these functions indeed only consist of $\log(x)$, $\log(1-x)$ and rational functions of x. Plugging this into the crossing equations (5.4.4) and (5.4.7) and the constraint (5.4.10), we get the following crossing equations for p^0, p^+ and r^0, r^+ :

$$p^{+}(1/x) + p^{+}((x-1)/x) = -x^{2}p^{0}(x) , \quad p^{+}((x-1)/x) = x^{2}p^{0}(1-x) ,$$

$$p^{+}(x/(x-1)) = (1-x)^{2}p^{0}(x) , \quad p^{+}(1/(1-x)) + p^{+}(x/(x-1)) = -(1-x)^{2}p^{0}(1-x) ,$$

$$r^{0}(1-x) = r^{0}(x) , \quad r^{+}(1/x) = x^{2}r^{0}(x) , \quad r^{+}(1/(1-x)) = (1-x)^{2}r^{0}(x) . \quad (5.4.12)$$

We then find the minimal solution:

$$a^{0}(x) = \frac{\log x^{2}}{1-x} + \frac{\log(1-x)^{2}}{x} = -\bar{D}_{1111}^{\text{hol}}(x) \quad \text{for } x \in (0,1) , \qquad (5.4.13)$$

where $\overline{D}_{1111}^{\text{hol}}(x)$ is the holomorphic box function. Note that (5.4.12) yields that $a^0(x)$ and $a^+(x)$ have the same functional form:

$$a^+(x) = -\bar{D}_{1111}^{\text{hol}}(x) \quad \text{for } x \in (1,\infty)$$
. (5.4.14)

Using (5.3.3) we write the interacting theory at tree-level as

$$H_{1111}^{\text{sugra}} = -u\bar{D}_{1111}^{\text{hol}} . \tag{5.4.15}$$

We have now determined the lowest-charge supergravity correlator from which we will obtain all higher-charge correlators from the 4d conformal symmetry in the following subsection.

5.4.2 4d Conformal Symmetry

As was found in [41] for $\mathcal{N} = 4$ SYM, the higher-dimensional conformal symmetry allows us to obtain the whole infinite tower of half-BPS correlators of all charges described by tree-level supergravity from the lowest correlator alone. The starting point is H_{1111}^{sugra} given in (5.4.15) which is lifted to four dimensions by simply lifting the 1d cross-ratios and holomorphic \bar{D} -functions to the usual cross-ratios and \bar{D} - functions, using the relations (5.2.3) for the cross-ratios. To get an object that transforms as a 4d conformal correlator, divide by $x_{12}^2 x_{34}^2$ and then replace $x_{ij}^2 \rightarrow x_{ij}^2 (1 + g_{ij}^2)$ to get the generating function:

$$\frac{H_{1111}^{\text{sugra}}}{x_{12}^2 x_{34}^2} |_{4d} = \frac{-u\bar{D}_{1111}(u,v)}{x_{12}^2 x_{34}^2} |_{4d} \to \frac{-u_{4d}\bar{D}_{1111}(u_{4d},v_{4d})}{x_{12}^2 x_{34}^2} \frac{1}{(1+g_{12}^2)(1+g_{34}^2)} , \quad (5.4.16)$$

where

$$u_{4d} = \frac{x_{12}^2 x_{34}^2}{x_{13}^2 x_{24}^2} \frac{(1+g_{12}^2)(1+g_{34}^2)}{(1+g_{13}^2)(1+g_{24}^2)} , \quad v_{4d} = \frac{x_{14}^2 x_{23}^2}{x_{13}^2 x_{24}^2} \frac{(1+g_{14}^2)(1+g_{23}^2)}{(1+g_{13}^2)(1+g_{24}^2)} . \tag{5.4.17}$$

This 4d object includes all higher-charge 1d supergravity correlators. To obtain the specific correlators, expand in small g_{ij} and take the coefficients of the appropriate powers of g_{ij}^2 . Before deriving a general formula describing all correlators, let us consider the explicit expansion for a few examples. The interacting piece of the correlators corresponds to the piece proportional to $\frac{x-y}{xy}$ and for the first few cases with charges $\{p_i\} = ppqq$ this looks like:

$$\begin{split} \langle \mathcal{O}_{1}\mathcal{O}_{2}\mathcal{O}_{2}\rangle_{1/c}|_{\frac{x-y}{xy}} = g_{12}g_{34} \times g_{34}^{2} \left(u\bar{D}_{1122}\right) , \\ \langle \mathcal{O}_{1}\mathcal{O}_{1}\mathcal{O}_{3}\mathcal{O}_{3}\rangle_{1/c}|_{\frac{x-y}{xy}} = g_{12}g_{34} \times \frac{1}{2}g_{34}^{4} \left(-u\bar{D}_{1133}\right) , \\ \langle \mathcal{O}_{2}\mathcal{O}_{2}\mathcal{O}_{2}\mathcal{O}_{2}\rangle_{1/c}|_{\frac{x-y}{xy}} = g_{12}g_{34} \times \left(ug_{12}^{2}g_{34}^{2} + g_{13}^{2}g_{24}^{2} + vg_{14}^{2}g_{23}^{2}\right) \left(-u\bar{D}_{2222}\right) , \\ \langle \mathcal{O}_{2}\mathcal{O}_{2}\mathcal{O}_{3}\mathcal{O}_{3}\rangle_{1/c}|_{\frac{x-y}{xy}} = g_{12}g_{34} \times \left(\frac{1}{2}ug_{12}^{2}g_{34}^{4} + g_{13}^{2}g_{24}^{2}g_{34}^{2} + vg_{14}^{2}g_{23}^{2}g_{34}^{2}\right) \left(u\bar{D}_{2233}\right) , \\ \langle \mathcal{O}_{3}\mathcal{O}_{3}\mathcal{O}_{3}\mathcal{O}_{3}\rangle_{1/c}|_{\frac{x-y}{xy}} = g_{12}g_{34} \times \left(\frac{1}{4}u^{2}g_{12}^{4}g_{34}^{4} + \frac{1}{4}g_{13}^{4}g_{24}^{4} + \frac{1}{4}v^{2}g_{14}^{4}g_{23}^{4} + ug_{12}^{2}g_{34}^{2}g_{12}^{2}g_{24}^{2} \\ + uvg_{12}^{2}g_{34}^{2}g_{14}^{2}g_{23}^{2} + vg_{13}^{2}g_{24}^{2}g_{14}^{2}g_{23}^{2}\right) \left(-u\bar{D}_{3333}\right) . \quad (5.4.18) \end{split}$$

To get the one-dimensional correlators, go back to 1d by taking the holomorphic limit of the \overline{D} -functions and considering the 1d cross-ratios (5.2.12). The supergravity correlators can be conveniently written in terms of the decomposition in (5.3.3) using (5.2.22) in 1d:

$$H_{1122}^{\text{sugra}} = u\bar{D}_{1122}^{\text{hol}} , \quad H_{1133}^{\text{sugra}} = -\frac{1}{2}u\bar{D}_{1133}^{\text{hol}} , \quad H_{2222}^{\text{sugra}} = -2u^2\frac{1-y+y^2}{y^2}\bar{D}_{2222}^{\text{hol}} , \\ H_{2233}^{\text{sugra}} = \frac{1}{2}u^2\frac{4-4y+3y^2}{y^2}\bar{D}_{2233}^{\text{hol}} , \quad H_{3333}^{\text{sugra}} = -\frac{3}{2}u^3\frac{(1-y+y^2)^2}{y^4}\bar{D}_{3333}^{\text{hol}} .$$
(5.4.19)

Note that the supergravity correlators with general charges show a simple structure and are proportional to a single \bar{D} -function with the same indices: $H_{p_1p_2p_3p_4}^{\text{sugra}} \propto \bar{D}_{p_1p_2p_3p_4}^{\text{hol}}$ which will become obvious in the general considerations below.

The expansion in g_{ij}^2 can be rewritten in terms of differential operators acting on D-functions. Therefore, to find a general expression for all higher-charge correlators let us formulate everything in terms of D-functions. These can be easily converted to \overline{D} -functions when needed to perform explicit calculations in position space using (2.3.17). Note that here we use a d-independent function defined in terms of the normalised D-function in (2.3.16) as $D'_{\{p_i\}} = (-2)^{-\Sigma_p} D_{\{p_i\}}^3$. Again, start from the 4d uplift (5.4.16) and notice that it exactly corresponds to D'_{1111} :

$$\frac{H_{1111}^{\text{sugra}}}{x_{12}^2 x_{34}^2} \Big|_{4d} = \frac{-u\bar{D}_{1111}}{x_{12}^2 x_{34}^2} \Big|_{4d} = -D'_{1111} \Big|_{4d} .$$
(5.4.20)

To get the coefficient of $(g_{ij}^2)^n$ in the g_{ij}^2 expansion of the uplifted correlator with $(u, v) \rightarrow (u_{4d}, v_{4d})$, instead of performing the expansion we take

$$\frac{\left(x_{ij}^2\right)^n}{n!} \frac{d^n}{d\left(x_{ij}^2\right)^n} D'_{\Delta_1 \Delta_2 \Delta_3 \Delta_4} , \qquad (5.4.21)$$

where (see [60])

$$\frac{d}{dx_{12}^2} D'_{\Delta_1 \Delta_2 \Delta_3 \Delta_4} = -D'_{\Delta_1 + 1 \Delta_2 + 1 \Delta_3 \Delta_4} .$$
 (5.4.22)

Thus the coefficient of $(g_{12}^2)^n$ is

$$\frac{(x_{12}^2)^n}{n!} \frac{d^n}{d(x_{12}^2)^n} D'_{\Delta_1 \Delta_2 \Delta_3 \Delta_4} = \frac{(-x_{12}^2)^n}{n!} D'_{\Delta_1 + n \Delta_2 + n \Delta_3 \Delta_4}$$
(5.4.23)

and similar for all x_{ij}^2 . Therefore, a general coefficient of $\prod_{i < j}^4 (g_{ij}^2)^{d_{ij}}$ is given by $\prod_{i < j} \frac{(-x_{ij}^2)^{d_{ij}}}{d_{ij}!} D'_{\{\Delta_i + \sum_{i < j} d_{ij}\}}$, where $0 \le d_{ij} = d_{ji}$, $d_{ii} = 0$.

³The factor $(-2)^{\Sigma_p}$ comes from the fact that in chapter 3 we usually factor out (-2) from $(-2X_i X_j) = x_{ij}^2$ whereas here we work in terms of x_{ij}^2 instead of $(X_i X_j)$ and similar for the spherical coordinates.

Hence a general correlator with charges p_i is given as

$$\frac{H_{p_1p_2p_3p_4}^{\text{sugra}}}{x_{12}^2 x_{34}^2} |_{4d} = \sum_{\{d_{ij}\}} \prod_{i < j} \left((-1)^{d_{ij}} \left(g_{ij}^2 \right)^{d_{ij}} \frac{\left(x_{ij}^2 \right)^{d_{ij}}}{d_{ij}!} \right) (-D'_{p_1p_2p_3p_4}) \\
= \sum_{\{d_{ij}\}} \prod_{i < j} \left(\frac{\left(y_{ij}^2 \right)^{d_{ij}}}{d_{ij}!} \right) (-1)^{\Sigma_p + 1} D'_{p_1p_2p_3p_4} ,$$
(5.4.24)

with

$$\sum_{i < j} d_{ij} = p_i - 1, \ 0 \le d_{ij} = d_{ji}, \ d_{ii} = 0 \ .$$
(5.4.25)

We recognise the factor depending on the internal coordinates y_{ij} as the analogue of the Mellin transform on the sphere we introduced in chapter 3 as *B*-functions. Note that the relevant object here is $B'_{\{p_i\}} = (-2)^{\sum_p} B_{\{p_i\}}$, given in terms of the *d*-independent *B*-function defined in (3.1.31) (see footnote 3). The d_{ij} correspond to the sphere-analogues of Mellin variables and $y_{ij}^2 = (-2Y_i \cdot Y_j)$. Equation (5.4.24) is a four-dimensional object which generates all 1d half-BPS correlators described by tree-level supergravity. To obtain general correlators in terms of *D*-functions it is useful to define the interacting piece of the correlator as

$$\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle_{\text{int}} = \frac{\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle|_{\frac{x-y}{xy}}}{x_{12} x_{34} y_{12} y_{34}} ,$$
 (5.4.26)

where we divide by $x_{12}x_{34}y_{12}y_{34}$ which is the 1d analogue of the Intriligator polynomial in 4d defined in appendix B. Note that the interacting correlator in (5.4.26) agrees with the definition of the interacting correlator in (3.1.14). The lowest-charge interacting correlator is then simply

$$\langle \mathcal{O}_1 \mathcal{O}_1 \mathcal{O}_1 \mathcal{O}_1 \rangle_{\text{int}} = -D'_{1111} = \frac{-uD_{1111}}{x_{12}^2 x_{34}^2}$$
 (5.4.27)

Finally, a general supergravity correlator is given by:

$$\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle_{\text{int}} = (-1)^{\sum_{p+1}} D'_{p_1 p_2 p_3 p_4} \times B'_{p_1 - 1 p_2 - 1 p_3 - 1 p_4 - 1} .$$
 (5.4.28)

To obtain explicit correlators in terms of conformal cross-ratios, rewrite the correlators in terms of \overline{D} -functions and take the holomorphic limit. In the next subsection we derive the same correlators from a 4d scalar effective action and compare the two results.

That all supergravity correlators can be obtained from the lowest-charge correlator alone is remarkable and shows that the higher-dimensional conformal symmetry is very powerful. Starting from just the box integral in the 1d case we can deduce the whole tower of spherical harmonics by acting with simple differential operators on it.

Decomposition into 4d conformal blocks

The four-dimensional conformal symmetry of the supergravity correlators suggests an expansion in 4d conformal blocks. This expansion will show that the lowest-charge supergravity correlator lifted to 4d (by using the usual \overline{D} -functions and cross-ratios, not the holomorphic ones) corresponds to a single 4d conformal block. This result is expected from the higher-dimensional conformal symmetry, there should be only a single block contributing for each spin l. Furthermore, 2d kinematics implies that only spin-0 blocks contribute and thus the 4d uplift of the lowest-charge supergravity correlator corresponds to a single 4d spin-0 block, which will be explained below.

The decomposition into 4d conformal blocks $G_{\Delta,l}(u, v)$ can be written as [154, 155],

$$\langle \mathcal{O}_1(x_1, y_1) \mathcal{O}_1(x_2, y_2) \mathcal{O}_1(x_3, y_3) \mathcal{O}_1(x_4, y_4) \rangle_{4d} = \frac{1}{x_{12}^2 x_{34}^2} \sum_{\Delta, l} A_{\Delta, l} G_{\Delta, l}(u, v) , \quad (5.4.29)$$

with

$$G_{\Delta,l}(z,\bar{z}) = \frac{1}{z-\bar{z}} u^{\frac{1}{2}(\Delta-l)} \left(\left(-\frac{1}{2}z \right)^{l} z_{2}F_{1}\left(\frac{1}{2}(\Delta+l), \frac{1}{2}(\Delta+l); \Delta+l; z \right) \right. \\ \left. \times {}_{2}F_{1}\left(\frac{1}{2}(\Delta-l-2), \frac{1}{2}(\Delta-l-2); \Delta-l-2; \bar{z} \right) - \left(z \leftrightarrow \bar{z} \right) \right).$$
(5.4.30)

First, expand the correlator at $\mathcal{O}(c^0)$, which is given by the 4d lift of \tilde{H}_{1111} in (5.3.12). The leading scaling dimension is $\Delta_0 = 2n + l + 2$, with the labels n = 0, 1, ... and spin l = 0, 2, ... The conformal block expansion at leading order is then given by

$$u + \frac{u}{v} = \sum_{n,l} A_{\Delta_0,l}^{(0)} G_{\Delta_0,l}(u,v) .$$
 (5.4.31)

This gives the free OPE coefficients

$$A_{2,l}^{(0)} = \frac{2^{1+l}(l!)^2}{(2\,l)!} , \qquad (5.4.32)$$

which only has $\Delta_0 = 2$ contributions. To decompose the 4d lift of the correlator at $\mathcal{O}(1/c)$ given in (5.4.15) we first expand the OPE data, as well as the decomposition in (5.4.29) to order 1/c:

$$\Delta = 2n + l + 2 + \frac{1}{c}\gamma_{\Delta_0,l} + \dots , \quad A_{\Delta_0,l} = A^{(0)}_{\Delta_0,l} + \frac{1}{c}A^{(1)}_{\Delta_0,l} + \dots , \qquad (5.4.33)$$

the block expansion up to $\mathcal{O}(1/c)$ is then

$$\langle \mathcal{O}_{1}(x_{1}, y_{1}) \mathcal{O}_{1}(x_{2}, y_{2}) \mathcal{O}_{1}(x_{3}, y_{3}) \mathcal{O}_{1}(x_{4}, y_{4}) \rangle_{4d} = \frac{1}{x_{12}^{2} x_{34}^{2}} \sum_{n,l} \left[A_{\Delta_{0},l}^{(1)} G_{\Delta_{0},l}(u, v) + \frac{1}{c} A_{\Delta_{0},l}^{(0)} \gamma_{\Delta_{0},l} \frac{1}{2} \frac{\partial}{\partial n} G_{\Delta_{0},l}(u, v) + \dots \right] .$$
 (5.4.34)

We wish to extract the anomalous dimensions γ , and this can be done by noticing that $\partial_n G_{\Delta_0,l}$ gives an expression of the form $(\log u G_{\Delta_0,l} + \text{ non-log terms})$. The correlator which is given in terms of \overline{D} -functions also has a contribution proportional to $\log u$. Therefore, we can isolate the relevant contributions to solve for the $\gamma_{\Delta_0,l}$ by taking the pieces proportional to $\log u$ and this yields

$$-u\bar{D}_{1111}|_{\log u} = \sum_{\Delta_0,l} A^{(0)}_{\Delta_0,l} \gamma_{\Delta_0,l} G_{\Delta_0,l}(u,v) . \qquad (5.4.35)$$

It turns out that the log-piece of the supergravity correlator indeed only has a single spin-0 block contribution with $\Delta_0 = 2$. The anomalous dimension is

$$\gamma_{2,0} = 1$$
, (5.4.36)

where we divided by $A_{2,0}^{(0)}$. This can also be seen from 4d supergravity scattering in flat space, analogously to the discussion in section 5.1 where the supergravity anomalous dimension agrees with the conformal partial wave coefficient of the flat space scattering amplitude. In general dimensions the expectation is that the anomalous dimension is ~ $\frac{1}{(l+1)_{(d-4)}}$ which for 4d reduces to a constant agreeing with (5.4.36). We will adapt the discussion in section 5.1 to 4d in the following.

Physical interpretation

The 4d supergravity amplitude in flat space is given by

$$A_4 = G_N \delta^4(Q) \to G_N s , \qquad (5.4.37)$$

where we have taken the scalar component. Note that $G_N \delta^4(Q)$ is dimensionless in 4d, so this amplitude has 4d conformal symmetry as expected. The factor of *s* indicates two-derivative interactions, as one would expect for supergravity. In contrast to the 10d amplitude (5.1.4), the 4d one has no θ dependence which implies that only l = 0 contributes in the partial wave expansion (5.1.5). This is expected for 2d massless scattering. Hence, there is only one partial wave coefficient:

$$\mathcal{A}_0^4(s) \sim 1 + \frac{R^2}{c} s ,$$
 (5.4.38)

where the Newton constant in 4d $G_N \sim R^2/c$, with the AdS radius R and the central charge c, and the one is put in by hand. This explains why supergravity correlators correspond to the 4d spin-0 block as was confirmed in (5.4.36). Based on this argument we can make a prediction for the supergravity anomalous dimensions obtained from a conformal block analysis in 1d and solving the mixing problem. Analogous to the $\mathcal{N} = 4$ SYM case, we expect the anomalous dimensions to be of the form

$$e^{\frac{1}{c}\gamma_{1d}^{\text{sugra}}} \sim 1 + \frac{1}{c}\delta^{(2)}$$
, (5.4.39)

where we note that γ^{sugra} contributes at order 1/c and $\delta^{(2)}$ is the eigenvalue of the second-order differential operator $\Delta^{(2)}$ acting on the 1d blocks. The eigenvalue $\delta^{(2)}$ encodes s and importantly there is no dependence on the spin l, which has to be the case because there is no spin in 1d CFTs. We will solve the mixing problem for anomalous dimensions of double-trace operators at $\mathcal{O}(1/c)$ and show that the result is consistent with this prediction in section 5.6. Before doing so, we will derive all half-BPS supergravity correlators from the 4d scalar effective action in the following subsection.

5.4.3 Effective Action

In general dimensions, $\operatorname{AdS}_q \times \operatorname{S}^q$, supergravity is not expected to be directly computed from the higher-dimensional scalar effective action, because it is not dual to contact interactions alone but rather is also described by exchange diagrams. But it turns out that in $\operatorname{AdS}_2 \times \operatorname{S}^2$ supergravity is described by the ϕ^4 interaction in the 4d effective action (5.2.27). This can be seen from the discussions above, that the 4d uplift of the lowest-charge supergravity correlator corresponds to the 4d spin-0 block (or rather the log *u*-piece of the correlator does). Hence, the conformal block expansion in 4d is truncated in spin and therefore supergravity can be described by contact diagrams alone, see also subsection 2.3.2. Thus, in $\operatorname{AdS}_2 \times \operatorname{S}^2$ we can deduce all correlators at $\mathcal{O}(1/c)$ from the effective field theory.

All four-point half-BPS correlators described by tree-level supergravity can be directly computed from the 4d effective action (5.2.27) by evaluating zero-derivative 4d Witten diagrams, which is very similar to the zero-derivative corrections at order α'^3 described in section 3.2. To obtain correlators from the 4d effective action we evaluate AdS₂×S² Witten diagrams, closely following the procedure explained in chapter 3. The supergravity term is the first term in the effective action (5.2.27) and is simply a ϕ^4 interaction:

$$S_{\text{sugra}} = \frac{1}{4!} A \times \int_{\text{AdS} \times \text{S}} d^2 \hat{X} d^2 \hat{Y} \phi(\hat{X}, \hat{Y})^4 , \qquad (5.4.40)$$

where we use embedding space formalism reviewed for AdS coordinates in subsection 2.3.1 and for spherical coordinates in 3.1.4. We obtain the corresponding CFT correlators by following the standard AdS/CFT procedure for computing correlators from AdS, but in a fully 4d covariant way, including the two-sphere manifestly. Using the generalised bulk-to-boundary propagators in (3.1.33) we obtain the AdS₂×S² Witten diagram for this ϕ^4 contact interaction, leading to the following proposal for the supergravity correlators: Chapter 5. $AdS_2 \times S^2$ Correlators: Effective Action and 4d Conformal 142 Symmetry

$$\langle \mathcal{OOOO} \rangle_{\text{int}}^{\text{sugra}} = \frac{1}{4!} A \frac{(\mathcal{C}_1)^4}{(-2)^4} \int_{\text{AdS} \times \text{S}} \frac{d^2 \hat{X} d^2 \hat{Y}}{(P_1 + Q_1)(P_2 + Q_2)(P_3 + Q_3)(P_4 + Q_4)}$$

= $\frac{1}{4!} A (\mathcal{C}_1)^4 \times D_{1111}^{\text{AdS}_2 \times \text{S}^2} .$ (5.4.41)

Recall that $P_i = \hat{X}.X_i$, $Q_i = \hat{Y}.Y_i$ and the AdS×S *D*-functions are defined in (3.1.37). The correlator (5.4.41) contains all supergravity half-BPS correlators of all spherical harmonics and to extract any specific correlator one expands in the appropriate powers in Y_i using the Taylor expansion of the 4d bulk-to-boundary propagators

$$(P_i + Q_i)^{-1} = \sum_{p=1}^{\infty} (-1)^{p-1} (P_i)^{-p} (Q_i)^{p-1} .$$
 (5.4.42)

The individual correlators are then given by 4:

$$\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle_{\text{int}}^{\text{sugra}} = \frac{1}{4!} A \frac{(\mathcal{C}_1)^4}{(-2)^4} \int_{\text{AdS}_2} d^2 \hat{X} \prod_i \frac{1}{(P_i)^{p_i}} \times \int_{\text{S}^2} d^2 \hat{Y} \prod_i (Q_i)^{p_i - 1} = \frac{1}{4!} A (\mathcal{C}_1)^4 D_{p_1 p_2 p_3 p_4}^{(1)} (X_i) \times B_{p_1 - 1 p_2 - 1 p_3 - 1 p_4 - 1}^{(1)} (Y_i) = A'' (-1)^{\Sigma_p + 1} D_{p_1 p_2 p_3 p_4} (X_i) \times B_{p_1 - 1 p_2 - 1 p_3 - 1 p_4 - 1} (Y_i) = A' (-1)^{\Sigma_p + 1} D'_{p_1 p_2 p_3 p_4} \times B'_{p_1 - 1 p_2 - 1 p_3 - 1 p_4 - 1} ,$$
 (5.4.43)

where going from the third to penultimate line we rewrote everything in terms of the normalised D- and B-functions defined in (2.3.16) and (3.1.31). Further, to compare to the results obtained from the 4d conformal symmetry in (5.4.28), rewrite D- and B-functions depending on $X_i.X_j, Y_i.Y_j$ as D'- and B'-functions in terms of x_{ij}^2, y_{ij}^2 which have factors of $(-2)^{\pm \Sigma_p}$ between them. We absorb all numerical factors into the coefficient A' which is unfixed and we set it to one in subsequent calculations, in agreement with the choice of normalisation in the previous subsection. In this case the log u piece of the lowest-charge correlator is exactly equal to the holomorphic limit of the 4d spin-0 block. The above expression includes all tree-level supergravity half-BPS correlators of any charge and it agrees with what we found from the

⁴This is (3.1.40) with $\Delta_i = d = 1$ and with $p_i \to p_i - 1$ to account for the fact that the lowest correlator is labelled with $p_i = 1$ rather than $p_i = 0$. We do not need to worry about the minus signs in the factors $(-1)^p$ in (5.4.42) since $B_{p_1p_2p_3p_4} = 0$ if $p_1 + p_2 + p_3 + p_4$ is odd.

4d conformal symmetry in (5.4.28), notably $\langle \mathcal{O}_{p_1} \mathcal{O}_{p_2} \mathcal{O}_{p_3} \mathcal{O}_{p_4} \rangle_{\text{int}}^{\text{sugra}} \propto D_{p_1 p_2 p_3 p_4}$. To obtain explicit expressions in position space convert D' to \overline{D} -functions and take the holomorphic limit.

5.5 Conformal Block Analysis and Double-Trace Spectrum

In the previous sections we have determined all half-BPS correlators for disconnected free theory and in the supergravity limit. Next, we wish to perform a conformal block analysis of these correlators in the large-c expansion and specifically compute the anomalous dimensions of double-trace operators in the spectrum. Before doing so in section 5.6, in this section, we will analyse the double-trace spectrum and discuss the operators that can be exchanged in the conformal block expansion (5.2.19) for each pair of scaling dimensions and SU(2) charges (Δ, p). We discuss the spectrum for free theory and supergravity, as well as for higher-derivative corrections.

A lot of progress has been made in the study of the conformal block analysis of four-point half-BPS correlators in the large-c expansion in the context of $\mathcal{N} = 4$ SYM [11, 14, 16, 84–91, 144, 145] and the analysis in this and following sections is inspired by these works. It is expected that short and long operators contribute to the OPE of the four-point functions of half-BPS operators \mathcal{O}_q and that the only unprotected operators which contribute at leading order in 1/c are double-trace operators which consist of a product of derivatives of two of the operators \mathcal{O}_q . All other multi-trace operators are expected to be subleading in 1/c. We discuss the double-trace operators which are present in the free theory and at leading order in 1/c. We will find that only specific operators are present in the spectrum in the supergravity limit, whereas more operators contribute at the order of free theory and higher-derivative corrections.

Let us start by considering all long double-trace operators in the OPE of two half-

BPS operators:

$$\mathcal{O}_{q_1q_2} = \mathcal{O}_{q_1}\partial^{\Delta+1-q_1-q_2}\mathcal{O}_{q_2} , \qquad q_1 \le q_2 , \qquad (5.5.1)$$

where the operators \mathcal{O}_{q_1} and \mathcal{O}_{q_2} have half-integer scaling dimensions and are labelled by integers $\Delta + 1/2$, the dimensions of the descendants, in terms of q_1 and q_2 according to the conventions introduced in (5.2.17) and used in the previous sections. We denote the scaling dimensions and SU(2) charges of exchanged operators $\mathcal{O}_{q_1q_2}$ by Δ and p respectively. Note that from now on throughout the rest of this chapter, when writing Δ and p we will always refer to the dimensions and charges of the exchanged double-trace operators (not the external ones).

The interacting piece of the correlator, corresponding to H in the decomposition (5.3.3), can be written as a double expansion in terms of large central charge cand small expansion parameter a, where a describes the higher-derivative corrections:

$$H(x,y) = H^{(0)} + \frac{1}{c} \left(H^{\text{sugra}} + a H^{4\text{-deriv}} + \dots \right) + \mathcal{O}(c^{-2}) \quad . \tag{5.5.2}$$

We wish to study the double-trace spectrum at different orders in c and a. From (5.5.1) together with considering the possible R-symmetry charges, one can predict which double-trace operators can in principle contribute to the spectrum at each weight (Δ, p) . Subsequently, by performing the conformal block analysis for free theory at $\mathcal{O}(1)$, supergravity at $\mathcal{O}(1/c)$ and for four-derivative corrections at $\mathcal{O}(a/c)$, we can determine which operators are really present in the double-trace spectrum at (Δ, p) , i.e. have non-zero conformal block coefficients at the order considered. It turns out that at $\mathcal{O}(1/c)$ only part of the operators contribute, while more operators contribute to free theory and to higher-derivative corrections. This will be explained by the fact that 1d correlators in the supergravity limit are truncated to 4d spin-0. Generally, there are many operators contributing at each weight (Δ, p) and to resolve this degeneracy we have to solve a so-called mixing problem. This will be studied in subsequent sections.

Before we describe the conformal block analysis in detail and solve the mixing problems for supergravity and four-derivative corrections in sections 5.6 and 5.8

respectively, we will first present the operators which are found to contribute to the double-trace spectrum. At the end of this section these will be interpreted in terms of an effective 4d spin similar to the one in 10d described in section 5.1. We notice that there are two classes of operators in the spectrum, we call them class A and class B. Both, class A and class B operators are present in the free theory, but only class A operators acquire anomalous dimensions and mix with other class A operators at order $\mathcal{O}(1/c)$, while class B operators decouple at $\mathcal{O}(1/c)$, i.e. they do not mix and do not acquire anomalous dimensions at the level of supergravity. Going to higherderivative corrections, particularly at four-derivatives which is the case studied in this chapter, both class A and class B operators acquire anomalous dimensions and mix. Note that class A and class B operators only mix with operators of the same class and class A operators never mix with class B operators. Additionally, it is important to note that there are different operators present for the cases where $t = \Delta - p$ odd or $t = \Delta - p$ even. At the level of supergravity there are only operators present when t is odd, these are the class A operators. In free theory and at the level of four-derivative corrections, there are operators mixing and acquiring anomalous dimensions for both odd and even t and we distinguish between them by labelling them with B^{to} and B^{te} respectively.

The fact that only specific operators acquire anomalous dimensions and mix at the order of supergravity can be explained from the point of view of the 4d conformal symmetry. Correlators in the supergravity limit correspond to the 4d spin-0 block only, which only allows for class A operators. This can be seen from the 4d effective spin we will conjecture in (5.5.6). Whereas for the 4d uplift of correlators corresponding to four-derivative corrections, there are contributions from spin-0 as well as spin-2 blocks and the decomposition of free theory correlators is not truncated in spin at all, thus allowing for more operators with non-zero anomalous dimensions, which includes all class B operators. Note that this is specific to 1d, because the correlators at $\mathcal{O}(1/c)$ correspond to a ϕ^4 interaction. In the $\mathcal{N} = 4$ SYM case, super-gravity correlators have no spin-truncated block decomposition and their spectrum

contains all possible double-trace operators.

We label the allowed values of pairs of q_1 and q_2 for class A and class B operators with odd or even t with (q_1^A, q_2^A) , $(q_1^{B^{to}}, q_2^{B^{to}})$ and $(q_1^{B^{te}}, q_2^{B^{te}})$ respectively. There are $d = d_A + d_{B^{to}} + d_{B^{te}}$ operators $\mathcal{O}_{q_1q_2}$ labelled by sets of pairs (q_1, q_2) , where d_A counts class A and $d_{B^{to}}$, $d_{B^{te}}$ class B operators with odd and even t respectively. The labels run over a set of operators $\mathcal{D}_{\Delta,p}$ and we parametrise this set by i_A , r_A , $i_{B^{to}}$, $r_{B^{te}}$, $r_{B^{te}}$, $r_{B^{te}}$ as follows:

$t = \Delta - p \text{ odd}$

For t odd we have both class A and class B operators:

$$q_{1}^{A} = 1 + i_{A} + r_{A} , \qquad q_{1}^{B^{\text{to}}} = 1 + i_{B^{\text{to}}} + r_{B^{\text{to}}} ,$$

$$q_{2}^{A} = 1 + i_{A} + p - r_{A} , \qquad q_{2}^{B^{\text{to}}} = 1 + i_{B^{\text{to}}} + (p - 1) - r_{B^{\text{to}}} ,$$

$$i_{A} = 0, \dots, \frac{t - 1}{2} , \qquad i_{B^{\text{to}}} = 1, \dots, \frac{t - 1}{2} ,$$

$$r_{A} = 0, \dots, \frac{\mu_{A} - 1}{2} , \qquad r_{B^{\text{to}}} = 0, \dots, \frac{\mu_{B^{\text{to}}} - 1}{2} , \qquad (5.5.3)$$

where

$$t = \Delta - p , \qquad \mu_A = \begin{cases} p+1 & p \text{ even} \\ p & p \text{ odd} \end{cases}, \qquad \mu_{B^{\text{to}}} = \begin{cases} p-1 & p \text{ even} \\ p & p \text{ odd} \end{cases}, \qquad (5.5.4)$$

with $d_A = \frac{1}{4}(\mu_A + 1)(t+1)$ and $d_{B^{to}} = \frac{1}{4}(\mu_{B^{to}} + 1)(t-1)$. Class A operators have even numbers of derivatives while class B operators have odd numbers of derivatives at t odd.

$t = \Delta - p$ even

There are only class B operators for t even, meaning these operators contribute to the free theory and higher-derivative corrections but not to supergravity. These operators are parametrised as

$$q_1^{B^{\text{te}}} = i_{B^{\text{te}}} + \left\lfloor \frac{r_{B^{\text{te}}} + 1}{2} \right\rfloor \,,$$

$$q_2^{B^{\text{te}}} = i_{B^{\text{te}}} + p - \left\lfloor \frac{r_{B^{\text{te}}}}{2} \right\rfloor,$$

$$i_{B^{\text{te}}} = 1, \dots, \frac{t}{2},$$

$$r_{B^{\text{te}}} = 0, \dots, p - 1,$$

(5.5.5)

with $d_{B^{\text{te}}} = \frac{1}{2} t p$.

We illustrate the exchanged operators at weight (Δ, p) for odd and even t in figure 5.1 and 5.2 respectively before we interpret them in terms of the effective 4d spin. In figure 5.1 the exchanged operators with odd t are illustrated in terms of the pairs (q_1, q_2) with the parametrisation (5.5.3). The black nodes denote class A operators while the white nodes denote class B^{to} operators. In the supergravity approximation, operators which are connected by vertical lines have the same anomalous dimensions after unmixing which are all zero at $\mathcal{O}(1/c)$, except for a single operator with non-zero anomalous dimension which is the operator denoted by A. When including higher-derivative corrections, the degeneracy is broken and focussing on four-derivatives the operators which acquire non-zero (and non-equal) anomalous dimensions after unmixing are highlighted by a grey rectangle. Denoted by E is the one class B operator which acquires non-zero anomalous dimension at $\mathcal{O}(a/c)$.

In figure 5.2 exchanged operators with even t are illustrated in terms of pairs (q_1, q_2) parametrised as in (5.5.5). For even t, operators mix and acquire anomalous dimensions only starting from the order of higher-derivative corrections, thus they all belong to class B. The operators are parametrised following (5.5.5) and it turns out that they split into two groups, operators with even $r_{B^{te}}$ are denoted by black nodes while operators with odd $r_{B^{te}}$ are denoted by white nodes. The two types of operators do not mix. There is one operator with even and one with odd $r_{B^{te}}$ which has non-zero anomalous dimension after unmixing at $\mathcal{O}(a/c)$ and they are highlighted by a grey rectangle.



Figure 5.1: The exchanged operators $\mathcal{O}_{q_1q_2}$ which contribute at (Δ, p) for t odd are illustrated in terms of the pairs (q_1^A, q_2^A) and $(q_1^{B^{\text{to}}}, q_2^{B^{\text{to}}})$. The operators are parametrised as described in (5.5.3). The black nodes denote operators of class A while white nodes denote operators of class B^{to} . The only operator acquiring non-zero anomalous dimension at the order of supergravity is the one denoted by A. Furthermore, the nodes in the grey rectangle correspond to operators which acquire non-zero anomalous dimensions at the order of four-derivative corrections.





4d effective spin

Similar to the 10d effective spin (5.1.3) which was justified from the similarity between (5.1.7) and (5.1.8), we conjecture a 4d effective spin in terms of 1d quantum numbers as follows

$$l_{4d} = 2\left(i_A + r_A + i_{B^{\text{to}}} + i_{B^{\text{te}}} + r_{B^{\text{te}}} - \frac{1 + (-1)^{r_{B^{\text{te}}} + 1}}{2}\right) .$$
 (5.5.6)

This means that at each order in a, this 4d spin will predict how many and which operators have non-zero anomalous dimensions after unmixing. As we know, the conformal block expansion of the 4d lift of 1d superconformal correlators in the supergravity limit is truncated to spin zero. Therefore, the only operators with nonzero anomalous dimensions are predicted to be those whose quantum numbers give $l_{4d} = 0$ and this can only be satisfied by class A operators. Hence, for supergravity the 4d spin simplifies significantly to

$$l_{4d}^{\text{sugra}} = 2\left(i_A + r_A\right) \ . \tag{5.5.7}$$

For higher-derivative corrections also class B operators are allowed since the dual bulk contact interactions correspond to spin-L corrections with $L \ge 2$. The four-derivative corrections whose conformal block analysis and unmixing we study in section 5.8 correspond to spin-2 corrections and thus we predict that operators whose quantum numbers satisfy $l_{4d} = 0$ and $l_{4d} = 2$ have non-zero anomalous dimensions at this order. We will solve the mixing problem for supergravity in the next section and for fourderivative corrections in section 5.8 and we will also interpret the results in terms of l_{4d} . First, let us start by describing how to solve the mixing problem at general orders in c and a in the following subsection.

5.5.1 Solving the Mixing Problem

In [144] the authors study the double-trace spectrum of $\mathcal{N} = 4$ SYM in the supergravity limit, where the double-trace operators exhibit degeneracy, as explained in the previous subsection for 1d. To solve the mixing problem we need to consider free theory and tree-level supergravity contributions to the correlators of four half-BPS operators. In the 4d case the degeneracy can be resolved for a large family of operators, and only a small residual degeneracy is left. In 1d this residual degeneracy is not obvious since the degenerate anomalous dimensions are all zero. We will find that there is only one non-zero anomalous dimension at each weight. It was later found in [14] that the residual degeneracy in 4d is lifted when considering higher-derivative contributions to the correlators and we will come to similar conclusions for 1d in section 5.8. In this subsection we will describe how to solve the mixing problem up to $\mathcal{O}(a/c)$ which corresponds to four-derivative corrections, this discussion can be easily extended to higher orders in a to include higher-derivative corrections.

To perform the conformal block analysis we need the free theory correlator which is the leading contribution to the large-c expansion and was described in section 5.3 and the first subleading contribution in the 1/c expansion, which we obtained both from the 4d conformal symmetry in subsection 5.4.2 and from a 4d scalar effective action in subsection 5.4.3. Even though we have not obtained the higher-derivative correlators yet, we will assume that we have them at our disposal and describe the unmixing at $\mathcal{O}(a/c)$. We will later derive all four-derivative corrections to all half-BPS correlators described by tree-level supergravity from the 4d scalar effective action in subsection 5.7.2.

We have seen above that there can be many double-trace operators with different quantum numbers i_A , r_A , $i_{B^{to}}$, $r_{B^{to}}$, $i_{B^{te}}$, $r_{B^{te}}$ contributing at the same weight (Δ, p) . To solve this mixing problem we perform the operator product expansion of $(\mathcal{O}_{p_1} \times \mathcal{O}_{p_2})$ and $(\mathcal{O}_{p_3} \times \mathcal{O}_{p_4})$, where the pairs (p_1, p_2) and (p_3, p_4) range over the same set $\mathcal{D}_{\Delta,p}$, described in (5.5.3) and (5.5.5). The conformal block expansion of the interacting piece of the correlator in terms of long blocks is

$$H(x,y) = \frac{xy}{x-y} \sum_{\Delta,p} A^{p_i}_{\Delta,p} B^{\text{long}}_{\Delta,p,p_{12},p_{34}}(x,y) , \qquad (5.5.8)$$

where the superconformal blocks are given in (5.2.23). The coefficients of the decom-

position are given as a sum of squares of OPE coefficients as follows

$$A^{p_i}_{\Delta,p} = \sum_{\mathcal{O}^{\Delta,p}} C_{p_1 p_2 \mathcal{O}} C_{p_3 p_4 \mathcal{O}} , \qquad (5.5.9)$$

where the sum goes over the degenerate operators because of operator mixing. Expanding the OPE-data to order 1/c we get

$$\Delta_{\mathcal{O}} = \Delta^{(0)} + \frac{1}{c} \left(\gamma^{\text{sugra}} + a \gamma^{4\text{-deriv}} + \dots \right) + \mathcal{O}(c^{-2}) ,$$

$$C_{pp\mathcal{O}} = C_{pp\mathcal{O}}^{(0)} + \left(a C_{pp\mathcal{O}}^{4\text{-deriv}} + \dots \right) + \mathcal{O}(c^{-1}) , \qquad (5.5.10)$$

where the anomalous dimensions γ depend on Δ , p and the degeneracy labels i and r. Plugging the expansion of the dimensions and OPE-coefficients back into (5.5.8) gives

$$H(x,y) = \frac{x y}{x - y} \sum_{\Delta^{(0)}, p} \left[A^{(0)}_{\Delta^{(0)}, p} B^{\log}_{\Delta^{(0)}, p, p_{12}, p_{34}}(x, y) + \frac{1}{c} \log u \sum_{\Delta^{(0)}, p} \left(M^{\text{sugra}}_{\Delta^{(0)}, p} + a M^{4\text{-deriv}}_{\Delta^{(0)}, p} + \dots \right) B^{\log}_{\Delta^{(0)}, p, p_{12}, p_{34}}(x, y) + \dots \right],$$
(5.5.11)

where the dots denote terms analytic in u which do not play a role for our analysis. We define the OPE-coefficients at orders $\mathcal{O}(c^0)$, $\mathcal{O}(1/c)$ and $\mathcal{O}(a/c)$ as follows (note that from now on we denote the classical scaling dimension $\Delta^{(0)}$ of the exchanged operators by Δ for simplicity):

$$A_{\Delta,p}^{(0)} = \sum_{\mathcal{O}^{\Delta,p}} C_{p_1 p_2 \mathcal{O}}^{(0)} C_{p_3 p_4 \mathcal{O}}^{(0)} , \qquad M_{\Delta,p}^{\text{sugra}} = \sum_{\mathcal{O}^{\Delta,p}} \gamma^{\text{sugra}} C_{p_1 p_2 \mathcal{O}}^{(0)} C_{p_3 p_4 \mathcal{O}}^{(0)} ,$$

$$M_{\Delta,p}^{4\text{-deriv}} = \sum_{\mathcal{O}^{\Delta,p}} \left(\gamma^{4\text{-deriv}} C_{p_1 p_2 \mathcal{O}}^{(0)} C_{p_3 p_4 \mathcal{O}}^{(0)} + \gamma^{\text{sugra}} C_{p_1 p_2 \mathcal{O}}^{(0)} C_{p_3 p_4 \mathcal{O}}^{4\text{-deriv}} + \gamma^{\text{sugra}} C_{p_1 p_2 \mathcal{O}}^{4\text{-deriv}} C_{p_1 p_2 \mathcal{O}}^{(0)} C_{p_3 p_4 \mathcal{O}}^{(0)} \right) .$$
(5.5.12)

The left hand side of equation (5.5.11) are the explicit forms of the correlators and comparing the equation to the double-expansion in (5.5.2), it can be seen that $A^{(0)}_{\Delta,p}$ is determined from the free theory contribution, $M^{\text{sugra}}_{\Delta,p}$ is determined from the correlators in the supergravity limit and $M^{4\text{-deriv}}_{\Delta,p}$ will be determined from the four-derivative corrections to the correlators. To decide which operators we have to consider in the conformal block analysis to solve the mixing problem, at each weight (Δ, p) let us arrange a $((d_A + d_{B^{to}}) \times (d_A + d_{B^{to}}))$ matrix of correlators running over $\mathcal{D}_{\Delta,p}$ for t odd and a $(d_{B^{te}} \times d_{B^{te}})$ matrix for t even respectively, where we use the parametrisation from earlier in this section. Note that the $((d_A + d_{B^{to}}) \times (d_A + d_{B^{to}}))$ matrix is block-diagonal, so class A and B operators can be treated independently, and similarly $(d_{B^{te}} \times d_{B^{te}})$ is block-diagonal where operators with $r_{B^{te}}$ even or odd can be treated independently. We then perform a conformal block analysis and arrange the coefficients into matrices $\hat{A}^{(0)}_{\Delta,p}$, $\hat{M}^{sugra}_{\Delta,p}$ and $\hat{M}^{4\text{-deriv}}_{\Delta,p}$ for free theory, supergravity and four-derivative corrections respectively. Comparing (5.5.11) to (5.5.2), keeping terms up to $\mathcal{O}(a/c)$ and writing (5.5.12) in matrix form then leads to the following unmixing equations:

$$\mathcal{O}(1): \quad \hat{A}_{\Delta,p}^{(0)} = \mathbb{C}^{(0)} \left(\mathbb{C}^{(0)}\right)^{T} ,$$

$$\mathcal{O}(1/c): \quad \hat{M}_{\Delta,p}^{\text{sugra}} = \mathbb{C}^{(0)} \hat{\gamma}^{\text{sugra}} \left(\mathbb{C}^{(0)}\right)^{T} ,$$

$$\mathcal{O}(a): \qquad 0 = \mathbb{C}^{(0)} \left(\mathbb{C}^{4\text{-deriv}}\right)^{T} + \mathbb{C}^{4\text{-deriv}} \left(\mathbb{C}^{(0)}\right)^{T} ,$$

$$\mathcal{O}(a/c): \quad \hat{M}_{\Delta,p}^{4\text{-deriv}} = \mathbb{C}^{(0)} \hat{\gamma}^{4\text{-deriv}} \left(\mathbb{C}^{(0)}\right)^{T} + \mathbb{C}^{(0)} \hat{\gamma}^{\text{sugra}} \left(\mathbb{C}^{4\text{-deriv}}\right)^{T} + \mathbb{C}^{4\text{-deriv}} \hat{\gamma}^{\text{sugra}} \left(\mathbb{C}^{(0)}\right)^{T} ,$$

$$(5.5.13)$$

where $\hat{\gamma}$ is a diagonal matrix of the anomalous dimensions. Note that $\hat{A}^{(0)}_{\Delta,p}$ is a diagonal matrix as can be easily seen from the form of the free theory correlators and $\hat{M}^{\text{sugra}}_{\Delta,p}$ and $\hat{M}^{4\text{-deriv}}_{\Delta,p}$ are symmetric matrices.

To solve the unmixing equations for the supergravity anomalous dimensions and the free three-point functions ($\mathcal{O}(1/c)$ and $\mathcal{O}(1)$) for general (Δ, p) it is useful to define the matrix \tilde{c}

$$\tilde{c}\tilde{c}^{T} = \mathrm{Id} , \quad \mathbb{C}^{(0)} = \left(\hat{A}^{(0)}\right)^{\frac{1}{2}} \cdot \tilde{c} .$$
 (5.5.14)

The unmixing equations then become:

$$\tilde{c} \cdot \hat{\gamma}^{\text{sugra}} \cdot \tilde{c}^T = \left(\hat{A}^{(0)}\right)^{-\frac{1}{2}} \cdot \hat{M}^{\text{sugra}} \cdot \left(\hat{A}^{(0)}\right)^{-\frac{1}{2}} .$$
(5.5.15)

The columns of \tilde{c} are eigenvectors of the matrix $(\hat{A}^{(0)})^{-\frac{1}{2}} \cdot \hat{M}^{\text{sugra}} \cdot (\hat{A}^{(0)})^{-\frac{1}{2}}$ and the

corresponding eigenvalues are the anomalous dimensions at $\mathcal{O}(1/c)$. So the mixing problem can be solved by solving the eigenvalue problem of the corresponding matrix. The anomalous dimensions after unmixing at the order of four-derivative corrections are then the $\mathcal{O}(a)$ eigenvalues of $(\hat{M}^{\text{sugra}} + a\hat{M}^{4\text{-deriv}}) \cdot (\hat{A}^{(0)})^{-1}$. We will illustrate this with examples below.

We now know how to solve the mixing problem in the supergravity limit and for four-derivative corrections. In the following section we focus on the supergravity unmixing and we will later perform the unmixing for four-derivative corrections in section 5.8.

5.6 Unmixing Supergravity

In this section we will solve the unmixing equations for supergravity at each weight (Δ, p) , where for every (Δ, p) we use (5.5.3) to determine the list of double-trace operators in the spectrum. Recall that only class A operators are in the double-trace spectrum of supergravity and thus from the 4d spin (5.5.6) we predict that there is only one non-zero anomalous dimension per weight, the one corresponding to $l_{4d} = 0$. Additionally, from the 4d conformal symmetry we predict that the value of the anomalous dimension should be the eigenvalue $\delta^{(2)}$ of the differential operator $\Delta^{(2)}$ (5.2.15) acting on the blocks. This can be seen from the arguments around (5.4.39). We determine the anomalous dimensions after unmixing for many values of (Δ, p) and finally, predict a general formula which is indeed $\delta^{(2)}$. Let us look at a few examples, starting with the sector with SU(2) charge p = 0.

5.6.1 p = 0 Sector

For the p = 0 sector, we only need to consider correlators of the form $H_{q_1q_1q_2q_2}$ and thus for free theory we only need equal charge correlators. We start with a conformal block analysis of free theory, where we also need to consider contributions from half-BPS blocks (5.2.24). Therefore, it is best to consider $G_{qqqq}^{(0)}$ (5.3.10) instead of H for free theory. $G_{qqqq}^{(0)}$ decomposes into one half-BPS block with $\Delta = p = 0$ which corresponds to the identity operator and long blocks with coefficients $A_{qqqq}^{(0)}$. Evaluating the free coefficients for many correlators and many values of Δ, p we obtain a general formula for the free coefficients $A_{qqqq}^{(0)}(\Delta, p)$ for all Δ, p, q :

$$A_{qqqq}^{(0)}(\Delta, p) = \frac{(1 + (-1)^{\Delta + p + 1}) \Delta! (2(p+1))! (\Delta + 2q - 1)!}{2 (2\Delta)! (\Delta + p + 1)p! (p+1)! (p+2q-1)! (-p+2q-2)! (1-2q)^2} \times (\Delta - p + 1)_p (\Delta - 2q + 2)_{-p+2q-2} .$$
(5.6.1)

These coefficients are non-zero only for odd t, which is expected from the parametrisation of operators in (5.5.5) because for even t only operators $\mathcal{O}_{q_1q_2}$ with $q_1 \neq q_2$ contribute.

For the simplest case $\Delta = 1$ there is only one exchanged operator of the form (5.5.1) contributing, $\mathcal{O}_1 \mathcal{O}_1$, thus we perform the conformal block analysis of the correlator H_{1111}^{sugra} . We spell out the supergravity coefficient together with the free theory coefficient from (5.6.1):

$$A_{1111}^{(0)}(1,0) = 1$$
, $M_{1111}^{\text{sugra}}(1,0) = \frac{4(\Delta!)^2}{(2\Delta)!}|_{\Delta=1} = 2$, (5.6.2)

The unmixing equations are

$$A_{1111}^{(0)}(1,0) = \left(C_{1,0}^{(0)}\right)^2 , \quad M_{1111}^{\text{sugra}}(1,0) = \gamma_{1,0}^{\text{sugra}} \left(C_{1,0}^{(0)}\right)^2 .$$
 (5.6.3)

Solving these equations we get

$$\gamma_{1,0}^{\text{sugra}} = 2 , \quad C_{1,0}^{(0)} = 1 .$$
 (5.6.4)

At weight $\Delta = 3$ there are two possible exchanged operators, $\mathcal{O}_2\mathcal{O}_2$ and $\mathcal{O}_1\partial^2\mathcal{O}_1$, which gives the matrices of OPE coefficients for free theory and supergravity:

$$\hat{A}_{3,0}^{(0)} = \begin{pmatrix} A_{1111}^{(0)} & 0\\ 0 & A_{2222}^{(0)} \end{pmatrix}_{(3,0)} = \begin{pmatrix} \frac{1}{10} & 0\\ 0 & \frac{1}{18} \end{pmatrix}, \quad \hat{M}_{3,0}^{\text{sugra}} = \begin{pmatrix} M_{1111}^{\text{sugra}} & M_{1122}^{\text{sugra}}\\ M_{1122}^{\text{sugra}} & M_{2222}^{\text{sugra}} \end{pmatrix}_{(3,0)} = \begin{pmatrix} \frac{1}{5} & \frac{1}{3}\\ \frac{1}{3} & \frac{5}{9} \end{pmatrix}, \quad (5.6.5)$$

where the new coefficients for general odd Δ are

$$M_{1122}^{\text{sugra}}(\Delta, 0) = \frac{2(\Delta - 1)\Delta!(\Delta + 2)!}{3(\Delta + 1)(2\Delta)!} , \quad M_{2222}^{\text{sugra}}(\Delta, 0) = \frac{(\Delta - 1)^2(\Delta + 2)^2(\Delta!)^2}{9(2\Delta)!} , \quad (5.6.6)$$

the rest were already spelled out in (5.6.2). To obtain the $\mathcal{O}(1/c)$ contribution to the anomalous dimensions and the leading contributions to the three-point functions, we solve the unmixing equations in matrix form (5.5.14) and (5.5.15):

$$\hat{\gamma}_{3,0}^{\text{sugra}} = \begin{pmatrix} 12 & 0\\ 0 & 0 \end{pmatrix}, \quad \mathbb{C}_{3,0}^{(0)} = \begin{pmatrix} \frac{1}{2\sqrt{15}} & \frac{1}{2\sqrt{3}}\\ \frac{\sqrt{5}}{6\sqrt{3}} & -\frac{1}{6\sqrt{3}} \end{pmatrix}, \quad (5.6.7)$$

where $\hat{\gamma}^{\text{sugra}}$ are the eigenvalues of the matrix $\hat{M}^{\text{sugra}} \cdot (\hat{A}^{(0)})^{-1}$. The eigenvectors are the columns of the orthonormal matrix \tilde{c} which gives the three-point functions $\mathbb{C}^{(0)} = (\hat{A}^{(0)})^{\frac{1}{2}} \cdot \tilde{c}$, as explained above. Note that there is only one non-zero anomalous dimension at $(\Delta, p) = (3, 0)$, namely $\gamma_{3,0}^{\text{sugra}} = 12$. Going to higher Δ and pthis structure will continue, at each weight there is only one non-zero anomalous dimension, as expected from the higher-dimensional symmetry considerations. Let us look at a few more examples.

At weight (5,0), there are three exchanged operators in the double-trace spectrum, $\mathcal{O}_3\mathcal{O}_3$, $\mathcal{O}_2\partial^2\mathcal{O}_2$ and $\mathcal{O}_1\partial^4\mathcal{O}_1$ and conformal block analysis of the appropriate correlators gives the symmetric matrices

$$\hat{A}_{5,0}^{(0)} = \begin{pmatrix} A_{1111}^{(0)} & 0 & 0 \\ A_{2222}^{(0)} & 0 \\ A_{3333}^{(0)} \end{pmatrix}_{(5,0)} = \begin{pmatrix} \frac{1}{126} & 0 & 0 \\ \frac{4}{81} & 0 \\ & \frac{1}{75} \end{pmatrix},$$

$$\hat{M}_{5,0}^{\text{sugra}} = \begin{pmatrix} M_{1111}^{\text{sugra}} & M_{1122}^{\text{sugra}} & M_{1133}^{\text{sugra}} \\ M_{2222}^{\text{sugra}} & M_{2233}^{\text{sugra}} \\ & M_{3333}^{\text{sugra}} \end{pmatrix}_{(5,0)} = \begin{pmatrix} \frac{1}{63} & \frac{2}{27} & \frac{1}{15} \\ \frac{28}{81} & \frac{14}{45} \\ & \frac{7}{25} \end{pmatrix}, \quad (5.6.8)$$

where the expressions for general odd Δ for the new coefficients are

$$M_{1133}^{\text{sugra}}(\Delta, 0) = \frac{(\Delta - 3)(\Delta - 1)(\Delta + 4)\Delta!(\Delta + 2)!}{30(\Delta + 1)(2\Delta)!} ,$$

$$M_{2233}^{\text{sugra}}(\Delta, 0) = \frac{(\Delta - 3)(\Delta - 1)^2(\Delta + 2)(\Delta + 4)\Delta!(\Delta + 2)!}{180(\Delta + 1)(2\Delta)!} ,$$

$$M_{3333}^{\text{sugra}}(\Delta, 0) = \frac{(\Delta - 3)^2(\Delta - 1)^2(\Delta + 2)(\Delta + 4)^2\Delta!(\Delta + 2)!}{3600(\Delta + 1)(2\Delta)!} .$$
(5.6.9)

Solving the unmixing equations (5.5.13) for the anomalous dimensions, we find that all are zero but one:

$$\gamma_{5,0}^{\text{sugra}} = 30$$
 . (5.6.10)

Starting from 3×3 matrices as in the present case, there is a zero-degeneracy, because more than one operator have zero anomalous dimension. Due to this degeneracy the free three-point functions cannot be completely fixed but we are left with two free parameters b_1 and b_2 . In particular, of the eigenvectors forming the matrix \tilde{c} , the two which correspond to the zero eigenvalues are not unique. Thus, the leading three-point functions is:

$$\mathbb{C}_{5,0}^{(0)} = \begin{pmatrix}
\frac{1}{3\sqrt{210}} & -\frac{3\sqrt{5}b_1 + b_2}{3\sqrt{690}n_{12}} & \frac{\sqrt{5}b_1 - 15b_2}{15\sqrt{138}n_{12}} \\
\frac{\sqrt{14}}{9\sqrt{15}} & \frac{\sqrt{46}b_2}{9\sqrt{15}n_{12}} & -\frac{\sqrt{46}b_1}{9\sqrt{15}n_{12}} \\
\frac{\sqrt{7}}{5\sqrt{30}} & \frac{2\sqrt{5}b_1 - 7b_2}{5\sqrt{690}n_{12}} & \frac{7\sqrt{5}b_1 + 10b_2}{25\sqrt{138}n_{12}}
\end{pmatrix},$$
(5.6.11)

where $n_{12} = \sqrt{b_1^2 + b_2^2}$ and $b_2 b_3 > b_1 b_4$. One of the coefficients b_1 and b_2 will be fixed when solving the mixing problem for four-derivative corrections, since the supergravity data goes into the higher-derivative corrections unmixing equations, see (5.5.13).

We solve the mixing problem for higher Δ analogously to the previous cases. Let us look at a few examples for p = 1 before analysing the results.

5.6.2 *p* = 1 Sector

Let us start by determining the free theory coefficients. For p > 0 we need to consider conformal block expansions of not only G_{qqqq} but also $G_{q_1q_2q_1q_2}$ (5.3.17). These correlators decompose into one half-BPS block with $\Delta = p = q_1 + q_2 - 1$ and long blocks with coefficients $A_{q_1q_2q_1q_2}^{(0)}$. The free theory coefficients for $q_1 = q_2$ were given in (5.6.1) and the coefficients for mixed charges are:

$$A_{q_{1}q_{2}q_{1}q_{2}}^{(0)} = \frac{(-1)^{q_{1}+q_{2}} 2^{-\left\lfloor \frac{p-|q_{12}|}{2} \right\rfloor + \frac{p+1}{2} - \frac{|q_{12}|}{2}}}{((\sqrt{2}-1)(-1)^{p} + (1+\sqrt{2})(-1)^{|q_{12}|})(2\Delta)!(p+1)!(\Delta-p)(\Delta+p+1)} \times \frac{(2(p+1))!(\Delta-|q_{12}|)!(\Delta+|q_{12}|)!(\Delta+q_{1}+q_{2}-1)!}{(\Delta-q_{1}-q_{2}+1)!(2q_{1}-1)(2q_{2}-1)(-p+q_{1}+q_{2}-2)!(p+q_{1}+q_{2}-1)!} \times \frac{\left(\left\lfloor \frac{1}{2}(p-|q_{12}|+1)\right\rfloor + |q_{12}|\right)!\left(\left\lfloor \frac{p-|q_{12}|}{2}\right\rfloor + \left\lfloor \frac{1}{2}(p-|q_{12}|+1)\right\rfloor + |q_{12}|\right)!}{(p-|q_{12}|)!\left(\left\lfloor \frac{p-|q_{12}|}{2}\right\rfloor + |q_{12}|\right)!\left(2\left(\left\lfloor \frac{1}{2}(p-|q_{12}|+1)\right\rfloor + |q_{12}|\right)\right)!}, (5.6.12)$$

where $q_{12} = q_1 - q_2$. Note that in the denominator of the first line, the eigenvalue $\delta^{(2)} = (\Delta - p) (\Delta + p + 1)$ (see (5.6.18)) appears, this was also observed in 4d free theory in [41] and suggests that $\Delta^{(2)} H^{\text{free}}$ should be a simple object as was confirmed in subsection 5.3.1.

For the simplest case $\Delta = 2$, p = 1 there is only one exchanged operator in the double-trace spectrum, $\mathcal{O}_1\mathcal{O}_2$, and we perform the conformal block analysis for the correlator H_{1212} . The free theory (see (5.6.12)) and supergravity coefficients are

$$A_{1212}^{(0)}(2,1) = \frac{1}{4} , \quad M_{1212}^{\text{sugra}}(2,1) = \frac{2(1+\Delta)\Delta!(1+\Delta)!}{3(2\Delta)!}|_{\Delta=2} = 1$$
 (5.6.13)

and solving the unmixing equations gives

$$\gamma_{2,1}^{\text{sugra}} = 4 , \quad C_{2,1}^{(0)} = \frac{1}{2} .$$
 (5.6.14)

Next, let us solve the mixing problem for (4,1) where two types of operators are

exchanged, $\mathcal{O}_1 \partial \mathcal{O}_2$ and $\mathcal{O}_2 \mathcal{O}_3$. The conformal block coefficients are:

$$\hat{A}_{4,1}^{(0)} = \begin{pmatrix} A_{1212}^{(0)} & 0\\ 0 & A_{2323}^{(0)} \end{pmatrix}_{(4,1)} = \begin{pmatrix} \frac{5}{84} & 0\\ 0 & \frac{1}{30} \end{pmatrix}, \quad \hat{M}_{4,1}^{\text{sugra}} = \begin{pmatrix} M_{1212}^{\text{sugra}} & M_{1223}^{\text{sugra}}\\ M_{1223}^{\text{sugra}} & M_{2323}^{\text{sugra}} \end{pmatrix}_{(4,1)} = \begin{pmatrix} \frac{5}{21} & \frac{1}{3}\\ \frac{1}{3} & \frac{7}{15} \end{pmatrix}, \quad (5.6.15)$$

where the new coefficients in terms of general even Δ are given by:

$$M_{1223}^{\text{sugra}} = \frac{(\Delta - 2)(\Delta + 1)(\Delta + 3)\Delta!(\Delta + 1)!}{15(2\Delta)!} ,$$

$$M_{2323}^{\text{sugra}} = \frac{(\Delta + 1)((\Delta + 3)(\Delta - 2))^2 \Delta!(\Delta + 1)!}{150(2\Delta)!} .$$
(5.6.16)

Solving the unmixing equations gives the anomalous dimensions and three-point functions

$$\gamma_{4,1}^{\text{sugra}} = 18 \ , \quad \mathbb{C}_{4,1}^{(0)} = \begin{pmatrix} \frac{\sqrt{5}}{3\sqrt{42}} & -\frac{\sqrt{5}}{6\sqrt{3}} \\ \frac{\sqrt{7}}{3\sqrt{30}} & \frac{1}{3\sqrt{15}} \end{pmatrix} \ . \tag{5.6.17}$$

Again there is only one non-zero anomalous dimension, $\gamma_{4,1}^{\rm sugra}=18$, as expected.

5.6.3 Anomalous Dimensions after Unmixing

We can solve the mixing problem for any pair (Δ, p) analogously to the above examples, where for every (Δ, p) we use (5.5.3) to determine the list of double-trace operators in the spectrum. Solving the unmixing equations for supergravity for many pairs of (Δ, p) we find that, as expected from the 4d conformal symmetry considerations in subsection 5.4.2 and the 4d effective spin conjecture (5.5.7), there is only one operator with non-zero anomalous dimension exchanged. Hence, there is no non-zero degeneracy in the supergravity limit in 1d. The value of the anomalous dimension is as expected which is the eigenvalue of $\Delta^{(2)}$ (5.2.15) acting on the superconformal blocks:

$$\Delta^{(2)}\left(\frac{x\,y}{x-y}B^{\log}_{\Delta,p,p_{12},p_{34}}\right) = \delta^{(2)}\left(\frac{x\,y}{x-y}B^{\log}_{\Delta,p,p_{12},p_{34}}\right) \,. \tag{5.6.19}$$

This was predicted from the 4d conformal symmetry in (5.4.39).

We have now studied the double-trace spectrum of correlators in the supergravity limit and their anomalous dimensions in detail. We have seen that they do agree with the predictions from 4d conformal symmetry and thus that supergravity correlators in $AdS_2 \times S^2$ indeed have 4d conformal symmetry. In the next sections we will consider higher-derivative corrections and how the higher-dimensional conformal symmetry breaks (except for an infinite number of correlators of specific spherical harmonics). On the other hand, all higher-derivative corrections can be deduced from a higherdimensional scalar effective field theory.

5.7 Higher-Derivative Corrections

In this section we study higher-derivative corrections described by a small-*a* expansion. We first derive the form of lowest-charge higher-derivative corrections, with any number of derivatives, from crossing symmetry and $x \rightarrow 0$ behaviour in subsection 5.7.1. In subsection 5.7.2 we obtain all four-derivative corrections to all half-BPS four-point correlators described by tree-level supergravity from a 4d effective action by evaluating generalised $AdS_2 \times S^2$ Witten diagrams with derivatives. Finally, in subsection 5.7.3 we briefly discuss how by acting with inverse higher-order differential operators on the higher-derivative corrections one can construct an object that transforms like a 4d conformal correlator in specific cases, as discussed in subsection 5.1.1. We will see that the 4d conformal symmetry is broken for general four-derivative corrections, however it is intact for an infinite tower of correlators of operators with SU(2) charges $\{p_i\} = pp11$ and crossing versions.

5.7.1 Lowest-Charge Higher-Derivative Corrections

Even though we mainly focus on the four-derivative corrections in this chapter, let us also discuss the general structure of any higher-derivative corrections in 1d SCFTs. Recall from the review in subsection 2.3.2 that in [30] the authors argued that for $AdS_{d>2}$ solutions to the crossing equations with a conformal block expansion truncated to spin L = 0, 2, 4, ... are dual to quartic contact interactions in the bulk. For a fixed spin L, there are L/2 + 1 independent interactions with number of derivatives running from 2L, 2L + 2, ..., 3L. In this subsection we will find that reducing these solutions to AdS_2 by taking the holomorphic limit, the L/2+1 solutions collapse to a single solution proportional to \overline{D}_{jjjj} , where j = L/2 + 1. This solution corresponds to a 2L-derivative interaction, i.e. the interaction with the lowest number of derivatives in the spin-L tower of solutions. This is the case for the lowestcharge correlators, when going to higher charges, this degeneracy will break. In this subsection, we focus on the lowest-charge correlators but consider interactions with any number of derivatives. For later sections however, we focus on the four-derivative corrections but consider all higher-charge correlators.

Let us deduce the form of higher-derivative corrections to the $p_i = 1$ correlator. Let us start with an ansatz for a higher-derivative correction at $\mathcal{O}(a^{\#}/c)$ analogous to supergravity in $(5.4.3)^5$:

$$\langle \mathcal{O}_1 \mathcal{O}_1 \mathcal{O}_1 \mathcal{O}_1 \rangle_{a^{\#}/c} = g_{12} g_{34} \frac{x}{y} (x - y) a(x) , \qquad (5.7.1)$$

where we get the same conditions from crossing invariance and the $x \rightarrow 0$ behaviour:

$$a(1-x) = a(x)$$
, $a\left(\frac{1}{x}\right) = x^2 a(x)$, $a(x) = \mathcal{O}(1)$. (5.7.2)

Again we make an ansatz for a(x) of the following form

$$a(x) = p(x)\log x^{2} + p(1-x)\log(1-x)^{2} + r(x) , \qquad (5.7.3)$$

⁵See subsection 5.4.1 for a more careful discussion of four-point crossing in 1d which could easily be extended to higher-derivative corrections to see that $a^0(x)$ and $a^+(x)$ have the same functional form. In this subsection we will consider a(x) with x in the regions $x \in (0, 1)$ and $x \in (1, \infty)$.

where p, r are rational functions and r(1 - x) = r(x). Further we have

$$p(x) = \frac{1}{x-1} \frac{1}{x^k (1-x)^k} \sum_{i=0}^m b_i x^i , \quad r(x) = \frac{1}{x-1} \frac{1}{x^k (1-x)^k} \sum_{i=0}^m c_i x^i , \quad (5.7.4)$$

where k and m are integers. Plugging this into (5.7.2), there are only new solutions for k = 2q and m = 3k = 6q with an integer q. Solutions for odd k reduce to superpositions of lower-k solutions in 1d. The q = 0 case corresponds to supergravity. Going to higher q we find solutions that are superpositions of a new correction and lower-q solutions (supergravity and lower-derivative terms). The new corrections are of the form

$$H_{1111}^{q} = x^{2}a(x) = -u\left(1 + u^{q} + v^{q}\right)\bar{D}_{q+1\,q+1\,q+1}^{\text{hol}} + \dots , \qquad (5.7.5)$$

where the dots denote terms with lower q. As we have seen in chapter 4 the averaged anomalous dimensions of double-trace operators in the conformal block decomposition of these correlators encode the number of derivatives of the dual quartic interactions in the bulk [30]. First, decomposing the free theory $G_{1111}^{(0)}$ (5.3.10) gives the identity operator and the coefficients of the long blocks $A_{1111}^{(0)}(\Delta, 0)$ (5.6.1):

$$A_{1111}^{(0)}(\Delta, 0) = \frac{(1 + (-1)^{\Delta+1}) (\Delta!)^2}{(2\Delta)!} .$$
 (5.7.6)

Expanding the higher-derivative corrections (5.7.5) at $\mathcal{O}(1/c)$ in terms of the long blocks according to (5.5.11) we get the averaged anomalous dimensions

$$\gamma(q) = N(q)P^{(q)}(\Delta) , \qquad (5.7.7)$$

with

$$N(q) = \frac{4^{q+1}(1+2q)!}{(2(1+2q))!}$$
(5.7.8)

and $P^{(q)}(\Delta)$ are polynomials of degree Δ^{4q} . For the first few cases we get

$$\begin{split} P^{(0)}(\Delta) &= 1 \ , \\ P^{(1)}(\Delta) &= \frac{1}{2} \left(\Delta^4 + 2 \, \Delta^3 + \Delta^2 + 6 \right) \ , \end{split}$$

$$P^{(2)}(\Delta) = \frac{1}{8} \left(\Delta^8 + 4\,\Delta^7 + 10\,\Delta^6 + 16\,\Delta^5 + 289\,\Delta^4 + 556\,\Delta^3 + 276\,\Delta^2 + 864 \right) \dots$$
(5.7.9)

The fact that the averaged anomalous dimensions scale like Δ^{4q} in the large- Δ limit implies that the corrections (5.7.5) correspond to a 4q-derivative interaction in the dual bulk field theory. As mentioned before, the q = 0 case corresponds to the supergravity correlator which is dual to a spin-0 or ϕ^4 interaction. The next case, q = 1, is given by $-u(1 + u + v)\bar{D}_{2222}^{\text{hol}}$ plus a spin-0 term and corresponds to a spin-2 correction and is dual to a four-derivative interaction (see (5.7.17)). As predicted in the beginning of this subsection, there is only one solution per spin L in 1d. In principle there is another spin-2 solution corresponding to six derivatives and to k = 3, but this reduces to the four-derivative solution plus supergravity. This is only true for the lowest-charge correlators though and going to higher charges, this degeneracy will be lifted.

We have now discussed the general structure of higher-derivative corrections to the lowest-charge supergravity correlators and in the next section we will obtain all four-derivative corrections to the half-BPS correlators for all spherical harmonics.

5.7.2 Effective Action

In this section we compute all four-derivative corrections from the 4d scalar effective action (5.2.27). We follow the procedure outlined in chapter 3, in particular section 3.4 where the α'^5 corrections, which correspond to four-derivative interactions, were computed. The only independent contact terms with four derivatives one can write down are the main contribution $(\nabla \phi . \nabla \phi)^2$ and the ambiguity $\nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2$. The complete effective action at this order is then with

$$S_{4-\text{deriv}}^{\text{main}} = \frac{3}{4!} \int_{\text{AdS}\times\text{S}} d^2 \hat{X} d^2 \hat{Y} \left(\nabla\phi,\nabla\phi\right) \left(\nabla\phi,\nabla\phi\right) ,$$

$$S_{4-\text{deriv}}^{\text{amb}} = \frac{6}{4!} \int_{\text{AdS}\times\text{S}} d^2 \hat{X} d^2 \hat{Y} \nabla^2 \nabla_\mu \phi \nabla^\mu \phi \phi^2 . \qquad (5.7.11)$$

Note that the ϕ^4 interaction is not considered as an ambiguity here, since it describes supergravity and not a zero-derivative correction arising from quantum gravity as in other dimensions.

The Witten diagram expression for the main correction is

$$\left\langle \mathcal{OOOO} \right\rangle_{\text{int}}^{4\text{-deriv;main}} = \frac{1}{4!} \frac{(\mathcal{C}_1)^4}{(-2)^4} \int_{\text{AdS}\times\text{S}} d^2 \hat{X} d^2 \hat{Y} \frac{N_{12}N_{34} + N_{13}N_{24} + N_{14}N_{23}}{(P_1 + Q_1)^2 (P_2 + Q_2)^2 (P_3 + Q_3)^2 (P_4 + Q_4)^2} , \quad (5.7.12)$$

where

$$N_{ij} = X_i X_j + Y_i Y_j + P_i P_j - Q_i Q_j . (5.7.13)$$

Furthermore, the ambiguity is

$$\left\langle \mathcal{OOOO} \right\rangle_{\text{int}}^{4\text{-deriv;amb}} = -\frac{1}{4!} \frac{(\mathcal{C}_1)^4}{(-2)^4} \int_{\text{AdS}\times\text{S}} \frac{d^2 \hat{X} d^2 \hat{Y}}{\prod_i (P_i + Q_i)} \sum_{i < j} \frac{L_{ij}}{(P_i + Q_i) (P_j + Q_j)} , \qquad (5.7.14)$$

where

$$L_{ij} = X_i \cdot X_j - Y_i \cdot Y_j + P_i P_j + Q_i Q_j .$$
 (5.7.15)

We get the explicit expressions in position space for the correlators at four derivatives by expanding (5.7.12) and (5.7.14) in terms of *D*- and *B*-functions, see (3.1.24)and (3.1.28). To expand a general decorated integral of the form (3.3.9) obtained from a contact interaction with any number of covariant derivatives, use (3.3.10)

$$(-2)^{2\Sigma_{X}+2\Sigma_{Y}} \sum_{p_{i}=0}^{\infty} \left(\prod_{i=1}^{4} (-1)^{p_{i}-1} \frac{(p_{i})_{\Delta_{i}+n_{i}-1}}{\Gamma(\Delta_{i})} D_{p_{i}-1+\Delta_{i}+n_{i}-n_{i}^{P}}^{(d)}(X_{i}) B_{p_{i}-1+n_{i}^{Q}}^{(d)}(Y_{i})\right) \times \left(\sum_{i< j} (X_{i}.X_{j})^{n_{ij}^{X}} (Y_{i}.Y_{j})^{n_{ij}^{Y}}\right),$$
(5.7.16)

where the different labels are explained in the discussion above (3.3.10). To perform explicit calculations in 1d position space one needs to express the correlators above in terms of holomorphic \overline{D} -functions and we spell out a few examples here. Note that the coefficients in (5.7.10) are unfixed and we choose the overall normalisation of the $H_{p_1p_2p_3p_4}^{4\text{-deriv}}$ below such that they agree with the supergravity correlators in (5.4.28). The explicit expressions for some of the correlators are:

$$\begin{aligned} H_{1111}^{4\text{-deriv}} &= 3u \left(\bar{D}_{1111}^{\text{hol}} - 5 \left(1 + u + v \right) \bar{D}_{2222}^{\text{hol}} \right) , \\ H_{pp11}^{4\text{-deriv}} &= \frac{\left(-1 \right)^{p+1} u^p}{(p-1)!} \left(f_1(p) \bar{D}_{pp11}^{\text{hol}} + f_2(p) u \bar{D}_{p+1\,p+1\,11}^{\text{hol}} + f_3(p) \left(1 + u + v \right) \bar{D}_{p+1\,p+1\,22}^{\text{hol}} \right) , \\ H_{p1p1}^{4\text{-deriv}} &= \frac{\left(-1 \right)^{p+1} u^{\frac{p+1}{2}}}{(p-1)! y^{p-1}} \left(f_1(p) \bar{D}_{p11p}^{\text{hol}} + f_2(p) \bar{D}_{p+1\,1p+1\,1}^{\text{hol}} + f_3(p) \left(1 + u + v \right) \bar{D}_{p+1\,2p+1\,2}^{\text{hol}} \right) , \\ H_{p11p}^{4\text{-deriv}} &= \frac{\left(-1 \right)^{p+1} u^{\frac{p+1}{2}}}{(p-1)! y^{p-1}} \left(f_1(p) \bar{D}_{p11p}^{\text{hol}} + f_2(p) \bar{D}_{p+1\,11\,p+1}^{\text{hol}} + f_3(p) \left(1 + u + v \right) \bar{D}_{p+1\,22\,p+1}^{\text{hol}} \right) , \\ H_{2222}^{4\text{-deriv}} &= 2 \frac{u^2}{y^2} \left(\left[-8 C_0 \left(1 - y + y^2 \right) + 2 \left(41 - 6y + 6y^2 \right) \right] \bar{D}_{2222}^{\text{hol}} + 35 \left(y^2 - 1 \right) \bar{D}_{3322}^{\text{hol}} \right) , \end{aligned}$$

$$(5.7.17)$$

where

$$f_1(p) = 2C_0 p(p-1) + p^2 (4p^2 - 8p + 1) , \quad f_2(p) = -(1+2p)(2p^2 - 3p + 1) ,$$

$$f_3(p) = (1+2p)(3+2p) . \quad (5.7.18)$$

Note that $H_{1111}^{4\text{-deriv}}$ agrees with the prediction in the previous subsection, see (5.7.5) for q = 1. We investigate the correlators in terms of the 4d conformal symmetry in the next subsection before we analyse the double-trace spectrum of these correlators and solve the mixing problem in the following section.

5.7.3 Breaking of 4d Conformal Symmetry

In this subsection we investigate the higher-derivative corrections in terms of the 4d conformal symmetry. In the previous section we have seen that one can derive all four-derivative corrections to all half-BPS correlators with any charges, described by tree-level supergravity, from a four-dimensional effective field theory. Thus, higher-
derivative corrections to correlators in $AdS_2 \times S^2$ have a four-dimensional symmetry. However, the four-dimensional *conformal* symmetry will generally be broken for higher-derivative corrections. As described in subsection 5.1.1 we will attempt to construct an object invariant under higher-dimensional conformal symmetry by acting with negative powers of differential operators related to $\Delta^{(2)}$ on the correlators. In this way, we obtain an object with the right dimensions to be invariant under 4d conformal symmetry, however, this only works for a subset of correlators with specific charge configurations, as we will explain shortly. In this subsection, we describe how this can be done for four-derivative corrections and show that there is an infinite tower of correlators of specific KK modes which can be obtained from the 4d conformal symmetry. Yet for general charges, the symmetry is broken. We will discuss the breaking of the symmetry and also briefly comment on implications for $AdS_5 \times S^5$.

$\langle \mathcal{O}_p \mathcal{O}_p \mathcal{O}_1 \mathcal{O}_1 \rangle$ and crossing versions

We start by studying correlators with external charges $\{p_i\} = pp11$ and their crossing versions. To get an object with the dimensions of a 4d conformal correlator, we have to act with negative powers of a differential operator on the correlator. By dimensional analysis, we saw in subsection 5.1.1 that for a vertex with k + 2 derivatives (where supergravity has two derivatives) one has to act with a general $(\Delta^{(2)})^{-k/2}$ on the corresponding correlator. Hence, for a four-derivative interaction (six derivatives in total when counting supergravity), we act with an inverse fourth-order differential operator on the correlator. In practice, we consider objects which have 4d conformal symmetry, expand them in internal coordinates to obtain all higher-charge correlators, analogous to the considerations in subsection 5.4.2, and act on them with a fourth-order differential operator to reconstruct the higher-charge higher-derivative corrections. The 4d conformal objects we have to consider are the 4d conformal blocks (5.4.30) with the appropriate twist and spin. We start from an ansatz for a general fourth-order Casimir and fix the coefficients by comparing to the explicit results obtained from the 4d effective action in (5.7.17).

By superconformal symmetry, all higher-order Casimir operators have to be of the form $\Delta^{(2)} \left(c_1 \mathcal{D}_x^a + c_2 \mathcal{D}_y^b \right)$ and thus a general fourth-order Casimir is given by

$$\Delta^{(4)} = c_1 + c_2 \Delta^{(2)} + c_3 \Delta^{(2)} \mathcal{D}_x + c_4 \Delta^{(2)} \mathcal{D}_y , \qquad (5.7.19)$$

where $\Delta^{(2)}$, \mathcal{D}_x , \mathcal{D}_y were defined in (5.2.15) and the c_i are unfixed coefficients. As explained in subsection 5.4.2 due to the 4d conformal symmetry there is only one 4d block contributing to the 4d uplift of the 1d correlators at each spin. For supergravity it is the spin-0 block and since the four-derivative correction corresponds to a spin-2 correction, we consider the 4d spin-0 and spin-2 blocks as the objects which play the leading role in the 4d conformal symmetry. Therefore, we propose that a subset of the four-derivative corrections can be reconstructed by acting with a fourth-order Casimir of the form (5.7.19) on the holomorphic limit of the 4d spin-0 and spin-2 blocks. Note that the blocks reproduce the log u piece of the correlators rather than the full functions. The higher-charge versions of the spin-L blocks can be obtained by replacing $x_{ij}^2 \to x_{ij}^2 (1 + g_{ij}^2)$, expanding in g_{ij}^2 and taking the coefficients of the appropriate powers in g_{ij}^2 analogous to subsection 5.4.2. Note that higher-charge versions of the spin-0 block exactly correspond to the log u piece of the supergravity correlators. The relevant equation at different charges p_i is then

$$H_{p_1p_2p_3p_4}^{4\text{-deriv, main}}|_{\log u} = \Delta_{\text{spin-0}}^{(4)} \left(4\text{d spin-0 block} \right)_{p_1p_2p_3p_4}^{\text{hol}} + \Delta_{\text{spin-2}}^{(4)} \left(4\text{d spin-2 block} \right)_{p_1p_2p_3p_4}^{\text{hol}} ,$$

$$(5.7.20)$$

where we compare our ansatz to the four-derivative corrections deduced from the effective action given in (5.7.17) at different charges. Note that we focus on the main contribution in (5.7.10) here.

Solving (5.7.20) for $H_{pp11}^{4\text{-deriv, main}}$ partially fixes the coefficients in (5.7.19) to

$$\Delta_{\text{spin-0}}^{(4)} = \frac{1}{6}\Delta^{(2)} + \frac{5}{12}\Delta^{(2)}\mathcal{D}_x + a_3\Delta^{(2)}\mathcal{D}_y , \quad \Delta_{\text{spin-2}}^{(4)} = -\frac{1}{45}\Delta^{(2)} + \frac{1}{90}\Delta^{(2)}\mathcal{D}_x + b_3\Delta^{(2)}\mathcal{D}_y , \quad (5.7.21)$$

where a_3 and b_3 remain unfixed since there is no y-dependence for $\{p_i\} = pp11$.

We can fix the remaining coefficients by considering $H_{p1p1}^{4\text{-deriv, main}}$ and $H_{p11p}^{4\text{-deriv, main}}$. We thus solve (5.7.20) by comparing to the results from (5.7.17) at the corresponding charges. This completely fixes the differential operators to

$$\Delta_{\text{spin-0}}^{(4)} = \frac{1}{12} \left(2 \,\Delta^{(2)} + 5 \,\left(\Delta^{(2)} \right)^2 \right) \,, \quad \Delta_{\text{spin-2}}^{(4)} = \frac{1}{90} \left(-2 \,\Delta^{(2)} + \left(\Delta^{(2)} \right)^2 \right) \,. \tag{5.7.22}$$

To summarise, these Casimir operators can lift the lowest-charge four-derivative correction correctly for $\{p_i\} = pp11$ and crossing versions. So the 4d conformal symmetry is satisfied by an infinite tower of four-derivative corrections. This is highly non-trivial, since the 4d blocks themselves are not crossing symmetric and also the Casimir operators do not preserve crossing symmetry (only under exchange of $1 \leftrightarrow 2$). Hence, an operator which does not preserve crossing symmetry acts on an object which is not crossing symmetric, and one obtains a crossing symmetric correlator. However, going to different charges there are no crossing symmetric solutions to (5.7.20) and thus the symmetry is broken. We will further discuss this breaking of the hidden conformal symmetry below and comment on implications for $AdS_5 \times S^5$.

Breaking of the symmetry and implications for $AdS_5 \times S^5$

Recall that the higher-dimensional conformal symmetry arises when a correlator corresponds to a conformally invariant amplitude in flat space which is connected to a scaleless coupling. When the correlator does not correspond to such an amplitude we can rescale it by acting with differential operators of appropriate powers on it, and this works well for free theory. When the dual bulk interaction vertices contain derivatives, which is the case for general higher-derivative corrections, the higherdimensional conformal symmetry breaks down (however, it might still be intact for an infinite tower of higher-derivative corrections with specific charges). This breaking of the symmetry can be anticipated because a reduction of derivative terms on the sphere will give a number of terms with different numbers of derivatives in AdS. Thus, we get terms which scale differently in AdS and we can therefore not rescale the corresponding correlator consistently.

We have investigated this in $AdS_2 \times S^2$ for the four-derivative corrections above. Let us now comment on the implications for the 10d conformal symmetry and its breaking in $AdS_5 \times S^5$. As discussed in subsection 5.1.1, in the supergravity approximation the correlator itself has higher-dimensional conformal symmetry whereas for free theory acting with the eighth-order differential operator $\Delta^{(8)}$ on the correlator rescales it such that it has 10d conformal symmetry. Now, consider the α'^3 corrections which correspond to a ϕ^4 interaction in the scalar effective action. We obtained all α'^3 corrections from the 10d scalar effective action in section 3.2. Based on the discussions above, since these corrections correspond to a zero-derivative interaction in the effective field theory, we expect the corresponding correlators to enjoy 10d conformal symmetry analogous to supergravity correlators. And this conjecture is indeed strongly supported by the results in [13] where the anomalous dimensions after unmixing associated with α'^3 corrections were obtained. The resulting anomalous dimensions are rational numbers which suggests that the 10d conformal symmetry is intact at this order.

To reconstruct these correlators from the 10d conformal symmetry by acting on them with inverse differential operators consider dimensional analysis. In the low-energy effective action in (2.3.18) supergravity is described by \mathcal{R} which has two derivatives while the first quartic correction corresponds to \mathcal{R}^4 which therefore has six more derivatives than supergravity. So to obtain an object with the correct dimensions we act with an inverse sixth-order differential operator $(\Delta^{(6)})^{-1}$ on the α'^3 corrections to the correlators. In practice, one would reconstruct the higher-charge α'^3 correlators from objects which do have 10d conformal symmetry, i.e. the 10d conformal blocks with appropriate spin and twist, and act with a differential operator $\Delta^{(6)}$ on them. Since the ϕ^4 interaction corresponds to a spin-0 correction (see subsection 2.3.2), the correct object to consider is the 10d spin-0 block uplifted to higher charges analogous to the supergravity case. We leave further discussions of this to [44]. Considering the α'^5 corrections which were obtained in section 3.4, we expect the 10d conformal symmetry to be broken. Similar to the four-derivative corrections in 1d discussed above, one might expect that by acting with a higher-order differential operator on the 10d spin-0 and spin-2 blocks uplifted to higher charges, one could reconstruct a subset of α'^5 corrections with specific charges, we do not consider this in detail here. The breaking of the symmetry for α'^5 corrections was also observed from a different point of view in [14]. The authors found that the anomalous dimensions of operators in the OPE spectrum of α'^5 corrections contain square roots in some cases. This indicates a breaking of the 10d conformal symmetry. In more detail, consider the unmixing of the anomalous dimensions of double-trace operators in the OPE spectrum of α'^3 and α'^5 corrections. The anomalous dimensions after unmixing associated with $\alpha^{\prime 3}$ are rational [13], which suggests that the 10d conformal symmetry is intact, as mentioned before. An important point in these calculations is that the $\mathcal{O}(\alpha'^3 c^0)$ correction to the three-point functions is absent. In [14], the authors found that for the α'^{5} corrections however, the $\mathcal{O}(\alpha'^{5}c^{0})$ correction to the three-point functions (analogous to the $\mathcal{O}(ac^0)$ corrections in (5.5.10)) is non-zero for some quantum numbers and this leads to square roots in some of the anomalous dimensions, which indicates breaking of the 10d conformal symmetry. We will discuss the mixing problem for four-derivative corrections in the 1d case in the following section.

5.8 Unmixing Four-Derivative Corrections

In this section, we solve the mixing problem for four-derivative corrections to half-BPS correlators in 1d SCFTs which are described by tree-level 4d $\mathcal{N} = 2$ supergravity. We have predicted all four-derivative corrections including all spherical harmonics in subsection 5.7.2. To solve the mixing problem we use the large-*c* and small-*a* expansion of the conformal block decomposition (5.5.11). We obtain a list of all operators in the double-trace spectrum at each weight (Δ, p) by using the parametrisations in (5.5.3) and (5.5.5). Further, to resolve the degeneracy of operators with different 1d quantum numbers contributing at the same weight (Δ, p), we use the unmixing equations given in (5.5.13). We perform the unmixing for many (Δ, p) , predict general formulas for the anomalous dimensions and check that they agree with the prediction in terms of the 4d effective spin (5.5.6). Recall the conjecture for the 4d spin in terms of 1d quantum numbers is

$$l_{4d} = 2\left(i_A + r_A + i_{B^{\text{to}}} + i_{B^{\text{te}}} + r_{B^{\text{te}}} - \frac{1 + (-1)^{r_B^{\text{te}+1}}}{2}\right) .$$
(5.8.1)

The four-derivative interaction corresponds to a spin-2 correction, thus we predict that we can have 4d spin-0 and spin-2. We therefore expect that only operators with quantum numbers satisfying $l_{4d} = 0, 2$ acquire non-zero anomalous dimensions and indeed we will find that our results agree with this prediction. We start by analysing the unmixing for $t = \Delta - p$ odd and even separately and presenting some examples. The first example where square roots appear is $(\Delta, p) = (5, 2)$ discussed around (5.8.20).

5.8.1 Unmixing for Odd t

We start by analysing the unmixing at weights (Δ, p) with $t = \Delta - p$ odd, where we have both class A and B^{to} operators. For each (Δ, p) use the parametrisation in (5.5.3) to list all operators in the double-trace spectrum at the given weight. Then construct $((d_A + d_{B^{\text{to}}})) \times ((d_A + d_{B^{\text{to}}}))$ matrices of correlators running over $\mathcal{D}_{\Delta,p}$ and solve the unmixing equations (5.5.13). Recall that class A and B operators do not mix, so they can be treated completely separately. The anomalous dimensions after unmixing are labelled as $(\gamma^A)_{i_A,r_A}^{\Delta,p}$ for class A operators and $(\gamma^{B^{\text{to}}})_{i_{B^{\text{to}}}}^{\Delta,p}$ for class B. Let us discuss the same examples as for the supergravity limit in section 5.6, starting with the singlet sector.

p = 0 sector

For p = 0, there are only class A operators in the spectrum, so the operators contributing are the same as for supergravity. Starting with the simplest case $\Delta = 1$, only one exchanged operator $\mathcal{O}_1\mathcal{O}_1$ contributes, thus we perform the conformal block analysis of $H_{1111}^{4\text{-deriv}}$ and get

$$M_{1111}^{4\text{-deriv}}(1,0) = \frac{2\Delta^2(\Delta+1)\Delta!(\Delta+1)!}{(2\Delta)!}|_{\Delta=1} = 4 , \qquad (5.8.2)$$

where the free theory coefficients are given in (5.6.1) and the supergravity contributions were obtained in (5.6.2). The unmixing equations are

$$M_{1111}^{4\text{-deriv}}(1,0) = \gamma_{1,0}^{4\text{-deriv}} \left(C_{1,0}^{(0)}\right)^2 + 2\gamma_{1,0}^{\text{sugra}} C_{1,0}^{(0)} C_{1,0}^{4\text{-deriv}} , \quad 0 = C_{1,0}^{(0)} C_{1,0}^{4\text{-deriv}}. \quad (5.8.3)$$

To solve these equations, we use the supergravity results (5.6.4) which yields

$$\left(\gamma_A^{4\text{-deriv}}\right)_{0,0}^{1,0} = 4 , \quad C_{1,0}^{4\text{-deriv}} = 0 .$$
 (5.8.4)

At weight $\Delta = 3$ there are two possible exchanged operators, $\mathcal{O}_2 \mathcal{O}_2$ and $\mathcal{O}_1 \partial^2 \mathcal{O}_1$ and the matrix of OPE coefficients for the four-derivative correction is

$$\hat{M}_{3,0}^{4\text{-deriv}} = \begin{pmatrix} M_{1111}^{4\text{-deriv}} & M_{1122}^{4\text{-deriv}} \\ M_{1122}^{4\text{-deriv}} & M_{2222}^{4\text{-deriv}} \end{pmatrix}_{(3,0)} = \begin{pmatrix} \frac{72}{5} & \frac{4}{3}(15+C_0) \\ \frac{4}{3}(15+C_0) & \frac{8}{9}(38+5C_0) \end{pmatrix}, \quad (5.8.5)$$

where the coefficients for general Δ are

$$M_{1122}^{4\text{-deriv}}(\Delta, 0) = \frac{(\Delta+2)(\Delta-1)\Delta!(8C_0\Delta! + (\Delta-1)\Delta(\Delta+2)!)}{3(2\Delta)!},$$

$$M_{2222}^{4\text{-deriv}}(\Delta, 0) = \frac{(\Delta+2)(\Delta-1)(\Delta!)^2}{18(2\Delta)!} \times \left[16C_0(\Delta+2)(\Delta-1) + \Delta^6 + 3\Delta^5 - 3\Delta^4 - 11\Delta^3 + 26\Delta^2 + 32\Delta - 32\right],$$
(5.8.6)

and the coefficients for free theory and supergravity were given in (5.6.5). We solve the unmixing equations in matrix form at orders $\mathcal{O}(a/c)$ and $\mathcal{O}(a)$, using the supergravity results (5.6.7):

$$\gamma_{3,0,i}^{4\text{-deriv}} = \left\{ \frac{16}{3} (137 + 15C_0), \frac{64}{3} \right\} , \ \mathbb{C}_{3,0}^{4\text{-deriv}} = (-1 + 3C_0) \begin{pmatrix} \frac{-\sqrt{5}}{9\sqrt{3}} & \frac{1}{9\sqrt{3}} \\ \frac{\sqrt{5}}{27\sqrt{3}} & \frac{5}{27\sqrt{3}} \end{pmatrix} , \quad (5.8.7)$$

where $\gamma_{3,0,i}^{4\text{-deriv}}$ are the eigenvalues of the matrix $(\hat{M}^{\text{sugra}} + a\hat{M}^{4\text{-deriv}}) \cdot (\hat{A}^{(0)})^{-1}$ at $\mathcal{O}(a)$ and *i* is labelling the non-zero anomalous dimensions. Recall that C_0 is the coefficient of the ambiguity in (5.7.10). While for supergravity there is only one anomalous dimension at every weight, there are two non-zero anomalous dimensions at $\mathcal{O}(a/c)$ for (3,0), $(\gamma_A^{4\text{-deriv}})_{0,0}^{3,0} = \frac{16}{3}(137 + 15C_0)$ and $(\gamma_A^{4\text{-deriv}})_{1,0}^{3,0} = \frac{64}{3}$. We will explain the labels, interpret the result from a 4d point of view and give general formulas in subsection 5.8.3.

At weight (5,0), there are three exchanged operators in the double-trace spectrum, $\mathcal{O}_3\mathcal{O}_3$, $\mathcal{O}_2\partial^2\mathcal{O}_2$ and $\mathcal{O}_1\partial^4\mathcal{O}_1$ and the conformal block analysis of the appropriate correlators gives the four-derivative contribution

$$\hat{M}_{5,0}^{4\text{-deriv}} = \begin{pmatrix} M_{1111}^{4\text{-deriv}} & M_{1122}^{4\text{-deriv}} & M_{1133}^{4\text{-deriv}} \\ & M_{2222}^{4\text{-deriv}} & M_{2233}^{4\text{-deriv}} \\ & & M_{3333}^{4\text{-deriv}} \end{pmatrix}_{(5,0)} \\ = \begin{pmatrix} \frac{50}{7} & \frac{8}{27}(105 + C_0) & \frac{4}{5}(30 + C_0) \\ & \frac{32}{81}(352 + 7C_0) & \frac{16}{45}(321 + 14C_0) \\ & & \frac{24}{25}(111 + 7C_0) \end{pmatrix}, \quad (5.8.8)$$

where the free theory and supergravity coefficients are given in (5.6.8) and the expressions for general Δ for the new coefficients are

$$M_{1133}^{4\text{-deriv}}(\Delta, 0) = \frac{(\Delta - 1) (\Delta - 3) \Delta! (24 C_0 (\Delta + 4) (\Delta + 2)! + \Delta (\Delta + 1) (\Delta - 2) (\Delta + 4)!)}{60 (\Delta + 1) (2\Delta)!} ,$$

$$M_{2233}^{4\text{-deriv}}(\Delta, 0) = \frac{(\Delta + 2) (\Delta + 4) (\Delta - 1) (\Delta - 3) (\Delta!)^2}{360 (2\Delta)!} \times \left[32 C_0 (\Delta + 2) (\Delta - 1) + \Delta (\Delta + 1) (\Delta (\Delta + 1) (\Delta^2 + \Delta - 10) + 88) - 96 \right] ,$$

$$M_{3333}^{4\text{-deriv}}(\Delta, 0) = \frac{(\Delta - 1) ((\Delta + 4) (\Delta - 3))^2 \Delta! (\Delta + 2)!}{7200 (\Delta + 1) (2\Delta)!} \times \left[48 C_0 (\Delta + 2) (\Delta - 1) + \Delta^6 + 3\Delta^5 - 11\Delta^4 - 27\Delta^3 + 226\Delta^2 + 240\Delta - 288 \right] .$$
(5.8.9)

We can now solve the unmixing equations (5.5.13) for the contributions to the anomalous dimensions at $\mathcal{O}(a/c)$ and the $\mathcal{O}(a)$ corrections to the three-point functions:

$$\gamma_{5,0,i}^{4\text{-deriv}} = \left\{ \frac{80}{3} (424 + 21C_0), \frac{1204}{3} \right\} ,$$

$$\mathbb{C}_{5,0}^{4\text{-deriv}} = (-1+3C_0) \begin{pmatrix} -\frac{2\sqrt{14}}{27\sqrt{15}} & \frac{-1}{486\sqrt{42}}c & \frac{504-5c}{1458\sqrt{210}} \\ -\frac{16\sqrt{14}}{81\sqrt{15}} & \frac{-336+c}{243\sqrt{42}} & \frac{2(504-5c)\sqrt{2}}{2187\sqrt{105}} \\ \frac{4\sqrt{14}}{45\sqrt{15}} & -\frac{2016+c}{2430\sqrt{42}} & \frac{-504+5c}{1215\sqrt{210}} \end{pmatrix},$$
(5.8.10)

with one unfixed parameter. Additionally, the unmixing equations at this order fix one of the coefficients b_1, b_2 in the leading three-point function (5.6.11) to

$$b_1 = -\frac{9}{4\sqrt{5}}b_2 \ . \tag{5.8.11}$$

There are two non-zero anomalous dimensions, which we will again discuss later from the point of view of the 4d conformal symmetry, they are given by $(\gamma_A^{4\text{-deriv}})_{0,0}^{5,0} = \frac{80}{3}(424 + 21C_0)$ and $(\gamma_A^{4\text{-deriv}})_{1,0}^{5,0} = \frac{1204}{3}$.

We solve the mixing problem for higher Δ analogously to the previous cases. Let us look at a few examples for p = 1, 2, where class *B* operators are relevant, before discussing *t* even.

p = 1 sector

Starting from p = 1 there are class A and class B operators exchanged, in the present case where t is odd, they are class B^{to} operators. These operators contribute from $t \ge 3$ and $p \ge 1$.

In the simplest case $\Delta = 2, p = 1$, there is only one exchanged class A operator in the double-trace spectrum, $\mathcal{O}_1\mathcal{O}_2$ and we perform the conformal block analysis at $\mathcal{O}(a/c)$ for the correlator H_{1212} and the OPE-coefficients are

$$M_{1212}^{4\text{-deriv}, t \text{ odd}}(2, 1) = \frac{1}{3\Delta(2\Delta)!} \times \left[\left(\Delta^5 + \Delta^4 - 4\Delta^3 - \Delta^2 + 7\Delta - 4 \right) \Delta! (\Delta + 2)! + 8 C_0 \left((\Delta + 1)! \right)^2 \right]_{\Delta = 2} = \frac{2}{3} (11 + 6C_0) , \qquad (5.8.12)$$

where the free theory coefficients are given in (5.6.12) and the supergravity coefficients were obtained in (5.6.13). Solving the unmixing equations gives

$$\left(\gamma_A^{4\text{-deriv}}\right)_{0,0}^{2,1} = \frac{8}{3}(11+6C_0) , \quad C_{2,1}^{4\text{-deriv}} = \frac{1}{2} .$$
 (5.8.13)

Next, let us solve one more mixing problem for t odd. At (4,1) two class A and one class B^{to} operators are exchanged, $\mathcal{O}_1 \partial^2 \mathcal{O}_2$, $\mathcal{O}_2 \mathcal{O}_3$ and $\mathcal{O}_2 \partial \mathcal{O}_2$. Recall that for todd, the class A operators have even numbers of derivatives while class B operators have odd numbers of derivatives. The conformal block analysis of the appropriate correlators gives

$$\begin{split} \hat{M}_{4,1}^{4\text{-deriv}} &= \begin{pmatrix} M_{1212}^{4\text{-deriv}} & M_{1223}^{4\text{-deriv}} & M_{1222}^{4\text{-deriv}} \\ M_{1223}^{4\text{-deriv}} & M_{2323}^{4\text{-deriv}} & M_{2322}^{4\text{-deriv}} \\ M_{1222}^{4\text{-deriv}} & M_{2322}^{4\text{-deriv}} & M_{2222}^{4\text{-deriv}} \end{pmatrix}_{(4,1)} \\ &= \begin{pmatrix} \frac{2}{21}(387 + 10C_0) & \frac{2}{15}(333 + 25C_0) & 0 \\ \frac{2}{15}(333 + 25C_0) & \frac{28}{75}(171 + 20C_0) & 0 \\ 0 & 0 & \frac{56}{3} \end{pmatrix}, \end{split} (5.8.14)$$

which is block-diagonal as expected and the new coefficients in terms of general Δ are given by:

$$M_{1223}^{4\text{-deriv}, t \text{ odd}}(\Delta, 1) = \frac{(\Delta - 2)}{30\Delta(2\Delta)!} \times \left[20 C_0 \Delta(\Delta + 3)((\Delta + 1)!)^2 + (\Delta^5 + \Delta^4 - 7\Delta^3 - \Delta^2 + 22\Delta - 16) \Delta!(\Delta + 3)! \right],$$

$$M_{2323}^{4\text{-deriv}, t \text{ odd}}(\Delta, 1) = \frac{(\Delta^2 + \Delta - 6)^2 (\Delta + 1)!}{300\Delta(2\Delta)!} \times \left[32 C_0 \Delta(\Delta + 1)! + (\Delta^6 + 3\Delta^5 - 8\Delta^4 - 21\Delta^3 + 89\Delta^2 + 100\Delta - 128) \Delta! \right],$$

$$M_{2222}^{4\text{-deriv}, t \text{ odd}}(\Delta, 1) = \frac{\Delta (\Delta^2 + \Delta - 6)^2 \Delta!(\Delta + 1)!}{3(2\Delta)!}.$$
(5.8.15)

The free theory and supergravity coefficients are given in (5.6.15). Note that the class B operators $\mathcal{O}_2\partial\mathcal{O}_2$ have zero supergravity OPE-coefficients but their free theory coefficients are non-zero. The free theory coefficients are relevant for the unmixing at $\mathcal{O}(a/c)$, for operators \mathcal{O}_{qq} they are given in (5.6.1) and the relevant coefficient at this weight is

$$A_{2222}^{\text{free}}(4,1) = \frac{2}{9}$$
, (5.8.16)

solving the $\mathcal{O}(1)$ unmixing equation gives the leading three-point function

$$\left(C_{4,1}^{(0)}\right)_{B^{\text{to}}} = \frac{\sqrt{2}}{3}$$
 (5.8.17)

We can now go on and solve the unmixing equations at $\mathcal{O}(a/c)$ and $\mathcal{O}(a)$, which gives the anomalous dimensions and three-point functions

$$\gamma_{4,1,i}^{4\text{-deriv}} = \left\{ 8(307+30C_0), \ \frac{392}{5}, \ 84 \right\} , \quad \mathbb{C}_{4,1}^{4\text{-deriv}} = \frac{1-3C_0}{27\sqrt{15}} \begin{pmatrix} 5\sqrt{14} & 10 & -\frac{\sqrt{5}}{2\sqrt{6}}c \\ -2\sqrt{14} & 14 & \frac{1}{\sqrt{30}}c \\ 0 & -c & 0 \end{pmatrix} ,$$

$$(5.8.18)$$

where we used the supergravity results (5.6.17). Note that c is unfixed and while there is no mixing between class A and class B operators at orders $\mathcal{O}(1)$, $\mathcal{O}(1/c)$, $\mathcal{O}(a/c)$, there could potentially be mixing at the level of the $\mathcal{O}(a)$ correction to the threepoint functions, depending on whether c is zero or not. This could be determined by studying higher-derivative corrections which mix with the four-derivative ones. There are three non-zero anomalous dimensions at (4, 1), two corresponding to class A and one to class B^{to} operators:

$$\left(\gamma_A^{4\text{-deriv}}\right)_{0,0}^{4,1} = 8(307 + 30C_0) , \quad \left(\gamma_A^{4\text{-deriv}}\right)_{1,0}^{4,1} = \frac{392}{5} , \quad \left(\gamma_B^{4\text{-deriv}}\right)_1^{4,1} = 84 , \quad (5.8.19)$$

which will be discussed in terms of the 4d spin below.

Let us look at one more example for odd t in order to understand the emergence of square roots in the anomalous dimensions, which indicates the breaking of the 4d conformal symmetry for higher-derivative corrections at higher charges. We discuss the first example where square roots appear, which is (5,2). This can be seen from the parametrisation in (5.5.3) and the illustration in figure 5.1 where square roots are expected to appear once operators with $i_A = 0$, $r_A = 0$ and both $i_A = 1$, $r_A = 0$ and $i_A = 0$, $r_A = 1$ contribute. This is the case when $t \ge 3$ and $p \ge 2$. In figure 5.1 this corresponds to (q_1, q_2) where all three black nodes in the grey highlighted area are present. The grey area contains all operators which acquire non-zero anomalous dimensions at $\mathcal{O}(a/c)$ when they are present in the spectrum at the weight (Δ, p) considered. The operator at position A is the one which has non-zero anomalous dimension also in supergravity, the two black nodes in the grey area connected by a vertical line correspond to the square roots. The square roots thus lift the residual degeneracy that was there in supergravity, where the black nodes connected by a vertical line correspond to the same anomalous dimension, which in the 1d case is zero. The anomalous dimensions of class B operators, denoted by white nodes, do not acquire square roots since there is no degeneracy to lift as class B operators decouple in the supergravity limit.

We will focus on the anomalous dimensions and neglect the three-point functions for the following example since the anomalous dimensions are the relevant objects to discuss the 4d conformal symmetry. The three-point functions can be obtained analogously to the examples above and we will do so for a few more examples with even t. Note that the $\mathcal{O}(a)$ corrections to the three-point functions after unmixing are generally non-zero for higher-derivative corrections, as can be seen from the examples considered above. This agrees with the observations in [14] that for α'^5 corrections to $\mathcal{N} = 4$ SYM correlators in the supergravity approximation, the $\mathcal{O}(\alpha'^5)$ corrections to the three-point functions are non-zero for some cases which are related to square roots in the anomalous dimensions. Whereas for $\alpha^{\prime 3}$ it was found in [13] that the corrections to the three-point functions are absent and all anomalous dimensions are rational and thus the 10d conformal symmetry is expected to be intact. See also the discussion in subsection 5.7.3. As a consequence, we would expect that the $\mathcal{O}(a)$ corrections to the three-point functions corresponding to class B operators are absent, since there are no square roots in these cases. This is e.g. supported by (5.8.18) where the entry 2222 corresponding to the class B operator is zero and we will check this in a few more cases below.

p=2 sector

At weight (5,2) there are four class A operators, $\mathcal{O}_1\partial^2\mathcal{O}_3$, $\mathcal{O}_2\mathcal{O}_4$, $\mathcal{O}_2\partial^2\mathcal{O}_2$, $\mathcal{O}_3\mathcal{O}_3$ and one class B operator, $\mathcal{O}_2\partial\mathcal{O}_3$. The class A operators will lead to square roots in the unmixed anomalous dimensions and class A and class B operators do not mix at the levels of supergravity and four-derivative correction anomalous dimensions. The conformal block analysis of the relevant correlators at $\mathcal{O}(a/c)$ then leads to the symmetric matrix:

$$\begin{split} \hat{M}_{5,2}^{4\text{-deriv}} &= \begin{pmatrix} M_{1313}^{4\text{-deriv}} & M_{1324}^{4\text{-deriv}} & M_{1322}^{4\text{-deriv}} & M_{1333}^{4\text{-deriv}} & M_{1323}^{4\text{-deriv}} \\ M_{2424}^{4\text{-deriv}} & M_{2422}^{4\text{-deriv}} & M_{2433}^{4\text{-deriv}} & M_{2423}^{4\text{-deriv}} \\ M_{2223}^{4\text{-deriv}} & M_{2223}^{4\text{-deriv}} & M_{2223}^{4\text{-deriv}} \\ M_{3333}^{4\text{-deriv}} & M_{3323}^{4\text{-deriv}} \\ M_{3333}^{4\text{-deriv}} & M_{3323}^{4\text{-deriv}} \\ M_{2323}^{4\text{-deriv}} & M_{3323}^{4\text{-deriv}} \\ M_{2323}^{4\text{-deriv}} & M_{3323}^{4\text{-deriv}} \\ \end{pmatrix}_{(5,2)} \end{split}$$

$$= \begin{pmatrix} \frac{2}{25}(774 + 35C_0) & \frac{6}{35}(416 + 35C_0) & \frac{4}{5}(942 + 35C_0) & \frac{36}{5}(90 + 7C_0) & 0 \\ & \frac{6}{245}(3910 + 441C_0) & \frac{36}{35}(90 + 7C_0) & \frac{4688}{245}(62 + 7C_0) & 0 \\ & & \frac{8}{135}(1955 + 56C_0) & \frac{8}{105}(1707 + 112C_0) & 0 \\ & & \frac{72}{245}(657 + 56C_0) & 0 \\ & & \frac{672}{25} \end{pmatrix}, \end{split}$$

$$(5.8.20)$$

we spell out the coefficients for general Δ in appendix I. Furthermore, since we have not studied this example in the supergravity section 5.6, we also give the free theory and supergravity coefficients and discuss the mixing problem in the supergravity limit, which is necessary to solve the unmixing at $\mathcal{O}(a/c)$, in the appendix.

Let us solve the unmixing equations at $\mathcal{O}(a/c)$ to get the anomalous dimensions after unmixing:

$$\gamma_{5,2,i}^{4\text{-deriv}} = \left\{ \frac{32}{3} (545 + 51C_0), \ \frac{8}{15} (415 - \sqrt{2881}), \ \frac{8}{15} (415 + \sqrt{2881}), \ \frac{1008}{5} \right\} \ . \ (5.8.21)$$

We then label the non-zero anomalous dimensions as follows:

$$\left(\gamma_A^{4\text{-deriv}}\right)_{0,0}^{5,2} = \frac{32}{3} \left(545 + 51C_0\right), \quad \left(\gamma_A^{4\text{-deriv}}\right)_{1,0}^{5,2} = \frac{8}{15} \left(415 - \sqrt{2881}\right),$$

$$\left(\gamma_A^{4\text{-deriv}}\right)_{0,1}^{5,2} = \frac{8}{15} \left(415 + \sqrt{2881}\right) , \quad \left(\gamma_{B^{\text{to}}}^{4\text{-deriv}}\right)_1^{5,2} = \frac{1008}{5} .$$
 (5.8.22)

We will give formulas for general (Δ, p) and analyse them from a 4d perspective below. Note that for the first time, square roots appear in the anomalous dimensions and this will be the case for any (Δ, p) with $p \ge 2$ and $t \ge 3$, where t is odd. These square roots resolve a residual degeneracy from supergravity, which in the 1d case is not obvious since it is a zero-degeneracy. There are no square roots for even t, which is expected because there are no supergravity contributions for even t and thus no degeneracy to resolve.

Continuing to solve the mixing problem weight by weight for many (Δ, p) for t odd, there will always be one rational anomalous dimension corresponding to the operator with $i_A = 0$, $r_A = 0$ and two square roots if both $i_A = 1$, $r_A = 0$ and $i_A = 0$, $r_A = 1$ operators are present (when both $t \ge 3$ and $p \ge 2$). This collapses to one rational anomalous dimension when only one of the two operators is present and to zero when none of the two is present. Besides class A, for class B^{to} there is one non-zero rational anomalous dimension when an operator with $i_{B^{\text{to}}} = 1$ is present (when $t \ge 3$ and $p \ge 1$). Next, let us study a few examples with t even before presenting the general formulas for all non-zero anomalous dimensions at $\mathcal{O}(a/c)$.

5.8.2 Unmixing for Even t

Let us study a few examples of unmixing for even t where only class B^{te} operators contribute to the double-trace spectrum. To get a list of exchanged operators at each weight, see (5.5.5) and figure 5.2. Class B^{to} operators only start contributing from p = 1, and they split into two groups, operators with even or odd $r_{B^{\text{te}}}$ respectively. These are illustrated in figure 5.2 with black and white nodes for even and odd $r_{B^{\text{te}}}$ respectively. The operators with (q_1, q_2) which are in the grey area are the ones which obtain non-zero anomalous dimensions at $\mathcal{O}(a/c)$. Let us start by looking at some examples with p = 1.

p=1 sector

At weight (3,1) there is only one operator present, $\mathcal{O}_1 \partial \mathcal{O}_2$ and performing the conformal block expansion we get the free (see (5.6.12)) and four-derivative coefficients:

$$A_{1212}^{(0)}(3,1) = \frac{2}{15} ,$$

$$M_{1212}^{4\text{-deriv}}(3,1) = \frac{(\Delta^3 - 5\Delta + 4)(\Delta - 1)!(\Delta + 2)!}{3(2\Delta)!}|_{\Delta = 3} = \frac{16}{9} , \qquad (5.8.23)$$

solving the unmixing equations (where there are no contributions from supergravity) we get the anomalous dimension

$$\left(\gamma_{B^{\text{te}}}^{4\text{-deriv}}\right)_{1,0}^{3,1} = \frac{40}{3}$$
 (5.8.24)

Next, let us obtain the three-point functions after unmixing to check whether the $\mathcal{O}(a)$ correction is zero as we would expect for class *B* operators. Solving the unmixing equation at $\mathcal{O}(1)$ and plugging the result into the equation at $\mathcal{O}(a)$ indeed gives:

$$C_{3,1}^{(0)} = \sqrt{\frac{2}{15}} , \quad C_{3,1}^{4\text{-deriv}} = 0 .$$
 (5.8.25)

At next highest weight (5, 1) there are two operators that contribute, $\mathcal{O}_1 \partial^2 \mathcal{O}_2$ and $\mathcal{O}_2 \partial \mathcal{O}_3$ and performing the conformal block expansion of the corresponding correlators gives

$$\hat{A}_{5,1}^{(0)} = \begin{pmatrix} A_{1212}^{(0)} & 0 \\ 0 & A_{2323}^{(0)} \end{pmatrix}_{(5,1)} = \begin{pmatrix} \frac{1}{42} & 0 \\ 0 & \frac{9}{175} \end{pmatrix} ,$$
$$\hat{M}_{5,1}^{4\text{-deriv}} = \begin{pmatrix} M_{1212}^{4\text{-deriv}} & M_{1223}^{4\text{-deriv}} \\ M_{1223}^{4\text{-deriv}} & M_{2323}^{4\text{-deriv}} \end{pmatrix}_{(5,1)} = \begin{pmatrix} \frac{52}{45} & \frac{104}{25} \\ \frac{104}{25} & \frac{1872}{125} \end{pmatrix} , \qquad (5.8.26)$$

where the new coefficients in terms of general odd Δ are

$$M_{2323}^{4\text{-deriv}, t \text{ even}}(\Delta, 1) = \frac{\left(\Delta^2 + \Delta - 4\right) \left(\Delta^3 - 13\Delta + 12\right)^2 (\Delta - 2)! (\Delta + 2)!}{75(2\Delta)!} ,$$

$$M_{1223}^{4\text{-deriv}, t \text{ even}}(\Delta, 1) = \frac{\left(\Delta - 3\right) (\Delta - 1) (\Delta + 4) \left(\Delta^2 + \Delta - 4\right) (\Delta - 1)! (\Delta + 2)!}{15(2\Delta)!} .$$
(5.8.27)

Solving the unmixing equations in matrix form for this case gives the anomalous

dimension

$$\left(\gamma_{B^{\text{teriv}}}^{4\text{-deriv}}\right)_{1,0}^{5,1} = \frac{5096}{15}$$
 (5.8.28)

For this case again, as for the whole p = 1 sector, there is exactly one non-zero anomalous dimension, which corresponds to the operator with $i_{B^{\text{te}}} = 1$, $r_{B^{\text{te}}} = 0$. Going to $p \ge 2$, there will be two non-zero anomalous dimensions, the additional one being for the operator with $i_{B^{\text{te}}} = 1$, $r_{B^{\text{te}}} = 1$. Furthermore, solving the equations (5.5.13) at order $\mathcal{O}(1)$ and $\mathcal{O}(a/c)$ (where $\gamma^{\text{sugra}} = 0$) and plugging the result into $\mathcal{O}(a)$ we get the three-point functions:

$$\mathbb{C}_{5,1}^{(0)} = \begin{pmatrix} \frac{1}{7\sqrt{6}} & \frac{3\sqrt{6}}{35} \\ \frac{1}{7} & -\frac{3}{35} \end{pmatrix}, \quad \mathbb{C}_{5,1}^{4\text{-deriv}} = \begin{pmatrix} \frac{3}{355} \left(-97 + 35\sqrt{6}\right) c & c \\ -\frac{2}{355} \left(105 + 58\sqrt{6}\right) c & \frac{1}{426} \left(-210 + 97\sqrt{6}\right) c \end{pmatrix},$$

$$(5.8.29)$$

where c is unfixed. However, we would expect c to be zero based on the discussions above, since for even t all operators acquire rational anomalous dimensions. This could be checked by considering higher-derivative corrections. We have now seen in a few examples that the $\mathcal{O}(a)$ corrections to the three-point functions are zero (or unfixed) for class B operators, as expected. For the following examples we will focus on the anomalous dimensions only, and the corrections to the three-point functions can be obtained analogously to the examples above. Let us now consider an example in the p = 2 sector where we expect two non-zero anomalous dimensions.

p=2 sector

Consider the simplest case (4,2), where there are two operators in the spectrum $\mathcal{O}_1\partial\mathcal{O}_3$ and $\mathcal{O}_2\mathcal{O}_3$. Conformal block expansion of the appropriate correlators gives the coefficients:

$$\hat{A}_{4,2}^{(0)} = \begin{pmatrix} A_{1313}^{(0)} & 0\\ 0 & A_{2323}^{(0)} \end{pmatrix}_{(4,2)} = \begin{pmatrix} \frac{3}{28} & 0\\ 0 & \frac{2}{21} \end{pmatrix}$$

$$\hat{M}_{4,2}^{4\text{-deriv}} = \begin{pmatrix} M_{1313}^{4\text{-deriv}} & M_{1323}^{4\text{-deriv}} \\ M_{1323}^{4\text{-deriv}} & M_{2323}^{4\text{-deriv}} \end{pmatrix}_{(4,2)} = \begin{pmatrix} \frac{21}{5} & 0 \\ 0 & \frac{224}{45} \end{pmatrix} .$$
(5.8.30)

Note that in this case also $\hat{M}_{4,2}^{4\text{-deriv}}$ is a diagonal matrix, this is because operators with even or odd $r_{B^{\text{te}}}$ do not mix at the level of free theory and four-derivative corrections, going to higher (Δ, p) will give block-diagonal matrices where the mixing problem could be solved independently for even or odd $r_{B^{\text{te}}}$.

The $\mathcal{O}(a/c)$ conformal block coefficients for general even Δ are:

$$M_{1313}^{4\text{-deriv, t even}}(\Delta, 2) = \frac{(\Delta - 2)^3 (\Delta + 3) (\Delta^2 + \Delta - 8) (\Delta - 3)! (\Delta + 3)!}{20(2\Delta)!} ,$$

$$M_{2323}^{4\text{-deriv, t even}}(\Delta, 2) = \frac{4 (\Delta^2 + \Delta - 6)^2 (\Delta^2 + \Delta - 4) \Delta! (\Delta + 1)!}{45\Delta(2\Delta)!} .$$
(5.8.31)

After unmixing, there are indeed two non-zero rational anomalous dimensions as expected:

$$\gamma_{4,2,i}^{4\text{-deriv}} = \left\{ \frac{196}{5}, \frac{784}{15} \right\} ,$$
 (5.8.32)

which we label by $\left(\gamma_{B^{\text{te}}}^{4\text{-deriv}}\right)_{1,0}^{4,2} = \frac{196}{5}$ and $\left(\gamma_{B^{\text{te}}}^{4\text{-deriv}}\right)_{1,1}^{4,2} = \frac{784}{15}$ and discuss further below, see subsection 5.8.3. The two operators with non-zero anomalous dimensions correspond to the two nodes in the grey highlighted rectangle in figure 5.2. We present an additional example for even t, (6,2) where more than one operator contributes for even and odd $r_{B^{\text{te}}}$ each, in appendix I.

One can solve the mixing problem for any (Δ, p) analogous to the given examples. Solving it for many cases, we conjecture general formulas and we spell them out and discuss them from a 4d perspective in the following subsection.

5.8.3 Anomalous Dimensions after Unmixing

We present general formulas for the anomalous dimensions at $\mathcal{O}(a/c)$ for any (Δ, p) for odd and even t. From the examples discussed above, we can see that there are three different non-zero class A and one class B^{to} anomalous dimensions for odd t and two different class B^{te} ones for even t (which agrees with the predictions from the 4d spin (5.5.6)). Solving for the anomalous dimensions after unmixing for many weights up to high (Δ, p) , we can conjecture general formulas. We start by presenting the results for odd t which are interesting because the breaking of the 4d conformal symmetry becomes obvious due to the appearance of square roots. Finally we also present the conjectured formulas for even t.

Anomalous dimensions for odd t

Recall that the four-derivative interaction corresponds to correlators with a conformal block expansion that is truncated to 4d spin-2. This can be understood in terms of the effective 4d spin given in (5.5.6). Firstly, analysing the 4d spin-0 sector it is easy to see that $l_{4d} = 2(i_A + r_A) = 0$ can only be satisfied by class A operators with quantum numbers $i_A = r_A = 0$, ($i_{B^{to}}$ only has values $i_{B^{to}} = 1, \ldots$ and thus does not contribute at 4d spin-0). The anomalous dimensions are

$$\left(\gamma^{A}\right)_{0,0}^{\Delta,p} = \delta^{(2)} \left(\frac{2}{9}\left(1 - 6C_{0}\right) + \frac{1}{18}\left(1 + 12C_{0}\right)\delta^{(2)} + \frac{5}{12}\left(\delta^{(2)}\right)^{2} + \frac{1}{9}\left(1 + 12C_{0}\right)\delta^{(y)}\right),$$
(5.8.33)

where $\delta^{(2)}$ and $\delta^{(y)}$ are the eigenvalues of the differential operators $\Delta^{(2)}$, \mathcal{D}_y in (5.2.15) acting on conformal blocks

$$\delta^{(2)} = (\Delta - p)(\Delta + p + 1) , \quad \delta^{(y)} = p(p + 1) . \tag{5.8.34}$$

For the 4d spin-2 sector $l_{4d} = 2(i_A + r_A + i_{B^{to}}) = 2$ there are three possible non-zero anomalous dimensions with quantum numbers $i_A = 1$, $r_A = 0$ and $i_A = 0$, $r_A = 1$ for class A and $i_A = 0$, $r_A = 0$, $i_{B^{to}} = 1$ for class B, this is the first case where class B operators play a role. These anomalous dimensions are given by:

$$\left(\gamma^{A}\right)_{1,0}^{\Delta,p} = \frac{1}{180} \,\delta^{(2)} \left(4 - 2\,\delta^{(2)} + 3\,\left(\delta^{(2)}\right)^{2} - 4\,\delta^{(y)} - 4\,\sqrt{8\,p^{3} + 4\,p^{4} - 4\,p^{2}\Delta(1+\Delta) - 4\,p\,(1+\Delta+\Delta^{2}) + (-1+2\,\Delta+2\,\Delta^{2})^{2}}\right),$$

$$(5.8.35)$$

$$(\gamma^{A})_{0,1}^{\Delta,p} = \frac{1}{180} \,\delta^{(2)} \Big(4 - 2\,\delta^{(2)} + 3\,\left(\delta^{(2)}\right)^{2} - 4\,\delta^{(y)} + 4\,\sqrt{8\,p^{3} + 4\,p^{4} - 4\,p^{2}\Delta(1+\Delta) - 4\,p\,(1+\Delta+\Delta^{2}) + (-1+2\,\Delta+2\,\Delta^{2})^{2}} \Big) ,$$
(5.8.36)

$$\left(\gamma^{B^{\text{to}}}\right)_{1}^{\Delta,p} = \frac{1}{60} \,\delta^{(2)} \left(-2\,\delta^{(2)} + \left(\delta^{(2)}\right)^{2} - 4\,\delta^{(y)}\right) \,. \tag{5.8.37}$$

The square roots resolve the degeneracy at supergravity, where in the 1d case several anomalous dimensions are zero (while in the 4d/10d case there is a non-zero degeneracy). The operators with the same anomalous dimensions in supergravity correspond to the black nodes (class A operators) connected by vertical lines in figure 5.1. The operators highlighted by a grey rectangle are the ones acquiring nonzero anomalous dimensions at the order of four derivatives. The two black nodes which are connected by a vertical line correspond to the operators with anomalous dimensions with square roots (5.8.35), (5.8.36) while the white node represents the operators with anomalous dimension (5.8.37).

Anomalous dimensions for even t

For even t, operators only start to contribute at 4d spin-2 and we get two different non-zero anomalous dimensions, labelled as $(\gamma^{B^{\text{te}}})_{i_{B^{\text{te}}},r_{B^{\text{te}}}}^{\Delta,p}$ with quantum numbers $i_{B^{\text{te}}} = 1$, $r_{B^{\text{te}}} = 0$ and $i_{B^{\text{te}}} = 1$, $r_{B^{\text{te}}} = 1$:

$$\left(\gamma^{B^{\text{te}}}\right)_{1,0}^{\Delta,p} = \frac{1}{60} \left(\delta^{(2)}\right)^2 \left(\delta^{(2)} - 2\right) , \qquad (5.8.38)$$

$$\left(\gamma^{B^{\text{te}}}\right)_{1,1}^{\Delta,p} = \frac{1}{60} \left(\delta^{(2)}\right)^2 \left(\delta^{(2)} + 2\right) . \tag{5.8.39}$$

The operators at even t are illustrated in figure 5.2 and the two operators acquiring non-zero anomalous dimensions are highlighted by a grey rectangle, where the op-

erator with anomalous dimension (5.8.38) is represented by the black node and the one corresponding to (5.8.39) by the white node.

Note that there are some relations between these anomalous dimensions, in particular the eigenvalues of Casimir operators in the anomalous dimensions, and the Casimir operators which act on 4d conformal objects to uplift the four-derivative corrections to higher charges in subsection 5.7.3. In particular (5.8.38) is proportional to the eigenvalue of $\Delta_{\text{spin-2}}^{(4)}$ in (5.7.22). The anomalous dimensions for odd t corresponding to 4d spin-0 and spin-2 are also partially given in terms of eigenvalues of Casimirs. However, there is no exact correspondence to eigenvalues of $\Delta_{\text{spin-0}}^{(4)}$ or $\Delta_{\text{spin-2}}^{(4)}$ due to the square roots and extra dependence on $\delta^{(y)}$. This also indicates a breaking of the 4d conformal symmetry.

We have now studied the mixing problem for four-derivative corrections in detail and have obtained general formulas for all $\mathcal{O}(a/c)$ anomalous dimensions after unmixing. These anomalous dimensions can be organised according to a 4d effective spin inspired by the 4d conformal symmetry and they indicate a breaking of the symmetry for higher-derivative corrections due to the appearance of square roots. We will conclude this chapter in the next section and discuss some interesting open questions for future research.

5.9 Conclusions and Future Directions

In this chapter we investigate holographic correlators in $AdS_2 \times S^2$. We start with a review of the hidden 10d conformal symmetry discovered in [41] where it is conjectured that all half-BPS four-point correlators described by tree-level supergravity can be obtained from one single 10d conformally invariant object. This conjecture is true for the supergravity approximation and also extends to free theory, as well as loop corrections (which we do not discuss in this thesis). However, it is generally broken for higher-derivative corrections as was confirmed in [14] from a study of anomalous dimensions after unmixing. An appearance of square roots in these

Chapter 5. $AdS_2 \times S^2$ Correlators: Effective Action and 4d Conformal 186 Symmetry

results indicates the breaking of the symmetry, however there are still many rational anomalous dimensions which suggests that some of the symmetry is conserved even for higher-derivative corrections. The hidden conformal symmetry arises when the corresponding flat space superamplitude is conformally invariant, hence when its coupling, i.e. $G_N \delta^{16}(Q)$ in the $\mathrm{AdS}_5 \times \mathrm{S}^5$ case, is dimensionless. Therefore, only specific four-point correlators enjoy this conformal symmetry and only on backgrounds which are conformally flat, such as $\mathrm{AdS}_q \times \mathrm{S}^q$. We have explained that in the supergravity approximation the four-point half-BPS correlator itself enjoys the 10d conformal symmetry, while for free theory the correlator has to be rescaled by acting on it with an eighth-order differential operator derived from the superconformal Casimir. This yields the correlator of superdescendants which exhibits 10d conformal symmetry.

In addition to studying 1d correlators in the context of the 4d hidden conformal symmetry we also derive four-point half-BPS correlators from a 4d scalar effective action in $AdS_2 \times S^2$ similar to the one introduced in chapter 3. While in $AdS_5 \times S^5$ the effective action only describes higher-derivative corrections, in $AdS_2 \times S^2$ the effective action describes both, supergravity and higher-derivative corrections. Note that we propose the existence of this effective action and deduce the consequences without proving it. We then compare the results to those from the 4d conformal symmetry where possible. Throughout this chapter we consider both these approaches and show that they nicely complement each other.

We discuss the free disconnected correlators and show that they exhibit a 4d conformal symmetry. Recall that for free theory one has to consider the correlator of descendants which is obtained by acting with $\Delta^{(2)}$ on the correlators. This yields a generating function which contains all higher-charge free theory correlators.

Supergravity correlators in $AdS_2 \times S^2$ are especially interesting because they exhibit 4d conformal symmetry and can also be deduced from the proposed 4d effective action. We start by deriving the lowest-charge correlator from crossing symmetry and small x behaviour alone and then show that one can obtain all higher-charge tree-level correlators by uplifting it to 4d and expanding in the internal variables. We derive a general formula for all four-point supergravity correlators and find exact agreement when deriving the same correlators from the 4d effective action (up to an overall unfixed coefficient), which suggests that the proposed 4d effective field theory indeed generates all tree-level half-BPS four-point correlators.

The first hint towards a 10d conformal symmetry was found in [144] when solving the mixing problem for anomalous dimensions in the supergravity limit where a remarkably simple structure was uncovered. To show that a similar structure (and even simpler due to the absence of spin in 1d CFTs) is satisfied for $AdS_2 \times S^2$ we analyse the double-trace spectrum in the conformal block decomposition of the fourpoint half-BPS correlators. We then explain how to solve the mixing problem for supergravity as well as higher-derivative corrections. Further, we show that the supergravity anomalous dimensions after unmixing are indeed given by a very simple formula, namely the eigenvalue $\delta^{(2)}$ of $\Delta^{(2)}$ acting on the 1d superconformal blocks. This simple structure can be explained from an effective 4d spin. Therefore, we have seen that 1d correlators in free theory and in the supergravity approximation indeed exhibit 4d conformal symmetry.

Finally, we investigate higher-derivative corrections. We start by deriving general higher-derivative corrections with lowest charge from crossing symmetry before we focus on the four-derivative corrections. We derive all four-derivative corrections to the half-BPS correlators with any charge configuration from the 4d effective field theory. Using these results, we go on to discuss the breaking of the higher-dimensional conformal symmetry. We propose that some of the correlators can be reconstructed by rescaling the higher-derivative four-point correlators by acting with an inverse fourth-order differential operator to obtain an object invariant under 4d conformal symmetry. In this way one can reproduce an infinite set of four-derivative corrections with specific charges, however the symmetry is broken for general charge configurations. This breaking can be anticipated because when dimensionally reducing interaction terms with covariant AdS×S derivatives on the sphere one gets

several terms with different numbers of derivatives in AdS which then cannot all be rescaled simultaneously. This strongly suggests that the α'^3 corrections in AdS₅×S⁵, which are described by zero-derivative interactions, have 10d conformal symmetry, which is in agreement with the rational anomalous dimensions found in [13].

Considering the four-derivative corrections we obtain from the proposed 4d effective action, we perform a conformal block analysis and solve the mixing problem for anomalous dimensions at $\mathcal{O}(a/c)$. We obtain general formulas for all anomalous dimensions and similar to [14] we find that a part of these anomalous dimensions contain square roots, whereas there are also many rational anomalous dimensions. Again we can explain the structure of these anomalous dimensions from an effective 4d spin. The fact that there are square roots in the unmixed anomalous dimensions indicates that the 4d conformal symmetry is broken for higher-derivative corrections. However, there are also many operators obtaining rational anomalous dimensions and this suggests that a part of the 4d conformal symmetry structure remains. This agrees with the findings above that a subset of correlators can still be constructed from the hidden conformal symmetry.

To summarise, both, the higher-dimensional conformal symmetry and the higherdimensional scalar effective field theory are very powerful approaches which nicely complement each other. While the hidden conformal symmetry describes all free theory and supergravity correlators (as well as loop corrections), it is generally broken for higher-derivative corrections. On the other hand, the scalar effective action approach generates all higher-derivative corrections for any number of derivatives and any charge configuration, but does not describe free theory and loop corrections. Whereas the effective action in $AdS_5 \times S^5$ only describes higher-derivative corrections, the effective action in $AdS_2 \times S^2$ also describes supergravity.

Future directions

There are a number of interesting open questions that arise form this research.

- We have seen that the hidden conformal symmetry and the higher-dimensional effective action approach nicely complement each other and that both are very powerful in obtaining four-point half-BPS correlators at different orders in *a* and 1/*c*. It would be very interesting to understand the breaking of the hidden conformal symmetry for higher-derivative corrections more precisely. It would be interesting to identify specific pieces of the integrands of Witten diagrams derived from the effective action that cause the breaking of the symmetry. It is not straightforward to identify these contributions from (5.7.12) and one idea might be to perform a Weyl transformation of the derivative terms in the effective action from flat space to AdS×S to see whether this could give more insight.
- Another very interesting research direction would be to study implications for 4d black holes. As mentioned before, $AdS_2 \times S^2$ is the near-horizon geometry of extremal black holes in 4d. Fixing the coefficients in the effective action would specify the underlying theory of quantum gravity and it would be interesting to see whether these coefficients could be constrained from the weak gravity conjecture [156, 157].
- It would also be interesting to consider higher-loop corrections. These can be constrained from the higher-dimensional conformal symmetry and we leave a discussion of this to [44] (also see [41]).
- Extending the conformal block analysis to higher-derivative corrections with more than four derivatives is another possible extension of our research. It would be interesting to compute the anomalous dimensions after unmixing and check that the results agree with the 4d effective spin conjecture in (5.5.6). Furthermore, one could consider constructing part of the correlators from the 4d conformal symmetry by acting with inverse differential operators on the correlators analogous to the discussion in subsection 5.7.3, where the power of the differential operators depends on the number of derivatives.

Chapter 6

Conclusion

In this thesis we have studied holographic correlators in three different examples of the AdS/CFT correspondence. In chapter 2 we started with a review of important concepts of conformal field theories, including correlators, OPE expansion and the conformal bootstrap. Furthermore, we have briefly reviewed the AdS/CFT correspondence and holographic correlators including higher-derivative corrections.

The first case we have studied was the classical canonical example of the AdS/CFT correspondence, the duality between string theory in $AdS_5 \times S^5$ and $\mathcal{N} = 4$ SYM in the boundary. The goal is to generalise the flat space Virasoro-Shapiro amplitude, describing stringy corrections to the supergravity approximation, to curved backgrounds. We have proposed a 10d scalar effective action in $AdS_5 \times S^5$ describing α' corrections to the correlators. From there we have described a general algorithm to compute all four-point half-BPS correlators described by tree-level string theory to any order in α' by evaluating new 10d Witten diagrams which manifestly include AdS_5 and S^5 coordinates. We have justified the existence of this effective field theory by reproducing known results for α'^3 and α'^5 and have then made new predictions for α'^6 and α'^7 . The coefficients of these interaction terms are fixed by comparing to the flat space VS amplitude. There are ambiguities in the AdS×S effective action which vanish in the flat space limit and can therefore not be fixed by our approach. However, most of the ambiguities at the orders considered in this thesis can be

determined by comparing to results obtained from localisation.

Next, in chapter 4 we have considered holographic correlators in $AdS_7 \times S^4$ which is another canonical example of the AdS/CFT correspondence but much less understood since the bulk theory is not known beyond the supergravity approximation and the boundary theory is non-Lagrangian. Our goal was to better understand the worldvolume theory of M5-branes, the 6d (2,0) theory, which is dual to M-theory in $AdS_7 \times S^4$. To study M-theory away from the supergravity approximation we consider correlators in the 6d (2,0) theory away from the strict large-c limit. Since the 6d (2,0) theory is non-perturbative we have approached this through conformal bootstrap methods. In particular, we have derived recursion relations for anomalous dimensions of double-trace operators in the conformal block expansion of four-point functions of primaries of the stress tensor multiplet. To do this, we have expanded the crossing equation in 1/c and taken the light-cone limit of the conformal cross-ratios to isolate the contributions from anomalous dimensions. Moreover, we have made use of orthonormality relations of the hypergeometric functions in the conformal blocks and have truncated the conformal block expansion to spin L. We have derived these recursion relations for a bosonic toy 6d CFT as well as for the supersymmetric 6d (2,0) theory. Further, we have described an algorithm for solving the recursion relations for the anomalous dimensions at any spin-truncation and for any twist. Finally, the anomalous dimensions encode the higher-derivative corrections to the low-energy effective action of M-theory. In particular, the large-twist behaviour of the anomalous dimensions of a specific spin-L truncated solution to the crossing equation gives the number of derivatives of the corresponding interaction vertex while the coefficients of the terms in the low-energy effective action cannot be fixed from our approach.

The third case we have investigated are holographic correlators in $AdS_2 \times S^2$, where quantum gravity on this background is dual to a one-dimensional superconformal field theory with SU(1,1|2) symmetry. This case is much less well understood but very interesting due to its relation to black hole physics. We have approached correlators in this example of the AdS/CFT correspondence from different points of view, the higher-dimensional conformal symmetry first observed for $\mathcal{N} = 4$ SYM and the higher-dimensional scalar effective action proposed in a previous chapter. The hidden conformal symmetry of correlators arises when the corresponding flat space superamplitude is connected to a dimensionless coupling and is thus conformally invariant. It is thus only satisfied for specific four-point correlators and moreover, requires a conformally flat background. We have seen that 1d correlators in free theory and in the supergravity approximation have 4d conformal symmetry, where one has to act with a differential operator $\Delta^{(2)}$ on the free theory correlators to obtain a 4d conformally invariant object. In these cases, all higher-charge correlators can be obtained from the lowest-charge correlator alone using the hidden conformal symmetry and we have shown that in the supergravity limit these results agree with the correlators derived from the 4d effective action by evaluating $AdS_2 \times S^2$ Witten diagrams. That the supergravity correlators satisfy 4d conformal symmetry was also confirmed through the analysis of anomalous dimensions of double-trace operators in the spectrum of the correlators after unmixing, which show the expected simple structure $\delta^{(2)}$.

We have then discussed higher-derivative corrections and specifically obtained all four-derivative corrections from the proposed 4d effective field theory. The higherdimensional conformal symmetry is generally broken in this case but through rescaling the correlators by acting on them with inverse powers of differential operators we can reconstruct an infinite tower of four-derivative corrections with specific charges. We have also solved the mixing problem for anomalous dimensions of double-trace operators in the spectrum of the four-derivative corrections and have obtained general formulas. While most of the anomalous dimensions are rational, there are operators which acquire anomalous dimensions with square roots and this indicates the breaking of the 4d conformal symmetry. To conclude, the hidden conformal symmetry conjecture is valid for free theory and supergravity and is generally broken for higher-derivative corrections whereas the scalar effective field theory approach generates all higher-derivative corrections but does not describe free theory or supergravity (except in the 1d case). Hence, using both of these approaches together puts very powerful constraints on the four-point correlators of half-BPS operators in conformally flat backgrounds.

Detailed conclusions including discussions of open problems are given at the end of each chapter. There are many interesting future directions arising from the research in this thesis, such as for example proving the existence of the scalar effective action proposed in chapter 3, or deriving it directly from the CFT side of the duality. It would also be interesting to understand systematically how to fix the ambiguities and how to resum the terms in the effective action. Furthermore, it would be interesting to extend this approach to different conformally flat backgrounds, to higher-point correlators or to loop corrections. Following on from chapter 4 an interesting direction would be to fix the coefficients of the higher-derivative corrections to the M-theory low-energy effective action using the chiral algebra conjecture [132]. Related to this, it would be interesting to derive higher-charge correlators and compute the corresponding anomalous dimensions. Based on chapter 5 it would be very interesting to explore the implications for higher-derivative corrections in black hole backgrounds, as well as to extend the study to loop corrections or to interactions with more derivatives. An interesting direction to pursue, combining both chapter 3 and chapter 5, would be to study the relation between the higher-dimensional scalar effective action and the higher-dimensional conformal symmetry further and understand the breaking of the conformal symmetry for higher-derivative corrections more precisely.

Appendix A

Mellin Space

In this appendix we will review a few important concepts for the Mellin space formalism. In [57] the author proposed a representation of conformal correlators that makes their duality to scattering amplitudes more apparent, Mellin space representation. Mellin space was first brought into the context of scattering amplitudes in [158, 159], where a duality between the Mellin amplitude and flat space scattering amplitudes as a function of Mandelstam invariants was pointed out. The flat space correlator of four primary scalar operators can be written as

$$\langle \mathcal{O}_1(x_1)\mathcal{O}_2(x_2)\mathcal{O}_3(x_3)\mathcal{O}_4(x_4)\rangle = \frac{\mathcal{N}}{(2\pi i)^2} \int d\delta_{ij} \prod_{i\leq j}^n \frac{\Gamma(\delta_{ij})}{\left(x_{ij}^2\right)^{\delta_{ij}}} M(\delta_{ij}) , \qquad (A.0.1)$$

with some normalisation \mathcal{N} . The integration variables satisfy the constraints

$$\sum_{i \le j} \delta_{ij} = \Delta_i \quad , \tag{A.0.2}$$

and the integrand is conformal with scaling dimension Δ_i at the point x_i . $M(\delta_{ij})$ is the Mellin amplitude and the integration contour runs parallel to the imaginary axis with $\operatorname{Re}(\delta_{ij}) > 0$.

In [57], the author then proposed that the Mellin amplitude is the scattering amplitude also in AdS spacetime (see (2.3.14)) and we will discuss this further in the following.

Contact Witten diagrams and Mellin space

In this subsection we include additional information about contact diagrams in Mellin space and their relations to the position space versions, which is useful in the context of chapter 3. Note that we do not discuss Mellin space in the context of 1d CFTs in chapter 5 because in 1d the Mandelstam variables are not independent (since u = 0) and thus the Mellin transforms are not unique (see also [160]).

In [57] the author computes the Mellin amplitude $M(\delta_{ij})$ for different Witten diagrams and finds that they resemble to flat space scattering amplitudes. Specifically contact Witten diagrams, which are the relevant cases for our considerations, give polynomial Mellin amplitudes which agree with flat space scattering amplitudes as functions of the Mandelstam invariants. The minimal contact diagram with no covariant derivatives always has constant Mellin amplitude¹ whereas derivative contact terms give polynomial Mellin amplitudes. More precisely a AdS contact interaction with 2k covariant derivatives corresponds to a Mellin amplitude which is a polynomial of degree k.

As we have seen in subsection 2.3.1 the contact Witten diagrams for interactions with no derivatives, which are described by *D*-functions, can be written in Mellin space as (see [57]):

$$D_{p_1 p_2 p_3 p_4}^{(d)}(X_i) = \mathcal{N}_{p_i}^{\text{AdS}_{d+1}} \times \int \frac{d\delta_{ij}}{(2\pi i)^2} \prod_{i < j} \frac{\Gamma(\delta_{ij})}{(X_i \cdot X_j)^{\delta_{ij}}} , \quad \text{with } \sum_{i < j} \delta_{ij} = p_j , \quad (A.0.3)$$

where the normalisation is given in (2.3.15).

The constraints $\sum_i \delta_{ij} = p_j$ can be solved most naturally by

$$\delta_{ij} = \frac{1}{2} (p_i + p_j - s_{ij}) , \qquad (A.0.4)$$

where $s_{12} = s_{34} = s$, $s_{14} = s_{23} = t$, $s_{13} = s_{24} = u = p_1 + p_2 + p_3 + p_4 - s - t$ which can be interpreted as kinematic invariants of an auxiliary momentum space k_i with

¹In any dimension d there is a contact diagram with constant Mellin transform, namely the one with $\Delta = d$ for $D_{\Delta\Delta\Delta\Delta}$. In 4d this special case is D_{4444} which corresponds to the α'^3 correction, while in 1d it is D_{1111} corresponding to supergravity.

 $s_{ij} = -(k_i + k_j)^2$. We can use this to rewrite the normalised *D*-functions (2.3.16) in terms of \overline{D} -functions with a simple relation to the Mellin transform as:

$$P' \times D_{p_1 p_2 p_3 p_4}(X_i) = \int \frac{ds \, dt}{(2\pi i)^2} u^{\frac{s}{2}} v^{\frac{t}{2}} \prod_{i < j} \Gamma(\delta_{ij})$$
$$= u^{\frac{1}{2}(p_1 + p_2)} v^{\frac{1}{2}(p_2 + p_3)} \bar{D}_{p_1 p_2 p_3 p_4}(u, v) , \qquad (A.0.5)$$

where the prefactor is

$$P' = \frac{(X_1.X_2)^{\frac{1}{2}(p_1+p_2)}(X_1.X_4)^{\frac{1}{2}(p_1+p_4)}(X_2.X_3)^{\frac{1}{2}(p_2+p_3)}(X_3.X_4)^{\frac{1}{2}(p_3+p_4)}}{(X_1.X_3)^{\frac{1}{2}(p_2+p_4)}(X_2.X_4)^{\frac{1}{2}(p_1+p_3)}}$$
(A.0.6)

and to go from the normalised D to \overline{D} use (2.3.16) and (2.3.17), where $x_{ij}^2 = -2X_i \cdot X_j$. Thus, one can easily convert between position space in terms of \overline{D} -functions and Mellin space. Note that in chapter 3 we give all correlators in Mellin space, nevertheless we can directly expand the 10d Witten diagrams in terms of D- and B-functions to get the correlators in position space (analogously for 4d Witten diagrams in chapter 5). Hence, we can get the correlators both in position space and in Mellin space directly from the effective action, yet it is interesting to understand how to convert between them.

Shifting the variables $s \rightarrow s + p_1 + p_2$, $t \rightarrow t + p_2 + p_3$ in (A.0.5) we get

$$\begin{split} \bar{D}_{p_1 p_2 p_3 p_4} &= \int \frac{ds \, dt}{(2\pi i)^2} u^{\frac{s}{2}} v^{\frac{t}{2}} \\ &\times \Gamma\left(\frac{-s}{2}\right) \Gamma\left(\frac{-p_1 - p_2 + p_3 + p_4 - s}{2}\right) \Gamma\left(\frac{-t}{2}\right) \Gamma\left(\frac{p_1 - p_2 - p_3 + p_4 - t}{2}\right) \Gamma\left(\frac{2p_2 + s + t}{2}\right) \Gamma\left(\frac{p_1 + p_2 + p_3 - p_4 + s + t}{2}\right) . \end{split}$$

$$(A.0.7)$$

Conversely the Mellin transform of any function which has the form of six Gammas in this form can be written as a \overline{D} -function as follows:

$$\int \frac{ds \, dt}{(2\pi i)^2} u^{\frac{s}{2}} v^{\frac{t}{2}} \Gamma\left(a_1 - \frac{s}{2}\right) \Gamma\left(a_2 - \frac{s}{2}\right) \Gamma\left(b_1 - \frac{t}{2}\right) \Gamma\left(b_2 - \frac{t}{2}\right) \Gamma\left(c_1 + \frac{s+t}{2}\right) \Gamma\left(c_2 + \frac{s+t}{2}\right)$$
$$= u^{a_1} v^{b_1} \bar{D}_{a_1 + b_2 + c_2, a_1 + b_1 + c_1, a_2 + b_1 + c_2, a_2 + b_2 + c_1} .$$
(A.0.8)

Note that this mapping is not unique because of the symmetries $a_1 \leftrightarrow a_2$, $b_1 \leftrightarrow b_2$, $c_1 \leftrightarrow c_2$, which generate various identities amongst the \overline{D} -functions.

Appendix B

The Polynomial $I(X_i, Y_i)$

The polynomial $I(X_i, Y_i)$, the so-called Intriligator polynomial, which factors out of all half-BPS four-point functions (3.1.14) is:

$$I(X_{i}, Y_{i}) = (x - y)(\bar{x} - y)(x - \bar{y})(\bar{x} - \bar{y})(X_{1}.X_{3})^{2}(X_{2}.X_{4})^{2}(Y_{1}.Y_{3})^{2}(Y_{2}.Y_{4})^{2} ,$$

with $x\bar{x} = \frac{X_{1}.X_{2}X_{3}.X_{4}}{X_{1}.X_{3}X_{2}.X_{4}} ,$ $(1 - x)(1 - \bar{x}) = \frac{X_{1}.X_{4}X_{2}.X_{3}}{X_{1}.X_{3}X_{2}.X_{4}} ,$
 $y\bar{y} = \frac{Y_{1}.Y_{2}Y_{3}.Y_{4}}{Y_{1}.Y_{3}Y_{2}.Y_{4}} ,$ $(1 - y)(1 - \bar{y}) = \frac{Y_{1}.Y_{4}Y_{2}.Y_{3}}{Y_{1}.Y_{3}Y_{2}.Y_{4}} .$ (B.0.1)

I is crossing symmetric under simultaneously permuting X_i, Y_i with X_j, Y_j . It is also a polynomial and written out fully in terms of the SO(2,4) and SO(6) invariants X_i, X_j and Y_i, Y_j is given as

$$\begin{split} I(X_i, Y_i) &= (X_1.X_4)^2 (X_2.X_3)^2 Y_1.Y_2 Y_1.Y_3 Y_2.Y_4 Y_3.Y_4 \\ &+ X_1.X_2 X_1.X_4 X_3.X_4 X_2.X_3 (Y_1.Y_3)^2 (Y_2.Y_4)^2 \\ &+ X_1.X_3 X_1.X_4 X_2.X_4 X_2.X_3 (Y_1.Y_2)^2 (Y_3.Y_4)^2 \\ &- X_1.X_2 X_1.X_4 X_3.X_4 X_2.X_3 Y_1.Y_3 Y_1.Y_4 Y_2.Y_3 Y_2.Y_4 \\ &- X_1.X_3 X_1.X_4 X_2.X_4 X_2.X_3 Y_1.Y_2 Y_1.Y_4 Y_2.Y_3 Y_3.Y_4 \\ &- X_1.X_3 X_1.X_4 X_2.X_4 X_2.X_3 Y_1.Y_2 Y_1.Y_3 Y_2.Y_4 Y_3.Y_4 \\ &- X_1.X_2 X_1.X_4 X_3.X_4 X_2.X_3 Y_1.Y_2 Y_1.Y_3 Y_2.Y_4 Y_3.Y_4 \\ &+ X_1.X_2 X_1.X_3 X_2.X_4 X_3.X_4 (Y_1.Y_4)^2 (Y_2.Y_3)^2 + (\dots) \end{split}$$

$$(\dots) + (X_{1}.X_{2})^{2} (X_{3}.X_{4})^{2} Y_{1}.Y_{3}Y_{1}.Y_{4}Y_{2}.Y_{3}Y_{2}.Y_{4}$$

- $X_{1}.X_{2}X_{1}.X_{3}X_{2}.X_{4}X_{3}.X_{4}Y_{1}.Y_{3}Y_{1}.Y_{4}Y_{2}.Y_{3}Y_{2}.Y_{4}$
+ $(X_{1}.X_{3})^{2} (X_{2}.X_{4})^{2}Y_{1}.Y_{2}Y_{1}.Y_{4}Y_{2}.Y_{3}Y_{3}.Y_{4}$
- $X_{1}.X_{2}X_{1}.X_{3}X_{2}.X_{4}X_{3}.X_{4}Y_{1}.Y_{2}Y_{1}.Y_{4}Y_{2}.Y_{3}Y_{3}.Y_{4}$. (B.0.2)

Appendix C

Contact Diagrams in AdS×S and AdS, and Supergravity

Although we will not use this fact in the rest of the thesis, it is worth pointing out an intriguing relation between the AdS×S contact diagrams and the better known standard AdS contact diagrams. This relation can be seen by comparing their respective Mellin transforms (3.1.41) and (3.1.25).

First, consider the special case $\Sigma_{\Delta} = d + 1$. In this case the final Pochhammer in (3.1.41) is absent and the Mellin transform becomes proportional to

$$\sum_{p_i=0}^{\infty} (-1)^{\Sigma_p} \int \frac{d\delta_{ij}}{(2\pi i)^2} \sum_{\{d_{ij}\}} \left(\prod_{i
$$= \int \frac{d\boldsymbol{\delta}_{ij}}{(2\pi i)^2} \left(\prod_{iwhere $\sum_{i, (C.0.1)$$$$$

where the equality is obtained by performing the sums over p_i and then changing variables from $\delta_{ij} \rightarrow \delta_{ij} = \delta_{ij} - d_{ij}$ and δ_{ij} contains AdS and S Mellin variables.

Comparing this with the Mellin transform of the AdS contact term (3.1.25) we see that this is proportional to a *D*-function with $X_i X_j \rightarrow X_i X_j + Y_i Y_j$. In other words it is proportional to a pure AdS contact term with embedding coordinates
Appendix C. Contact Diagrams in AdS×S and AdS, and Supergravity 202

 $X_i^{\mu} = (X_i^A, Y_i^I)$, corresponding to a (2d + 2)-dimensional bulk. More precisely we have the relation¹

$$D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}^{\mathrm{AdS}_{d+1} \times \mathrm{S}^{d+1}}(X_i, Y_i) = \frac{\pi^{d+1}}{(-2)^{\Sigma_\Delta} \prod_i \Gamma(\Delta_i)} \times D_{\Delta_1 \Delta_2 \Delta_3 \Delta_4}(X_i, Y_i) , \quad \text{where } \Sigma_\Delta = d+1 .$$
(C.0.2)

Note that this case $\Sigma_{\Delta} = d + 1$ corresponds precisely to the case of a dimensionless contact term in the flat space limit, $\int d^{2d+2}x \phi_{\Delta_1} \dots \phi_{\Delta_4}$. The above relation (C.0.2) is an example of the enhanced higher-dimensional conformal symmetry observed in [41]. We will look at this explicitly for $\mathcal{N} = 4$ SYM in the next subsection.

Now let us modify the above discussion for the case with $\Sigma_{\Delta} \neq d+1$. Here the direct relation between AdS×S and AdS contact terms is spoiled by the presence of the Pochhammer at the end of (3.1.41) which depends on Σ_p that we are summing over. A simple way of reproducing this Pochhammer whilst still having a summed up formula is then to rescale all the Y variables and differentiate. Concretely, we can write

$$D_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}^{\mathrm{AdS}_{d+1}\times\mathrm{S}^{d+1}}(X_{i},\sqrt{r}Y_{i}) = \frac{2}{\Gamma(\Sigma_{\Delta}-d-1)}\frac{1}{r^{d/2}}\left(\frac{d}{dr}\right)^{\Sigma_{\Delta}-d-1}r^{\Sigma_{\Delta}-d/2-1}D_{\Delta_{1}\Delta_{2}\Delta_{3}\Delta_{4}}^{(2d+2)}(X_{i},\sqrt{r}Y_{i}) ,$$
(C.0.3)

where the *D*-function is for a (2d + 2)-dimensional bulk.

Tree-level supergravity in $\mathcal{N} = 4$ SYM

While the main focus in chapter 3 is obtaining tree-level string corrections to $\mathcal{N} = 4$ SYM correlators from an effective action involving massless scalars in 10d, it is interesting to also look at the tree-level supergravity prediction following the approach described in this appendix. While we do not expect this to arise from an effective superpotential, all single-trace half-BPS correlators were shown in [41] to possess a 10d conformal structure and in particular can be obtained by expanding

¹We here compare with the dimension-independent, normalised *D*-function (3.1.26), since in 2d + 2 dimensions the $D^{(d)}$ -function itself diverges when $\Sigma_{\Delta} = d + 1$ due to the Γ -function in the numerator of (2.3.15).

out D_{2422} . For D_{2422} the simple relation between AdS×S and standard AdS contact diagrams described above is valid because $\Sigma_{\Delta} = 5 = d + 1$ and hence (C.0.2) applies. The tree-level supergravity result can be written [41]

$$\langle \mathcal{OOOO} \rangle_{\text{sugra}} \propto \frac{1}{(X_1 \cdot X_3 + Y_1 \cdot Y_3)} \frac{1}{(X_1 \cdot X_4 + Y_1 \cdot Y_4)} \frac{1}{(X_3 \cdot X_4 + Y_3 \cdot Y_4)} D_{2422}(X_i, Y_i) .$$

(C.0.4)

Inserting the Mellin representation of D_{2422} (3.1.25) and changing variables $\delta_{ij} \rightarrow \delta_{ij} - 1$ for i, j = 1, 3, 4 and δ_{ij} unchanged otherwise, this can be written in the form (3.1.42) with $\Delta_i = 4$ with the Mellin amplitude

$$\mathcal{M}_{\text{sugra}} \propto \frac{1}{(\boldsymbol{\delta}_{13}-1)(\boldsymbol{\delta}_{14}-1)(\boldsymbol{\delta}_{34}-1)} = \frac{1}{(\delta_{13}-d_{13}-1)(\delta_{14}-d_{14}-1)(\delta_{34}-d_{34}-1)} \quad (C.0.5)$$

The denominator in the above equations can be understood from the supergravity piece of the Virasoro-Shapiro amplitude which is $\frac{1}{STU}$, which acts like inverse derivatives on D_{2422} . Acting with STU on (C.0.4) gives back the zero-derivative contact term D_{4444} which has Mellin transform 1 and can be obtained from a 10d superpotential.

Appendix D

α'^7 Ambiguities in AdS₅×S⁵

The ambiguities at order α'^7 were introduced in (3.6.2) and we spell out their Witten diagram expressions and the corresponding Mellin amplitudes in the following.

The first ambiguity at α'^7 contributes to the effective action with

$$S^{\text{amb}_{1}}_{\alpha'^{7}} = \frac{6}{4!} \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} \left(\nabla^{\mu}\nabla^{\nu}\nabla_{\mu}\nabla^{\rho}\nabla^{\sigma}\nabla_{\rho}\phi\right) \left(\nabla_{\nu}\nabla_{\sigma}\phi\right) \phi^{2} , \qquad (\text{D.0.1})$$

it corresponds to a four-derivative interaction and its contribution to the half-BPS correlator is given as

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7}; \text{amb}_{1}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS} \times \text{S}} \frac{d^{5} \hat{X} d^{5} \hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \sum_{i < j} \frac{K_{ij}^{\text{amb}_{1}}}{(P_{i} + Q_{i})^{2} (P_{j} + Q_{j})^{2}} \times 4^{3} \times 5 ,$$
(D.0.2)

where

$$K_{ij}^{\text{amb}_{1}} = 45 \left[(X_{i}.X_{j} + P_{i}P_{j})^{2} + (Y_{i}.Y_{j} + Q_{i}Q_{j})^{2} \right] - 9 \left(P_{i}P_{j} + Q_{i}Q_{j} \right)^{2} - 180 Q_{i}Q_{j} Y_{i}.Y_{j} + 26 P_{i}P_{j} Q_{i}Q_{j} .$$
(D.0.3)

We write the corresponding Mellin amplitude as

$$\mathcal{M}_{\alpha'^{7}}^{\text{amb}_{1}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{1}} + 204\mathcal{M}_{\alpha'^{5}}^{\text{amb}} + 12288\mathcal{M}_{\alpha'^{3}}^{\text{main}} , \qquad (\text{D.0.4})$$

where

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha'^{7}}^{\operatorname{amb}_{1}} &= 288 \left[\left(\Sigma_{p} - 1 \right)_{5} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) \right. \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \frac{1}{2} \left(\mathbf{s} \, c_{s}^{2} + \mathbf{t} \, c_{t}^{2} + \mathbf{u} \, c_{u}^{2} \right) \\ &- \left(\Sigma_{p} - 1 \right)_{4} \left(\Sigma_{p} + 3 \right) \left[2 \left(\mathbf{s} \, \tilde{s} + \mathbf{t} \, \tilde{t} + \mathbf{u} \, \tilde{u} \right) + \left(\mathbf{s} \, c_{s} + \mathbf{t} \, c_{t} + \mathbf{u} \, c_{u} \right) \right] \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(\frac{1}{12} \left(c_{s}^{4} + c_{t}^{4} + c_{u}^{4} \right) - \Sigma_{p} \left(c_{s}^{2} \, \tilde{s} + c_{t}^{2} \, \tilde{t} + c_{u}^{2} \, \tilde{u}^{2} \right) - \frac{1}{2} \Sigma_{p} \left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(+ \frac{29}{36} \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) + \frac{1}{18} \left(c_{s}^{2} \, c_{t}^{2} + c_{s}^{2} \, c_{u}^{2} + c_{t}^{2} \, c_{u}^{2} \right) - \frac{5}{6} \, c_{s} \, c_{t} \, c_{u} \, \Sigma_{p} \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(2 \left(\Sigma_{p}^{2} + 6 \right) \left[\left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) + \left(\tilde{s} \, c_{s} + \tilde{t} \, c_{t} + \tilde{u} \, c_{u} \right) \right] - \frac{1}{6} \Sigma_{p}^{2} \left(\Sigma_{p}^{2} + 72 \right) \right) \right]. \end{aligned} \tag{D.0.5}$$

The next three ambiguities also correspond to four-derivative terms. The contribution to the effective action from the second ambiguity is

$$S^{\text{amb}_2}_{\alpha'^7} = \frac{6}{4!} \int_{\text{AdS}\times\text{S}} d^5 \hat{X} d^5 \hat{Y} \left(\nabla^2 \nabla^\mu \nabla^\nu \nabla^\rho \nabla_\nu \phi \right) \left(\nabla_\mu \nabla_\rho \phi \right) \phi^2 , \qquad (\text{D.0.6})$$

which corresponds to the correlator

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7}; \text{amb}_{2}}$$

$$= \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS} \times S} \frac{d^{5} \hat{X} d^{5} \hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \sum_{i < j} \frac{K_{ij}^{\text{amb}_{2}}}{(P_{i} + Q_{i})^{2} (P_{j} + Q_{j})^{2}} \times 4^{3} \times 5^{2} \times 2 , \quad (\text{D.0.7})$$

where

$$K_{ij}^{\text{amb}_2} = 5 \left[\left(X_i \cdot X_j + P_i P_j \right)^2 + \left(Y_i \cdot Y_j + Q_i Q_j \right)^2 \right] - \left(P_i P_j + Q_i Q_j \right)^2 - 20 Q_i Q_j Y_i \cdot Y_j + 2 P_i P_j Q_i Q_j .$$
(D.0.8)

The Mellin amplitude is

$$\mathcal{M}_{\alpha'^{7}}^{\text{amb}_{2}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{2}} + 248\mathcal{M}_{\alpha'^{5}}^{\text{amb}} + 14336\mathcal{M}_{\alpha'^{3}}^{\text{main}} , \qquad (\text{D.0.9})$$

$$\hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{2}} = 320 \left[(\Sigma_{p} - 1)_{5} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) + (\Sigma_{p} - 1)_{4} \frac{1}{2} \left(\mathbf{s} c_{s}^{2} + \mathbf{t} c_{t}^{2} + \mathbf{u} c_{u}^{2} \right) - (\Sigma_{p} - 1)_{4} (\Sigma_{p} + 3) \left[2 \left(\mathbf{s} \tilde{s} + \mathbf{t} \tilde{t} + \mathbf{u} \tilde{u} \right) + \left(\mathbf{s} c_{s} + \mathbf{t} c_{t} + \mathbf{u} c_{u} \right) \right]$$

$$+ (\Sigma_{p}-1)_{3} \left(\frac{3}{40} \left(c_{s}^{4} + c_{t}^{4} + c_{u}^{4} \right) - \Sigma_{p} \left(c_{s}^{2} \tilde{s} + c_{t}^{2} \tilde{t} + c_{u}^{2} \tilde{u}^{2} \right) \right) + (\Sigma_{p}-1)_{3} \left(\frac{1}{20} \left(c_{s}^{2} c_{t}^{2} + c_{s}^{2} c_{u}^{2} + c_{t}^{2} c_{u}^{2} \right) - \frac{1}{2} \Sigma_{p} \left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3} \right) + \frac{4}{5} \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) \right) + (\Sigma_{p}-1)_{3} \left(2 \left(\Sigma_{p}^{2} + 6 \right) \left[\left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) + \left(\tilde{s} c_{s} + \tilde{t} c_{t} + \tilde{u} c_{u} \right) \right] \right) + (\Sigma_{p}-1)_{3} \left(-\frac{9}{10} c_{s} c_{t} c_{u} \Sigma_{p} - \frac{1}{40} \Sigma_{p}^{2} \left(7 \Sigma_{p}^{2} + 480 \right) \right) \right].$$
(D.0.10)

The third ambiguity contributes to the effective action with

$$S^{\text{amb}_3}_{\alpha'^7} = \int_{\text{AdS}\times\text{S}} d^5 \hat{X} d^5 \hat{Y} \left(\nabla^2 \nabla^\mu \nabla^\nu \nabla^\rho \nabla_\nu \phi \right) \left(\nabla_\rho \phi \right) \left(\nabla_\mu \phi \right) \phi , \qquad (\text{D.0.11})$$

and the prediction for its contribution to the half-BPS correlator is given by

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7}; \text{amb}_{3}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} \frac{d^{5}\hat{X}d^{5}\hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \left[\frac{K_{123}^{\text{amb}_{3}}}{(P_{1} + Q_{1})^{2} (P_{2} + Q_{2}) (P_{3} + Q_{3})} + \text{perms} \right] \times 4^{4} \times 10 ,$$

$$(D.0.12)$$

where we sum over all permutations and

$$K_{ijk}^{\text{amb}_3} = P_i^2 \left(4 P_j P_k - X_j . X_k \right) + Q_i^2 \left(4 Q_j Q_k + Y_j . Y_k \right) + 5 P_i \left(P_k X_i . X_j + P_j X_i . X_k \right) - 5 Q_i \left(Q_k Y_i . Y_j + Q_j Y_i . Y_k \right) + 5 \left(X_i . X_j X_i . X_k + Y_i . Y_j Y_i . Y_k \right) .$$
(D.0.13)

The corresponding Mellin amplitude is

$$\mathcal{M}_{\alpha'^{7}}^{\text{amb}_{3}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{3}} - 704 \mathcal{M}_{\alpha'^{5}}^{\text{amb}} - 32768 \mathcal{M}_{\alpha'^{3}}^{\text{main}} , \qquad (D.0.14)$$

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{3}} &= -640 \left[\left(\Sigma_{p} - 1 \right)_{5} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) \right. \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left. \frac{1}{2} \left(\mathbf{s} c_{s}^{2} + \mathbf{t} c_{t}^{2} + \mathbf{u} c_{u}^{2} \right) \right. \\ &- \left(\Sigma_{p} - 1 \right)_{4} \left(\Sigma_{p} + 3 \right) \left[2 \left(\mathbf{s} \tilde{s} + \mathbf{t} \tilde{t} + \mathbf{u} \tilde{u} \right) + \left(\mathbf{s} c_{s} + \mathbf{t} c_{t} + \mathbf{u} c_{u} \right) \right] \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(\frac{1}{20} \left(c_{s}^{4} + c_{t}^{4} + c_{u}^{4} \right) - \Sigma_{p} \left(c_{s}^{2} \tilde{s} + c_{t}^{2} \tilde{t} + c_{u}^{2} \tilde{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(- \frac{1}{10} \left(c_{s}^{2} c_{t}^{2} + c_{s}^{2} c_{u}^{2} + c_{t}^{2} c_{u}^{2} \right) - \frac{1}{2} \Sigma_{p} \left(c_{s}^{3} + c_{t}^{3} + c_{u}^{3} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(\frac{13}{20} \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) - \frac{3}{10} c_{s} c_{t} c_{u} \Sigma_{p} \right) \end{aligned}$$

$$+ (\Sigma_p - 1)_3 \left(2 \left(\Sigma_p^2 + 6 \right) \left[\left(\tilde{s}^2 + \tilde{t}^2 + \tilde{u}^2 \right) + \left(\tilde{s} c_s + \tilde{t} c_t + \tilde{u} c_u \right) \right] - \frac{1}{5} \Sigma_p^2 \left(\Sigma_p^2 + 60 \right) \right) \right].$$
(D.0.15)

The next ambiguity contributes to the effective action as

$$S^{\text{amb}_4}_{\alpha'^7} = \int_{\text{AdS}\times\text{S}} d^5 \hat{X} d^5 \hat{Y} \left(\nabla^\mu \nabla^\nu \nabla^\rho \nabla_\nu \nabla_\rho \phi \right) \left(\nabla^\sigma \nabla_\mu \phi \right) \left(\nabla_\sigma \phi \right) \phi , \qquad (\text{D.0.16})$$

and the corresponding correlator is given by

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7}; \text{amb}_{4}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} \frac{d^{5}\hat{X}d^{5}\hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \left[\frac{K_{123}^{\text{amb}_{4}}}{(P_{1} + Q_{1})^{3} (P_{2} + Q_{2})^{2} (P_{3} + Q_{3})} + \text{perms} \right] \times 4^{4} \times 10 ,$$

$$(D.0.17)$$

where

$$\begin{aligned} K_{ijk}^{amb_{4}} &= P_{i}^{2}Q_{i}\left[5\left(5P_{j}+Q_{j}\right)\left(X_{j}.X_{k}+Y_{j}.Y_{k}\right)-4Q_{j}Q_{k}\left(6P_{j}+Q_{j}\right)+20P_{j}^{2}P_{k}\right] \\ &\quad -P_{i}Q_{i}^{2}\left[5\left(P_{j}+5Q_{j}\right)\left(X_{j}.X_{k}+Y_{j}.Y_{k}\right)+4P_{j}P_{k}\left(P_{j}+6Q_{j}\right)-20Q_{j}^{2}Q_{k}\right] \\ &\quad -P_{i}^{2}\left[Q_{j}\left(P_{j}+Q_{j}\right)Y_{i}.Y_{k}+5Y_{i}.Y_{j}\left(X_{j}.X_{k}+Y_{j}.Y_{k}+P_{j}P_{k}-Q_{j}Q_{k}\right)\right] \\ &\quad +Q_{i}^{2}\left[P_{j}\left(P_{j}+Q_{j}\right)X_{i}.X_{k}-5X_{i}.X_{j}\left(X_{j}.X_{k}+Y_{j}.Y_{k}+P_{j}P_{k}-Q_{j}Q_{k}\right)\right] \\ &\quad +P_{i}Q_{i}\left[25\left(P_{j}P_{k}-Q_{j}Q_{k}\right)\left(X_{i}.X_{j}+Y_{i}.Y_{j}\right)+\left(P_{j}+Q_{j}\right)\left(5Q_{j}Y_{i}.Y_{k}-5P_{j}X_{i}.X_{k}\right) \\ &\quad +25\left(X_{i}.X_{j}+Y_{i}.Y_{j}\right)\left(X_{j}.X_{k}+Y_{j}.Y_{k}\right)\right]. \end{aligned}$$

The contribution of this ambiguity to the Mellin amplitude is

$$\mathcal{M}_{\alpha'^{7}}^{\text{amb}_{4}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{4}} - 128\mathcal{M}_{\alpha'^{5}}^{\text{main}} , \qquad (D.0.19)$$

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{4}} &= 32 \left[\left(\Sigma_{p} - 1 \right)_{5} \Sigma_{p}^{2} \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{5} \left(\left(\mathbf{s}^{2} c_{s}^{2} + \mathbf{t}^{2} c_{t}^{2} + \mathbf{u}^{2} c_{u}^{2} \right) + \left[\mathbf{s}^{2} \left(c_{t}^{2} + c_{u}^{2} \right) + \mathbf{t}^{2} \left(c_{s}^{2} + c_{u}^{2} \right) + \mathbf{u}^{2} \left(c_{s}^{2} + c_{t}^{2} \right) \right] \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(-5 \left(\mathbf{s} c_{s}^{3} + \mathbf{t} c_{t}^{3} + \mathbf{u} c_{u}^{3} \right) - 10 \left(\mathbf{s} c_{s}^{2} \tilde{s} + \mathbf{t} c_{t}^{2} \tilde{t} + \mathbf{u} c_{u}^{2} \tilde{u} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(-10 \Sigma_{p}^{2} \left(\mathbf{s} \tilde{s} + \mathbf{t} \tilde{t} + \mathbf{u} \tilde{u} \right) - 5 \Sigma_{p}^{2} \left(\mathbf{s} c_{s} + \mathbf{t} c_{t} + \mathbf{u} c_{u} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(-10 \left[\mathbf{s} \tilde{s} \left(c_{t}^{2} + c_{u}^{2} \right) + \mathbf{t} \tilde{t} \left(c_{s}^{2} + c_{u}^{2} \right) + \mathbf{u} \tilde{u} \left(c_{s}^{2} + c_{t}^{2} \right) \right] \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(-5 \left[\mathbf{s} c_{s} \left(c_{t}^{2} + c_{u}^{2} \right) + \mathbf{t} c_{t} \left(c_{s}^{2} + c_{u}^{2} \right) + \mathbf{u} c_{u} \left(c_{s}^{2} + c_{t}^{2} \right) \right] \right) \end{aligned}$$

$$+ (\Sigma_{p}-1)_{3} \left(4 \left(c_{s}^{4} + c_{t}^{4} + c_{u}^{4} \right) + 20 \left(\tilde{s}^{2} c_{s}^{2} + \tilde{t}^{2} c_{t}^{2} + \tilde{u}^{2} c_{u}^{2} \right) + 8 \left(c_{s}^{2} c_{t}^{2} + c_{s}^{2} c_{u}^{2} + c_{t}^{2} c_{u}^{2} \right) \right)$$

$$+ (\Sigma_{p}-1)_{3} \left(20 \left(c_{s}^{3} \tilde{s} + c_{t}^{3} \tilde{t} + c_{u}^{3} \tilde{u} \right) + 20 \Sigma_{p}^{2} \left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) \right)$$

$$+ (\Sigma_{p}-1)_{3} \left(-8 \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) + 20 \Sigma_{p}^{2} \left(\tilde{s} c_{s} + \tilde{t} c_{t} + \tilde{u} c_{u} \right) \right)$$

$$+ (\Sigma_{p}-1)_{3} \left(20 \left[\tilde{s}^{2} \left(c_{t}^{2} + c_{u}^{2} \right) + \tilde{t}^{2} \left(c_{s}^{2} + c_{u}^{2} \right) + \tilde{u}^{2} \left(c_{s}^{2} + c_{t}^{2} \right) \right] \right)$$

$$+ (\Sigma_{p}-1)_{3} \left(20 \left[\tilde{s} c_{s} \left(c_{t}^{2} + c_{u}^{2} \right) + \tilde{t} c_{t} \left(c_{s}^{2} + c_{u}^{2} \right) + \tilde{u} c_{u} \left(c_{s}^{2} + c_{t}^{2} \right) \right] - 12 \Sigma_{p}^{4} \right) \right] .$$

$$(D.0.20)$$

Finally, the fifth ambiguity contributes to the effective action with

$$S^{\text{amb}_{5}}_{\alpha'^{7}} = \int_{\text{AdS}\times\text{S}} d^{5}\hat{X} d^{5}\hat{Y} \left(\nabla^{\mu}\nabla^{\nu}\nabla^{\rho}\nabla^{\sigma}\nabla_{\rho}\phi\right) \left(\nabla_{\mu}\nabla_{\sigma}\phi\right) \left(\nabla_{\nu}\phi\right)\phi , \qquad (\text{D.0.21})$$

which corresponds to a six-derivative interaction and its contribution to the half-BPS correlator is

$$\langle \mathcal{OOOO} \rangle |_{\alpha'^{7}; \text{amb}_{5}} = \frac{1}{4!} \frac{(\mathcal{C}_{4})^{4}}{(-2)^{16}} \int_{\text{AdS}\times\text{S}} \frac{d^{5}\hat{X}d^{5}\hat{Y}}{\prod_{i} (P_{i} + Q_{i})^{4}} \left[\frac{K_{123}^{\text{amb}_{5}}}{(P_{1} + Q_{1})^{3} (P_{2} + Q_{2})^{2} (P_{3} + Q_{3})} + \text{perms} \right] \times (-4^{4}) ,$$
(D.0.22)

$$\begin{split} K_{ijk}^{amb5} &= P_i^3 \left[P_j \left(4 \left(Q_j + 21 P_j \right) P_k - 45 X_j . X_k \right) \right] + Q_i^3 \left[Q_j \left(4 \left(P_j + 21 Q_j \right) Q_k + 45 Y_j . Y_k \right) \right] \\ &+ P_i^2 Q_i \left[8 \left(Q_j - 4 P_j \right) P_j P_k + 4 \left(Q_j - 5 P_j \right) \left(Q_j + 6 P_j \right) Q_k - 40 P_j X_j . X_k \\ &- 5 Q_j Y_j . Y_k \right] + P_i Q_i^2 \left[8 \left(P_j - 4 Q_j \right) Q_j Q_k + 4 \left(P_j - 5 Q_j \right) \left(P_j + 6 Q_j \right) P_k \\ &+ 40 Q_j Y_j . Y_k + 5 P_j X_j . X_k \right] + P_i^2 \left[P_j \left(129 P_j + 4 Q_j \right) X_i . X_k + 120 P_j^2 Y_i . Y_k \\ &+ 15 X_i . X_j \left(17 P_j P_k - 3 X_j . X_k \right) + Q_j \left(-5 Q_k Y_i . Y_j + \left(-4 P_j + Q_j \right) Y_i . Y_k \right) \\ &+ 5 Y_i . Y_j Y_j . Y_k \right] + Q_i^2 \left[-Q_j \left(129 Q_j + 4 P_j \right) Y_i . Y_k - 15 Y_i . Y_j \left(17 Q_j Q_k + 3 Y_j . Y_k \right) \\ &- 120 Q_j^2 X_i . X_k - P_j \left(-5 P_k X_i . X_j + \left(-4 Q_j + P_j \right) X_i . X_k \right) + 5 X_i . X_j X_j . X_k \right] \\ &+ P_i Q_i \left[8 \left(P_j + Q_j \right) \left(P_j X_i . X_k - Q_j Y_i . Y_k \right) \\ &+ 20 \left(- X_i . X_j \left(P_j \left(2P_k + 15 Q_k \right) + 2 X_j . X_k \right) + Y_i . Y_j \left(Q_j \left(2Q_k + 15 P_k \right) - 2 Y_j . Y_k \right) \right) \right] \\ &+ P_i \left[150 P_k \left(\left(X_i . X_j \right)^2 - \left(Y_i . Y_j \right)^2 \right) + 300 P_j X_i . X_j \left(X_i . X_k + Y_i . Y_k \right) \right] \\ &+ Q_i \left[-150 Q_k \left(\left(X_i . X_j \right)^2 - \left(Y_i . Y_j \right)^2 \right) + 300 Q_j Y_i . Y_j \left(X_i . X_k + Y_i . Y_k \right) \right] \\ &+ 150 \left(\left(X_i . X_j \right)^2 - \left(Y_i . Y_j \right)^2 \right) \left(X_i . X_k + Y_i . Y_k \right) . \end{split}$$

The corresponding Mellin amplitude is

$$\mathcal{M}_{\alpha'^{7}}^{\text{amb}_{5}} = \hat{\mathcal{M}}_{\alpha'^{7}}^{\text{amb}_{5}} - \frac{11}{2} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{1}} + 5 \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{2}} + \frac{1}{8} \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{3}} - \mathcal{M}_{\alpha'^{7}}^{\text{amb}_{4}} + 64 \mathcal{M}_{\alpha'^{5}}^{\text{main}} + 66 \mathcal{M}_{\alpha'^{5}}^{\text{amb}} + 4096 \mathcal{M}_{\alpha'^{3}}^{\text{main}} , \qquad (D.0.24)$$

$$\begin{aligned} \hat{\mathcal{M}}_{\alpha^{\prime7}}^{amb_{5}} &= 128 \left[\left(\Sigma_{p} - 1 \right)_{6} \left(\mathbf{s}^{3} + \mathbf{t}^{3} + \mathbf{u}^{3} \right) \right. \\ &+ \left(\Sigma_{p} - 1 \right)_{5} \left(\frac{1}{2} \left(\mathbf{s}^{2} c_{s}^{2} + \mathbf{t}^{2} c_{t}^{2} + \mathbf{u}^{2} c_{u}^{2} \right) + \frac{1}{2} \Sigma_{p} \left(\Sigma_{p} + 8 \right) \left(\mathbf{s}^{2} + \mathbf{t}^{2} + \mathbf{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{5} \left(- \left(\Sigma_{p} + 7 \right) \left[2 \left(\mathbf{s}^{2} \tilde{s} + \mathbf{t}^{2} \tilde{t} + \mathbf{u}^{2} \tilde{u} \right) + \left(\mathbf{s}^{2} c_{s} + \mathbf{t}^{2} c_{t} + \mathbf{u}^{2} c_{u} \right) \right] \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(-\frac{5}{2} \left(\mathbf{s} c_{s}^{3} + \mathbf{t} c_{t}^{3} + \mathbf{u} c_{u}^{3} \right) - 5 \left(\mathbf{s} c_{s}^{2} \tilde{s} + \mathbf{t} c_{t}^{2} \tilde{t} + \mathbf{u} c_{u}^{2} \tilde{u} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(6 \left(4 \Sigma_{p} + 7 \right) \left[\left(\mathbf{s} \tilde{s}^{2} + \mathbf{t} \tilde{t}^{2} + \mathbf{u} \tilde{u}^{2} \right) + \left(\mathbf{s} \tilde{s} c_{s} + \mathbf{t} \tilde{t} c_{t} + \mathbf{u} \tilde{u} c_{u} \right) \right] \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(6 \left(4 \Sigma_{p} + 7 \right) \left[\left(\mathbf{s} \tilde{s}^{2} + \mathbf{t} \tilde{t}^{2} + \mathbf{u} \tilde{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(\frac{1}{2} \left(14 \Sigma_{p} + 19 \right) \left(\mathbf{s} c_{s}^{2} + \mathbf{t} \tilde{c}^{2} + \mathbf{u} c_{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{4} \left(\frac{1}{2} \left(14 \Sigma_{p} + 19 \right) \left(\mathbf{s} c_{s}^{2} + \mathbf{t} c_{t}^{2} + \mathbf{u} c_{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(10 \left(c_{s}^{3} \tilde{s} + c_{t}^{3} \tilde{t} + c_{u}^{3} \tilde{u} \right) + 10 \left(\tilde{s}^{2} c_{s}^{2} + \tilde{t}^{2} c_{t}^{2} + \tilde{u}^{2} c_{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(-90 \Sigma_{p} \left(\tilde{s}^{2} c_{s} + \tilde{t}^{2} c_{t} + \tilde{u}^{2} c_{u} \right) - 47 \Sigma_{p} \left(c_{s}^{2} \tilde{s} + c_{t}^{2} \tilde{t} + c_{u}^{2} \tilde{u} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(\frac{51}{4} \Sigma_{p}^{2} \left(c_{s}^{2} + c_{t}^{2} + c_{u}^{2} \right) + 50 \Sigma_{p}^{2} \left(\tilde{s}^{2} + \tilde{t}^{2} + \tilde{u}^{2} \right) \right) \\ &+ \left(\Sigma_{p} - 1 \right)_{3} \left(50 \Sigma_{p}^{2} \left(\tilde{s} c_{s} + \tilde{t} c_{t} + \tilde{u} c_{u} \right) - \frac{1}{8} \Sigma_{p}^{2} \left(81 \Sigma_{p}^{2} + 352 \right) \right) \right] \right]. \tag{D.0.25}$$

Appendix E

Conformal Blocks in 6d

Conformal blocks for four-point correlators of scalar operators of arbitrary scaling dimensions Δ_i , i = 1, ..., 4, in any even dimension were derived by Dolan and Osborn in [46]. In 6d, the blocks are given by

$$G^{\text{DO}}(\Delta, l, \Delta_{12}, \Delta_{34}) =
 \mathcal{F}_{00} - \frac{l+3}{l+1} \mathcal{F}_{-11} \\
 - \frac{\Delta - 4}{\Delta - 2} \frac{(\Delta + l - \Delta_{12})(\Delta + l + \Delta_{12})(\Delta + l + \Delta_{34})(\Delta + l - \Delta_{34})}{16(\Delta + l - 1)(\Delta + l)^{2}(\Delta + l + 1)} \mathcal{F}_{11} \\
 + \frac{(\Delta - 4)(l+3)}{(\Delta - 2)(l+1)} \\
 \times \frac{(\Delta - l - \Delta_{12} - 4)(\Delta - l + \Delta_{12} - 4)(\Delta - l + \Delta_{34} - 4)(\Delta - l - \Delta_{34} - 4)}{16(\Delta - l - 5)(\Delta - l - 4)^{2}(\Delta - l - 3)} \mathcal{F}_{02} \\
 + 2(\Delta - 4)(l+3)\frac{\Delta_{12}\Delta_{34}}{(\Delta + l)(\Delta + l - 2)(\Delta - l - 4)(\Delta - l - 6)} \mathcal{F}_{01}, \quad (\text{E.0.1})$$

where (Δ, l) are the scaling dimension and spin of a primary operator in the conformal block expansion, $\Delta_{ij} = \Delta_i - \Delta_j$, and

$$\mathcal{F}_{ab} = \frac{(z\bar{z})^{\frac{1}{2}(\Delta-l)}}{(z-\bar{z})^3} \Big\{ z^{l+a+3}\bar{z}^b \\ \times {}_2F_1\left(\frac{1}{2}(\Delta+l-\Delta_{12})+a,\frac{1}{2}(\Delta+l+\Delta_{34})+a;\Delta+l+2a,z\right) \\ \times {}_2F_1\left(\frac{1}{2}(\Delta-l-\Delta_{12})-3+b,\frac{1}{2}(\Delta-l+\Delta_{34})-3+b;\Delta-l-6+2b;\bar{z}\right) \\ -z \leftrightarrow \bar{z} \Big\} .$$
(E.0.2)

For the toy model analysed in section 4.2, the blocks are given by

$$G^{\rm B}_{\Delta,l}(z,\bar{z}) = (l+1) G^{\rm DO}(\Delta,l,0,0) ,$$
 (E.0.3)

where $\Delta = 2n + l + 2\Delta_0 + \mathcal{O}(1/c)$. Moreover, for the 6d (2,0) theory analysed in section 4.3, the blocks are given by [25, 47]

$$G_{\Delta,l}^{\rm S}(z,\bar{z}) = \frac{4(l+1)}{(l+2)^2 - \Delta^2} \frac{(z-\bar{z})^3}{u^5} G^{\rm DO}(\Delta+4,l,0,-2) , \qquad (E.0.4)$$

where $\Delta = 2n + l + 8 + \mathcal{O}(1/c)$ with $n \ge 0$.

Appendix F

Orthogonality of Hypergeometrics

In this appendix we derive orthogonality relations for hypergeometric functions used in chapter 4, explicating a brief argument in [30] which then allows us to obtain a new case relevant for the supersymmetric 6d theory. Our starting point will be the differential operator¹

$$D_z = z^2 (1-z) \partial_z^2 - (a+b+1) z^2 \partial_z - a b z .$$
 (F.0.2)

This operator has eigenfunctions satisfying

$$D_z H_m(z) = m(m-1)H_m(z)$$
, (F.0.3)

where

$$H_m(z) = z^m {}_2F_1(m+a, m+b; 2m; z) .$$
 (F.0.4)

First consider a = b = 0. In this case, the differential operator in (F.0.2) reduces to $D_z = z^2 \partial_z (1-z) \partial_z$. Let us look at the object $H_m H_{1-m'}$ (we will omit the arguments (z) in the following). Using the symmetry of the differential operator D_z , after

$$D_{\epsilon} = z^{2} (1-z) \partial_{z}^{2} + \bar{z}^{2} (1-\bar{z}) \partial_{\bar{z}}^{2} - (a+b+1) (z^{2} \partial_{z} + \bar{z}^{2} \partial_{\bar{z}}) - a b (z+\bar{z}) + \epsilon \frac{z \bar{z}}{z-\bar{z}} ((1-z) \partial_{z} - (1-\bar{z}) \partial_{\bar{z}}) , \qquad (F.0.1)$$

¹Note that this operator is closely related to the conformal Casimir. In d dimensions this is [46]

where a, b are arbitrary constants and $\epsilon = d - 2$. The non-interacting part (i.e ϵ -independent part) reduces to $D_z + D_{\bar{z}}$.

integrating by parts twice and using (F.0.3) we find that

$$0 = \oint \frac{dz}{2\pi i} \frac{1}{z^2} \left[(D_z H_m) H_{1-m'} - H_m (D_z H_{1-m'}) \right]$$

= $\left[m(m-1) - m'(m'-1) \right] \oint \frac{dz}{2\pi i} \frac{1}{z^2} H_m H_{1-m'}$, (F.0.5)

where the contour encircles the origin. It follows that H_m and $H_{1-m'}$ are orthogonal with respect to the inner product defined above if $m \neq m'$. Plugging in (F.0.4) and shifting (m, m') to (m + 2, m' + 2) then implies the inner product in (4.2.16), where we fix the normalisation by noting that $_2F_1(\alpha, \beta, \gamma, z) = 1 + \mathcal{O}(z)$ and evaluating the residue at z = 0. This relation was first obtained in [30].

Next, consider a = 0, b = -1, in which case (F.0.2) reduces to $D_z = z^2 (1 - z) \partial_z^2$. Following the same arguments as above we find that

$$0 = \oint \frac{dz}{2\pi i} \frac{1}{z^2(1-z)} \left[(D_z H_m) H_{1-m'} - H_m (D_z H_{1-m'}) \right]$$

= $\left[m(m-1) - m'(m'-1) \right] \oint \frac{dz}{2\pi i} \frac{1}{z^2(1-z)} H_m H_{1-m'}$, (F.0.6)

so H_m and $H_{1-m'}$ are orthogonal with respect to the inner product defined above if $m \neq m'$. Plugging in (F.0.4) and shifting (m, m') to (m + 2, m' + 2) then proves the inner product in (4.3.17), where the normalisation is once again fixed by evaluating the residue at z = 0.

Appendix G

Tree-Level Supergravity in the 6d (2,0) **Theory**

In this appendix we will discuss the 6d (2,0) tree-level supergravity prepotential and derive the anomalous dimensions of the double-trace operators contributing to its conformal block expansion. For large c, the four-point prepotential can be derived from tree-level Witten diagrams for supergravity in AdS₇×S⁴. In [139] the authors showed that the supergravity contribution consists of a contribution from free theory (including c^0 and 1/c terms) plus the remaining supergravity contribution, which we call dynamical contribution and which goes like 1/c. We denote the corresponding prepotential terms by F^{free} and F^{sugra} respectively. Note that in section 4.3 we only considered the leading disconnected free contribution at c^0 , now we have to consider an extra piece. The free contribution is then given by:

$$F^{\text{free}} = 1 + \frac{1}{u^2} + \frac{1}{v^2} + \frac{1}{c} \left(\frac{1}{u} + \frac{1}{v} + \frac{1}{uv} \right) , \qquad (G.0.1)$$

which satisfies the crossing condition (4.3.3). Decomposing F^{free} according to (4.3.4) yields

$$A = 1$$
, $g(z) = \frac{z}{c} \left(1 + \frac{1}{1-z} \right)$. (G.0.2)

In addition to expanding G according to (4.3.5) one can expand g in terms of short conformal blocks as follows

$$g(x) = \sum_{m=0}^{\infty} B_m g_m(x) , \quad g_m(x) = x^{m+1} {}_2F_1(m+2,m+1,2m+4,x) , \qquad (G.0.3)$$

where the first contribution with m = 0 corresponds to the conformal block of the stress tensor supermultiplet (see [26] for more details on this). However, we focus on the expansion of G in terms of long blocks which gives

$$A_{n,l}^{\text{free}} = \frac{(l+2)(n+3)!(n+4)!(l+2n+9)(l+n+5)!(l+n+6)!}{(2n+5)!(2l+2n+9)!} \times \left(\frac{1}{72}(l+2n+10) + \frac{1}{c}\frac{(-1)^n}{2(l+1)(l+3)(l+2n+8)}\right), \quad (G.0.4)$$

for l even and zero otherwise. Note that the c^0 -piece was already obtained in (4.3.8).

Next, we consider the dynamical supergravity contribution at $\mathcal{O}(1/c)$. This is given by the following prepotential derived from AdS contact diagrams:

$$F^{\text{sugra}} = -\frac{1}{c} \frac{(z - \bar{z})^2}{u \, v} \bar{D}_{3337} \ . \tag{G.0.5}$$

Decomposing this according to (4.3.4) gives A = 0 and

$$g(z) = \frac{1}{c} \left(2 z_2 F_1(2, 1, 4, z) - z \left(1 + \frac{1}{1 - z} \right) \right) . \tag{G.0.6}$$

The first term corresponds to the conformal block of the stress tensor supermultiplet, which can be seen from (G.0.3) with m = 0. Importantly, the second term cancels with the free theory contribution (G.0.2). These terms correspond to twist-4 states (see [26] for more details). This cancellation is required because it is expected that only operators in the singlet representation with twist eight or higher develop anomalous dimensions. The twist-4 states which contribute in the free theory should be absent in the supergravity limit because from the AdS/CFT correspondence it is known that there are no supergravity states with twist four, which corresponds to the minimal twist representation.

Finally, the OPE coefficients and anomalous dimensions of the long operators in

the conformal block expansion of the dynamical supergravity contribution can be obtained from an expansion of the form (4.3.10) but with $A_{n,l}^{\text{free}}$ which depends on 1/c instead of $A_{n,l}^{(0)}$ and without the crossing version. Let us look more closely at the contributions from the dynamical part. Since the conformal block expansion at order 1/c is proportional to $\log u$ (see (4.3.10)) it is useful to decompose G^{sugra} as follows:

$$G^{\text{sugra}}(z,\bar{z}) = \log u \ G_{\log}(z,\bar{z}) + G_{\text{non-log}}(z,\bar{z}) \ , \tag{G.0.7}$$

where only the $\log u$ piece is important for the computation of the $\gamma_{n,l}$ and the non-log piece is analytic as $u \to 0$. Now performing the conformal block expansion of G_{\log} gives the anomalous dimensions

$$\gamma_{n,l}^{\text{sugra}} = -\frac{3}{c} \left(\frac{n(n+3)}{2(l+3)(l+2n+8)} + 1 \right) \frac{(n+1)(n+2)(n+3)(n+4)(n+5)(n+6)}{(l+1)(l+2)(l+2n+9)(l+2(n+5))} , \tag{G.0.8}$$

where we divided by $A_{n,l}^{\text{free}}$. The supergravity OPE coefficients can then be computed from (4.2.9).

Note that in the large-*n* limit $\gamma_{n,l}^{\text{sugra}}$ goes like n^5 . In subsection 4.3.2 we compare the large-twist behaviour of the anomalous dimensions from spin-*L* truncations to this scaling in order to deduce the additional number of derivatives the spin-*L* contact interactions obtain compared to supergravity.

Appendix H

Quadratic Super Casimir and Correlator of Descendants

In this appendix we will derive the quadratic super Casimir of SU(1,1|2). This contains the second-order differential operator $\Delta^{(2)}$ which plays a leading role in the hidden four-dimensional conformal symmetry. Furthermore, we will sketch the calculation of the correlator of descendants introduced in subsection 5.2.1.

The superblocks (see (5.2.23) and (5.2.24)) are eigenfunctions of the quadratic super Casimir at points 1 and 2 acting on the correlator. Note that the formalism and super Casimir outlined below generalise naturally from the supergroup SU(1,1|2)to any supergroup of the form SU(m,m|2n). This was done in the bosonic case SU(m,m) in [49].

Consider the super Grassmannian Gr(1|1, 2|2), the space of $(1|1) \times (2|2)$ matrices $u^{\alpha}{}_{A}$. Here the small Greek indices refer to the local isotropy group GL(1|1) whilst the big Latin indices refer to the global group GL(2|2). Explicitly, one can put coordinates on this Grassmannian as

$$(u_i)^{\alpha}{}_B = \left(\delta^{\alpha}_{\beta}, (X_i)^{\alpha}{}_{\dot{\beta}}\right) , \quad \left(\bar{u}_i\right)^{B}{}_{\dot{\alpha}} = \left(\begin{array}{c} -(X_i)^{\beta}{}_{\dot{\alpha}} \\ \delta^{\dot{\beta}}_{\dot{\alpha}} \end{array}\right) , \quad (\text{H.0.1})$$

where X_i is a matrix containing the spacetime and internal coordinates x, y, and

Grassmann odd variables $\theta, \bar{\theta}$:

$$(X_i)^{\alpha}{}_{\dot{\beta}} = \begin{pmatrix} x_i & \theta_i \\ \bar{\theta}_i & y_i \end{pmatrix} \text{ and } X_{ij} = X_i - X_j .$$
(H.0.2)

We thus have $(u_i)^{\alpha}_{B} (\bar{u}_j)^{B}_{\dot{\alpha}} = (X_{ij})^{\alpha}_{\dot{\alpha}}$. Then the generators of the superconformal group SU(1,1|2) at point *i* are given as

$$D_{iB}^{A} = (u_{i})_{A}^{\alpha} \frac{\partial}{\partial (u_{i})_{B}^{\alpha}} , \qquad (\text{H.0.3})$$

The quadratic Casimir operator acting on the four-point function at points 1 and 2 is then given as

$$\mathcal{C}_{1,2}^{SU(1,1|2)} = \frac{1}{2} (D_{1B}^A + D_{2B}^A) (D_{1A}^B + D_{2A}^B) . \tag{H.0.4}$$

Superconformal symmetry SU(1, 1|2) fixes the correlator in terms of a conjugation invariant function of a cross-ratio matrix (see [49, 161]) to the form

$$\langle \Psi_{p_1}\Psi_{p_2}\Psi_{p_3}\Psi_{p_4}\rangle = P_{p_i} \times f(Z) \tag{H.0.5}$$

where the matrix of cross-ratios is given by

$$Z \coloneqq X_{12} X_{24}^{-1} X_{43} X_{31}^{-1} = \begin{pmatrix} x & \xi \\ \bar{\xi} & y \end{pmatrix}, \qquad (\text{H.0.6})$$

with Grassmann odd variables $\xi, \bar{\xi}$ and note that x, y here are different to the crossratios used in the main text, see [49, 161] for more details. That f(Z) is a conjugation invariant function of the cross-ratio matrix means that $f(Z) = f(G^{-1}ZG)$ where $G \in GL(2,2)$. Therefore, it can be diagonalised and the correlator can be written in terms of a function $f(\hat{x}, \hat{y})$ of the eigenvalues of Z, \hat{x}, \hat{y} only. These are given by

$$\hat{x} = x + \frac{\xi \bar{\xi}}{x - y}$$
 $\hat{y} = y + \frac{\xi \bar{\xi}}{x - y}$. (H.0.7)

That the eigenvalues of Z are given by (H.0.7) can be checked by verifying that the diagonal matrix of eigenvalues gives the same supertrace and superdeterminant as

the matrix of cross-ratios Z:

$$\operatorname{str}(Z) = x - y = \hat{x} - \hat{y}$$
, $\operatorname{sdet}(Z) = \frac{x - \xi \xi / y}{y} = \frac{\hat{x}}{\hat{y}}$. (H.0.8)

Now, acting with the Casimir on the supercorrelator in terms of a function depending on the eigenvalues of Z, $f(\hat{x}, \hat{y})$ (H.0.5), and commuting through the prefactor gives

$$\mathcal{C}_{1,2}^{SU(1,1|2)} \langle \Psi_{p_1} \Psi_{p_2} \Psi_{p_3} \Psi_{p_4} \rangle = P_{p_i} \times \left[\left((p_{12} - p_{34}) \left(x^2 \frac{\partial}{\partial x} + y^2 \frac{\partial}{\partial y} \right) + p_{34} p_{12} (x - y) \right) f + \frac{1}{2} \mathcal{C}_{1,2}^{SU(1,1|2)} f \right], \quad (\text{H.0.9})$$

with $p_{ij} = p_i - p_j$. To obtain this equation we used that $\operatorname{sdet}(M) = \exp(\operatorname{str}\log(M))$ to deal with differentiating the propagators $g_{ij} := \operatorname{sdet}(X_{ij}^{-1})$ and then applied the double derivative directly, using $D_{12B}^A(u_i)_C^\alpha = (u_i)_B^\alpha \delta_C^A$ and $D_{12A}^B(\bar{u}_i)_{\delta}^C = -\delta_A^C(\bar{u}_i)_{\delta}^B$ for i = 1 or 2.

Next, consider the Casimir acting on the conjugation invariant function f(Z) = f(x, y) of the cross-ratio matrix Z, which is a function of the eigenvalues x, y of Z only, as discussed above, where we drop the hat for simplicity. By examining the action of the Casimir on arbitrary products of traces of powers of Z, $\prod_i \operatorname{tr}(Z^i)^{a_i}$, and the corresponding expressions as polynomials of eigenvalues we find that

$$\mathcal{C}_{1,2}^{SU(1,1|2)}f(x,y) = \left[\left(x^2 \frac{\partial}{\partial x} - \frac{2xy}{x-y} \right) (1-x) \frac{\partial}{\partial x} - \left(y^2 \frac{\partial}{\partial y} - \frac{2yx}{y-x} \right) (1-y) \frac{\partial}{\partial y} \right] f(x,y)$$
$$= \frac{x-y}{xy} \left(x^2 \partial_x (1-x) \partial_x - y^2 \partial_y (1-y) \partial_y \right) \frac{xyf(x,y)}{x-y} .$$
(H.0.10)

Finally, combining equations (H.0.9) and (H.0.10), we obtain the action of the quadratic super Casimir on the correlator

$$\mathcal{C}_{1,2}^{SU(1,1|2)} = P_{p_i} \times \frac{x-y}{xy} \Delta^{(2)} \frac{xy}{x-y} P_{p_i}^{-1} ,$$

$$\Delta^{(2)} = \mathcal{D}_x^{(p_{12}, p_{43})} - \mathcal{D}_y^{(-p_{12}, -p_{43})} ,$$

$$\mathcal{D}_x^{(p_{12}, p_{43})} = x^2 \partial_x (1-x) \partial_x + (p_{12} + p_{43}) x^2 \partial_x - p_{12} p_{43} x .$$
(H.0.11)

The second-order differential operator $\Delta^{(2)}$ is the 1d analogue of $\Delta^{(8)}$ in $\mathcal{N} = 4$ SYM and is essential for the higher-dimensional conformal symmetry.

Performing similar, simpler computations one can also obtain the Casimirs of the subgroups SU(1|1) and SU(2) acting on the correlator as

$$C_{1,2}^{SU(1,1)} = P_{p_i} \times \mathcal{D}_x^{(p_{12}, p_{43})} P_{p_i}^{-1} ,$$

$$C_{1,2}^{SU(2)} = P_{p_i} \times \mathcal{D}_y^{(-p_{12}, -p_{43})} P_{p_i}^{-1} .$$
(H.0.12)

The action of the superconformal Casimir on the correlator can then be written directly in terms of those of the subgroups as:

$$\mathcal{C}_{1,2}^{SU(1,1|2)} = \frac{x-y}{xy} \left(\mathcal{C}_{1,2}^{SU(1|1)} - \mathcal{C}_{1,2}^{SU(2)} \right) \frac{xy}{x-y} . \tag{H.0.13}$$

Correlator of descendants

We will now compute the correlator of superconformal descendants

$$\partial_{\theta_1} \partial_{\theta_2} \partial_{\bar{\theta}_3} \partial_{\bar{\theta}_4} \langle \Psi_{p_1} \Psi_{p_2} \Psi_{p_3} \Psi_{p_4} \rangle |_{\theta_i = \bar{\theta}_i = 0} , \qquad (H.0.14)$$

introduced in subsection 5.2.1. As explained in (H.0.5) the supercorrelator is fixed by superconformal symmetry in terms of $f(\hat{x}, \hat{y})$ which only depends on the eigenvalues of the cross-ratio matrix [49, 161]. To compute the derivatives acting on the correlator of the form (H.0.5) using the definitions (H.0.7) one can use Mathematica with the help of the grassmann package [162] to deal with the Grassmann odd variables. One then finds that the Grassmann odd derivatives acting on f are consistent with the following differential operator:

$$\partial_{\theta_1} \partial_{\theta_2} \partial_{\bar{\theta}_3} \partial_{\bar{\theta}_4} f(\hat{x}, \hat{y}) |_{\theta_i = \bar{\theta}_i = 0} = \frac{1}{x_{12} x_{34} y_{12} y_{34}} \left(\mathcal{D}_{\hat{x}}^{(0,0)} - \mathcal{D}_{\hat{y}}^{(0,0)} \right) \frac{\hat{x} \hat{y}}{\hat{x} - \hat{y}} f(\hat{x}, \hat{y}) . \quad (\text{H.0.15})$$

Pulling this through the prefactor then gives the action on the correlator itself:

$$\begin{aligned} \partial_{\theta_{1}} \partial_{\theta_{2}} \partial_{\bar{\theta}_{3}} \partial_{\bar{\theta}_{4}} \langle \Psi_{p_{1}} \Psi_{p_{2}} \Psi_{p_{3}} \Psi_{p_{4}} \rangle |_{\theta_{i} = \bar{\theta}_{i} = 0} \\ &= \frac{P_{p_{i}}}{x_{12} x_{34} y_{12} y_{34}} \left(\mathcal{D}_{x}^{(p_{12}, p_{43})} - \mathcal{D}_{y}^{(-p_{12}, -p_{43})} \right) P_{p_{i}}^{-1} \frac{xy}{x - y} \langle \Psi_{p_{1}} \Psi_{p_{2}} \Psi_{p_{3}} \Psi_{p_{4}} \rangle |_{\theta_{i} = \bar{\theta}_{i} = 0} \\ &= \mathcal{I}^{-1} \mathcal{C}_{1,2}^{SU(1,1|2)} \langle \Psi_{p_{1}} \Psi_{p_{2}} \Psi_{p_{3}} \Psi_{p_{4}} \rangle |_{\theta_{i} = \bar{\theta}_{i} = 0} , \qquad (H.0.16) \end{aligned}$$

with \mathcal{I} defined in (5.2.16). The final line above relates the descendant correlator to the action of the superconformal Casimir and comes directly from (5.2.15). We thus obtain

$$\langle \phi_{\Delta_{1}+\frac{1}{2}}\phi_{\Delta_{2}+\frac{1}{2}}\phi_{\Delta_{3}+\frac{1}{2}}\phi_{\Delta_{4}+\frac{1}{2}} \rangle = \mathcal{I}^{-1}\mathcal{C}_{1,2}^{SU(1,1|2)} \langle \psi_{\Delta_{1}}\psi_{\Delta_{2}}\psi_{\Delta_{3}}\psi_{\Delta_{4}} \rangle$$

$$= \frac{1}{x_{12}x_{34}y_{12}y_{34}} \left(\mathcal{C}_{1,2}^{SU(1|1)} - \mathcal{C}_{1,2}^{SU(2)}\right) \frac{xy}{x-y} \langle \psi_{\Delta_{1}}\psi_{\Delta_{2}}\psi_{\Delta_{3}}\psi_{\Delta_{4}} \rangle ,$$

$$(H.0.17)$$

as written in (5.2.14).

Appendix I

Further Results for Unmixing of Four-Derivative Corrections

In this appendix we spell out some additional results relevant for section 5.8.

Unmixing at odd t for (5,2)

We start by considering the OPE coefficients for $(\Delta, p) = (5, 2)$, where the operators in the double-trace spectrum of the OPE are $\mathcal{O}_1\partial^2\mathcal{O}_3$, $\mathcal{O}_2\mathcal{O}_4$, $\mathcal{O}_2\partial^2\mathcal{O}_2$, $\mathcal{O}_3\mathcal{O}_3$ and $\mathcal{O}_2\partial\mathcal{O}_3$. Performing a conformal block analysis for free theory and supergravity gives the coefficients:

$$\hat{A}_{5,2}^{(0)} = \begin{pmatrix} A_{1313}^{(0)} & 0 & 0 & 0 & 0 \\ & A_{2424}^{(0)} & 0 & 0 & 0 \\ & & A_{2222}^{(0)} & 0 & 0 \\ & & & A_{3333}^{(0)} & 0 \\ & & & & & A_{3333}^{(0)} \end{pmatrix}_{(5,2)} = \begin{pmatrix} \frac{7}{120} & 0 & 0 & 0 & 0 \\ & \frac{5}{168} & 0 & 0 & 0 \\ & & \frac{5}{168} & 0 & 0 & 0 \\ & & & \frac{5}{168} & 0 \\ & & & \frac{5}{1$$

We spell out formulas for the coefficients at the order of supergravity and fourderivative corrections for general odd Δ (see (5.6.12) for general $A_{q_1q_2q_1q_2}^{(0)}$):

$$\begin{split} M_{1313}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta+1)(\Delta+2)\Delta!(\Delta+2)!}{30(2\Delta)!}, \\ M_{1324}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)(\Delta+1)(\Delta+2)(\Delta+4)\Delta!(\Delta+2)!}{420(2\Delta)!}, \\ M_{1322}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-1)(\Delta+2)\Delta!(\Delta+2)!}{15(2\Delta)!}, \\ M_{1333}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)(\Delta-1)(\Delta+2)(\Delta+4)\Delta!(\Delta+2)!}{210(2\Delta)!}, \\ M_{2424}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)^2(\Delta+1)(\Delta+2)(\Delta+4)^2\Delta!(\Delta+2)!}{5880(2\Delta)!}, \\ M_{2422}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)(\Delta-1)(\Delta+2)(\Delta+4)\Delta!(\Delta+2)!}{210(2\Delta)!}, \\ M_{2433}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)^2(\Delta-1)(\Delta+2)(\Delta+4)^2\Delta!(\Delta+2)!}{2940(2\Delta)!}, \\ M_{2233}^{\text{sugra}}(\Delta,2) &= \frac{2(\Delta-1)^2(\Delta+2)^2(\Delta!)^2}{15(2\Delta)!}, \\ M_{2333}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta-3)(\Delta-1)^2(\Delta+2)(\Delta+4)\Delta!(\Delta+2)!}{105(\Delta+1)(2\Delta)!}, \\ M_{3333}^{\text{sugra}}(\Delta,2) &= \frac{(\Delta+2)(\Delta^3-13\Delta+12)^2\Delta!(\Delta+2)!}{1470(\Delta+1)(2\Delta)!}, \end{split}$$
(I.0.2)

$$\begin{split} M_{1313}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{1}{60(2\Delta)!} \times \left[24 \, C_0 \left((\Delta + 2)! \right)^2 \\ &\quad (\Delta + 3) \left(\Delta(\Delta + 1) \left(\Delta^2 + \Delta - 5 \right) + 24 \right) (\Delta - 2)^3 (\Delta - 3)! (\Delta + 3)! \right], \\ M_{1324}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)(\Delta + 4)!}{840(2\Delta)!} \times \left[\frac{40 \, C_0 \left(\Delta + 2 \right)!}{\Delta + 3} \\ &\quad + (\Delta - 2) \left(\Delta(\Delta + 1) \left(\Delta(\Delta + 1) \left(\Delta^2 + \Delta - 15 \right) + 118 \right) - 400 \right) (\Delta - 2)! \right], \end{split}$$

$$\begin{split} M_{1322}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{\Delta!(\Delta + 2)!}{30(2\Delta)!} \times \left[20 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \left(\Delta - 2 \right) (\Delta + 3) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 10 \right) + 28 \right) \right], \\ M_{1333}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)\Delta!(\Delta + 4)!}{620(\Delta + 3)(2\Delta)!} \times \left[36 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \left(\Delta - 2 \right) (\Delta + 3) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 14 \right) + 60 \right) \right], \\ M_{2424}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)^2}{35280(2\Delta)!} \times \left[\frac{168 C_0 \left((\Delta + 4)! \right)^2}{(\Delta + 3)^2} + (\Delta + 4)^2 (\Delta - 2)!(\Delta + 2)! \right. \\ &\quad \times \left(\Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(3\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 25 \right) \right) \right. \\ &\quad + 1192 \right) - 8220 \right) + 20000 \right], \\ M_{2422}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)\Delta!(\Delta + 4)!}{420(\Delta + 3)(2\Delta)!} \times \left[36 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \left(\Delta - 2 \right) (\Delta + 3) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 14 \right) + 60 \right) \right], \\ M_{2423}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)^2 (\Delta + 4)\Delta!(\Delta + 4)!}{5880(\Delta + 3)(2\Delta)!} \times \left[52 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 24 \right) + 280 \right) - 904 \right], \\ M_{2222}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta^2 + \Delta - 2) (\Delta 1)^2}{15(2\Delta)!} \times \left[16 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 15 \right) + 74 \right) - 80 \right], \\ M_{2233}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)^2 (\Delta - 1)(\Delta + 4)\Delta!(\Delta + 2)!}{210(\Delta + 1)(2\Delta)!} \times \left[32 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 19 \right) + 130 \right) - 144 \right], \\ M_{333}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 3)^2 (\Delta - 1)(\Delta + 4)^2 \Delta!(\Delta + 2)!}{2940(\Delta + 1)(2\Delta)!} \times \left[48 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 23 \right) + 330 \right) - 432 \right], \\ M_{2323}^{4\text{-deriv}, t \text{ odd}}(\Delta, 2) &= \frac{(\Delta - 1)^3 (\Delta + 2) \left(\Delta^2 + \Delta - 12 \right)^2 (\Delta - 2)! (\Delta - 2)!}{2940(\Delta + 1)(2\Delta)!} \times \left[48 C_0 \left(\Delta^2 + \Delta - 2 \right) \right. \\ &\quad + \Delta (\Delta + 1) \left(\Delta (\Delta + 1) \left(\Delta^2 + \Delta - 23 \right) + 330 \right) - 432 \right], \end{aligned}$$

Now we can solve the mixing problem in the supergravity limit. This yields the anomalous dimension

$$\gamma_{5,2}^{\text{sugra}} = 24$$
, (I.0.4)

which is $\delta_{5,2}^{(2)}$ as expected. The mixing problem at $\mathcal{O}(a/c)$ is solved in section 5.8.

Unmixing at even t for (6,2)

We present an additional example for even t here. At weight (6,2) there are four operators in the spectrum, $\mathcal{O}_1\partial^3\mathcal{O}_3$, $\mathcal{O}_2\partial\mathcal{O}_4$, $\mathcal{O}_2\partial^2\mathcal{O}_3$ and $\mathcal{O}_3\mathcal{O}_4$, two with even and odd $r_{B^{\text{te}}}$ respectively. Performing a conformal block analysis of the relevant correlators gives the following free theory and four-derivative coefficients:

$$\hat{A}_{6,2}^{(0)} = \begin{pmatrix} A_{1313}^{(0)} & 0 & 0 & 0 \\ A_{2424}^{(0)} & 0 & 0 \\ & A_{2323}^{(0)} & 0 \\ & & A_{2323}^{(0)} & 0 \\ & & & A_{2323}^{(0)} & 0 \\ & & & & A_{2323}^{(0)} & 0 \\ & & & & & A_{2424}^{(0)} & M_{3434}^{4\text{-deriv}} & M_{1324}^{4\text{-deriv}} \\ & & & & & & M_{1324}^{4\text{-deriv}} & M_{1323}^{4\text{-deriv}} & M_{1334}^{4\text{-deriv}} \\ & & & & & & & M_{2423}^{4\text{-deriv}} & M_{2434}^{4\text{-deriv}} \\ & & & & & & & M_{2323}^{4\text{-deriv}} & M_{2334}^{4\text{-deriv}} \\ & & & & & & & & M_{3434}^{4\text{-deriv}} \\ & & & & & & & & M_{3434}^{4\text{-deriv}} \\ \end{pmatrix}_{(6,2)}^{408} = \begin{pmatrix} \frac{1224}{275} & \frac{408}{35} & 0 & 0 \\ \frac{408}{35} & \frac{1496}{499} & 0 & 0 \\ 0 & 0 & \frac{304}{55} & \frac{456}{35} \\ 0 & 0 & \frac{456}{35} & \frac{7524}{245} \end{pmatrix}.$$
(I.0.5)

As expected, $\hat{M}_{6,2}^{4\text{-deriv}}$ is a block-diagonal matrix, because operators with even or odd $r_{B^{\text{te}}}$ do not mix. Note that each block is symmetric, as it should be from the structure of the correlators. The free coefficients for general Δ , p, q_1 , q_2 are given in (5.6.12), some $\mathcal{O}(a/c)$ coefficients for general even Δ were given in (5.8.31) and the additional ones are

$$\begin{split} M_{1324}^{4\text{-deriv}, t \text{ even}}(\Delta, 2) &= \frac{(\Delta - 4)(\Delta - 2)^3(\Delta + 3)(\Delta + 5)(\Delta - 3)!(\Delta + 3)!(\Delta^2 + \Delta - 8)}{168(2\Delta)!} , \\ M_{2424}^{4\text{-deriv}, t \text{ even}}(\Delta, 2) &= \frac{5(\Delta - 2)^3(\Delta + 3)(\Delta - 3)!(\Delta + 3)!(\Delta^2 + \Delta - 20)^2(\Delta^2 + \Delta - 8)}{7056(2\Delta)!} , \\ M_{2334}^{4\text{-deriv}, t \text{ even}}(\Delta, 2) &= \frac{(\Delta - 4)(\Delta + 5)\Delta!(\Delta + 1)!(\Delta^2 + \Delta - 6)^2(\Delta^2 + \Delta - 4)}{105\Delta(2\Delta)!} , \\ M_{3434}^{4\text{-deriv}, t \text{ even}}(\Delta, 2) &= \frac{(\Delta - 4)^2(\Delta - 2)^2(\Delta + 3)^2(\Delta + 5)^2\Delta!(\Delta + 1)!(\Delta^2 + \Delta - 4)}{980\Delta(2\Delta)!} . \end{split}$$

(I.0.6)

Solving the unmixing equations, as expected, one finds two non-zero anomalous dimensions:

$$\gamma_{4,2,i}^{4\text{-deriv}} = \left\{ \frac{3672}{5}, \ \frac{4104}{5} \right\} , \qquad (I.0.7)$$

which are labelled by $\left(\gamma_{B^{\text{te}}}^{4\text{-deriv}}\right)_{1,0}^{6,2} = \frac{3672}{5}$ and $\left(\gamma_{B^{\text{te}}}^{4\text{-deriv}}\right)_{1,1}^{6,2} = \frac{4104}{5}$. Conjectured formulas for general (Δ, p) are discussed in subsection 5.8.3.

Bibliography

- Juan Martin Maldacena. 'The Large N limit of superconformal field theories and supergravity'. In: Adv. Theor. Math. Phys. 2 (1998), pp. 231–252. doi:10.1023/A:1026654312961. arXiv:hep-th/9711200.
- [2] E. Cremmer, B. Julia and Joel Scherk. 'Supergravity Theory in Eleven-Dimensions'. In: *Phys. Lett. B* 76 (1978), pp. 409–412. doi:10.1016/0370-2693(78)90894-8.
- [3] Ofer Aharony, Oren Bergman, Daniel Louis Jafferis and Juan Maldacena.
 'N=6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals'. In: *JHEP* 10 (2008), p. 091.
 doi:10.1088/1126-6708/2008/10/091.
 arXiv:0806.1218 [hep-th].
- B. Bertotti. 'Uniform electromagnetic field in the theory of general relativity'. In: *Phys. Rev.* 116 (1959), p. 1331. doi:10.1103/PhysRev.116.1331.
- [5] I. Robinson. 'A Solution of the Maxwell-Einstein Equations'. In: Bull. Acad. Pol. Sci. Ser. Sci. Math. Astron. Phys. 7 (1959), pp. 351–352.
- [6] M. A. Virasoro. 'Alternative constructions of crossing-symmetric amplitudes with regge behavior'. In: *Phys. Rev.* 177 (1969), pp. 2309–2311. doi:10.1103/PhysRev.177.2309.
- Joel A. Shapiro. 'Electrostatic analog for the virasoro model'. In: *Phys. Lett.* B 33 (1970), pp. 361–362. doi:10.1016/0370-2693(70)90255-8.
- [8] Michael B. Green, J. H. Schwarz and Edward Witten. SUPERSTRING THEORY. VOL. 1: INTRODUCTION. Cambridge Monographs on Mathematical Physics. July 1988.

- [9] Edward Witten. 'Anti-de Sitter space and holography'. In: Adv. Theor. Math. Phys. 2 (1998), pp. 253–291. doi:10.4310/ATMP.1998.v2.n2.a2. arXiv:hep-th/9802150.
- [10] Vasco Gonçalves. 'Four point function of N = 4 stress-tensor multiplet at strong coupling'. In: JHEP 04 (2015), p. 150. doi:10.1007/JHEP04(2015)150. arXiv:1411.1675 [hep-th].
- [11] Luis F. Alday, Agnese Bissi and Eric Perlmutter. 'Genus-One String Amplitudes from Conformal Field Theory'. In: JHEP 06 (2019), p. 010. doi:10.1007/JHEP06(2019)010. arXiv:1809.10670 [hep-th].
- [12] Damon J. Binder, Shai M. Chester, Silviu S. Pufu and Yifan Wang. 'N = 4 Super-Yang-Mills correlators at strong coupling from string theory and localization'. In: JHEP 12 (2019), p. 119. doi:10.1007/JHEP12(2019)119. arXiv:1902.06263 [hep-th].
- [13] J. M. Drummond, D. Nandan, H. Paul and K. S. Rigatos. 'String corrections to AdS amplitudes and the double-trace spectrum of N = 4 SYM'. In: JHEP 12 (2019), p. 173. doi:10.1007/JHEP12(2019)173. arXiv:1907.00992 [hep-th].
- J. M. Drummond, H. Paul and M. Santagata. 'Bootstrapping string theory on AdS₅ × S⁵'. In: (Apr. 2020). arXiv:2004.07282 [hep-th].
- [15] Francesco Aprile and Pedro Vieira. 'Large p explorations. From SUGRA to big STRINGS in Mellin space'. In: JHEP 12 (2020), p. 206. doi:10.1007/JHEP12(2020)206. arXiv:2007.09176 [hep-th].
- F. Aprile, J. M. Drummond, H. Paul and M. Santagata. 'The Virasoro-Shapiro amplitude in AdS₅×S⁵ and level splitting of 10d conformal symmetry'. In: (Dec. 2020). arXiv:2012.12092 [hep-th].
- [17] P. K. Townsend. 'The eleven-dimensional supermembrane revisited'. In: *Phys. Lett. B* 350 (1995), pp. 184–187. doi:10.1016/0370-2693(95)00397-4. arXiv:hep-th/9501068.

- [18] Edward Witten. 'String theory dynamics in various dimensions'. In: Nucl. Phys. B 443 (1995), pp. 85–126.
 doi:10.1016/0550-3213(95)00158-O.
 arXiv:hep-th/9503124.
- [19] Neil Lambert, Arthur Lipstein and Paul Richmond. 'Non-Lorentzian M5-brane Theories from Holography'. In: JHEP 08 (2019), p. 060. doi:10.1007/JHEP08(2019)060. arXiv:1904.07547 [hep-th].
- [20] Edward Witten. 'Geometric Langlands From Six Dimensions'. In: (May 2009).
 arXiv:0905.2720 [hep-th].
- S. Ferrara, A. F. Grillo and R. Gatto. 'Tensor representations of conformal algebra and conformally covariant operator product expansion'. In: Annals Phys. 76 (1973), pp. 161–188. doi:10.1016/0003-4916(73)90446-6.
- [22] S. Ferrara, A. F. Grillo, G. Parisi and Raoul Gatto. 'Covariant expansion of the conformal four-point function'. In: *Nucl. Phys. B* 49 (1972). [Erratum: Nucl.Phys.B 53, 643–643 (1973)], pp. 77–98. doi:10.1016/0550-3213(73)90467-7.
- [23] A. M. Polyakov. 'Nonhamiltonian approach to conformal quantum field theory'. In: Zh. Eksp. Teor. Fiz. 66 (1974), pp. 23–42.
- [24] Riccardo Rattazzi, Vyacheslav S. Rychkov, Erik Tonni and Alessandro Vichi.
 'Bounding scalar operator dimensions in 4D CFT'. In: *JHEP* 12 (2008),
 p. 031.
 doi:10.1088/1126-6708/2008/12/031.
 arXiv:0807.0004 [hep-th].
- [25] Christopher Beem, Madalena Lemos, Leonardo Rastelli and Balt C. van Rees. 'The (2, 0) superconformal bootstrap'. In: *Phys. Rev. D* 93.2 (2016), p. 025016. doi:10.1103/PhysRevD.93.025016. arXiv:1507.05637 [hep-th].
- [26] Paul Heslop and Arthur E. Lipstein. 'M-theory Beyond The Supergravity Approximation'. In: JHEP 02 (2018), p. 004. doi:10.1007/JHEP02(2018)004. arXiv:1712.08570 [hep-th].

- Shai M. Chester, Silviu S. Pufu and Xi Yin. 'The M-Theory S-Matrix From ABJM: Beyond 11D Supergravity'. In: JHEP 08 (2018), p. 115. doi:10.1007/JHEP08(2018)115. arXiv:1804.00949 [hep-th].
- Shai M. Chester and Eric Perlmutter. 'M-Theory Reconstruction from (2,0)
 CFT and the Chiral Algebra Conjecture'. In: JHEP 08 (2018), p. 116.
 doi:10.1007/JHEP08(2018)116.
 arXiv:1805.00892 [hep-th].
- [29] Shai M. Chester. 'AdS₄/CFT₃ for unprotected operators'. In: *JHEP* 07 (2018), p. 030.
 doi:10.1007/JHEP07(2018)030.
 arXiv:1803.01379 [hep-th].
- [30] Idse Heemskerk, Joao Penedones, Joseph Polchinski and James Sully.
 'Holography from Conformal Field Theory'. In: JHEP 10 (2009), p. 079. doi:10.1088/1126-6708/2009/10/079. arXiv:0907.0151 [hep-th].
- [31] Andrew Strominger. 'AdS(2) quantum gravity and string theory'. In: JHEP 01 (1999), p. 007.
 doi:10.1088/1126-6708/1999/01/007.
 arXiv:hep-th/9809027.
- [32] G. W. Gibbons and P. K. Townsend. 'Black holes and Calogero models'. In: *Phys. Lett. B* 454 (1999), pp. 187–192. doi:10.1016/S0370-2693(99)00266-X. arXiv:hep-th/9812034.
- [33] R. Jackiw. 'Lower Dimensional Gravity'. In: Nucl. Phys. B 252 (1985).
 Ed. by R. Baier and H. Satz, pp. 343–356.
 doi:10.1016/0550-3213(85)90448-1.
- [34] C. Teitelboim. 'Gravitation and Hamiltonian Structure in Two Space-Time Dimensions'. In: *Phys. Lett. B* 126 (1983), pp. 41–45. doi:10.1016/0370-2693(83)90012-6.
- [35] Subir Sachdev and Jinwu Ye. 'Gapless spin-fluid ground state in a random quantum Heisenberg magnet'. In: *Physical Review Letters* 70.21 (May 1993), pp. 3339–3342. http://dx.doi.org/10.1103/PhysRevLett.70.3339
 doi:10.1103/physrevlett.70.3339.

- [36] Alexei Kitaev and S. Josephine Suh. 'The soft mode in the Sachdev-Ye-Kitaev model and its gravity dual'. In: Journal of High Energy Physics 2018.5 (May 2018). http://dx.doi.org/10.1007/JHEP05(2018)183 doi:10.1007/jhep05(2018)183.
- [37] Phil Saad, Stephen H. Shenker and Douglas Stanford. 'JT gravity as a matrix integral'. In: (Mar. 2019). arXiv:1903.11115 [hep-th].
- [38] Douglas Stanford and Edward Witten. 'JT Gravity and the Ensembles of Random Matrix Theory'. In: (July 2019). arXiv:1907.03363 [hep-th].
- [39] Gábor Sárosi. 'AdS₂ holography and the SYK model'. In: *PoS* Modave2017 (2018), p. 001.
 doi:10.22323/1.323.0001.
 arXiv:1711.08482 [hep-th].
- [40] Upamanyu Moitra, Sunil Kumar Sake, Sandip P. Trivedi and V. Vishal.
 'Jackiw-Teitelboim Gravity and Rotating Black Holes'. In: *JHEP* 11 (2019), p. 047. doi:10.1007/JHEP11(2019)047. arXiv:1905.10378 [hep-th].
- [41] Simon Caron-Huot and Anh-Khoi Trinh. 'All tree-level correlators in AdS₅×S₅ supergravity: hidden ten-dimensional conformal symmetry'. In: JHEP 01 (2019), p. 196. doi:10.1007/JHEP01(2019)196. arXiv:1809.09173 [hep-th].
- [42] Theresa Abl, Paul Heslop and Arthur E. Lipstein. 'Towards the Virasoro-Shapiro amplitude in $AdS_5 \times S^5$ '. In: *JHEP* 04 (2021), p. 237. doi:10.1007/JHEP04(2021)237. arXiv:2012.12091 [hep-th].
- [43] Theresa Abl, Paul Heslop and Arthur E. Lipstein. 'Recursion relations for anomalous dimensions in the 6d (2,0) theory'. In: JHEP 04 (2019), p. 038. doi:10.1007/JHEP04(2019)038. arXiv:1902.00463 [hep-th].
- [44] Theresa Abl, Paul Heslop and Arthur E. Lipstein. In: (2021). In preparation.

- [45] Slava Rychkov. 'EPFL Lectures on Conformal Field Theory in $D \ge 3$ Dimensions'. In: SpringerBriefs in Physics (2016). http://inspirehep.net/record/1415968/files/arXiv:1601.05000.pdf doi:10.1007/978-3-319-43626-5. arXiv:1601.05000 [hep-th].
- [46] F. A. Dolan and H. Osborn. 'Conformal partial waves and the operator product expansion'. In: *Nucl. Phys. B* 678 (2004), pp. 491–507. doi:10.1016/j.nuclphysb.2003.11.016. arXiv:hep-th/0309180.
- [47] P. J. Heslop. 'Aspects of superconformal field theories in six dimensions'. In: JHEP 07 (2004), p. 056. doi:10.1088/1126-6708/2004/07/056. arXiv:hep-th/0405245.
- [48] Matthijs Hogervorst. 'Dimensional Reduction for Conformal Blocks'. In: JHEP 09 (2016), p. 017. doi:10.1007/JHEP09(2016)017. arXiv:1604.08913 [hep-th].
- [49] Reza Doobary and Paul Heslop. 'Superconformal partial waves in Grassmannian field theories'. In: JHEP 12 (2015), p. 159. doi:10.1007/JHEP12(2015)159. arXiv:1508.03611 [hep-th].
- [50] S. S. Gubser, Igor R. Klebanov and Alexander M. Polyakov. 'Gauge theory correlators from noncritical string theory'. In: *Phys. Lett. B* 428 (1998), pp. 105–114. doi:10.1016/S0370-2693(98)00377-3. arXiv:hep-th/9802109.
- [51] K. Becker, M. Becker and J. H. Schwarz. String theory and M-theory: A modern introduction. Cambridge University Press, Dec. 2006.
- [52] Eric D'Hoker and Daniel Z. Freedman. 'Supersymmetric gauge theories and the AdS / CFT correspondence'. In: Strings, Branes and Extra Dimensions: TASI 2001: Proceedings. 2002, pp. 3–158. arXiv:hep-th/0201253 [hep-th].
- [53] Joseph Polchinski. 'S matrices from AdS space-time'. In: (Jan. 1999). arXiv:hep-th/9901076.
- [54] Leonard Susskind. 'Holography in the flat space limit'. In: AIP Conf. Proc. 493.1 (1999). Ed. by C. P. Burgess and Robert C. Myers, pp. 98–112. doi:10.1063/1.1301570.
 arXiv:hep-th/9901079.

- [55] Steven B. Giddings. 'The Boundary S matrix and the AdS to CFT dictionary'. In: *Phys. Rev. Lett.* 83 (1999), pp. 2707–2710. doi:10.1103/PhysRevLett.83.2707. arXiv:hep-th/9903048.
- [56] Steven B. Giddings. 'Flat space scattering and bulk locality in the AdS / CFT correspondence'. In: *Phys. Rev. D* 61 (2000), p. 106008. doi:10.1103/PhysRevD.61.106008. arXiv:hep-th/9907129.
- [57] Joao Penedones. 'Writing CFT correlation functions as AdS scattering amplitudes'. In: JHEP 03 (2011), p. 025. doi:10.1007/JHEP03(2011)025. arXiv:1011.1485 [hep-th].
- [58] Charlotte Sleight. 'Interactions in Higher-Spin Gravity: a Holographic Perspective'. In: J. Phys. A 50.38 (2017), p. 383001. doi:10.1088/1751-8121/aa820c. arXiv:1610.01318 [hep-th].
- [59] Igor R. Klebanov and Edward Witten. 'AdS / CFT correspondence and symmetry breaking'. In: *Nucl. Phys. B* 556 (1999), pp. 89–114. doi:10.1016/S0550-3213(99)00387-9. arXiv:hep-th/9905104.
- [60] G. Arutyunov, F. A. Dolan, H. Osborn and E. Sokatchev. 'Correlation functions and massive Kaluza-Klein modes in the AdS / CFT correspondence'. In: *Nucl. Phys. B* 665 (2003), pp. 273–324. doi:10.1016/S0550-3213(03)00448-6. arXiv:hep-th/0212116.
- [61] Luis F. Alday, Agnese Bissi and Tomasz Lukowski. 'Lessons from crossing symmetry at large N'. In: JHEP 06 (2015), p. 074. doi:10.1007/JHEP06(2015)074. arXiv:1410.4717 [hep-th].
- [62] G. Arutyunov and S. Frolov. 'On the correspondence between gravity fields and CFT operators'. In: *JHEP* 04 (2000), p. 017. doi:10.1088/1126-6708/2000/04/017. arXiv:hep-th/0003038.
- [63] Gleb Arutyunov, Rob Klabbers and Sergei Savin. 'Four-point functions of 1/2-BPS operators of any weights in the supergravity approximation'. In: JHEP 09 (2018), p. 118. doi:10.1007/JHEP09(2018)118. arXiv:1808.06788 [hep-th].
- [64] F. Aprile, J. M. Drummond, P. Heslop, H. Paul, F. Sanfilippo, M. Santagata and A. Stewart. 'Single particle operators and their correlators in free N = 4 SYM'. In: JHEP 11 (2020), p. 072. doi:10.1007/JHEP11(2020)072. arXiv:2007.09395 [hep-th].
- [65] Hong Liu and Arkady A. Tseytlin. 'On four point functions in the CFT / AdS correspondence'. In: *Phys. Rev. D* 59 (1999), p. 086002. doi:10.1103/PhysRevD.59.086002. arXiv:hep-th/9807097.
- [66] Eric D'Hoker, Daniel Z. Freedman, Samir D. Mathur, Alec Matusis and Leonardo Rastelli. 'Graviton exchange and complete four point functions in the AdS / CFT correspondence'. In: Nucl. Phys. B 562 (1999), pp. 353–394. doi:10.1016/S0550-3213(99)00525-8. arXiv:hep-th/9903196.
- [67] G. Arutyunov and S. Frolov. 'Scalar quartic couplings in type IIB supergravity on AdS(5) x S⁵'. In: *Nucl. Phys. B* 579 (2000), pp. 117–176. doi:10.1016/S0550-3213(00)00210-8. arXiv:hep-th/9912210.
- [68] G. Arutyunov and S. Frolov. 'Four point functions of lowest weight CPOs in N=4 SYM(4) in supergravity approximation'. In: *Phys. Rev. D* 62 (2000), p. 064016. doi:10.1103/PhysRevD.62.064016. arXiv:hep-th/0002170.
- [69] G. Arutyunov and Emery Sokatchev. 'On a large N degeneracy in N=4 SYM and the AdS / CFT correspondence'. In: Nucl. Phys. B 663 (2003), pp. 163–196. doi:10.1016/S0550-3213(03)00353-5. arXiv:hep-th/0301058.
- [70] F. A. Dolan, M. Nirschl and H. Osborn. 'Conjectures for large N superconformal N=4 chiral primary four point functions'. In: *Nucl. Phys. B* 749 (2006), pp. 109–152. doi:10.1016/j.nuclphysb.2006.05.009. arXiv:hep-th/0601148.
- [71] Leon Berdichevsky and Pieter Naaijkens. 'Four-point functions of different-weight operators in the AdS/CFT correspondence'. In: JHEP 01 (2008), p. 071. doi:10.1088/1126-6708/2008/01/071. arXiv:0709.1365 [hep-th].

- [72] Linda I. Uruchurtu. 'Four-point correlators with higher weight superconformal primaries in the AdS/CFT Correspondence'. In: JHEP 03 (2009), p. 133. doi:10.1088/1126-6708/2009/03/133. arXiv:0811.2320 [hep-th].
- [73] Linda I. Uruchurtu. 'Next-next-to-extremal Four Point Functions of N=4 1/2 BPS Operators in the AdS/CFT Correspondence'. In: JHEP 08 (2011), p. 133. doi:10.1007/JHEP08(2011)133. arXiv:1106.0630 [hep-th].
- [74] Gleb Arutyunov, Rob Klabbers and Sergei Savin. 'Four-point functions of all-different-weight chiral primary operators in the supergravity approximation'. In: *JHEP* 09 (2018), p. 023. doi:10.1007/JHEP09(2018)023. arXiv:1806.09200 [hep-th].
- [75] R. R. Metsaev and Arkady A. Tseytlin. 'Type IIB superstring action in AdS(5) x S⁵ background'. In: *Nucl. Phys. B* 533 (1998), pp. 109–126. doi:10.1016/S0550-3213(98)00570-7. arXiv:hep-th/9805028.
- [76] Gleb Arutyunov and Sergey Frolov. 'Foundations of the $AdS_5 \times S^5$ Superstring. Part I'. In: J. Phys. A 42 (2009), p. 254003. doi:10.1088/1751-8113/42/25/254003. arXiv:0901.4937 [hep-th].
- [77] Nathan Berkovits. 'Super Poincare covariant quantization of the superstring'. In: JHEP 04 (2000), p. 018. doi:10.1088/1126-6708/2000/04/018. arXiv:hep-th/0001035.
- [78] Nathan Berkovits and Thiago Fleury. 'Harmonic Superspace from the $AdS_5 \times S^5$ Pure Spinor Formalism'. In: *JHEP* 03 (2013), p. 022. doi:10.1007/JHEP03(2013)022. arXiv:1212.3296 [hep-th].
- [79] Thales Azevedo and Nathan Berkovits. 'Open-closed superstring amplitudes using vertex operators in $AdS_5 \times S^5$ '. In: *JHEP* 02 (2015), p. 107. doi:10.1007/JHEP02(2015)107. arXiv:1412.5921 [hep-th].

- [80] Osvaldo Chandia and Brenno Carlini Vallilo. 'A superfield realization of the integrated vertex operator in an AdS₅ × S⁵ background'. In: JHEP 10 (2017), p. 178.
 doi:10.1007/JHEP10(2017)178.
 arXiv:1709.05517 [hep-th].
- [81] Leonardo Rastelli and Xinan Zhou. 'Mellin amplitudes for $AdS_5 \times S^5$ '. In: *Phys. Rev. Lett.* 118.9 (2017), p. 091602. doi:10.1103/PhysRevLett.118.091602. arXiv:1608.06624 [hep-th].
- [82] Leonardo Rastelli and Xinan Zhou. 'How to Succeed at Holographic Correlators Without Really Trying'. In: JHEP 04 (2018), p. 014. doi:10.1007/JHEP04(2018)014. arXiv:1710.05923 [hep-th].
- [83] Luis F. Alday and Xinan Zhou. 'All Holographic Four-Point Functions in All Maximally Supersymmetric CFTs'. In: *Phys. Rev. X* 11.1 (2021), p. 011056. doi:10.1103/PhysRevX.11.011056. arXiv:2006.12505 [hep-th].
- [84] Luis F. Alday and Agnese Bissi. 'Loop Corrections to Supergravity on $AdS_5 \times S^5$ '. In: *Phys. Rev. Lett.* 119.17 (2017), p. 171601. doi:10.1103/PhysRevLett.119.171601. arXiv:1706.02388 [hep-th].
- [85] F. Aprile, J. M. Drummond, P. Heslop and H. Paul. 'Quantum Gravity from Conformal Field Theory'. In: JHEP 01 (2018), p. 035. doi:10.1007/JHEP01(2018)035. arXiv:1706.02822 [hep-th].
- [86] Luis F. Alday and Simon Caron-Huot. 'Gravitational S-matrix from CFT dispersion relations'. In: JHEP 12 (2018), p. 017. doi:10.1007/JHEP12(2018)017. arXiv:1711.02031 [hep-th].
- [87] F. Aprile, J. M. Drummond, P. Heslop and H. Paul. 'Loop corrections for Kaluza-Klein AdS amplitudes'. In: *JHEP* 05 (2018), p. 056. doi:10.1007/JHEP05(2018)056. arXiv:1711.03903 [hep-th].
- [88] Francesco Aprile, James Drummond, Paul Heslop and Hynek Paul.
 'One-loop amplitudes in AdS₅×S⁵ supergravity from N = 4 SYM at strong coupling'. In: JHEP 03 (2020), p. 190. doi:10.1007/JHEP03(2020)190. arXiv:1912.01047 [hep-th].

- [89] Luis F. Alday and Xinan Zhou. 'Simplicity of AdS Supergravity at One Loop'. In: JHEP 09 (2020), p. 008. doi:10.1007/JHEP09(2020)008. arXiv:1912.02663 [hep-th].
- [90] Luis F. Alday. 'On Genus-one String Amplitudes on $AdS_5 \times S^5$ '. In: (Dec. 2018). arXiv:1812.11783 [hep-th].
- [91] J. M. Drummond and H. Paul. 'One-loop string corrections to AdS amplitudes from CFT'. In: JHEP 03 (2021), p. 038. doi:10.1007/JHEP03(2021)038. arXiv:1912.07632 [hep-th].
- [92] J. M. Drummond, R. Glew and H. Paul. 'One-loop string corrections for AdS Kaluza-Klein amplitudes'. In: (Aug. 2020). arXiv:2008.01109 [hep-th].
- [93] Paul S. Howe and Peter C. West. 'The Complete N=2, D=10 Supergravity'. In: Nucl. Phys. B 238 (1984), pp. 181–220. doi:10.1016/0550-3213(84)90472-3.
- [94] Michael B. Green and Congkao Wen. 'Modular Forms and SL(2,Z)-covariance of type IIB superstring theory'. In: JHEP 06 (2019), p. 087. doi:10.1007/JHEP06(2019)087. arXiv:1904.13394 [hep-th].
- [95] Michael B. Green. 'Interconnections between type II superstrings, M theory and N=4 supersymmetric Yang-Mills'. In: *Lect. Notes Phys.* 525 (1999).
 Ed. by L. Brink and R. Marnelius, p. 22. doi:10.1007/BFb0104240.
 arXiv:hep-th/9903124.
- [96] Heng-Yu Chen and Jun-ichi Sakamoto. 'Superconformal Block from Holographic Geometry'. In: JHEP 07 (2020), p. 028. doi:10.1007/JHEP07(2020)028. arXiv:2003.13343 [hep-th].
- [97] Shai M. Chester and Silviu S. Pufu. 'Far beyond the planar limit in strongly-coupled $\mathcal{N} = 4$ SYM'. In: *JHEP* 01 (2021), p. 103. doi:10.1007/JHEP01(2021)103. arXiv:2003.08412 [hep-th].

- [98] Shai M. Chester, Michael B. Green, Silviu S. Pufu, Yifan Wang and Congkao Wen. 'New Modular Invariants in N = 4 Super-Yang-Mills Theory'. In: (Aug. 2020).
 arXiv:2008.02713 [hep-th].
- [99] Henriette Elvang and Yu-tin Huang. Scattering Amplitudes in Gauge Theory and Gravity. Cambridge University Press, Apr. 2015.
- [100] Johannes M. Henn and Jan C. Plefka. Scattering Amplitudes in Gauge Theories. Vol. 883. Berlin: Springer, 2014. doi:10.1007/978-3-642-54022-6.
- [101] Stanley Mandelstam. 'Light Cone Superspace and the Ultraviolet Finiteness of the N=4 Model'. In: Nucl. Phys. B213 (1983), pp. 149–168.
- [102] Lars Brink, Olof Lindgren and Bengt E. W. Nilsson. 'The Ultraviolet Finiteness of the N=4 Yang-Mills Theory'. In: *Phys. Lett.* B123 (1983), pp. 323–328.
- [103] Niklas Beisert et al. 'Review of AdS/CFT Integrability: An Overview'. In: Lett. Math. Phys. 99 (2012), pp. 3–32. doi:10.1007/s11005-011-0529-2. arXiv:1012.3982 [hep-th].
- [104] Lance J. Dixon, Johannes M. Henn, Jan Plefka and Theodor Schuster. 'All tree-level amplitudes in massless QCD'. In: JHEP 01 (2011), p. 035. doi:10.1007/JHEP01(2011)035. arXiv:1010.3991 [hep-ph].
- [105] Lars Brink, John H. Schwarz and Joel Scherk. 'Supersymmetric Yang-Mills Theories'. In: *Nucl. Phys. B* 121 (1977), pp. 77–92. doi:10.1016/0550-3213(77)90328-5.
- [106] Michael B. Green and John H. Schwarz. 'Supersymmetrical String Theories'. In: *Phys. Lett. B* 109 (1982), pp. 444–448. doi:10.1016/0370-2693(82)91110-8.
- [107] John H. Schwarz. 'Covariant Field Equations of Chiral N=2 D=10 Supergravity'. In: Nucl. Phys. B 226 (1983). Ed. by A. Salam and E. Sezgin, p. 269. doi:10.1016/0550-3213(83)90192-X.
- [108] Sangmin Lee, Shiraz Minwalla, Mukund Rangamani and Nathan Seiberg.
 'Three point functions of chiral operators in D = 4, N=4 SYM at large N'. In: Adv. Theor. Math. Phys. 2 (1998), pp. 697–718. doi:10.4310/ATMP.1998.v2.n4.a1. arXiv:hep-th/9806074.

- [109] Burkhard Eden, Anastasios C. Petkou, Christian Schubert and Emery Sokatchev. 'Partial nonrenormalization of the stress tensor four point function in N=4 SYM and AdS / CFT'. In: Nucl. Phys. B 607 (2001), pp. 191–212. doi:10.1016/S0550-3213(01)00151-1. arXiv:hep-th/0009106.
- [110] Michael B. Green and Congkao Wen. 'Maximal U(1)_Y-violating n-point correlators in $\mathcal{N} = 4$ super-Yang-Mills theory'. In: *JHEP* 02 (2021), p. 042. doi:10.1007/JHEP02(2021)042. arXiv:2009.01211 [hep-th].
- [111] Michael B. Green, Michael Gutperle and Pierre Vanhove. 'One loop in eleven-dimensions'. In: *Phys. Lett. B* 409 (1997), pp. 177–184. doi:10.1016/S0370-2693(97)00931-3. arXiv:hep-th/9706175.
- [112] Michael B. Green and Savdeep Sethi. 'Supersymmetry constraints on type IIB supergravity'. In: *Phys. Rev. D* 59 (1999), p. 046006. doi:10.1103/PhysRevD.59.046006. arXiv:hep-th/9808061.
- Michael B. Green, Hwang-h. Kwon and Pierre Vanhove. 'Two loops in eleven-dimensions'. In: *Phys. Rev. D* 61 (2000), p. 104010. doi:10.1103/PhysRevD.61.104010. arXiv:hep-th/9910055.
- Michael B. Green and Pierre Vanhove. 'Duality and higher derivative terms in M theory'. In: JHEP 01 (2006), p. 093. doi:10.1088/1126-6708/2006/01/093. arXiv:hep-th/0510027.
- [115] Michael B. Green, Stephen D. Miller and Pierre Vanhove.
 'SL(2,Z)-invariance and D-instanton contributions to the D⁶R⁴ interaction'. In: Commun. Num. Theor. Phys. 09 (2015), pp. 307–344. doi:10.4310/CNTP.2015.v9.n2.a3. arXiv:1404.2192 [hep-th].
- [116] Paul S. Aspinwall. 'Some relationships between dualities in string theory'. In: Nucl. Phys. B Proc. Suppl. 46 (1996). Ed. by E. Gava, K. S. Narain and C. Vafa, pp. 30–38. doi:10.1016/0920-5632(96)00004-7. arXiv:hep-th/9508154.

- [117] John H. Schwarz. 'The power of M theory'. In: *Phys. Lett. B* 367 (1996), pp. 97–103. doi:10.1016/0370-2693(95)01429-2. arXiv:hep-th/9510086.
- [118] Shai M. Chester, Michael B. Green, Silviu S. Pufu, Yifan Wang and Congkao Wen. 'Modular invariance in superstring theory from N = 4 super-Yang-Mills'. In: JHEP 11 (2020), p. 016. doi:10.1007/JHEP11(2020)016. arXiv:1912.13365 [hep-th].
- [119] P. Heslop and Paul S. Howe. 'Chiral superfields in IIB supergravity'. In: *Phys. Lett. B* 502 (2001), pp. 259–264. doi:10.1016/S0370-2693(01)00149-6. arXiv:hep-th/0008047.
- [120] P. Heslop. PhD thesis. King's College London, 2001.
- [121] Sebastian de Haro, Annamaria Sinkovics and Kostas Skenderis. 'On a supersymmetric completion of the R4 term in 2B supergravity'. In: *Phys. Rev. D* 67 (2003), p. 084010.
 doi:10.1103/PhysRevD.67.084010.
 arXiv:hep-th/0210080.
- [122] N. Berkovits and P. Howe. '(private communication)'. In: ().
- [123] Arvind Rajaraman. 'On a supersymmetric completion of the R⁴ term in type IIB supergravity'. In: *Phys. Rev. D* 72 (2005), p. 125008. doi:10.1103/PhysRevD.74.085018. arXiv:hep-th/0505155.
- [124] Leonardo Rastelli, Konstantinos Roumpedakis and Xinan Zhou. 'AdS₃ × S³ Tree-Level Correlators: Hidden Six-Dimensional Conformal Symmetry'. In: JHEP 10 (2019), p. 140. doi:10.1007/JHEP10(2019)140. arXiv:1905.11983 [hep-th].
- [125] Stefano Giusto, Rodolfo Russo, Alexander Tyukov and Congkao Wen. 'The CFT₆ origin of all tree-level 4-point correlators in AdS₃ × S³'. In: *Eur. Phys. J. C* 80.8 (2020), p. 736. doi:10.1140/epjc/s10052-020-8300-4. arXiv:2005.08560 [hep-th].
- B. Eden, Paul S. Howe and Peter C. West. 'Nilpotent invariants in N=4 SYM'. In: *Phys. Lett. B* 463 (1999), pp. 19–26. doi:10.1016/S0370-2693(99)00705-4. arXiv:hep-th/9905085.

- [127] Paul S. Howe, C. Schubert, E. Sokatchev and Peter C. West. 'Explicit construction of nilpotent covariants in N=4 SYM'. In: *Nucl. Phys. B* 571 (2000), pp. 71–90. doi:10.1016/S0550-3213(99)00768-3. arXiv:hep-th/9910011.
- Burkhard Eden, Paul Heslop, Gregory P. Korchemsky and Emery Sokatchev.
 'Hidden symmetry of four-point correlation functions and amplitudes in N=4 SYM'. In: Nucl. Phys. B 862 (2012), pp. 193–231. doi:10.1016/j.nuclphysb.2012.04.007. arXiv:1108.3557 [hep-th].
- [129] Leonardo Rastelli and Xinan Zhou. 'Holographic Four-Point Functions in the (2, 0) Theory'. In: JHEP 06 (2018), p. 087. doi:10.1007/JHEP06(2018)087. arXiv:1712.02788 [hep-th].
- [130] Luis F. Alday and Xinan Zhou. 'All Tree-Level Correlators for M-theory on $AdS_7 \times S^4$ '. In: *Phys. Rev. Lett.* 125.13 (2020), p. 131604. doi:10.1103/PhysRevLett.125.131604. arXiv:2006.06653 [hep-th].
- [131] M. Henningson and K. Skenderis. 'The Holographic Weyl anomaly'. In: JHEP 07 (1998), p. 023. doi:10.1088/1126-6708/1998/07/023. arXiv:hep-th/9806087.
- [132] Christopher Beem, Leonardo Rastelli and Balt C. van Rees. 'W symmetry in six dimensions'. In: JHEP 05 (2015), p. 017. doi:10.1007/JHEP05(2015)017. arXiv:1404.1079 [hep-th].
- [133] Paul S. Howe, G. Sierra and P. K. Townsend. 'Supersymmetry in Six-Dimensions'. In: *Nucl. Phys. B* 221 (1983), pp. 331–348. doi:10.1016/0550-3213(83)90582-5.
- [134] Malcolm Perry and John H. Schwarz. 'Interacting chiral gauge fields in six-dimensions and Born-Infeld theory'. In: *Nucl. Phys. B* 489 (1997), pp. 47–64. doi:10.1016/S0550-3213(97)00040-0. arXiv:hep-th/9611065.
- [135] Paolo Pasti, Dmitri P. Sorokin and Mario Tonin. 'Covariant action for a D = 11 five-brane with the chiral field'. In: *Phys. Lett. B* 398 (1997), pp. 41–46. doi:10.1016/S0370-2693(97)00188-3. arXiv:hep-th/9701037.

- [136] Michael B. Green and Pierre Vanhove. 'D instantons, strings and M theory'. In: *Phys. Lett. B* 408 (1997), pp. 122–134. doi:10.1016/S0370-2693(97)00785-5. arXiv:hep-th/9704145.
- [137] J. G. Russo and Arkady A. Tseytlin. 'One loop four graviton amplitude in eleven-dimensional supergravity'. In: *Nucl. Phys. B* 508 (1997), pp. 245–259. doi:10.1016/S0550-3213(97)00631-7. arXiv:hep-th/9707134.
- [138] Fiorenzo Bastianelli and Roberto Zucchini. 'Three point functions of chiral primary operators in d = 3, N=8 and d = 6, N=(2,0) SCFT at large N'. In: *Phys. Lett. B* 467 (1999), pp. 61–66. doi:10.1016/S0370-2693(99)01179-X. arXiv:hep-th/9907047.
- [139] G. Arutyunov and E. Sokatchev. 'Implications of superconformal symmetry for interacting (2,0) tensor multiplets'. In: *Nucl. Phys. B* 635 (2002), pp. 3–32. doi:10.1016/S0550-3213(02)00359-0. arXiv:hep-th/0201145.
- [140] A. Liam Fitzpatrick and Jared Kaplan. 'Unitarity and the Holographic S-Matrix'. In: JHEP 10 (2012), p. 032. doi:10.1007/JHEP10(2012)032. arXiv:1112.4845 [hep-th].
- [141] Francis A. Dolan, Laurent Gallot and Emery Sokatchev. 'On four-point functions of 1/2-BPS operators in general dimensions'. In: *JHEP* 09 (2004), p. 056. doi:10.1088/1126-6708/2004/09/056.

doi:10.1088/1126-6708/2004/09/056 arXiv:hep-th/0405180.

- [142] Xinan Zhou. 'On Superconformal Four-Point Mellin Amplitudes in Dimension d > 2'. In: JHEP 08 (2018), p. 187. doi:10.1007/JHEP08(2018)187. arXiv:1712.02800 [hep-th].
- [143] Christopher Beem, Madalena Lemos, Pedro Liendo, Wolfger Peelaers, Leonardo Rastelli and Balt C. van Rees. 'Infinite Chiral Symmetry in Four Dimensions'. In: *Commun. Math. Phys.* 336.3 (2015), pp. 1359–1433. doi:10.1007/s00220-014-2272-x. arXiv:1312.5344 [hep-th].

- [144] F. Aprile, J. M. Drummond, P. Heslop and H. Paul. 'Unmixing Supergravity'. In: *JHEP* 02 (2018), p. 133. doi:10.1007/JHEP02(2018)133. arXiv:1706.08456 [hep-th].
- [145] Francesco Aprile, James Drummond, Paul Heslop and Hynek Paul.
 'Double-trace spectrum of N = 4 supersymmetric Yang-Mills theory at strong coupling'. In: *Phys. Rev. D* 98.12 (2018), p. 126008. doi:10.1103/PhysRevD.98.126008. arXiv:1802.06889 [hep-th].
- [146] Luis F. Alday, Shai M. Chester and Himanshu Raj. '6d (2,0) and M-theory at 1-loop'. In: JHEP 01 (2021), p. 133.
 doi:10.1007/JHEP01(2021)133.
 arXiv:2005.07175 [hep-th].
- [147] Julian Lee and Sangmin Lee. 'Mass spectrum of D = 11 supergravity on AdS(2) x S² x T⁷'. In: *Nucl. Phys. B* 563 (1999), pp. 125–149. doi:10.1016/S0550-3213(99)00598-2. arXiv:hep-th/9906105.
- [149] Dmitri Sorokin, Arkady Tseytlin, Linus Wulff and Konstantin Zarembo.
 'Superstrings in AdS(2)xS(2)xT(6)'. In: J. Phys. A 44 (2011), p. 275401. doi:10.1088/1751-8113/44/27/275401. arXiv:1104.1793 [hep-th].
- [150] Igor R. Klebanov and Arkady A. Tseytlin. 'Intersecting M-branes as four-dimensional black holes'. In: *Nucl. Phys. B* 475 (1996), pp. 179–192. doi:10.1016/0550-3213(96)00338-0. arXiv:hep-th/9604166.
- Steven B. Giddings and Mark Srednicki. 'High-energy gravitational scattering and black hole resonances'. In: *Phys. Rev. D* 77 (2008), p. 085025. doi:10.1103/PhysRevD.77.085025. arXiv:0711.5012 [hep-th].
- Simone Giombi, Radu Roiban and Arkady A. Tseytlin. 'Half-BPS Wilson loop and AdS₂/CFT₁'. In: *Nucl. Phys. B* 922 (2017), pp. 499–527. doi:10.1016/j.nuclphysb.2017.07.004. arXiv:1706.00756 [hep-th].

- [153] Dalimil Mazáč. 'A Crossing-Symmetric OPE Inversion Formula'. In: *JHEP* 06 (2019), p. 082.
 doi:10.1007/JHEP06(2019)082.
 arXiv:1812.02254 [hep-th].
- F. A. Dolan and H. Osborn. 'Conformal four point functions and the operator product expansion'. In: Nucl. Phys. B 599 (2001), pp. 459–496. doi:10.1016/S0550-3213(01)00013-X. arXiv:hep-th/0011040.
- F. A. Dolan and H. Osborn. 'Superconformal symmetry, correlation functions and the operator product expansion'. In: *Nucl. Phys. B* 629 (2002), pp. 3–73. doi:10.1016/S0550-3213(02)00096-2. arXiv:hep-th/0112251.
- Yevgeny Kats, Lubos Motl and Megha Padi. 'Higher-order corrections to mass-charge relation of extremal black holes'. In: JHEP 12 (2007), p. 068. doi:10.1088/1126-6708/2007/12/068. arXiv:hep-th/0606100.
- [157] Clifford Cheung, Junyu Liu and Grant N. Remmen. 'Proof of the Weak Gravity Conjecture from Black Hole Entropy'. In: JHEP 10 (2018), p. 004. doi:10.1007/JHEP10(2018)004. arXiv:1801.08546 [hep-th].
- [158] Gerhard Mack. 'D-independent representation of Conformal Field Theories in D dimensions via transformation to auxiliary Dual Resonance Models. Scalar amplitudes'. In: (July 2009). arXiv:0907.2407 [hep-th].
- [159] Gerhard Mack. 'D-dimensional Conformal Field Theories with anomalous dimensions as Dual Resonance Models'. In: *Bulg. J. Phys.* 36 (2009). Ed. by V. K. Dobrev, pp. 214–226. arXiv:0909.1024 [hep-th].
- [160] Pietro Ferrero, Kausik Ghosh, Aninda Sinha and Ahmadullah Zahed.
 'Crossing symmetry, transcendentality and the Regge behaviour of 1d CFTs'. In: JHEP 07 (2020), p. 170. doi:10.1007/JHEP07(2020)170. arXiv:1911.12388 [hep-th].
- [161] P. J. Heslop and P. S. Howe. 'Four point functions in N=4 SYM'. In: JHEP 01 (2003), p. 043.
 doi:10.1088/1126-6708/2003/01/043.
 arXiv:hep-th/0211252.

[162] M. Headrick. http://people.brandeis.edu/~headrick/Mathematica/grassmann.m. (Accessed March 2021).